CHAPTER ONE: INTRODUCTION AND BACKGROUND: GROUNDING THE QUESTION

Introduction

In this thesis the level of agreement between cancer patients' understanding of their quality of life (QoL) and their nurses' perception of their QoL will be examined. This will be done by conducting a survey using a questionnaire followed by an interpretive approach to obtain an in-depth understanding of the similarities and differences of QoL perception by cancer patients and their nurses. The purpose of this chapter is to provide the background concerning the topic and to justify the significance of the research study in both the clinical and research areas.

To achieve this goal, the impact of cancer internationally will be discussed from a statistical point of view, looking particularly at Australia and especially South Australia (SA). The importance of situating cancer research in the context of QoL and its potential benefits for cancer patients, health care professionals, and policy makers is discussed. Having emphasised the importance of the direct assessment of cancer patients' QoL, the importance of other people's perceptions, particularly nurses' perceptions, is recounted. Following this, the process through which research questions were developed from the researcher's clinical experiences will be described and finally, an overview of the thesis will be given.

The significance of cancer

Cancer is a major global health issue. Kamangar, Dores and Anderson (2006) reported that in 2002 approximately 25 million persons lived with different kinds of cancer worldwide. They stated that in 2002 lung, breast, colon and rectum, stomach, prostate, liver, cervix, and oesophagus were amongst the eight most common cancers internationally. The World Health Organisation (WHO 2007) estimated that in 2005, 7.6 million (out of 58 million) deaths worldwide (13.1 %) occurred due to cancer. WHO also anticipated that cancer death rates globally will increase to around 9 and 11.4 million in 2015 and 2030, respectively. Other statistical information indicates

that people at no stage of their life can be rid of the risk of cancer; however, it is most likely that individuals in the age group 65 years and older will be affected by cancer (Yancik 2005).

The Australian Institute of Health and Welfare (AIHW 2007) and Australian Association of Cancer Registries (AACR 2007) state that in 2003, one in three men and one in four women would be directly affected by cancer before the age of 75. They also estimated that in 2006 cancer was responsible for the deaths of more than 39,200 people in Australia. This incidence can be identified from statistics presented by the Australian Bureau of Statistics (ABS) about Causes of Death in Australia in 2005. Malignant neoplasms, at 29.4%, were the main cause of death in Australia followed by ischemic heart disease, which was responsible for 18% of all deaths. While ischemic heart disease has been one of the main causes of death amongst Australian men and women, according to ABS (2007) death rates from ischemic heart disease have decreased from 23.7% in 1995 to 18.0% in 2005. However, the corresponding figure for malignant neoplasms has increased from 27.0% to 29.4% over the same period (ABS 2007). Cancer is now the number one killer of Australian men and women.

Cancer is a burden for both Australian males and females and can impact on all organs of the body. The ABS (2006) shows that from those reporting a neoplasm or tumour in 2004-05, 87% reported a malignant neoplasm (cancer). According to AIHW and AACR (2007) in 2003, prostate cancer was the most commonly diagnosed cancer among males, followed by colorectal cancer, melanoma and lung cancer. Breast cancer was the most common cancer among females in 2003, followed by colorectal cancer, melanoma and lung cancer (AIHW and AACR 2007).

The South Australian Department of Health (2004) has reported that cancer is the second major cause of death in South Australia (SA). In addition, one in three South Australians will be diagnosed with cancer at some time during their lives. While both men and women in SA share similar rates of cancers, prostate cancer is the most common for men at 29.2% whereas for women it is breast cancer (28.4%) (South Australian Cancer Registry 2004). Comparison between the SA and national statistics shows that they match each other in terms of incidence and the most

common cancers in men and women. However, cancer deaths in SA are marginally lower than the national figures.

The significance of QoL research in cancer care

Reviewing the oncological literature indicates that investigators have been paying particular attention to QoL, particularly in the last two decades. A range of QoL trials can be accessed involving various types of cancer, such as: prostate (Galalae et al. 2004; Gall 2004), lung (Bottomley et al. 2003), gastrointestinal (Cense et al. 2004; de Liano et al. 2003), breast (Bardwell et al. 2004; Bloom et al. 2004; Bottomley et al. 2005), colorectal (Efficace et al. 2004; Guren et al. 2003), head and neck (Ferlito et al. 2003; Funk et al. 2004), testis (Fossa et al. 2003; Miyake et al. 2004) and gynaecological cancer (Hawighorst-Knapstein et al. 2004; Lutgendorf et al. 2002). This indicates that many researchers consider that QoL is an important consideration when delivering care to cancer patients.

Although many of the QoL research studies conducted with cancer patients have provided useful information, there are publications in which QoL has not been measured with any rigour. Gill and Feinstein (1994), for example, state that there are many articles published under the QoL title that do not have a primary focus on QoL. Added to this there is some evidence of using (or misusing) QoL as a marketing tool, particularly by the pharmaceutical industry (Aaronson 1988). Researchers therefore need to distinguish this bulk of less reliable literature from the more informative studies. The large number of QoL research studies suggests that QoL is an important matter for patients and increasingly so for health care professionals and policy makers.

Cancer patients and QoL

There are many reasons why QoL is considered important for cancer patients. Cancer has a deleterious effect on all aspects of a person's life (MacDonald 2001). Based on the researcher's teaching responsibilities and clinical experiences with cancer patients (described later in this chapter), he recognises that having a cancer can be a catastrophic experience and challenges the individual's reality and, of course, their

QoL. The QoL concept is explained more fully in the next chapter. In brief, this is a concept used to emphasise that different aspects of individuals' lives such as physical, psychological, and emotional are important in determining the experience of living. Quality of life needs to be taken into the consideration by health care professionals when caring for those whose life is under threat.

Even when cancer patients have undergone cancer treatments for cure or to increase their chance of survival, their lives may be distorted further (Isikhan et al. 2001). In other words, in attempting to give years to the life of cancer patients, the life during these years or their QoL has frequently been compromised. Research studies which contribute to a greater understanding of the impact of cancer on the individual's life and subsequent alleviation of this problem are worthwhile.

Health care professionals, policy makers, and QoL

QoL is not only an important issue for individual cancer patients; it is also a matter of great importance for health care professionals and policy makers. QoL information can potentially be used extensively in clinical practice and this matter has been addressed in the literature. Higginson and Carr (2001) highlighted that QoL information can be used for screening and prioritising potential problems in different aspects of health status such as the patient's physical, psychological and social condition. QoL information can help to facilitate communication with patients and identify their preferences, for example, to select a specific treatment or care plan (Jacobsen, Davis & Cella 2002). Patients can communicate their problems and priorities by filling out a QoL questionnaire or through an interview. Issues like sex life, personal relationships and financial issues, for example, are amongst those important matters patients usually do not express explicitly unless they are questioned. QoL assessment can prompt the process of revealing hidden problems more appropriately and lead to more holistic care.

QoL information also has implications for training new staff, clinical auditing and in clinical management (Higginson & Carr 2001). By looking at outcomes of QoL questionnaires along with other complementary information, policy makers, for example, can identify which aspects of patients' care need more attention. This also

has implications for training health care professionals. Knowledge about QoL assessment can be incorporated into the curriculum of undergraduate and postgraduate. QoL information can also be used as an indicator for assessing the cost-effectiveness of different treatments (Hendry & McVittie 2004; Movsas 2003; Schwartz & Sprangers 2002). Two different treatments, for example, may have the same effect on patients' survival, but the treatment decision may be made in favour of one that has a better effect on patients' QoL. This research study will suggest implications for clinical practice and other related fields and therefore will be potentially very useful for health care professionals and policy makers. Potential implications of QoL information are further explained in Chapter Two.

Although QoL information may be important for health care professionals, there are some significant problems that come with this need. One of these is if health care professionals can assess patients' QoL on their own behalf or whether they need to assess it directly with patients. That is, can the carer act effectively as a proxy for the patient or must patients provide QoL information themselves?

Who can provide the best information about patients' QoL?

Health care professionals generally agree that asking patients is the ideal way to assess QoL because QoL is a subjective and personal experience (Fayers & Machin 2000; Movsas 2003; Rebollo et al. 2004; Schwartz & Sprangers 2002). It is always difficult for a health carer to know exactly what QoL means for their patient unless they assess it directly with the patient. However, this "should not imply a wholesale rejection of alternative sources of such information" (Sprangers & Aaronson 1992, p.744). There are situations in which health care professionals' perceptions about patients' QoL are important. For example, in cases where the patient is unable to be consulted a proxy assessment may become the only source of QoL information. Nurses are intimately involved with cancer patients and their perceptions of cancer patients' QoL may be vital.

Significance of nurses' perceptions of cancer patients' QoL

There are situations when nurses' perceptions about cancer patients' QoL are important. Firstly, in the clinical area nurses need to identify different physiopsycho-social needs of patients using a holistic approach (Rebollo et al. 2004; Tallis 2005). The philosophy of nursing invites nurses to nurture people to achieve holistic health and to adopt an approach that incorporates and integrates all aspects of their life into care decision-making (Erickson, Tomlin & Swain 1983). Focusing on QoL is consistent with the philosophy of a holistic approach in nursing (King et al. 2002). Perceptions that nurses form about patients' QoL, provide nurses with the best possible opportunity to identify needs, make decisions and select appropriate actions to be more therapeutic in their supportive roles and to improve patients' QoL (King 2006a). Conversely, without a full understanding of patients' QoL, decision-making about patient care could be potentially haphazard (King et al. 1995; Nekolaichuk et al. 1999; Pickard & Knight 2005).

Secondly, in the clinical area patients may try to show themselves favourably to clinicians and may not genuinely express their feelings (Fayers & Jones 1983). In these situations, an experienced nurse can gain the patients' confidence and hear any unpleasant comments about disease, care or treatment. Therefore, it is possible that nurses' perceptions about patients' QoL can closely reflect what is really going on in the patients' lives. Patients' situations including concerns, expectations and hopes, might be seen through nurses' eyes and perspectives more fully than other practitioners.

Thirdly, although other health care professionals make many treatment decisions about patients (Aaronson 1991), they usually have time constraints, particularly in outpatient oncology settings (Fisch et al. 2003; von Essen 2004). They may rely on nurses to convey to them information about patients' QoL (Parker 1997). In these situations nurses become a voice for their patients. Therefore, this voice needs to be accurate enough to convey to other health care professionals the broad spectrum of patients' problems, needs, and expectations. Despite ongoing debates on the privileged position of nurses in the research and practice with patients (Cheek 1996; Lawler 1997), it appears that nurses generally are more knowledgeable about the

patients compared with other health care professionals. Empirical evidence from a survey with a group of Canadian oncologists, for instance, revealed that more than half of them (54%) identified nurses as the best source of QoL information (Bezjak et al. 1998).

Finally, there are situations in which a direct QoL assessment with the patient is not possible or is very difficult, such as when patients experience cognitive impairments, communication deficits, have symptoms that really frustrated them, where QoL assessment is physically or emotionally too burdensome or patients are very old or very young (Addington-Hall & Kalra 2001; Chaboyer & Forrester 2000; Rebollo et al. 2004; Sneeuw et al. 1998; von Essen 2004) or where patients personally do not like to provide QoL information (Aaronson 1991; Blazeby et al. 1995; Epstein et al. 1989; Magaziner et al. 1988; Sneeuw et al. 1998). Nurses as proxies may form perceptions about QoL of this group of 'non-respondent' patients (King et al. 1995; Slevin et al. 1988; Sneeuw et al. 1999).

When somebody else other than the patient forms a perception about a patient's QoL and reports on it, they are referred to as patients' proxies (Phillips et al. 2003; Tamim, McCusker & Dendukuri 2002). Armstrong (1989, cited in Nelson et al. 1990) states that in 271 studies conducted on the aetiology of non-infectious diseases published between 1980 and 1985 in the *American Journal of Epidemiology*, proxies were used in 8.8% of studies. Stineman et al. (2004) stated about one third of the information from the National Health Interview Survey (NHIS) on QoL was proxyprovided. Part of the results of a review by Sneeuw (2002) between 1991-2000 revealed that nurses were selected in many QoL research studies to act as patients' proxies for non-respondent patients. Accordingly, findings of the research reported in this thesis may also be useful and relevant for researchers who are interested in knowing whether nurses can provide reliable information in proxy research studies.

Whatever the circumstances, a reasonable degree of agreement needs to exist between the patient's and nurse's perceptions of the patient's QoL. Given this, assessing the level of agreement between the patient and the nurse over the patient's QoL is considered important and worthy of investigation.

The gap in QoL research

The need for a research study comparing nurses' and patients' perceptions on QoL was further reinforced when it was identified that there is still a gap in the QoL research literature. Evidence from previous research studies focussing on 'proxy rating' identified several general trends. Proxies, for example, appear to have a better understanding of objective aspects of QoL, such as the physical aspect, rather than subjective aspects such as the psychological and social QoL (Becchi et al. 2004; Herrman, Hawthorne & Thomas 2002; Lobchuk & Degner 2002; Wilson et al. 2000). However, there are gaps or omissions in such research studies.

Firstly, research studies in which the perceived QoL of cancer patients is compared with that of nurses worldwide appear to be inconsistent in their outcomes. For example, nurses' perceptions of cancer patients' QoL are considered inaccurate in some research studies (Brunelli et al. 1998; Horton 2002; Slevin et al. 1988; Zhao et al. 2003) whereas others reported that such perceptions are reasonably correct (Fisch et al. 2003; Geddes et al. 1990; Sneeuw et al. 1999). Generally, researchers recommend further studies to compare proxies' ratings of patients' QoL with that of patients' own rating (Lampic & Sjoden 2000; McPherson & Addington-Hall 2003; Pickard et al. 2004; Tang & McCorkle 2002b).

Secondly, few research studies adequately address the influence of major factors affecting agreement between patients and proxies (Magaziner et al. 1988; Novella et al. 2001; Sands et al. 2004; Tang & McCorkle 2002b; von Essen 2004). As stated by several researchers (Lobchuk & Degner 2002; Sneeuw, Sprangers & Aaronson 2002) the findings of research studies have not been consistent and there is need for further research work. For example, in one research study, the degree of QoL agreement between patients and their proxies was influenced predominantly by the patients' performance status (Horton 2002). Another study yielded no evidence of such a relationship (Sneeuw et al. 1997). Further research is necessary to identify the different variables affecting QoL agreement (Lampic & Sjoden 2000; Tang & McCorkle 2002b; von Essen 2004) using more accurate statistical tests including multivariate analysis (see Chapter Five - results of the first phase).

Thirdly, the majority of QoL proxy research studies are empirical and data have been analysed by different statistical tests. These are mainly Bivariate tests (such as the Pearson correlation). There is a lack of interpretive research in this field. Such research studies can explore nurses' perceptions of cancer patients' QoL in greater depth and uncover factors that may influence nurses' perceptions of cancer patients' QoL. Further interpretive research studies comparing different perspectives should be conducted.

Finally, the literature search for the research reported here yielded no Australian research comparing nurses' rating of patients' QoL with cancer patients' rating of their QoL. A number of empirical research studies have been conducted, but they used QoL tools developed in other countries including the United States of America and united Kingdom rather than the cross-culturally developed tools such as the World Health Organisation's Quality of Life Brief Questionnaire (WHOQoL-BREF) which incorporates a broad spectrum of QoL issues (The WHO Group 1995). Outcomes of research studies in which the WHOQoL-BREF is used can better be generalised to other situations or countries (see Chapter Four).

The literature review shows that a research study comparing patients' and nurses' understanding of cancer patients' QoL, and identifying variables influencing agreement between these groups in Australia, would be useful. The study needs to adopt a QoL tool with strong external validity such as the WHOQoL-BREF questionnaire, use more robust statistical tests, and incorporate interpretive methodology.

The researcher's experience and background

This research study uses an interpretive approach to evaluating the experiences of cancer patients and nurses about cancer patients' QoL. It is therefore appropriate to consider the researcher's experiences and background, particularly in relation to cancer patients. These experiences began with working as a qualified nurse and developed further as the researcher worked as a clinical educator, university lecturer and PhD student.

The researcher has worked with cancer patients in different clinical situations, particularly chemotherapy and radiotherapy departments, for more than 10 years in his country of origin, Iran. The researcher began working with cancer patients at the end of the second year of his Bachelor of Science in Nursing. Due to hospital work demands, nursing students at a Bachelor level could apply for work after finishing the second year of study. The researcher therefore had a professional position for two years as a qualified nurse and worked with cancer patients as well as other medical-surgical patients across different non-specialist oncology wards during his study commitments. These experiences encouraged the researcher to begin thinking about the different physical-psycho-social needs of cancer patients and how health care professionals managed them.

Working as a qualified nurse coincided with the researcher's studies for the Master of Science in Medical-Surgical Nursing Education for three years. To complete this degree, the researcher investigated early side effects of radiotherapy in cancer patients undergoing radiotherapy after a self-care education program in Iran. The researcher saw many patients dealing with difficulties related to diagnosis of cancer and their life expectancy in specialist oncology wards. Patients were concerned about the symptoms of cancer, cancer treatment side effects, and the social and psychological problems associated with cancer along with their mortality. The researcher gradually developed a feeling that both 'quantity and quality' of life for cancer patients should be paid particular attention. It brought to his mind that for many people affected by cancer, life during and after therapy is not just concerned with how long they live, but how well they live. It was during these times that the researcher thought more deeply about psycho-social needs of cancer patients as well as physical ones.

By spending time reading oncology books and research papers and teaching nursing students in the hospitals, the researcher learned more about the importance of QoL research for cancer patients and nurses. The researcher developed a closer relationship with patients during patient care with nursing students in cancer departments, and these led to some valuable experiences. One of these experiences for example, concerned a patient with a malignant lymphoma in his abdomen. The patient was not told about his diagnosis but his family was. The clinician started

chemotherapy in order to decrease the size of the tumour and to facilitate its removal later by surgery. The clinician thought that he made the best medical choice for the patient and withheld the truth from the patient. The clinician thought that the patient would accept his diagnosis and comply with the treatment later. A couple of weeks later the patient experienced alopecia and noticed his diagnosis. The patient realised the truth of having cancer, became distressed, and did not follow through to the surgical treatment. The patient died later due to the very severe side effects of the aggressive chemotherapy that was required. This experience encouraged the researcher again to consider more deeply if health care professionals particularly nurses, have a holistic understanding of the needs, expectations, and desires of patients who live with cancer.

The researcher received a scholarship to continue his studies in Australia leading to a PhD. In the light of that knowledge related to QoL and this clinical experience with cancer patients, the researcher decided to investigate QoL with cancer patients. He reflected on his first experiences with cancer patients' QoL, discussed the issue with his colleagues and thought more deeply about how perceptions of health care professionals and even family members are similar or different from those of the patients. The researcher has also pondered over the possible reasons why these perceptions are similar or different and raised a further question as how perceptual differences might affect cancer patients. For example, the researcher was interested in how different treatments that patients receive (such as chemotherapy and radiotherapy) might affect nurses' perceptions of cancer patients' QoL. What will happen to patients if nurses do not understand the full spectrum of patients' needs and problems when they, for instance, undergo chemotherapy. What about nurses' factors like their experiences, gender and age? Do more experienced or older nurses have a better grasp of the patients' situation or their QoL or not?

The researcher also reflected on his own position as an international student who is doing this research study in Australia. Cultural differences might affect his interpretations, particularly in the interpretive phase of the research study and undermine the usefulness of the study for Australia. The researcher has tried to resolve this by engaging more in Australian society and through ongoing consultations with his supervisors, other Australian friends, and a few patients. On the other hand, how can research outcomes in Australia be useful for his country of origin? This idea emerged because the researcher has always been personally interested in improving nursing care for cancer patients in Iran. This challenging question is addressed in the next section.

The significance of the research study for health care in Iran

How can the findings of this research study in Australia benefit and influence health care in Iran? Even though the research study is conducted in Australia, research outcomes will be still relevant to Iran for several reasons.

Firstly, cancer is also a major health problem in Iran. Unfortunately, obtaining reliable statistical information from Iran can be problematic. Based on the researcher's best knowledge, institutions or centres that are engaged in collecting or reporting cancer statistics and other relevant information are very limited in Iran compared with Australia.

As an international student it was surprising to see that in Australia different institutes or foundations are collecting and reporting critical statistical information about cancer separately or in relationship with each other. For example, the ABS collects information about cancer through the National Health Survey (NHS) but also uses complementary information from other sources including the Australian Institute of Health and Welfare (AIHW). AIHW works in conjunction with different centres, particularly those in their states in collecting, analysing and reporting healthrelated data. At the state level, Australia also benefits from the different government and non-government cancer related centres. The Cancer Council, for example, is a non-government institute and one of its main goals is to improve understanding and public knowledge of cancer in each state. Another major federal government centre working across government jurisdictions is the Australian Department of Health and Aging which is actively engaged in registering diagnosed cancers through a Cancer Registry.

Based on the researcher's knowledge, the only up-to-date information about cancer

in Iran has been reported by WHO (2006). It indicates that in 2005 cancer killed around 47000 people in the country. Cancer is the main cause of death (11.8%) in Iran after cardiovascular diseases and injuries. Other chronic diseases altogether constitute only 17.5% of deaths nation-wide. The rate of cancer death is projected to undergo a slight increase from 11.8% to 13.4% by 2030. In the WHO report on Iran, a general category is introduced as 'other malignant neoplasms' but it is not clear what kind of cancers fall into this category. In this report, stomach cancer and 'other malignant neoplasms' are the most common cancers in males in 2002. For females the situation changes in that 'other malignant neoplasms' are at the top of the list followed by stomach cancer.

Comparing available information about Iran with international figures indicates three points. The first is that general trends are only marginally less than the international statistics. For example, 13.1% of deaths worldwide are due to cancer and in Iran it is 11.8%. The second point is that the incidence of most diagnosed cancers in males and females in Iran varies slightly from the international statistics. In Australia, for instance, the most common cancers are breast cancer and prostate cancer in women and men, respectively. These cancers were ranked at the fourth and third category of most common cancers amongst Iranian men and women, respectively. The third point is that cancer trends in Iran might be closer to international figures if a more nation-wide data collection was conducted. It is therefore essential that this information require strategies to be planned, such as registration of diagnosed cancers in each province and then at the national level in Iran.

Secondly, the relevance of relating the research outcomes to Iran can more fully be recognised if the current research study is seen in the framework of a bigger research study that will be conducted by the researcher after finishing his PhD. In other words, outcomes of this research study can be beneficial by undertaking a comparative study in Iran in the near future. Comparing QoL data across different countries in the shape of cross-cultural research trials is one important line of research for investigators. This would help, for instance, to explore if cancer patients in a developed country like Australia have a better state of QoL compared with a developing country like Iran. Such research studies would be very useful in evaluating the underlying reasons for this and improving the health systems.

13

Thirdly, in the survey phase of the research study the WHOQoL-BREF questionnaire, discussed earlier, was chosen. The dimensions that are covered by this tool may be broad enough to cover many QoL issues that are relevant to Australia and Iran. Some empirical outcomes of the research study therefore might have external validity and applicable to Iran, but with necessary caveats.

Fourthly, the relative transferability of the qualitative data from this Australiancontext to Iran is a relevant topic to discuss. Transferability of interpretive findings from one area to another relies on the similarity of contexts and there is some literature that has done this. For example, cancer patients in developing countries like Iran are suffering from many QoL issues that may occur in Australia. The major cancers in developing countries appear in an advanced state and given the notion that such countries do not have enough resources for treating them well, it is important to provide opportunities to prevent, palliate and improve their QoL (Olweny 1992). Chaturvedi et al. (1997) also stated that in most Asian countries, QoL activities were very diverse and in the early stages of development, except in Japan. They also concluded that QoL deserves greater recognition as a part of the general public health landscape. Their conclusions were based on pre-designed forms answered by representatives from different Asian nations including one representative from Australia. Studies with a mixed approach like this research study (see Chapter Three) can potentially create outcomes that might have better 'inference transferability' to other places and situations (Teddlie & Tashakkori 2003, p. 37).

The research questions

This research project will examine how accurately nurses perceive and report cancer patients' QoL in the clinical area in an Australian population. The following main research questions will be answered:

- 1. What differences and/or similarities are there between cancer patients' and nurses' perceptions of cancer patients' quality of life?
- 2. Why do differences and/or similarities exist between cancer patients' and nurses' perceptions?

It would be ideal to undertake a research study involving all key persons who are engaged in caring for cancer patients such as other health care professionals and significant others as well as nurses. However, nurses are intimately involved with cancer patients more than other health care professionals and their understanding of cancer patients' QoL is important. Moreover, inclusion of other health care professionals or significant others was beyond the resources available and therefore the scope of this study.

The research study utilised rigorous methods to comprehensively compare nurses' understanding of cancer patients' QoL with that of patients' own perceptions.

Methodology

The research study was underpinned by a mixed approach. The first phase was based on an empirical methodology and utilised the WHOQoL-BREF questionnaire. The aim was to assess the level of agreement between patients and nurses about cancer patients' QoL and identify variables influencing agreement. The second phase of the study was founded on the principles of grounded theory. The aim of this phase was to include an interpretive perspective and explore reasons why patients and nurses may differ in their perceptions using semi-structured interviews. More details of the methodology and methods can be seen in Chapter Three (methodology chapter), Chapter Four (methods of the first phase), and Chapter Six (methods of the second phase).

Thesis structure

The thesis consists of nine chapters. Chapter One provides an introduction to the whole project and discusses the significance of the research topic. Chapter Two concerns a review of the literature which serves as the basis for generating the research questions.

Chapter Three describes the methodology and explains the philosophical issues underpinning each phase of the research study. This chapter emphasises the role of the mixed research approach.

Chapter Four outlines the method of the first phase and the results of this are presented in the next chapter, Chapter Five.

In Chapter Six and Chapter Seven, respectively, methods and results of the second phase are fully described.

In Chapter Eight, the discussion chapter, all the key information drawn from the two phases of the research study is summarised and discussed with reference to the research questions and findings. Limitations of the research approach are discussed at the end of the chapter.

Chapter Nine provides the conclusion and discusses the implications arising from comparing patients' and nurses' perceptions about cancer patients' QoL.

Conclusion

The importance of the QoL concept, particularly in the clinical area of cancer patients, has been considered. The key issue is that patients, while they can provide the best source of information about their own QoL, may not be able to provide this data. In the meantime, nurses also form perceptions about cancer patients' QoL. They may need to act on such perceptions to support their clinical decision-making, especially where patients cannot provide such information. Other health care professionals and researchers may also rely on nurses to provide them with proxy QoL information about patients. It is useful to determine the extent to which nurses' understanding of cancer patients' QoL agrees with that of patients themselves. The motivation for this study arises from the researcher's knowledge and experience as a qualified nurse and a nursing clinical educator and the need to improve the adequacy and accuracy of nurses' assessment of patients QoL.

In the next chapter, the literature that has been published on general and more specific aspects of the QoL concept (such as QoL development, definitions,

dimensions, and implications) will be reviewed. Furthermore, the key variables related to the level of agreement between cancer patients and nurses about cancer patients' QoL and how these variables may influence the level of agreement between the parties will be reviewed. This discussion will provide the basis for formulating the research questions.

CHAPTER TWO: LITERATURE REVIEW: SITUATING THE STUDY IN RELATION TO OTHER RESEARCH AND ASSOCIATED THEORY

Introduction

Chapter One argued that QoL research with cancer patients has potential benefits for cancer patients, health care professionals and policy makers. Another justification for conducting this research study was that the findings from comparing nurses' perceptions about cancer patients' QoL with that of patients' own perceptions will add new knowledge to the current literature.

The purpose of this chapter is to relate this research study to the most common issues and debates in the literature about nurses' perceptions of cancer patients' QoL. Although this research study has a mixed approach, most of the literature reviewed in this chapter is empirical in nature and this helps to establish a rationale for the research questions. A few existing interpretive studies are also reviewed in this chapter. Another literature search was conducted at the conclusion of the interpretive phase, focussing on the specific categories that emerged.

Concept of Quality of Life (QoL)

Understanding the concept of QoL is essential for this research project and it can be stated at the outset that this concept is the result of lengthy debate and discussions.

Development of the QoL concept

In the following historical survey, the development of the QoL concept has been divided into three stages: early times, 1940s-1960s, and 1960s-present. This narrative framework has been chosen because it reflects the relative quantity and nature of discussions that appeared during these eras.

Early times

The expression QoL is fairly new in academic literature (Szalai 1980) but the themes it dealt with were nonetheless familiar to some people. Strain (1990) and King (2003), for example, highlighted that people have always expressed concerns regarding having a good life particularly after their basic needs have been met. Some researchers believe that concepts such as human happiness and satisfaction have been given particular attention over a long period of time. Richardson Lear (2004), for instance, stated that the historical development of the QoL concept goes back millennia and possibly originated in the time of Aristotle in ancient Athens. Aristotle was one of the great Western philosophers who articulated about concepts such as happiness. In the book *Introduction to Aristotle* (McKeon, 1947), one single chapter focuses on issues concerning happiness. Consider, for example, the following extract:

...what is the highest of all goods achievable by action. Verbally there is very general agreement; for both the general run of men and people of superior refinement say that it is happiness, and identify living well and doing well with being happy; but with regard to what happiness is they differ...For the former think it is some plain and obvious thing, like pleasure, wealth, or honour....Now some thought that apart from these many goods there is another which is self-subsistent and causes the goodness of all these as well. (Aristotle, cited in McKeon 1947, p. 311)

Aristotle explained that happiness has different meanings for individuals but there should be an ultimate kind of happiness that satisfies all people. Despite this general interest about issues like happiness and pleasure, it appears that until recent decades there was less conceptualisation and clarification about QoL.

From the late 1940s to the 1960s

Several important publications can be identified indicating that in the late 1940s a shift in the conceptualisation about health and evaluating medical interventions occurred. Karnofsky and Burchenal (1949), for example, introduced the idea that in evaluating chemotherapeutic agents in cancer, both quantitative (such as survival rate) and qualitative (such as performance status, well-being) factors should be given particular attention. They described performance status as an important criterion indicating how patients can independently conduct their daily living activities or how much they rely on medical care. Karnofsky and Burchenal were pioneers in the

medical field and introduced a scale that emphasised the importance of measuring the performance status of patients rather than attempting to accurately determine patients' prognosis (Spitzer 1987). Similarly, the Index of Activities of Daily Living was developed by Katz et al. (1963); this instrument assessed patients' physical functioning. However, it can be argued that these measures are generally unidimensional in character and focus mainly on physical aspects. These publications marked a significant turning point in the discussion of QoL.

A similar conceptualisation about health can also be seen in the WHO definition of health that was released in 1948. WHO's constitution (1946, p.1315) defined health as "a state of complete physical, social and mental well-being, and not merely the absence of disease or infirmity". De Haes and van Knippenberg (1985) stated that psychological and social needs, as well as materialistic goals, become more apparent in the literature during the 1960s and 1970s. The period between the 1940s and 1960s witnessed a greater focus on health and a holistic approach to understanding one's health. This can be considered as an introductory phase to the formal introduction of the QoL concept that emerged later.

From the 1960s to the present

Szalai (1980) stated that the term QoL was used for the first time during the late 1950s or early 1960s in popular discussions or general publications like newspapers and magazines. This period saw the development of a number of QoL definitions, refined to the point in 1993 when WHO explicitly stated that all individuals have a right to psycho-social care for an adequate QoL in addition to physical care (World Health Organization 1993). Based on QoL definitions, a number of QoL tools were also created including those developed by WHO (WHOQoL-BREF and WHOQoL-100). The advantages and disadvantages of QoL tools are explained in more detail in the following chapters, particularly Chapter Four.

Nowadays the QoL concept is a very well-known concept though it continues to generate considerable discussion and debate. There are several key themes present in the literature concerning QoL and to some extent these can help us to identify the principal elements and issues surrounding it. The most common themes in the literature, particularly recently, include:

- 1. Quality and/or quantity (Chancellor, Coyle & Drummond 1997; McVilly, Burton & Davidson 2000; Steel, Geller & Carr 2005; Tang & McCorkle 2002a). In other words, when considering individuals, not only is the number of years of life important but also the way that a person lives during those years. There might be people who live for a longer time but have not fulfilled their potential in all those years. Conversely, a number of individuals have shorter lives than others but are quite happy and satisfied during these years.
- 2. QoL assessment along with other measurements like survival, tumour control, treatment toxicity, performance status, mortality and morbidity (Bottomley 2002; Costantini et al. 2000; Cox 2003; Garratt et al. 2002; O'Carroll et al. 2000; Roila & Cortesi 2001). While criteria like survival or mortality rate are important and might reflect patients' QoL, these criteria cannot adequately describe the impact of a disease on all aspects of individuals' lives. Conversely, a QoL assessment takes into account a range of important physical, psychological, and social issues particularly during the course of a disease like cancer.
- 3. QoL assessment as a means to consider both subjective and objective parts of health status (Bottomley 2002; Giesler 2000; McVilly, Burton & Davidson 2000; Skevington, Lotfy & O'Connell 2004; Testa & Simonson 1996). The focus of a QoL assessment is not only on objective (measurable) factors like nutritional status, sleep patterns, and physical activities. It must also take into account subjective aspects of experience like individuals' satisfaction with their family, community, society, and other relationships.

In summary, it is not clear when and who exactly created the concept of QoL. However, it did gain importance in the literature during the 1960s. QoL has been developed further since that time through the introduction of specific QoL definitions and tools. QoL assessment can potentially lead to a better understanding of the patient as a whole and their health status and provides a parameter by which various domains of health status, such as physical, psychological, and social, can be assessed or evaluated. This research study is conducted in the oncology field and therefore it is important to understand that QoL has implications for this area of expertise specifically.

Development of QoL concept in cancer care

Oncology is one of the main fields in which QoL began. Aaronson (1989) identified several reasons for QoL being developed in the oncology area:

1. The patient's needs and choices need to be considered before making treatment decisions.

Aaronson argues that cancer has been a long-term and not always curable disease. He concluded that it was quite important that in the selection of treatment choices and care strategies, patients can receive particular attention. In addition, before introducing the QoL concept, it was common for health care professionals to select a treatment choice for a cancer patient without any consideration of what patients wanted.

2. The need to identify less toxic treatments.

Despite significant advances in improving the survival of cancer patients and introducing a number of cancer treatments, Aaronson describes the concerns to identify less toxic cancer treatments. According to him, toxicities related to treatments can affect the QoL of cancer patients, and QoL assessment was a way to identify better treatments with less toxicity.

The implication of Aaronson's view is that cancer patients will benefit. For example, a specific group of cancer patients can potentially receive a chemotherapy or radiotherapy regimen as a treatment. Through QoL assessment patients might express, for example, that a radiotherapy regimen has a better impact on their QoL than chemotherapy. So it might be helpful for health care professionals to realise that the radiotherapy regimen has less treatment toxicities for patients and they continue with that treatment.

3. The need to consider patients' rights.

According to Aaronson, much more attention was paid to the concepts of

patients' rights and especially formal consent. These rights enable patients to be informed and QoL assessment was a unique way to translate medical information into understandable concepts for patients.

This movement as explained by Aaronson has been beneficial to cancer patients. For example, it became quite important that patients have a right to make choices about their treatments, how they let health care professionals treat them, and even they can say 'No' to a treatment if they think that they cannot or will not tolerate it. Health care professionals can provide cancer patients with a QoL tool, through which patients can understand a number of issues and have this opportunity to convey to the health care professionals their expectations and needs.

4. The need to approach treatment more holistically.

Finally, Aaronson describes the way in which more emphasis was put on treating cancer patients with a holistic approach. In fact, in cancer treatments both psychological and physical toxicities are important. QoL assessment was a useful approach, which increased awareness that with treatments, both the patient's physical and psychological toxicities should receive particular attention.

Aaronson suggests that when a patient is receiving a chemotherapy regimen, for instance, they are faced with a range of physical toxicities (e.g. stomatitis and diarrhoea) and psychological and emotional toxicities (e.g. a sense of isolation and body image disturbances) and all of them need to be taken into consideration. A health care professional cannot provide effective care and treatment to a patient experiencing stomatitis, for example, with a chemical component which provides only pain relief and comfort in the oral cavity but forget the preceding anxiety associated with it or the fear of recurrent stomatitis. The researcher's experience suggests that when managing a stomatits, a broad range of concerns can be explored; a small window to the holistic care of clients.

Altogether, the development of the concept of QoL concept is a response to patients who believe that they are not only physical bodies but also human beings who want to live and have personal needs and interests. QoL to this day continues to be a controversial concept and its definition has been a contested field.

QoL definitions

Several researchers (Cella & Tulsky 1990; Varricchio 1990) on the issue of QoL research in a clinical area like oncology have insisted on the need for a workable definition. This is a difficult task as there is no universally accepted definition and many debates exist (Bottomley 2002; Fayers & Machin 2000; Hendry & McVittie 2004; King et al. 1997). There are several reasons why there is a lack of consensus about the definition of QoL. Firstly, this concept has a highly individual nature related to one's personal values (Anderson & Burckhardt 1999; Carr, Gibson & Robinson 2001; Holmes & Dickerson 2003; King et al. 2002; Mast 1995; Zhan 1992). It is "a vague and ethereal entity, something that many people talk about, but which nobody very clearly knows what to do about" (Campbell, Converse & Rodgers 1976, p. 471). Assessment of QoL includes "the measurement of practically anything of interest to anybody" (Andrews & Withey 1976, p.6). A concept that is so highly subjective will inevitably make it difficult to offer a definition acceptable to all because it constitutes issues that are experienced by a broad range of people.

Secondly, the definition of QoL is often related to who is involved and from what perspective. For example, many philosophers emphasised that QoL is related to personal happiness (Holmes 1989). Chinese philosophers believe that quality of life is achieved if there is a balance between Yin and Yang (Zhan 1992). Economists define QoL as being associated with the allocation of resources to achieve aims (Anderson & Burckhardt 1999). So, different disciplines have different approaches to QoL definition.

Thirdly, some similarities can be observed between this concept and others such as health, well-being, and life satisfaction, to name a few, which makes it even more difficult to introduce an exact and consensual definition of QoL (Anderson & Burckhardt 1999; King et al. 1997; Zhan 1992). People, for example, may think that

life satisfaction is with the same as QoL, while others will argue that it is just a part of it which only takes into the account the subjective aspects of QoL (King 2003).

Due to these ambiguities, different definitions have been proposed. Some common definitions of QoL are as follows:

- individuals' perceptions of their position in life in the context of the culture and value systems in which they live and in relation to their goals, expectations, standards, and concerns (World Health Organisation 1996, p.6);
- a person's sense of well-being that stems from satisfaction or dissatisfaction with the areas of life that are important to him/her (Ferrans & Powers 1992, p.29);
- 3. subjective evaluation of the positiveness or negativeness of attributes that characterize one's life (Padilla et al. 1990, p.108).

Despite different definitions, some important characteristics emerge from such definitions of QoL. Firstly, QoL is an individual experience. No two persons can be assumed to have the same level of QoL before it can be assessed (Hinds & King 2003). Further, comparing two persons who appear to have the same level of QoL as a means of assessing one's QoL is problematic because the perception of QoL varies amongst individuals.

Secondly, QoL is a subjective judgement; a perception or sense. Even though there are some objective measures that can show individuals' perceptions of QoL (King 2003), no outsider can claim that one can truly assess somebody else's QoL based only on such measures. For example, daily living activities of patients, as an objective criterion, can be used to assess how patients are physically independent. While it may be logical to assume that patients who are more independent in their daily activities have higher levels of QoL, this would be an incomplete evaluation unless other subjective issues and concerns are also considered.

Finally, the level of QoL is related to different things in individuals' lives such as expectations, important life goals and chosen activities. In other words, achieving an appropriate level of QoL for a specific person depends on meeting different needs that are priorities for that person.

This current research study is mainly based on the first definition (above) developed by the WHO. The WHO definition supports the study because it includes all of the above mentioned factors (individuality, subjectivity, and breadth). In other words, in this research study QoL is considered as being the patients' perception, which is individual and subjective. It is acknowledged that the patient is the most important source of QoL information and other perceptions about patients' QoL need to be compared on that basis. This thesis also accepts the notion that QoL consists of different aspects that constitute a person's QoL. These are usually called QoL domains or dimensions. Research studies related to QoL need to clarify what QoL domains will be considered.

Quality of Life domains

Like QoL definitions, disagreements exist as what domains exactly contribute to individuals' QoL. This is not surprising because the domains are based on the QoL definitions. The terms QoL domain, QoL dimension, and QoL aspect are used interchangeably (Gill & Feinstein 1994; Glaus 1993; Trompenaars et al. 2005). In this research study these terms (QoL aspects, QoL domains, and QoL dimensions) are used interchangeably. Quality of life domains are considered to be "a set of elements to which a variable is limited, or the range over which the concept of QoL extends" (Schalock 2004, p.205). A QoL domain consists for particular components (Gill & Feinstein 1994). Generally speaking, there is not general consensus about the range of domains or aspects of QoL (Draper 1992; Ferrans & Powers 1985; Schumacher, Olschewski & Schulgen 1991).

While in the past only one QoL domain - such as physical - has been measured, most experts would agree that QoL consists of at least the physical, psychological and social domains (Boling, Fouladi & Basen-engquist 2003; Cox 2003; Tallis 2005). Additional domains have also been identified such as the functional (Brunelli et al. 1998; Cella et al. 2002; Fallowfield 2002; Wilson et al. 2000), emotional (Cella et al. 2002; Fallowfield 2002; Movsas 2003; Schumacher, Olschewski & Schulgen 1991), spiritual (Bonomi et al. 2000; Ferrell Dow & Grant 1995; King et al. 1997), family (Ferrans & Powers 1992), illness (Hawthorne, Richardson & Osborne 1999), independent living (Bonomi et al. 2000; Hawthorne, Richardson & Osborne 1999),

symptoms and side effects (King et al. 1997; Lindley 1992) and environmental domains (Bonomi et al. 2000).

It is important to consider what QoL domains are suggested by the WHO. Based on the WHO definition of QoL, QoL must be assessed across six domains. They are: physical health; psychological health; level of independence; social relationships; environment; and spirituality. In order to address these QoL domains, WHO constructed a QoL tool named WHOQoL-100 (The WHO Group 1998a).

The WHO later developed another QoL tool named the WHOQoL-BREF questionnaire. This tool is a short version of the WHOQoL-100, which is useful for assessing QoL in the clinical area. It assesses QoL across four domains: physical, psychological, social relationship, and environmental (World Health Organisation 1998). The domains concerning spirituality and the level of independence appearing in the original tool were merged into other domains. In the survey (first) phase of this research study, QoL is assessed across four domains as recommended by the WHOQoL-BREF questionnaire. More details about this tool and other QoL tools can be seen in Chapter Four.

Even though the WHOQoL-BREF questionnaire is generally an appropriate tool for use in clinical contexts, some criticism exists. For example, it has been emphasised that items in the social aspect of the tool may not be comprehensive enough to assess this domain (O'Carroll et al. 2000). So this research study will help to determine the appropriateness of this tool further by addressing one of its main psychometric properties, namely reliability, and through interpretive analyses, which emerge in the second phase of the research study. The critique of the WHOQoL-BREF questionnaire is discussed in Chapter Nine. QoL domains of the WHOQoL-BREF questionnaire and their related facets are demonstrated in Table 2-1. In this table, each facet is defined as a behaviour, state of being, a capacity or potential, or a subjective perception or experience (The WHOQOL Group 1994).

Domain	Facets incorporated within domains
1.Physical health	Activities of daily living
	Dependence on medicinal substances and medical
	aids
	Energy and fatigue
	Mobility
	Pain and discomfort
	Sleep and rest
	Work Capacity
2. Psychological	Bodily image and appearance
	Negative feelings
	Positive feelings
	Self-esteem
	Spirituality / Religion / Personal beliefs
	Thinking, learning, memory and concentration
3. Social relationships	Personal relationships
	Social support
	Sexual activity
4. Environment	Financial resources
	Freedom, physical safety and security
	Health and social care: accessibility and quality
	Home environment
	Opportunities for acquiring new information and
	skills
	Participation in and opportunities for recreation /
	leisure activities
	Physical environment (pollution / noise / traffic /
	climate)
	Transport

Table 2-1: Description of aspects of the WHOQoL-BREF questionnaire expressed as domains and facets of QoL

Source: World Health Organisation (1996, p.7)

To conclude, QoL is a broadly defined concept and located in multiple domains. In the survey phase, this research study uses the WHOQoL-BREF questionnaire. It assesses QoL across four domains as recommended by the WHOQoL-BREF questionnaire: physical, psychological, social relationship, and environmental. This research study acknowledges the very personal nature of the QoL concept and for this reason incorporates a more in-depth assessment of QoL during the interpretive phase of the research study.

When QoL research is conducted in the clinical area, it is also important to make a decision whether to use the term QoL or Health Related Quality of Life (HRQoL). This includes domains that are suitable for the clinical area but is again a debatable issue that will need to considered further in the next section.

Differences between Quality of Life (QoL) and Health Related Quality of Life (HRQoL)

Similar to debates about QoL definitions and domains, an ambiguity can also be observed in the literature about the difference between QoL and HRQoL. Some researchers (Boling, Fouladi & Basen-engquist 2003; Fayers & Machin 2000; Movsas 2003) believe that the term QoL is used in a variety of fields, but the term HRQoL is often used to indicate quality of life in relation to diseases and therapeutic interventions and argue that it is a more suitable term in clinical research trials and in the medical field. Boling et al. (2003) also indicated that HRQoL differs from the general concept of QoL, in that it usually includes the psychological, physical, and social functioning of patients, but excludes perceptions of environment, housing, or other external domains.

Hunt (1997) in an editorial paper, critically reviewed QoL definitions and domains and pointed out that the term HRQoL has been inserted to solve some criticisms related to the QoL concept. However, separating effects or problems related to health from those simultaneous issues related to finance, friendship, family life, responsibilities, expectations, occupation and aging is in reality not a well considered distinction to make when one takes into account the overlap between them. Family problems, for example, may worsen the process of patients' recovery whereas those patients with a more supportive family might receive better care and therefore more rapid improvement in their health status. Guyatt et al. (1993) also indicated HRQoL is a suitable term as a means of emphasising those aspects of QoL related to health. However, when a patient is so ill, all aspects of life such as income, freedom, and environmental issues not directly related to health, should be regarded as health-

related matters.

To address this ambiguity, in many research or review articles (Allison, Guichard & Gilain 2000; Bezjak et al. 2001; Cella, DF 1996; Dancey et al. 1997; Fayers et al. 1997; Stavem et al. 2000; Stephens et al. 1997; Trompenaars et al. 2005) both terms QoL and HRQoL have been used though without any rationale as to why. It also appears that even the WHO did not make a clear distinction between these two terms and prefers to define the term QoL. The reason might be that domains introduced by the WHO are comprehensive enough to consider any issues important in the clinical area or under other circumstances. So the WHO definition of QoL might imply that such a distinction between QoL and HRQoL may not be appropriate. A review study on 75 QoL articles was conducted by Gill and Feinstein (1994) to evaluate how well QoL is being measured in the medical literature. Results revealed no article differentiated 'overall' QoL from HRQoL (Gill & Feinstein 1994). It is also interesting to note that QoL and HRQOL are also considered to be synonymous terms in some websites (Phoenix 5 2002).

It can be concluded that while both terms, QoL and HRQoL, have been used extensively and interchangeably in this field, the term QoL is more suitable for this study. The reason is that the term QoL is an easier term to follow compared with HRQoL. It is a holistic concept because this research study was based on the WHO definition of QoL and QoL domains might be comprehensive enough to include all issues that might be important for individuals.

Implications of QoL information

The literature generally describes implicitly or explicitly the reasons behind assessing QoL (Aaronson 1989; Fayers & Machin 2000) and in the oncology area specifically (Jacobsen, Davis & Cella 2002; Roila & Cortesi 2001; Wright et al. 2003). Fayers and Machin (2000, pp.7-14) described seven subdivisions where QoL assessment existed in randomised clinical trials:

- 1) clinical trials of treatment with curative intent;
- 2) clinical trials of treatment with palliative intent;

- 3) improving symptom relief, care or rehabilitation;
- 4) facilitating communication with patients;
- 5) patient preferences;
- 6) late problems of psychological adaptation;
- 7) medical decision-making.

In the following pages these subdivisions will be discussed more fully with some examples from oncology and non-oncology literature used to clarify the importance and value of QoL data. While QoL implications are discussed in separate sections, overlap exists between these implications. For example, while a research study is conducted with the aim of improving care or communication with patients, it can also help researchers to identify patients' preferences. Another research study, for instance, might be conducted with a curative purpose but outcomes can also be useful for better care.

1-Clinical trials of treatment with curative intent

Several authors (Cella 1995; Gridelli, Cortesi & Roila 2001; McMillan 1996) state in many cancer cases, such as malignant lymphoma or testicular cancer, the goal of therapy is to cure and to remove disease but even in these examples, there are treatment side effects. In this context, QoL can be affected and should be considered. Regarding another example, where two treatment options have equal impact on the survival rate, the final decision in choosing the best option, may be in favour of the one with better impact on QoL (de Haes & van Knippenberg 1985; Litwin et al. 1999; Maguire & Selby 1989; Strain 1990).

For example, Testa et al. (1993, cited in Fayers & Machin 2000), in their study on an evaluation of two anti-hypertensive drugs, found that the effect of drugs on the sampling population was the same in the terms of efficacy and safety, but QoL scores were different. The results were useful for selecting one drug. This research study is similar in its purpose with another one conducted by Palmer et al. (1980). In this research study, researchers aimed to select one chemotherapy regimen (one drug versus five drugs) suitable for breast cancer survival with metastasis. Patients reported worsened QoL related to multi-drug chemotherapy so it was useful for

selecting a one drug regimen.

In a study on 224 locally advanced breast cancer survivors, the effect of two chemotherapy regimens (systemic standard chemotherapy versus dose-intensified chemotherapy) on QoL was measured so as to select one regimen. Research findings showed in a three-month period, intensified treatment had a worse effect on QoL than the standard one, but the QoL scores were the same after 12 months. The authors concluded that dose-intensive treatment has a temporary impact on QoL and needs more research to assess this regimen (Bottomley et al. 2005).

In another study, two kinds of treatments (amputation plus chemotherapy versus limb-sparing surgery plus radiation therapy plus chemotherapy) were compared in their effect on QoL in 26 patients with soft tissue sarcoma. Research findings revealed that the QoL scores of patients under limb-sparing surgery plus radiation were not higher than those under amputation as researchers had assumed it would be (Sugarbaker et al. 1982). This can show how perceptions of other people about patients' QoL vary with that of patients' own perceptions.

In the above cases the selection of one treatment from two groups is relatively easy to identify as the impact of two treatments on the survival rate is relatively equal, but one treatment can lead to a better or worse QoL. However, when one treatment leads to an increase in survival but decreases the QoL or, vice versa, improves the QoL but decreases survival, selecting one treatment from the two is more difficult (Roila & Cortesi 2001). In these situations, there are two solutions. The first solution is to consult with patients about all positive and negative effects of different treatment choices and consider the patient's preferences. This option gives more weight to patients' perspectives and places the patients central to decision making. The importance of patients' preferences will be discussed further in this chapter. The second solution is to use measures in which quality and quantity (duration of survival) combined together like Quality Adjusted Life Years (QALY) and Quality-Adjusted Time Without Symptoms and Toxicity (Q-TWIST) (Mandelblatt & Eisenberg 1995; Roila & Cortesi 2001). In this option the attention is given to both patients' and health care professionals' perspectives; patients' perspectives about their QoL are assessed using utility QoL tools. Chapter Four will elaborate on the types of QoL tools that exist.

One of the most common measures of combining the quantity with quality is the QALY (Mandelblatt & Eisenberg 1995). The basis of this approach is that if two options are offered to individuals to choose either a longer life with serious problems or to have a shorter but more healthy and satisfactory life, a rational person would choose the second option (Hunt 1997). For a patient, for example, conducting a chemotherapy regimen might improve the survival rate up to 10 years but the associated quality of life using a utility tool might be 0.5 due to severe chemotherapy toxicities and the QALY of the patient is only five years ($10 \times 0.5=5$ QALY). This chemotherapy might be superior to another treatment with the utility of 0.4 due to more toxicities but with a survival of 11 years ($11 \times 0.4=4.4$ QALY) (Giesler 2000).

Q-TWIST is another extension of the QALY in which time and the patients' preferences are considered together (Mosconi et al. 2001). Giesler (2000) suggests that in order to calculate the Q-TWIST, survival time needs to be divided into three health states: (a) a period of time that patients experience toxic effects of therapy (TOX); (b) a period of time after completion of therapy without symptoms or toxicities (TWIST); and (c) a period of time after the relapse (REL). QoL during these times (except TWIST) are measured using a utility tool. TWIST receives a maximal utility of 1 and then Q-TWIST is calculated based on the following formula:

Q-TWIST= (utility value during the TOX \times TOX) + TWIST + (utility value during the TOX \times REL)

It can be concluded that QoL information has important implications for choosing a treatment option. Even so, other criteria such as survival or treatment side effects are also important; QoL information arising from patients is central to such decision-making. It is therefore essential to get patients' perspectives about their QoL because it can reflect their treatment choices. The situation is similar when the decision is made for palliative purposes.

2-Clinical trials of treatment with palliative intent

The cited QoL studies so far have had a curative purpose; however, for many

patients, particularly those with advanced cancers, the goal of therapy is palliation and not to prolong survival or produce a cure (Gridelli, Cortesi & Roila 2001; Horton 2002; Mandelblatt & Eisenberg 1995; McMillan 1996; Moinpour et al. 2000; Payne 1992; Perez et al. 1997; Slevin et al. 1988). The aim of treatment is to palliate symptoms or to prolong the time without symptoms and in these situations QoL is considered as an outcome for symptom evaluation and management (Brunelli et al.; McMillan 1996; Michael & Tannock 1998; Schipper & Levitt 1985). Many research studies exist that show how outcomes of QoL research studies are useful for choosing a palliative treatment.

In a study by Osoba (1999), the effect of two kinds of drug regimens (prednisone alone versus mitoxantrone plus prednisone) was used to measure pain and QoL endpoints of metastatic prostate cancer patients. Results identified that treatment with mitoxantrone plus prednisone compared with only prednisone was associated with lower levels of pain and better QoL in several QoL domains.

Coates et al. (1987) did research on metastatic breast cancer patients to compare two kinds of chemotherapy regimens (continuous chemotherapy versus intermittent therapy) with non-curative purposes. The outcomes, surprisingly, found patients under continuous chemotherapy have better QoL and better palliative results than those with intermittent chemotherapy. For such a cancer the goal of treatment is not curative. Therefore in this research the important criterion for selection was the better scores of one regimen on physical well-being, mood, pain, and appetite and a quality-of-life index.

In a study of chemotherapy as palliative treatment by Earl et al. (1991), 300 smallcell lung cancer patients were selected and randomised to receive either regular 'planned' or 'as required' chemotherapy. The chemotherapy regimen was the same in both groups. The planned group received chemotherapy on a regular basis and the 'as required' group received it based on factors such as disease symptoms. QoL and symptom assessment measurements were conducted for both groups. Despite the fact that 'as required' chemotherapy is considered by health care professionals as a unique way of using chemotherapy for palliative purposes, patients receiving this treatment assessed their QoL lower and experienced more severe symptoms than patients receiving 'planned' treatment. It was concluded that 'as required' chemotherapy is a less satisfactory treatment compared with planned chemotherapy for palliative purposes.

In another research study, Tannock et al. (1988) selected a chemotherapy regimen for metastatic breast cancer patients and used two different doses, initially using the low dose and for those patients failing to respond to this, a higher-dose was used. QoL was measured using linear analogue self-assessment scales. While, higher-dose chemotherapy made more toxicity immediately after treatment, the QoL of patients generally improved after this period. This trial suggests that using a full-dose chemotherapy regimen can improve QoL of patients and is useful for palliative purposes.

These cases mentioned above indicate how patients' ideas about QoL may be considered when a plan is made to palliate patients. Often patients' palliative choices differ from routinely used care plans.

3-Improving symptom relief, care or rehabilitation

In many cases QoL data was used to provide useful information to relieve specific symptom(s) such as fatigue, dyspnoea, pain, nausea, vomiting and to improve care and rehabilitation (Fayers, Curran & Machin 1998; Isikhan et al. 2001; Jacobsen, Davis & Cella 2002; Movsas 2003; Osoba 2000).

Langendijk et al. (2000) conducted research with the aim to assess QoL and respiratory symptom change in a group of 65 patients with metastatic non-small-cell lung carcinoma (NSCLC) before and after three episodes of thoracic radiotherapy. Patients reported the following QoL response rates: 1-excellent for haemoptysis (79%); 2-good for arm/shoulder pain (56%) chest wall pain (53%) and cough (49%); 3-moderate for dyspnoea (39%); and minimal for the general symptoms fatigue (22%) and appetite loss (11%). Global QoL improved in 37% of the cases. This study confirmed that the radiotherapy regimen was useful for palliation of respiratory symptoms and could improve QoL in this patient population.

QoL information can also be used by health care professionals involved in patient care or rehabilitation, for a wide range of issues including assessing physical and psychological problems or morbidities and to plan, evaluate and modify a specific program of care (Grant et al. 1990; Wright et al. 2003). For example, results of recent research study demonstrate that lung cancer patients benefit from psychological support at the beginning of their treatment, due to the disease's impact on the social and emotional part of their QoL (Lheureux et al. 2004). Consequently, it is clear that QoL assessment has implications for the development of a suitable psychological support program for lung cancer patients.

Results of a longitudinal research study by Sweed et al. (2002) on oesophageal cancer patients before and after oesophagectomy revealed that global QoL decreased slightly over time and the intensity of several symptoms including hoarseness, reflux and diarrhoea after oesophagectomy increased and led to a gradual decrease in the QoL scores. They also concluded that where a decrease in symptom intensity is observed, it is very important to institute measures to increase QoL and nursing care plays a major role in this effort. Another study conducted by He and Liu (2005) on nasopharyngeal cancer patients revealed that such patients benefited from education about the side effects of radiation therapy and needed to be supported in the use of appropriate coping styles by nurses.

Another study by Deschler et al. (1999) compared QoL assessment scores between cancer patients and their primary lay caregiver (spouse, child or friend) before head and neck surgery. Research findings identified that, except for some areas of differences, such as pain and general health, there was agreement between patients and the primary lay caregiver in several aspects of a given QoL tool. The researchers concluded that such concordance has implications for outpatient rehabilitation of cancer patients by their relatives.

4-Facilitating communication with patients

Fayer (2000) stated that the aim of QoL assessment in some situations is to provide information to be used in caring for future patients. Meyerowitz (1993) mentioned that QoL information indirectly informs physicians and nurses about the common
concerns and psychological reactions of patients and this knowledge can be used to anticipate the problems, to communicate better or to discuss issues further that otherwise may be overlooked when caring for this group of patients. Such information can be collected by filling out a QoL questionnaire when patients, for example, are waiting to see a doctor or questionnaires can be completed easily through a computer (Schwartz & Sprangers 2002).

Jacobsen, Davis and Cella (2002) also agree that QoL assessment can be helpful to identify problems that staff may be reluctant to ask patients about or patients may be hesitant to express directly to staff, such as sexual concerns. While health care professionals are constantly being challenged in communication with cancer patients to tell the truth about having a cancer in order to maximise patients' QoL, results of QoL assessment before and after cancer diagnosis on two groups revealed that, generally, knowledge about the cancer and its treatment does not significantly affect patients' QoL (Montazeri et al. 2004).

Velikova et al. (2004) explored the effects of regular assessment and use of QoL assessment in an oncology area on the process of care and patient well-being. From 286 cancer patients, a number were randomly classified into three groups. The first group consisted of those who completed QoL assessment and other relevant forms by computer and gave the results to oncologists. The second group completed the questionnaires but did not provide any feedback and finally there was a control group for whom no questionnaire was filled in. They concluded that routine assessment of cancer patients' QoL can improve communication between physicians and patients and is beneficial, particularly for patients who, as a result, had better QoL and emotional functioning.

5-Patient preferences

Perspectives, expectations, attitudes and preferences of patients in some situations may differ from those judgements by health care workers such as physicians or nurses and even from other groups of patients or healthy persons (Belcher 1990; Tamburini 2001; von Essen 2004). For this reason, QoL assessment can provide an opportunity to understand these differences more clearly.

Some research findings identified that many cancer patients rated their QoL relatively highly and they were generally satisfied with their lives (Bush et al. 1995; Rustoen et al. 1999; Waldron et al. 1999). Staff and relatives, on the other hand, tended to underestimate the patients' psychological problems, exaggerated physical symptoms, misjudged the patients' trade off between quantity and quality and underestimated the patients' QoL (King et al. 1995; Molzahn, Northcott & Dossetor 1997; Tamim, McCusker & Dendukuri 2002). In a study conducted by Slevin et al. (1990), for example, findings revealed that cancer patients generally accepted positively an intensive chemotherapy regimen with less benefit than did healthy people or attending staff such as doctors and nurses.

A study was carried out by Montazeri et al. (2003) to better understand QoL of lung cancer patients using three types of QoL questionnaires before and after diagnosis and between cancer patients who received the treatment and those who did not. This research study reported on different kinds of health problems that this group of cancer patients experienced after diagnosis of cancer. It was concluded that QoL information can enhance our understanding of cancer patients' experiences of their treatment.

Another study (Koller et al. 2000) where 55 cancer patients under radiotherapy were assessed to understand the relationship between therapy expectations and global QoL before and after radiotherapy, revealed three kinds of expectations including: (a) pain/emotional control; (b) healing; and (c) tumour/symptom control. Nineteen patients had curative treatment but 35 patients reported "healing" as an expectation. The more expectation there was of healing, the better QoL was observed. In the group as a whole, radiotherapy decreased patients' QoL only a little, but it substantially worsened the QoL of those patients who had an expectation of healing but perceived that this had failed. These findings indicate that in cancer patients, the expectation of healing is one major component of a good global QoL, whereas more limited expectations (pain control, tumour control) relate to lower QoL. Further research about patients' expectations and how it can improve the patients' care and rating of their own QoL was recommended.

An interpretive phenomenological study was conducted by McGrath (2004) to

uncover positive spiritual outcomes for survivors of haematological malignancies. Results revealed that those patients who gain successful treatment results and are well supported can experience positive outcomes such as greater confidence and independence, better capability to express their personal needs, being more informed of body needs, and being less judgmental. In general, such patients were mindful that the treatment was the gift of extra life. They wanted to experience life at the fullest level and to be able to do those things they were interested in doing. McGrath (2004) noted that during the cancer journey, patients also experienced a heightened sense of family togetherness, awareness of reliable friends and family members, respect from others, and changes in work values. These changes were associated with an overall improvement in QoL.

6- Late problems of psychological adaptation

Many of the physical and psychological problems for patients surviving cancer may be similar to those experienced by patients with chronic diseases and will be found to exist over a long period of time (Aziz 2002; Movsas 2003). QoL information may help health care staff to understand these problems more clearly (Schwartz & Sprangers 2002; Thapa & Rowland 1989). QoL assessment may lead to better understanding of some aspects of patients' QoL, including psychosocial problems and pain intensity, if patients' QoL can be followed up over longer time periods by caregivers (Byar, Eilers & Nuss 2005). For example, an eleven year follow-up of long-term survivors of testicular cancer was undertaken in order to evaluate their QoL using two QoL tools and to assess social problems. A comparison was made between this group (case group) and a healthy control group. Generally speaking, no significant difference was identified between QoL scores of the case and the control groups. The study also revealed that, except for patients' sexuality problems, patients generally did not have a greater range of associated health problems than the 'healthy' population.

Bush et al. (1995) conducted research to assess the QoL and the health problems of 125 adults surviving 6–18 years after bone marrow transplantation. Research findings revealed that, except for a very small proportion, long-term survivors generally reported full and meaningful lives, with no major continuous

complications, and viewed themselves as cured and well.

7-Medical decision making

QoL information is valuable when making medical decisions (Belcher 1990; Strain 1990). For example, QoL scores in cancer patients can be a strong predictor of survival (Blazeby, Brookes & Alderson 2001; Butow, Coates & Dunn 1999; Montazeri, Gillis & McEwen 1998; Montazeri et al. 2001; Osoba 1999; Schwartz & Sprangers 2002; Shadbolt, Barresi & Craft 2002; Sprangers 2002), particularly the physical aspect of QoL assessment, which has been a major predictor of survival in cancer patients (Arraras et al. 2002; Cassileth et al. 1985; Cassileth, Walsh & Lusk 1988; Chang et al. 1998; Vigano et al. 2004).

There are many documents that emphasise the strong relationship between QoL scores, irrespective of other values, with survival rate. For example, a study conducted by Coates, Porzsolt and Osoba (1997) assessed the prognostic value of QoL scores in a group of advanced cancer patients. Results revealed that QoL scores carry prognostic information independent of other recorded factors. The value of QoL scores as a predictive factor for survival was also confirmed for breast cancer patients after disease relapse (Coates et al. 2000), with metastatic malignant melanoma (Coates et al. 1993), lung cancer (Ruckdeschel & Piantadosi 1994) and colorectal liver metastasis (Earlam et al. 1996). Therefore, QoL information arising from research studies has several implications. Such data will be useful in selecting an efficacious treatment incorporating curative and palliative intentions, and in improving communication with patients and understanding of their preferences.

This study also assesses the influence of some important variables on the nurses' understanding of patients' QoL. This is an important aspect of exploring why patients and nurses may differ in their perceptions. Research studies in which the influence of variables on the level of agreement on QoL were assessed are very limited, particularly those with nurses as proxies, oncology patients, and with interpretive methodologies. The relevant literature therefore was reviewed to address this issue.

Variables influencing agreement between patients and nurses about cancer patients' QoL

It is important to recognise that nurses' perceptions of cancer patients' QoL may differ from patients' own perceptions for a number of reasons. Important variables might be related to patients (such as age, sex, and education), proxies (such as age, sex, education, and time spent with subjects) and other general issues (such as QoL domains, QoL assessment) (Addington-Hall & Kalra 2001). These important issues affecting the level of QoL agreement will be explored further in the following subsections.

Patients' and proxies' demographics

It might seem obvious that different variables such as age, gender, marital status, and educational level might influence agreement between patients and proxies (McPherson & Addington-Hall 2003). However, previous research findings led to inconsistent outcomes when effects of such characteristics on agreement were explored.

For example, variables including age, gender, socioeconomic status, and marital status of patients have no effect on the level of agreement between patients and proxies about QoL (Epstein et al. 1989; McCusker & Stoddard 1984). Both Epstein et al. and McCusker and Stoddard used QoL tools other than the WHOQoL-BREF questionnaire and sample populations were less than 100, which is generally considered to be a small sample size. Larsson, von Essen and Sjoden (1998) compared the perceptions of seventeen patients with endocrine cancer and their staff by using the European Organization for Research and Treatment of Cancer (EORTC) QoL questionnaire. Findings from their study revealed that no specific pattern for the effect of patients' and staff members' demographic variables regarding agreement could be identified.

The same outcomes emerged when QoL of 103 cancer patients was compared with QoL perception of their significant others using the EORTC QoL questionnaire. No significant relationship was found between a range of patients' and proxies' background characteristics (such as age, gender, proxies' relationship with the

patient, and the living arrangement in relation to the patient) and the level of agreement between them (Sneeuw et al. 1997).

Brunelli et al. (1998) also conducted a research study to measure agreement between 159 terminal cancer patients and proxies using the Therapy Impact Questionnaire (TIQ). They demonstrated that none of the demographic factors for patients or proxies have a statistically significant effect on the level of QoL agreement between the two. Even though the patient sample size is not small, they do not state how many proxies took part in the study. However, other studies have led to conflicting results. In Sneeuw et al.'s (1998) analysis, the extent of agreement between 307 cancer patients and 224 proxies was compared using the EORTC QLQ questionnaire. In this research study several demographic variables of cancer patients and their significant others were found to be associated with the level of agreement. However, using multivariate analysis these factors together explained only less than 15% of the variance in patient-proxy differences (Sneeuw et al. 1998). This research study used a large sample size for both patients and proxies and had more precision to identify the influence of several variables on agreement.

Another research study was conducted by Kristjanson et al. (1998) in an Australian population. The two main objectives were to assess the level of agreement between terminal cancer patients and family caregivers about symptom distress and to identify the effect of demographic and disease and treatment related variables on the level of agreement. 78 dyads of patients and caregivers agreed to participate in the study. The outcomes identified a reasonable agreement between proxies' ratings of symptom distress and cancer patients' rating. These results have also been reported to be similar to those of a Canadian research study conducted by Lobchuk et al. (1997) and confirm that family members may provide reasonable proxy ratings of advanced cancer patients' symptom distress. A few demographic variables, such as marital status, gender of the patient, age of the family caregiver, the relationship between the patient and family caregiver and family income, were also identified to affect the level of agreement. This research study assessed agreement on a part of QoL (symptom distress) and was conducted on a small sample.

Relatively similar outcomes were explored in another research study in which cancer

patients' and caregivers' (a family member or a nurse) assessment of symptom distress were compared with each other. While no statistically significant relationships of patients' demographics of sex, age and education level and agreement were found, non-statistically significant tendencies were found for some caregiver demographics such as education, professional experience, and family care giver burden (Broberger, Tishelman & von Essen 2005).

The influence of educational level was also identified in another proxy research study conducted by Muhlenkamp (1986). In this research study, agreement between 30 hospitalised arthritis patients and 26 registered nurses on affective states (hostility, anxiety, and depression) was assessed. One of the main findings of the study was that agreement was better when nurses had a higher level of education. In another study, the level of agreement between 231 elderly patients and their primary care givers was assessed using the Euro QoL (EQ) questionnaire. Amongst the demographic variables, greater agreement was found between patients and proxies when the proxy's native language was English (Tamim, McCusker & Dendukuri 2002).

Research conducted to explore the influence of demographic variables on the level of agreement about QoL suggests that outcomes are varied and at times contradictory. No general trend can be identified that shows that particular variables influence the level of agreement in a specific direction. It appears that effects of such variables need to be explored further, particularly using larger sample sizes. An important issue is that a small sample size can lead to biased outcomes (von Essen 2004), particularly when samples are about 50 or less in each group of patients and proxies (Sneeuw, Sprangers & Aaronson 2002).

Patients' health status

The effect of patients' health status on patient-proxy QoL agreement is also an area of debate (McPherson & Addington-Hall 2003). In some studies looking at proxy rating of cancer patients, the degree of QoL agreement was influenced by the clinical and performance status of the patient as an indicator of health status. For example, a research study was conducted to evaluate the QoL rating amongst 90 cancer patients

and their significant others, physicians and nurses on a given QoL tool. Part of the outcomes indicated that agreement was better when patients have a very good or poor performance status (Sneeuw et al. 1999). Performance status using Eastern Cooperative Oncology Group (ECOG) shows how the disease is progressing, how it is affecting the daily living abilities of the patient, and determines appropriate treatment and prognosis. However, in another study no relationship was found between cancer patients' and significant others' agreement and overall performance status of patients (Sneeuw et al. 1997). So it is still necessary to investigate how indicators of patients' health status like patients' performance status can influence the level of agreement between patients and proxies.

Contact time between patients and proxies

It is generally accepted that more contact time between the proxy and the patient, results in better agreement in QoL ratings (Lobchuk & Degner 2002; Magaziner et al. 1988; Tamim, McCusker & Dendukuri 2002; Tang & McCorkle 2002b; von Essen 2004).

Steel, Geller and Carr (2005) found that agreement between cancer patients and caregivers including a family member or a physician was generally better after three months compared with the baseline assessment. Proxies had more contact with patients during this period of time and could better monitor changes in the patients' QoL. This effect can also be identified in another research study in which agreement between patients and proxies was better in a four month follow-up compared with the baseline due to spending more time with patients (Tamim, McCusker & Dendukuri 2002).

While the duration of contact between patients and nurses is a key issue in agreement, how proxies use their time in getting to know the patient may be most important. For example, a proxy research study was conducted by Broberger, Tishelman and von Essen (2005) to explore discrepancies between lung cancer patients and their caregivers (nurses and family caregivers) in assessment of symptom distress and symptom occurrence. Thirty three patient–nurse dyads and 54 patient–family caregiver dyads assessed patients' symptom distress and symptom

occurrence. In this research study, instead of asking about the length of time the proxy had known the patient as an indicator, nurses were asked to answer the question "How well do you know this patient?" Results demonstrated that nurses knew two patients very well, 26 patients well or fairly well, and five patients not at all. The result indicates that caregivers, including nurses, might spend a period of time with patients but not know them in-depth or well.

Another issue is that spending more time with patients might also lead to disagreement in some situations. In Magaziner et al.'s (1988) research study, agreement between 361 elderly hip fracture patients and their proxies was assessed on measures of patient health and functional status. The outcomes indicated that proxies with more contact with patients had better agreement with patients but when they did disagree with each other, proxies with the greatest patient contact tended to overestimate patient disability (Magaziner et al. 1988).

It can be concluded that when nurses spend enough time with patients they may better understand patients' QoL. In this research study, the effect of the time each nurse spends with cancer patients on their understanding of patients' QoL was examined.

Proxies' burden

As proxies usually are in a close relationship with patients, they are susceptible to many stressor burdens that can affect and disrupt their own QoL (Edwards & Ung 2002; Ergun, Oran & Bender 2005; Meyerowitz 1993; Tang & McCorkle 2002b). It is argued that such stressors in turn can influence their evaluation of the patient's QoL. For example, Sneeuw et al. (1998) found that QoL disagreement between cancer patients and significant others increased when caregivers had a higher level of intensity in their care-giving function.

In another research study, the effect of some demographic and clinical factors related to terminally ill cancer patients and family caregivers on the level of agreement in their rating of patients' QoL was assessed. As a part of the research study the caregiving burden was calculated using the amount of care needed and the impact of care giving on the caregiver's health. The results indicated that the care-giving burden has a negative influence on the level of agreement between patients and families about cancer patients' QoL (Tang 2006). This is an outcome that was present in another study where the level of agreement between 91 patients with dementia and their family caregivers about patients' QoL was assessed. Caregivers who reported higher levels of burden rated patients' QoL lower than did patients in all five domains of QoL (Sands et al. 2004).

Altogether it appears that nurses' assessment of their own QoL and other related issues, like their burden of care, can influence their understanding of patients' QoL. In this research study, nurses' QoL was assessed and then the influence of nurses' assessment of their own QoL on the level of agreement between cancer patients and nurses about patients' QoL was examined.

QoL domain

It is stated by many researchers that there is better QoL agreement between patients and proxies about patients' QoL when more concrete, objective and less private aspects of the QoL assessment such as the physical domain are explored (Addington-Hall & Kalra 2001; Broberger, Tishelman & von Essen 2005; Clipp & George 1992; Dorman et al. 1997; Lobchuk & Degner 2002; Magaziner et al. 1988; Magaziner et al. 1997; McPherson & Addington-Hall 2003; Pickard et al. 2004; Sneeuw et al. 1999; Tamim, McCusker & Dendukuri 2002; Tang & McCorkle 2002b; von Essen 2004; Wilson et al. 2000). Care providers usually focus more on the physical problems of patients and they have less accurate information concerning the psychosocial and social aspects of patients' QoL, like personal relationships (Brunelli et al. 1998; Stromgren et al. 2001). This research study assessed QoL across four domains and it is interesting to consider the extent to which the findings support this trend.

In summary, the outcomes of the literature review in relation to variables influencing the level of agreement between patients and proxies about QoL identified several issues. Firstly, the majority of research studies were empirical and assessed the influence of only a limited number of patients' and nurses' demographic and clinical variables on the level of agreement. Because variables influencing agreement are potentially very diverse, more complementary interpretive approaches are necessary to explore the whole range of possible variables. Interpretive research studies can also provide a more in-depth understanding as to why these variables are important in nurses' understanding of cancer patients' QoL.

Secondly, outcomes of the empirical studies in which the effect of clinical and demographic variables on the level of agreement have been tested generally show different and diverse outcomes. This means that the effect of these variables on the level of agreement still needs to be explored further. Finally, empirical research studies in which the influence of nurses' variables on the level of agreement was examined are quite limited. An improved understanding of the ways in which nurses' characteristics might facilitate nurses' understanding of cancer patients' QoL would be potentially very useful.

Conclusion

In this chapter, the main historical and developmental phases or characteristics of the QoL concept were described, and how the QoL is defined in terms of various domains/dimensions. While different QoL definitions have been identified, it is clear that QoL is an individual and broad concept. It is argued that while some commentators recommend using the HRQoL assessment in the medical area, full assessment of QoL is a more suitable approach and will be used in this research study. QoL has been introduced into the oncology area as a way to pay attention to patients' rights and preferences and to ensure that health care professionals consider all aspects of individuals' situation when planning their care, including physical and psychological parts.

QoL information has many applications in the clinical area. For example, QoL information can be useful for determining if a curative or palliative treatment is most suitable for patients. QoL information is very useful in: understanding specific symptom(s) and enhancing care and rehabilitation; improving communication with patients; providing a better understanding of patients' preferences; identifying longer

term problems of patients; and in supporting better decisions about patients' care plans. The next chapter details the methodological and philosophical issues underpinning the research process.

CHAPTER THREE: METHODOLOGY

Introduction

In the previous chapter, the literature in relation to nurses' perceptions about cancer patients' QoL was reviewed. It was emphasised that if nurses can accurately assess cancer patients' QoL, it can potentially lead to useful intervention, particularly for the clinical area. However, a number of variables (such as age and marital status) might influence the level of agreement between cancer patients and nurses about cancer patients' QoL which will be examined in the first phase. Other factors (such as how participants understand the meaning of QoL concept) can also influence nurses' understanding of cancer patients' QoL and will be explored in the second phase.

The purpose of this chapter is to present the philosophical and theoretical issues that underpin this research study comparing nurses' and patients' perceptions about cancer patients' QoL. This discussion, in turn, underpins the methods chosen for the study. These philosophical and theoretical issues are also known as a 'paradigm' which mainly consists of methodology, epistemology and ontology (Finlay 2006). While the concept of 'paradigm' is used widely, it is considered as an unclear and ambiguous term by some commentators (Wainwright 1997). Conversely, the term 'worldview' is an easier term to understand (Weaver & Olson 2006) and therefore is preferred for this research study. Because different meanings and interpretations have been suggested about the concepts of ontology, epistemology, and methodology, it is very important to know what these concepts mean more precisely in the context of this thesis.

Guba (1990, p.18) highlighted the notion that ontology is about "what is the nature of the knowable?" or "what is the nature of reality?", epistemology is about "what is the nature of relationship between the knower (the inquirer) and the known (or knowable)?" and methodology focuses on "the best means for gaining knowledge about the world". Therefore, choosing a methodology means that researchers take into consideration both questions of epistemology which talks about how the

knowledge will be created and ontology which demonstrate the way we think about the nature of the world and of our being.

Ontology is relevant to choosing a suitable methodology or perspective for the research project because if researchers assume a reality 'out there' (realism), they might choose an empirical approach which allows them to focus on the reality as precisely as possible by controlling possible confounding variables (Guba & Lincoln 1994). Conversely, researchers with a relativist ontological view accept different meanings and interpretations about reality and therefore might adopt an interpretive perspective.

Similarly, epistemological positions that researchers bring to the research endeavour influence the selection of methodology (Weaver & Olson 2006). Reality might seem to be out there and quantifiable (objectivism) compared with those who believe that separation of meanings from objectives is impossible (subjectivism). The former might select an empirical methodology whereas the latter would adopt an interpretive approach (Guba & Lincoln 1994). More discussion about the ontology and epistemology and their relationship with the methodology can be found later in this chapter.

The thesis as a whole has a mixed approach which incorporates empirical and interpretive ideas in two different phases. Therefore, this chapter explains why these two perspectives are used together and what this can add to the strength and usefulness of the research study. The methodology and related ontology and epistemology underpinning the first phase are discussed. For the second phase, why an interpretive approach is used will be justified and then why, from a variety of options within this approach, a grounded theory approach was chosen.

The methodology of the whole thesis: a mixed research approach

Given that several terms have been used to name research studies with mixed methodologies or mixed methods (Creswell 2003; Sandelowski 2000a; Teddlie & Tashakkori 2003), the term 'mixed research approach' was developed. It was

concluded that this term is more useful and can include either a study that is mixed at a method level or at a methodological level. The terms 'mixed method' or 'mixed methods (with plural s)' which are suggested in the literature seem to be confusing for the researcher because they imply that the combination is only at the method level.

There are several reasons why a mixed research approach was chosen as a framework for the research. Firstly, a number of precursor empirical research studies assessed the level of agreement between cancer patients and proxies about cancer patients' QoL and explored variables influencing this agreement (Sands et al. 2004; Schwenk et al. 2004). However, the outcomes of research studies assessing the level of agreement vary across studies and the variables that have been investigated are very diverse. Moreover, some of the empirical studies that consider variables influencing the level of agreement between patients and proxies are limited because they do not canvass all important variables.

It was concluded that an empirical study needs to be conducted to verify or confirm the relationship(s) between the level of agreement between cancer patients and nurses in an Australian context. In addition, an exploratory research study was required to explore in depth the interactions between cancer patients and nurses that might result in differences in nurses' perceptions of cancer patients' QoL. A mixed research approach is very useful in such situations because it allows different kinds of questions to be addressed. Teddlie and Tashakkori (2003, p.15) suggested that a mixed research approach is very useful when researchers want to "simultaneously answer confirmatory and exploratory questions and therefore verify and generate theory in the same study". This research study has both confirmatory (the level of agreement between patients and nurses about cancer patients' QoL) and exploratory questions (why perceptual differences exist).

The second reason why a mixed research approach was chosen is that this study deals with the complex issue of QoL. Nurses' perceptions of cancer patients' QoL is a complex issue and in this study such perceptions are seen from different perspectives using empirical and interpretive approaches. Empirical enquiries usually gather the information from a large number of participants and make interpretations of quantifiable data. Conversely, interpretive enquires usually deal with experiences of a limited number of participants in order to have a richer understanding of the phenomenon. These enquiries together can increase the breath and depth of the research study. Therefore, it is most likely that complexities of QoL perception will be understood more fully. Sandelowski (2000a) also argues that a mixed research approach allows researchers to better understand the complex issues like health or QoL.

The final reason for choosing a mixed research approach is that conclusions made from such studies may have more generalisations compared with those emerging from either qualitative or quantitative data alone. The reason is that experiences of participants related to nurses' understanding of cancer patients' QoL in response to the WHOQoL-BREF questionnaire, are compared with their subjective experiences determined through their responses to interview questions. Therefore, it is most likely that more consistent outcomes can be revealed by comparing quantifiable and qualitative data that are also useful and applicable to other situations and contexts. Teddlie and Tashakkori (2003, p.37) suggested the term "inference transferability" in mixed research approaches that covers both the quantitative term of "external validity" and the qualitative term of "transferability". Teddlie and Tashakkori (2003, p.42) opined that mixed research approaches can potentially lead to more 'inference transferability' due to "the *gestalt principle* that the whole is bigger than the sum of its parts" (Teddlie & Tashakkori 2003, p.42).

To be precise, a mixed research approach provides researchers with this opportunity to get an in-depth understanding of complex issues and questions (Williams-Cox 2004) and potentially can lead to more rigorous findings (Smith & Biley 1997). The researcher would suggest that other researchers, particularly nursing PhD students, deal with complex issues related to health by incorporating mixed research approaches into their research studies.

Despite the importance of studies with a mixed approach, a number of debates exist as to what philosophical and theoretical issues underpin a study with mixed approach. Teddlie and Tashakkori (2003, pp.17-22) mentioned that previous researchers using mixed research approaches, have taken the following main positions. The first position adopted is by those researchers who believe that mixed research approaches do not need to justify their worldview. A research study can be conducted provided that methods are chosen that best answer the research questions. Another option is introduced by those researchers who argue that conducting mixed research approaches are not possible because researchers cannot adopt two different perspectives or worldviews in one study. This option has been criticised by others arguing that different worldviews can be used in the same research study and even researchers might welcome opposite worldviews in one research study. Finally, there are researchers who are of the idea that a mixed study can use only one worldview. These latter ones mainly proposed the pragmatic worldview because mixed research studies can find practical answers for research questions.

Having considered the above options, it can be argued that like many other research studies, this research study uses two different worldviews in two separate phases of the research study. A single worldview did not select for this research study because while investigating the theoretical and methodological issues, no phase was dominant over another. Therefore, because each phase of the research study has its own theoretical and philosophical issues, in the following sections the theoretical and philosophical issues of this study are discussed separately.

The methodology and the philosophical issues underpinning the first phase

In the first phase of the research study, the main aim was to understand the degree or level of agreement between patients and nurses about cancer patients' QoL. It was realised that numerous empirical studies measuring this agreement in a population other than Australia needs to be extended into the Australian context due to the importance of QoL perceptions. Given that QoL is a culturally bonded concept, this could provide a number of Australian patients and nurses with the opportunity to reflect on their experiences and further empirical comparisons amongst such experiences can be made.

When this study began the researcher reflected on his worldview and recognised that he only conducted empirical studies and he is more or less an 'empirical person'. In other words, he was a person who saw the reality out there as being more quantifiable and objective. However, he also acknowledged that there might be other meanings and interpretations of reality and therefore he was interested in changing his worldview to a more balanced view gradually. This required the researcher to explore alternative methodologies.

The researcher was also new to the research environment and starting with an empirical study allowed him to become more familiar with the research area, participants and constructing the research questions for the interpretive phase. Creswell et al. (2003, p.277) argue that conducting an empirical study followed by an interpretive one can be potentially very useful for "quantitative researchers" who are unfamiliar with interpretive research studies. They stated that the quantitative phase allow researchers to familiarise themselves with interpretive studies while they start the empirical phase.

Therefore, in the first phase an empirical study was conducted which is based on the principles of a positivist worldview. It is quite important to consider this worldview in more detail to see on what grounds the first phase of the research study has been established.

Positivism

The first phase of the research study was based on the positivist view of knowledge. It is quite a rigorous worldview for research studies and has been developed over a long period of time (Lincoln & Guba 2000).

The first generation of positivists can be traced back to the 18th and 19th centuries when peoples' thinking about life moved away from traditional religious interpretations and focused on human experiences (Crossan 2003). Comte, the French philosopher, for instance, used the word 'positivist' for the first time and influenced social research in the twentieth century (Lewes 1853). As a result of this generation, more attention was given to the importance of investigation based on evidence, particularly that emerging from direct observation and away from the oppressive authority of religious belief systems. The second generation is related to

many philosophers in the early 20th century who are collectively named the Vienna Circle (Crossan 2003). These philosophers believed that experience is the only source of knowledge but such experience needs to be studied using logical inductive thinking (Suppe 2001). Logical inductive thinking will help researchers to draw meaningful and less speculative conclusions from data particularly when researchers study complex concepts associated with human health like QoL.

The next generation was developed by Karl Hempel in the post-war period (Crossan 2003). Hempel (1965) preferred to use the term empiricists for himself and similar philosophers rather than positivist and suggested that researchers need to have a focus on reasoning based on logical conclusions through deductive thinking rather than pre-accepted assumptions. The major feature of this generation appears to be that researchers use general principles to predict relationships between facts and then conduct research studies to see if these predictions are correct or not.

In brief, these generations of positivist thinkers developed important concepts including evidence-based inquiry, inductive and deductive thinking that can be used in this research study. However, these important aspects or figures of the positivist worldview can also be classified in another way. For example, the well-known classification is that the positivist approach focuses on realism in ontology, objectivism in epistemology, and is experimental regarding methodology (Guba & Lincoln 1994). In the following sections, these major elements of the positivist approach will be explained in more detail. While issues related to positivism are open to debate, the major characteristics of positivism are adapted from an often cited publication written by Lincoln and Guba (2000).

Positivism and the issue of realism

The ontology of the first phase of the research study is based on realism because patients and nurses will report on their experiences by answering a QoL survey tool that translates the internal state of QoL to QoL scores. The researcher assumes that this reflects the real state of QoL. Lincoln and Guba (2000) point out that in the positivist view, the reality is out there and the role of researchers is to find that truth. They mention that the knowledge emerging from such findings will be summarised in the form of time and context free generalisations that can be used to predict and control natural phenomena. In other words, in this worldview reality is seen as being external to researchers (Grbich 1999) and in fact things are experienced as they exist (Wainwright 1997).

Positivism and the issue of objectivism

The epistemology of the first phase of the research study is based on objectivism because the state of QoL has been measured by a QoL tool leading to quantifiable data which is then analysed by statistical tests. The assumption here is that QoL is accessible from the "outside" of a person when the inner experience is operationalised. Lincoln and Guba (2000) highlight that in a positivist worldview researchers stand in a distant position and see the natural situation without altering it, which can potentially exclude any confounding or biasing variables from the process of questioning. In other words, this epistemology accepts the possibility of making objective observations or measurements (Gerber & Moyle 2004).

Positivism and the issue of experimentalism

The methodology of the first phase is derived from the ontological and epistemological assumptions described above. An objectivist/positivist approach requires tools that allow accurate measurement of an empirical reality. This led to the choice of empirical experimentalism where patients and nurses were invited to answer a QoL questionnaire with prepared QoL questions to assess the state of QoL. Moreover, in order to control the situation (another feature of empirical experimentalism), there are inclusion and exclusion criteria. For example, participants are confined to only adult patients and nurses and they are asked to report on their QoL scores based on their experiences over the last two weeks. Guba and Lincoln (1994) assert that while researchers try to ask questions and to exclude any confounding variables, it is possible for them to be biased because nature is complex and tends to be confounded. Therefore, researchers prepare their questions in advance in a propositional form and nature also carefully verifies the hypothesis or to replicate findings.

In summary, the positivist worldview underpinning the first phase of the research

study is based on acceptance of the fact that knowledge has the following characteristics:

- is discovered, i.e. universal and external truths are grasped and justified
- arises from empirical processes which are reductionist, value neutral, quantifiable, objective and operationalizable statements are valid only if publicity verifiable by sense data.

(Higgs & Llewellyn 1998, p.61)

Given the fact that this thesis uses a complementary approach, the philosophical issues underpinning the second phase of the research study are discussed below.

The methodology and philosophical issues underpinning the second phase

An empirical study based on principles of a positivist approach is useful in guiding the first phase of the research study because it allows a large number of participants to report on QoL questions that express their internal state of QoL. However, the researcher was also interested in providing more opportunity to participants to give meanings and interpretations to QoL issues. This was done in the second phase of the research study using an interpretive study. Liamputtong and Ezzy (2005) emphasise that in contrast to the empirical studies that might test a specific theory or verify relationship between variables, in interpretive studies participants interpret and give meaning to events and things.

As well as participants' interpretations, the researcher also wanted to have in-depth descriptions of QoL issues all through the data collection and analysis rather than making interpretations of final data as it can be identified in empirical studies. In other words, an interpretive approach allowed the researcher to take a more active role and interpret experiences of participants as he was conducting interviews. This does not mean making interpretations that are not rooted in the actual data but rather to interpret the data more deeply. DePoy and Gitlin (1998) suggest that in interpretive studies, investigators work from the beginning to the end of the study process in a very active way to find the truth in natural situations.

Consequently, knowledge arising from the interpretive worldview underpinning the second phase of the research study has the following characteristics:

- comprises constructions arising from the minds of knowing, conscious and feeling beings
- is generated through a search for meaning, beliefs and values, and through looking for wholes and relationships with other wholes.

(Higgs & Llewellyn 1998, p.61)

Interpretive studies have a varied range of methodologies such as ethnography, grounded theory, and phenomenology. A couple of more recent methodologies also emerged in nursing named under titles such as 'descriptive, exploratory' (Annells 2007), 'descriptive' (Sandelowski 2000b), and 'interpretive descriptive' (Thorne, Kirkham & MacDonald-Emes 1997). It is important to reflect on the difficulties the researcher experienced in determining the best methodology to use. There is overlap between and divergent ideas and even definitions that are used by researchers when they describe methodology. For example, a researcher might label a research study as a grounded theory study whereas another might call it descriptive, interpretive, or descriptive-exploratory. It is not surprising to see these divergent views and controversies because interpretive studies are still in the early stages of development. Jeon (2004) also highlights that finding an interpretive methodology that can best answer the research question is difficult.

However, the researcher decided to base the second phase of the research study on the principles of grounded theory method, more specifically a classical version of it. The following section justifies the methodology chosen to underpin the second phase.

The rationales for choosing processes and principles of grounded theory

The following reasons are offered to justify why this research study was based on the principles of grounded theory.

1-Finding conceptual categories that inform practice

The researcher used 'the principles of grounded theory' to show that it is a modified version of grounded theory. This study presents only descriptions and a conceptual ordering and not an explanatory schema. In other words, this thesis is not concerned with developing a theory as has been done in other grounded theory studies. Annells

(2007) pointed out that many researchers in different areas of nursing have used grounded theory methods at two levels of analysis, being description and conceptual ordering, rather than seeking to develop a substantive theory about action and interaction.

The aim of this thesis is to obtain a fundamental understanding and description of the experience of QoL and differences between patients and nurses. Annells (2003) suggested that researchers might use some aspects of grounded theory, like coding up to a conceptual ordering level, and this is helpful in obtaining a basic understanding of the research area. Strauss and Corbin (1998, p.288) in their response to students' questions also point out that researchers can use some of their procedures in the studies aiming for "description, conceptual ordering, or discovery of categories to build measurement scales" rather than to build a theory. However, it can be argued that grounded theory studies with coding up to a conceptual ordering level are also useful for those researchers who are not able to develop a full theory due to their time limitation and availability of resources – both of which were factors in the current research project.

2-Lack of knowledge about the reasons, outcomes, and implications of agreement

Review on the literature identified that generally there is a lack of research in which nurses' perceptions about cancer patients' QoL have been explored in depth. Grounded theory is especially very useful for understanding those situations involving people about which very little is known (Jeon 2004; Stanley 2006). Other interpretive approaches are also helpful to uncover knowledge but they might open the door for interpretations that are less supported by the actual data. A research study based on a grounded theory approach, particularly a classical version of it, tries to develop knowledge on the basis of information gathered from exploring the issues with participants (Auerbach & Silverstein 2003). Grounded theory then requires the researcher to rigorously link interpretive conclusions with the actual data.

3-Focus of the research study on patients-nurses interactions

The main aim of the study is to know how nurses perceive cancer patients' QoL

when they interact with cancer patients in the oncology wards. What is going on between cancer patients and nurses that makes a difference in QoL perceptions? It is consistent with a grounded theory approach because it also focuses on actions and reactions (Annells 2006). Grounded theory is based "on the process of social interactions by which individuals make sense of the world" (Stanley 2006, p.64). It is quite important to briefly review interactionism because it is the main philosophy underpinning grounded theory and in turn the second phase of the research study.

Interactionism

The history of interactionism can be traced back to the American social psychologist George Herbert Mead who taught this concept at the University of Chicago (Morris 1977). Mead was in a close relationship with pragmatists like Dewey and he was greatly influenced by pragmatists (Morris 1962). Pragmatists generally think that individuals need to adapt themselves to social changes and the human mind makes this process possible (Jeon 2004). In other words, individuals through purposeful thinking can find suitable ways to answer their research questions or solve their problems. Creswell (2003) expresses his idea that pragmatism will give individuals a sense of freedom for choosing alternatives that meet their needs or expectations. Thinking in a pragmatic way can be seen in Mead's later publications in which he highlighted the importance of the mind.

In the introduction to Mead's *Mind, Self and Society,* Morris (1962) clarified Mead's beliefs. Morris points out that "mind and self are, without remainder, generated in a social process, and he [Mead] has for the first time isolated the mechanism of this genesis" (p. xv). This implies that the self can be understood only in interaction with other people and society because nobody lives in isolation. As individuals communicate with each other, they might accept that other people's beliefs and attitudes about them are correct and relevant. These attitudes and beliefs might shape how people think about themselves and in turn they might reflect in a certain way.

Similar interpretations of Mead's view about the relationship between the self and society have been made by other researchers. For example, Annells (1996) pointed out that the main point of Mead's thinking was "the essential defining of self through social roles, expectations, and perspectives cast on self by society and by those

within society. He argued that humans come to understand collective social definitions through a socializing process" (pp. 380-381). In other words, as individuals are in a continuous interaction with each other in the context of a complex society, they learn about their social roles and mutually they can also influence the society.

Mead's beliefs about the social process, the importance of self and mind were titled symbolic interactionism by one of his students (Blumer 1969). According to Blumer "what makes human beings unique is their symbolic ability to define their situations and shape their actions" (Wilson & Hutchinson 1991, p.267). So, in each society individuals develop a number of symbols such as words either for speaking or writing and non-verbals cues like gestures which are meaningful for them and facilitate interactions between them.

Although symbolic interactionism influenced a number of fields such as philosophy, psychology and sociology, one of the major goals of Blumer in developing Mead's philosophical work and creating the symbolic interactionism approach was to improve research practice (Blumer 1969). Wilson and Hutchinson (1991, p.267) argue that "research based on symbolic interactionism emphasises how people view their circumstances, how they interact, and how these processes change. One such research strategy is termed grounded theory methodology".

In summary, symbolic interactionism is based on the notion that meaning can be established through people's interaction. This is quite a supportive statement for choosing grounded theory as a methodological basis for this research study because nurses make perceptions about cancer patients' QoL when they interact with patients. Patients' signs and symptoms, verbal and non-verbal cues will help nurses to develop perceptions about patients as they interact with them. Nurses' perceptions about patients as they interact with them. Nurses' perceptions about patients think about their own QoL. Any issues that influence the way patients the workplace can also affect nurses' understanding of patients' QoL and need to be taken into consideration. Symbolic interactions provides a framework for understanding these aspects of the nurse-patient relationship.

4-To incorporate findings of the first phase into the construction of the second phase

Another advantage of using a grounded theory approach is that it allows researchers to incorporate empirical data from the first phase into the development of findings of the second phase. Grounded theory allows any form of collected data including survey, experiment, and case study to be combined and integrated (Glaser 1978). If researchers choose an interpretive approach that is not congruent with the empirical one and does not accept the quantitative data, it is most likely that researchers feel frustrated and can not synthesise any ideas from the divergent data. Conversely, two methodologies agreeing with each other provides researchers with more consistency in inferences and the conclusions they make. Annells (2006, p.59) also highlighted that without congruency in approaches that researchers choose, they may feel frustrated and "becoming lost in multiple fractures which would inevitably occur when trying to do the research".

5- Researcher's philosophical view and background

The more the researcher read and reflected he found that his views reflected a more realistic and objective perspective. In the ontology's continuum ranging from realist to relativist (Lincoln & Guba 2000), the researcher's view was near to the end of the continuum of the realist aiming to move toward the relativist. In the epistemology's continuum ranging from objectivist to subjectivist (Lincoln & Guba 2000), the researcher placed himself near to the objectivist aiming to move toward subjectivist.

On more reflection, as also identified by others (Sandelowski 2000b), the researcher realised after a while that it is not easy to exchange his strong belief system from a positivist view to a completely different worldview such as constructivism. In contrast to more dialectical and subjectivist modes of an interpretive worldview such as phenomenology, a research study based on principles of grounded theory, especially a classical mode, was closer to the researcher's worldview. While the research question needs to be one that grounded theory can answer, the researcher's worldview is also important (Annells 1996, p.379). Finlay (2006, p.14) is of the following opinion:

However, your preferences count too. You have to adopt a methodology to

which you can relate. To identify and clarify your preferences, you might find it useful to reflect on your values, beliefs and interests; your goals; your resources and opportunities; your skills and knowledge; and any academic/disciplinary demands that influence you.

Altogether, it seems that a research study based on principles of a classical mode of grounded theory methodology is more congruent and fits better with the researcher's worldview. Grounded theory allows researchers to move from empirical studies to interpretive ones more easily (Smith & Biley 1997). It is important to review different modes of grounded theory in more detail to see the differences that exist between modes of grounded theory in terms of their ontology and epistemology and why the classical mode of grounded theory was chosen as a base for this research study.

Different modes of grounded theory

The classical form of grounded theory method was introduced by the collaborative work of Barney Glaser and Anselm Strauss in 1967 (Annells 1997). Glaser had a more positivist background working in Columbia University whereas Strauss had more experience with fieldwork, symbolic interactionism and pragmatic philosophy at University of Chicago (Charmaz 2000).

The major concern of Glaser and Strauss leading to the development of grounded theory was that many social sciences research studies were more empirical, with fewer field work efforts, and a division between the theory and empirical research (Glaser & Strauss 1967). Before grounded theory was introduced, there was a tendency to develop a theory and then gather the data to verify it (Dey 1999, cited in Walker & Myrick 2006). The grounded theory method in fact originated from Glaser and Strauss's motivation to address the gap between theory and practice that:

had not been bridged by studies using logical deductive reasoning as the method of inquiry....Grounded theory is an approach for generating theory that is grounded in and systematically derived from data, with an emphasis on the comparative method of constant, concurrent data collection and analysis. (Jeon 2004, pp. 251-252)

The theory they were trying to generate was dependent on data coming from field work and in order to bridge the theory-practice gap, a "middle range" theory emerged (Charmaz 2000, p.510). Glaser and Strauss (1967, pp. 32-33) point out that such a middle range theory sits between the "minor working hypothesis" of every day life and "all-inclusive" grand theories. In other words, in contrast to grand theories (Bringer, Johnston & Brackenridge 2004), middle range theories try to conceptualise the "first hand-experiences" of people (Wilson, Hutchinson & Holzemer 2002, p.1313) and are very useful in practical situations.

Annells (1996) states that critical realism is the ontology of this mode of grounded theory that accepts that "the social and natural worlds have differing realities, but that both forms of reality are probabilistically apprehensible, albeit imperfectly". She also asserts that the epistemology of this version presents "a discernable post positivist suggestion that the method is independent of the researcher and has separate existence. This can be identified as a modified objectivist epistemological view about the nature of the relationship between the knower and what can be known". Finally, the methodology of this mode, as emphasised by Annells, is based on "theory discovery and generation in a detailed qualitative research process rather than logico-deductive a priori assumptions, which comprise the first step of positivist research" (1996, pp. 385-387). As a result of a grounded theory research using the classical approach, researchers can interpret the data but it also offers researchers the means and methods of thinking and analysing so that interpretations do not veer to far from the actual data. Altogether, the ontology and epistemology of the classical mode of grounded theory is very close to a positivist approach chosen for the first phase of this study.

Another mode of grounded theory was published in 1990 by Strauss and Corbin. The major difference between this mode and the classical one is the "suggestion of a conditional matrix" in the data analysis (Annells 1996, p.386). Strauss and Corbin (1998, pp.181-182) are of the opinion that a phenomenon needs to be located "contextually or within the full range of macro and micro conditions in which it is embedded and tracing out the relationships of subsequent actions/interactions through to their consequences". Such approaches to data analysis were criticised as forcing data and therefore deviating from the basics of the grounded theory method (Walker & Myrick 2006). However, in Strauss and Corbin's version of grounded theory it is not clear what exactly constitutes the macro and micro conditions in the

context of a research study and how different consequences of action/interaction can be identified. Therefore using such a matrix is not necessary and instead emerging conceptual categories from data are mainly dependent on researchers allowing the data to tell the story.

This research study is mainly based on the concepts developed by Glaser and Strauss (1967). These features consist of following items: theoretical sampling, theoretical sensitivity and issue of theoretical saturation; comparative data analysis; memo writing; and coding and identification of the core category. As stated earlier, these methods keep the researcher very close to the data and interpretations made through this process are most likely to be grounded in actual experiences of participants. While these features are interwoven, they have been considered as methods and discussed in the relevant sections in the method chapter of the second phase (Chapter Six). If there is any modification to these features, they are also addressed in those sections. The crucial thing is that if any modification to this methodology is made, they need to be addressed by researchers (Jeon 2004).

In summary, the interpretive phase of the research study was based on the principles of a grounded theory methodology in which the focus is mainly on patient-nurse interactions in order to make conceptual categories to inform the practice.

Conclusion

This research study as a whole has a mixed approach that is very helpful to deal with complex issues like QoL and the results should have greater generalisation to, and validity in, other situations and contexts. The first phase is an empirical study based on principles of a positivist view of knowledge. The second phase is an interpretive study based on principles of a classical mode of grounded theory. The researcher believes that philosophical issues underpinning grounded theory (mainly interactionism and pragmatism) are very supportive for the interpretive phase of this research study because the study has a focus on interactions between patients and nurses in order to develop conceptual categories to help the caring practice. Moreover, philosophical issues of grounded theory appear to be more congruent with

the researcher's philosophical view and background at this time. Finally, the researcher also found that methods suggested in grounded theory (comparative data analysis, memo writing, and theoretical sampling) are very useful because they constantly remind the researcher to stay with the data and avoid making interpretations that are not supported by the actual data.

Having clarified the methodology and philosophical issues underpinning this thesis, the methods and results of each phase are addressed separately in the following chapters. Chapter Four focuses on the methods while Chapter Five examines the results of the first phase. Chapters Six and Seven discuss the methods and results of the second phase, respectively.

CHAPTER 4: METHODS OF THE FIRST PHASE

Introduction

In the previous chapter, it was explained that this research study has a mixed approach. The first phase has an empirical approach in which QoL perceptions of patients and nurses are translated to QoL scores that are quantifiable and objective. To do justice to the complex nature of the concept of QoL, a range of statistical tools will be used in this phase to better understand the multi-dimensional nature of the QoL concept. The second phase of this study continues to acknowledge the multi-dimensional construct of QoL and includes subjective experiences of participants to provide richer interpretations of QoL perceptual differences. These two approaches together will increase the breath and depth of the research study.

This chapter describes and justifies the quantitative methods used in the first phase of this study. While the methodology is related to philosophical issues underpinning the research, methods relate to the techniques used to gather and analyse data (Wainwright 1997). Furthermore the specific aims of the first phase are introduced. After a brief introduction about different QoL tools, it will be argued why the WHOQoL-BREF questionnaire is an appropriate tool in this study. Then the actual procedure used for the data collection is introduced. A justification of the appropriateness of the sample size in the first phase of the research study is made at the end of the chapter.

Aims of the first phase

In Chapter One, two research questions for the whole project were presented enquiring about the level of agreement between patients and nurses about cancer patients' QoL and why QoL perceptual differences exist. In order to answer these questions, this phase sought to address the following specific aims:

 to determine the reliability of cancer patients' and nurses' QoL scores on the WHOQoL-BREF questionnaire using an internal consistency test (Cronbach alpha coefficient);

- 2. to explore the structure of domains of the WHOQoL-BREF questionnaire completed by cancer patients using exploratory factor analysis;
- to investigate the level of agreement between patients' and nurses' QoL scores on the WHOQoL-BREF questionnaire using different statistical tests (Pearson correlation, IntraClass Correlation Coefficient, Bland-Altman test, Proportion of exact agreement, Paired t-test);
- to establish the association between the level of agreement between patients' and nurses' QoL scores on the WHOQoL-BREF questionnaire and patients' level of QoL using Bland-Altman test; and
- 5. to investigate the influence of patients' and nurses' demographic and clinical variables on the level of agreement between patients' and nurses' QoL scores on the WHOQoL-BREF questionnaire using Bivariate (Pearson Correlation) and Multivariate (Mulitple regression) tests.

It is not the intention of the research study to explore for statistically significant relationships between clinical and demographic variables of patients and nurses and QoL scores given by patients and nurses separately but to identify if these variables influence the level of QoL agreement between these two chosen groups.

Having considered these specific aims, statistical procedures that were used to answer the above research questions will be explained.

Research Design and procedure

Instrument

The WHOQoL-BREF questionnaire used in this research study is classified as a generic health profile QoL tool. It is relevant to explain the classification of QoL tools to see what characteristics generally differentiate the WHOQoL-BREF questionnaire from other QoL tools.

Classification of the WHOQoL-BREF questionnaire

QoL tools have been classified in two major divisions: generic (or general) and specific tools (Soni & Cella 2002). Specific QoL instruments focus on problems that

are applicable only to a certain disease or population (for example cancer) a certain dimension of QoL (psychological) or a given condition (pain) or a specific treatment modality (Cella & Nowinski 2002). These instruments are usually more responsive than generic QoL tools, but often are not comprehensive and cannot be used to compare responses across different diseases or conditions (Fayers & Machin 2000).

The generic QoL instrument, on the other hand, is designed to measure the complete spectrum of dimensions relevant to QoL and can be used to compare results across different patients and may also be used for healthy people (Cella & Nowinski 2002; Movsas 2003). The two types of generic instruments are utility measures and health profiles (Movsas 2003). With utility instruments, QoL is measured as a single number along a continuum ranging from death (0) to full health (10) (Giesler 2000). In other words, such measures integrate multidimensional information related to QoL into only a single comprehensive item; it shows how patients value the state of their health.

Conversely, health profiles are single instruments that measure different aspects of QoL with different items (Fayers & Machin 2000). For example, the WHOQoL-BREF questionnaire as a health profile QoL tool uses 26 items or questions which assess the QoL across four domains including physical (seven items), psychological (six items), social relationship (three items), and environmental (eight items) domains and two items measuring overall quality of life and general health. All 26 items are rated on a five-point Likert-type scale (e.g. ranging from very poor, to poor, to neither poor nor good, to good, and to very good) (The WHO Group 1998a).

The aim of this research study was to compare patients' and nurses' perceptions about cancer patients' QoL across different domains. Moreover, the cancer patients who were selected in this research study had different types of cancer. So a generic QoL tool like the WHOQoL-BREF questionnaire was chosen because it might better do a comprehensive comparison between patients and nurses about cancer patients' QoL.

Patient and proxy versions of the WHOQoL-BREF questionnaire and instructions

The WHOQoL-BREF questionnaire was used by both the patient and nurse groups. The questionnaire for nurses (proxy version) was similar to that of patients with only small modifications made to QoL items as recommended by Hays et al. (1995). For example, the item 'how satisfied are you with your health?' in the patient version was modified in the proxy version to 'how satisfied is the patient with his (her) health?'

Participants also needed to be instructed as to how to complete the WHOQoL-BREF questionnaire. Patients were instructed to complete the questionnaire to reflect how they felt about their quality of life. The purpose of giving instructions to the proxy is different across research studies. For example in some QoL studies the proxy was asked to respond as they thought the subject would (Sneeuw et al. 1997) or try to view the situation from the patients' perspective (Sneeuw et al. 1998) while in other proxy studies (Dorman et al. 1997; Horton 2002; McVilly, Burton & Davidson 2000; Moinpour et al. 2000; Nekolaichuk et al. 1999; Wilson et al. 2000) no such statements were made. In this research study, in the instruction part of the proxy version, nurses were instructed to complete the survey specifically about the patient and the quality of life (QoL) they believe this patient has.

Given the fact that numerous QoL tools exist, a justification for choosing a specific QoL tool needs to be presented (Gill & Feinstein 1994). This is discussed in the next section.

Rationales for selecting the WHOQoL-BREF questionnaire

The literature review shows that many QoL tools exist. One article on the QoL of lung cancer patients, for example, stated that more than 50 instruments were actually used to assess QoL (Montazeri, Gillis & McEwen 1998). In another study that reviewed 75 articles about usage of QoL tools, 159 QoL tools were identified (Gill & Feinstein 1994). Salek (1998) compiled a compendium of 160 QoL instruments and stated that a large number of QoL measures are available. This number of QoL tools might indicate that generally there is no need to develop more QoL tools. Spilker et

al. (1990) also stated that having numerous QoL tools can show that QoL instruments probably have reached or are close to reaching the maximum number. Garratt et al. (2002) found that in some areas there are numerous QoL tools and recommendations for the selection of such tools are needed. Therefore, in this research study a decision was made not to develop any new QoL tool, but use one of the existing QoL tools and make some recommendations about how to use it in the clinical area.

Having a large number of QoL tools can also bring with it the difficulty of making a choice for several reasons. Firstly, each QoL tool might have some advantages and disadvantages. Verma (1997) commented that every QoL tool has its own strengths and weaknesses and as other researchers have stated (Cella 1996; Ganiats et al. 1997; Hawthorne, Richardson & Day 2001; Sloan et al. 2002), no QoL tool can claim to be the ideal one. In fact, when researchers tried to compare some QoL tools, the results showed that no one instrument generally could be replaced by others in the assessment of QoL (Kemmler et al. 1999; King, Dobson & Harnett 1996; Kopp et al. 2000).

Secondly, previous researchers have not always provided reasons for their selection. Such rationales can be beneficial for researchers who choose to conduct similar research. For example, a literature review conducted by Efficace, Bottomley and van Andel (2003) on a number of data bases from 1980 to 2001 showed that rationales for using a specific QoL tool were provided in only 44% of studies. The findings of Gill and Feinstein (1994) about QoL in the medical field also revealed that having reasons for selecting a QoL tool was addressed in only 36% of articles they studied. In a similar vein, when reviewing 265 QoL articles, only 15% of them provided a rationale for selecting the specific instrument (Kong & Gandhi 1997).

Having discussed these general issues with regard to the selection of a QoL tool for use in this study, there are several reasons why the WHOQoL-BREF questionnaire has initially been chosen for this research study from generic QoL tools.

1-It possesses good psychometric properties

One important issue during the process of assessing a QoL tool is to what extent

responses to a given tool are truthful (Giesler 2000). To maximise this, there is a psychometric validation process (Mystakidou et al. 2004). If a QoL tool has poor psychometric properties, this can compromise the outcomes of QoL assessment in clinical trials (McKenna & Doward 2004). Before introducing the psychometric properties of the WHOQoL-BREF questionnaire specifically, it is relevant to explain the main psychometric properties of QoL tools further. The most important psychometric properties of a QoL tool consist of validity and reliability

A valid tool must measure what it claims to measure (Boling, Fouladi & Basenengquist 2003). In fact, a valid tool helps to define the degree of confidence that a researcher can have in the scores derived from the instrument (McMillan 1996). For example, a valid QoL tool must measure QoL and not any other construct. The most important aspects of validity consist of criterion, content, and construct validity.

Criterion validity searches for a correlation between one tool and another instrument, particularly one which is assumed to be superior (Grant et al. 1990). There is no QoL tool that can claim to have superiority over other tools or questionnaires or to be the gold standard (Aaronson 1989). Therefore, other existing QoL tools can be chosen for making a correlation (Giesler 2000). For example, a new QoL tool can have criterion validity provided that a correlation between this tool and a previous existing QoL tool can be observed. Content validity indicates whether or not the items in the tool are representative of the domain it is intended to assess and adequately samples the content area (Boling, Fouladi & Basen-engquist 2003). For instance, the physical domain of a QoL tool must include items to measure exactly that and not contain items from other domains, otherwise it would be deemed as having poor validity.

Construct validity indicates to what extent a tool is related to other tools based on a series of hypotheses about the variables that the tools are intended to assess (Boling, Fouladi & Basen-engquist 2003). Two most common types of construct validity consist of convergent validity (positive relation with a related measure) and discriminate/divergent validity (evidence of less correlation with a measure of other constructs) (Boling, Fouladi & Basen-engquist 2003). For example, we might find a positive relationship and therefore a convergent validity between a QoL tool and a tool that claims to measure health satisfaction. In this case both constructs of QoL
and health satisfaction share similar characteristics. On the other hand, a QoL tool might have divergent validity if it does not have a significant relationship with a tool that measures, for example 'job seeking behaviours'.

Reliability is another important psychometric property of a measure that indicates to what extent the scores obtained by the instrument can be reproduced between one administration of the instrument to another time (McMillan 1996). When scores are unreliable, the researcher or clinician cannot rely on the accuracy of the scores in representing the phenomenon of interest. The two common forms of measuring reliability are internal reliability and test-re-test reliability (Boling, Fouladi & Basen-engquist 2003; Giesler 2000). Internal reliability or internal consistency indicates whether the items in the questionnaire measure the same concept and to what extent the hypothesised scale is free of random error (Aaronson 1989; Boling, Fouladi & Basen-engquist 2003). In test-re-test reliability, correlation between scores of a given tool in two points of time will be calculated (Giesler 2000).

Unfortunately, there is a lack of information related to psychometric properties of different QoL tools. For example, in a review of 14 QoL tools by Grant et al. (1990), their results revealed that while in many instances psychometric tests were reported, test-retest and internal consistency were cited in less than half of the measures and content validity indices in only one measure. In a similar work on 265 QoL articles, the reliability and validity of instruments were reported in 23.8% and 21.5% of cases, respectively (Kong & Gandhi 1997).

However, the psychometric properties of the WHOQoL-BREF questionnaire are very clear. As the WHO group (1998a) stated, high correlations exist between QoL domains of the WHOQoL-BREF questionnaire with the original long tool WHOQoL-100. The internal consistency of the WHOQoL-BREF ranges from 0.66 in the social relationship domain to 0.84 in the physical domain and test–retest reliability of the tool is 0.75 for all domains. All above correlations fall within intervals 0.61-0.80 and 0.81-1.00, which indicate almost substantial and perfect associations respectively (Landis & Koch 1977).

The WHOQoL-BREF questionnaire is therefore considered to be a psychometrically

robust tool for measuring the level of agreement between patients and nurses about cancer patients' QoL.

2-The tool is appropriate for use in the clinical area

Even though the WHOQoL-BREF questionnaire is a generic QoL tool, it can also be used in clinical studies. The WHO Group (1998a, 1998b) pointed out that the WHOQoL-BREF questionnaire is considered suitable by the WHO to best assess QoL for such areas as routine clinical work, epidemiological studies and clinical trials. In the clinical area the issue of time for completion of a QoL tool is important not only for patients but also for health care professionals. Although a number of generic QoL tools are lengthy and difficult to use for seriously ill patients (Salek 1998), the WHOQoL-BREF questionnaire takes about five minutes to be completed by well-literate people in the self-administrated mode (Department of Psychiatry of the University of Melbourne 2000). Note, however, that longer completion times have been reported (Phungrassami et al. 2004).

Sloan (1998) suggests using a minimalist approach in selecting a QoL tool for clinical studies and argues that a QoL tool having fewer but more critical questions is preferable to a long one, because it does not put too much burden on patients to complete it. It was also pointed out by Barofsky (1997) that a long QoL questionnaire may not be suitable because it has a higher drop-out rate of patients and therefore has a deleterious effect on the QoL data. Moinpour et al. (1990), basing their work on recommendations by the Southwest Oncology Group, also emphasised the use of brief QoL tools. It is important to select a QoL tool that takes no more than 10 to15 minutes to complete, particularly for patients undergoing palliative therapy (Cella 1995).

3- It can be used in cross-cultural studies

The WHOQoL-BREF questionnaire is also suitable for conducting cross-cultural research studies. While researchers are faced with many QoL tools, most of them have been developed in the United States and United Kingdom and suitable only for those countries (The WHO Group 1995). As the WHO clarified, the two main reasons for creating the WHO quality of life tools were its commitment to health with a holistic approach as well as having a QoL tool with much more consensus

between researchers engaged in cross-cultural research studies (The WHO Group 1995). The WHOQoL-BREF was developed through an international process and consists of a broad range of aspects or dimensions which are applicable to different countries and cultures (Skevington, Lotfy & O'Connell 2004).

For example, the WHOQoL-BREF questionnaire has been culturally adjusted for Australian people which in turn maximises the reliability for measuring QoL (Herrman, Hawthorne & Thomas 2002). The researcher also wishes to do a similar or trans-cultural research in the near future with Iranian cancer patients and make a comparison between the two countries. There is also a Farsi (Persian) version of this QoL tool available for this purpose. A culturally-sensitive QoL questionnaire provides researchers with the means to carry out multi-centre or multi-country QoL assessment with a comparative approach (Saxena et al. 2001).

However, according to Ecosse et al. (2005) even the results of cross-cultural research studies with the WHOQoL tools should be considered with caution. In fact, they analysed the longer WHOQoL-100 questionnaire with the RASCH model (named after Georg Rasch who developed the model) to construct a brief tool with a single score and suitable for cross-cultural studies. Results have identified that just 25 items out of 100 QoL items were culturally equivalent across the six countries under consideration.

4-QoL tool development is based on both experts' and individuals' views

Criticisms have been presented about some QoL tools that were created by people or organisations for private or commercial reasons (Hunt 1997) and developed from experts' consensus without consulting individual patients (Carr & Higginson 2001). Another reason to use the WHOQoL-BREF questionnaire in this research study is that the WHOQoL tool was developed by the WHO with collaboration from researchers, clinicians and scientists (Skevington & Suzanne 2002). Moreover, the WHOQoL-BREF tool is one of those few QoL tools which are based on a theoretical framework which includes the perceptions of healthy people and patients (Carr & Higginson 2001).

Accordingly, the WHOQoL-BREF questionnaire was used in this research because:

(1) it provides a broad assessment of QoL; (2) is not overbearing for patients and health care professionals who may be asked to complete the questionnaire; and (3) is culturally suitable to be used in Australia.

Data Collection

The collection of data for the first phase was completed between July 2005 and February 2007. The research study was conducted in three major hospitals and in different wards including two specialist oncology wards, five non-specialist oncology wards, three outpatient chemotherapy units, one radiotherapy centre and one palliative care area. Therefore, the procedures for data collection were modified based on how the research study can be conducted better in different wards. Having gained appropriate clearance from the relevant ethical committees, two methods were used for data collection.

In 18 cases, there were some practical issues that limited a direct access to the cancer patients, mainly in the palliative setting or in the outpatient wards. In these cases, all forms and questionnaires related to one case (patient-nurse) were put in an envelope and placed in a specific area in the ward. In each envelope, a step-by-step guide was provided for nurses on how to collect the data. Any registered nurse who provided nursing care for a cancer patient, had knowledge of that patient, and was happy to take part in the study gave the patient a 'Participant Information Sheet' that introduced the project to them. If the patient verbally indicated an interest in participating in the research, the nurse completed the 'Patient Characteristics Form', which obtained information including patient's age, gender, marital status, educational level, first language, current cancer diagnosis, most common current treatment, treatment setting and patient performance status. After this the nurse gave the 'WHOQoL-BREF questionnaire to be completed by patients to the patient and separately filled out the 'WHOQoL-BREF questionnaire-to be completed by the nurse about the patient. After that, another form ('Nurses Characteristic Form') that recorded characteristics of nurses such as age, gender, first language, marital status, and educational level, approximate contact time with patients, clinical experience and nurse's quality of life, was also completed by the nurse. All the above completed forms and questionnaires were then collected by the nurse and put into one envelope.

These nurses were informed to complete the questionnaires alone without asking patients any questions and without looking at the patients' questionnaires. The stepby-step guide for nurses and relevant questionnaires and forms can be observed in **Appendices B, C, D, E, F, G** and **H**.

The second method of collecting data was conducted by the researcher. Here the researcher went to the wards and first talked with the nurse manager in order to select the patient suitable for the study based on inclusion criteria and to identify the primary nurse of patient. Then the researcher talked with the primary nurse to see if the nurse was happy to take part in the research and if they knew the patient. If the nurse was satisfied, then the researcher talked with the patient about the research and gave the patient the 'Participant Information Sheet'. When the patient verbally indicated his or her consent, then the researcher gave the relevant forms and questionnaires to the patient and the nurse and after their completion separately, he picked them up and put them in the envelope which was allocated for that patient-nurse pair. This process facilitated the response rate.

To maximise the reliability of the data, orientation sessions were conducted by the researcher for nurses along with the Clinical Nurse Consultant (CNC) of the wards so that they could be well informed and instructed about the research project. A few individual sessions were also implemented as nurses requested. The questionnaire was generally completed by nurses on the same day at a time suitable for them based on their perceptions of cancer patients' QoL. However, there were a few limited cases (less than 10) where nurses filled out the forms next day due to a busy schedule. The nurses were not allowed to ask the patients any questions specific to the questionnaire before filling out their own questionnaire. Instead, they had to refer to the medical or nursing records.

Inclusion and exclusion criteria for the participants

The research study involved cancer patients as well as their attending nurses at three main public hospitals in Adelaide, South Australia. The selection was based on the following inclusion criteria.

Patients were eligible for inclusion in the study if they had confirmed diagnosis of any kind of cancer; reached the age of 18 years or older; and had the ability to read and write in English to be able to respond to the questionnaire appropriately; and agreed to participate in the study. Patients were selected from all inpatient and outpatient oncology centres and differed in their health status, disease severity and treatment modalities. This increases the variability in QoL ratings and allows outcomes of the study to be generalised to a wider group of patients (Sneeuw et al. 1997).

All registered nurses who provided nursing care for a patient were eligible to take part if they stated that they knew the patient and consented to take part in the study.

Ethical considerations

- 1. Undertaking of the first phase of the research study was approved by three Human Research Ethics Committees.
- 2. A letter confirming participant indemnity was provided by Flinders University.
- 3. Permission letters were obtained from the Director of the Nursing Division of the different hospitals.
- 4. Letters of approval from the Director of Medical Oncology Unit at the participating hospitals and the relevant subcommittee of the Palliative Care Unit were obtained.
- 5. There was no need for patients or nurses to write their name on the questionnaires or forms. Instead, the same number was recorded on the upper right-hand corner of all questionnaires and forms related to patients and nurses, and all of them put in an envelope. This allowed information to be properly matched and participant anonymity was assured.
- 6. The research data were stored according to The National Health and Medical Research Council (NHMRC) policy on research data storage. For example, original questionnaire forms were stored in a locked filing cabinet in the researcher's office. The original data will be kept up to 5 years in a secure research archive after which they will be destroyed. The coded data were entered into a password protected computer and will be archived as well.

7. In order to deal with patients' possible emotional distress in the process of completion of QoL tools, supportive care was negotiated with the Clinical Nurse Consultant on different wards so that health care professionals could help the patient if required. Nurses were not expected to experience any emotional distress by filling out a QoL questionnaire for patients. Fortunately, there was no report of such distress from patients and nurses.

Data sampling

One of the most important things in QoL research studies is to obtain a proper sample size based on all information available (Julious et al. 1997). For the purpose of estimating the necessary sample sizes, a consultation was made with a statistical consultant as well as the relevant WHO group involved with the development of the WHOQoL-BREF questionnaire. Based on previous relevant literature (McPherson & Addington-Hall 2003; Sneeuw et al. 1997), the comparisons between patients' and nurses' perceptions about cancer patients' QoL are done at two levels: individual and group level. Therefore, the minimum sample size was calculated in a way that these two levels of comparison were possible.

At the individual level, the research aim was to have a sample size that could estimate the proportion of exact agreement (see the next chapter for more details) between patients and nurses. Therefore, the following assumptions were made:

1. it was assumed that the desired significance level would be 0.05, which is the probability of the researchers making a Type I error ($\alpha = 0.05$; i.e. concluding there is a difference when there really is no difference); i.e. there is a 1/20 chance of committing a type I error. In clinical research it is usual to proceed with alpha = 0.05 (Cella 1996).

2. it was assumed that the proportion of exact agreement between patients and nurses in the WHOQoL-BREF items is 0.6 (60%). The researcher referred to previous studies (Fisch et al. 2003; Sneeuw et al. 1997) that stated that proxies had an exact agreement of about 60% with QoL responses of cancer patients. The proportion of exact agreement is explained in details in the next chapter.

3. it was assumed that the delta value (margin of error) would be 0.08 (8%) which is the proportion of error that can be expected around the exact agreement of 60%.

4. based on the above assumptions, at the individual level a minimum of 160 cases in each study cohort (patients and nurses) was needed. This sample size was estimated, using the normal approximation to the binomial distribution.

At the group level, for calculating the sample size the following assumptions were made:

- 1.a comparison between patients and nurses could be made between the mean scores on the four WHOQoL-BREF domains. The calculation was based on the average difference in domain mean scores across all four domains, arising from the two studies with a sample of people with psychosis (Herrman, Hawthorne & Thomas 2002) and schizophrenia (Becchi et al. 2004). These research studies reported mean and standard deviation (columns three, four, six, and seven, respectively) for each domain of the WHOQoL-BREF questionnaire and can be seen in Table 4-1.
- 2. the standard deviations of the difference scores for the quality of life subscales in the two research studies above were calculated by a statistical consultant stating that they range from 1 to 1.5 (not reported in the above table). A 3-point difference in QoL mean domain scores between the two groups of patients and proxies on any subscale therefore corresponds to a minimally interesting effect size of 2. This large effect size is detectable in a paired t-test with a sample size of only 5 (significance level 5%, power 80%).
- 4.based on the calculation of a statistical consultant, a sample of 160 patients (and relevant nurse-pairs), was required for agreement analysis at the individual patient level, which gave a detectable difference of 0.2 0.3 units between QoL mean domain scores of patients and nurses.

		Mean scores		Standard deviations			
Research studies	Domains	Cases	Proxies	Differences	Cases	Proxies	Mean
Becchi et al. (2004)	Physical	57.3	54.0	3.3	16.3	15.2	15.8
	Psychological	52.1	48.9	3.2	18.9	17.3	18.1
	Social	50.7	46.1	4.6	20.3	16.5	18.4
	Environment	55.4	55.8	0.4	14.6	14.9	14.8
Herrman et al. (2002)	Physical	60.7	57.0	3.7	15.4	12.5	14.0
	Psychological	56.8	51.1	5.7	17.4	13.0	15.2
	Social	51.3	43.4	7.9	20.3	18.8	19.6
	Environment	61.1	55.4	5.7	13.8	13.5	13.7

Table 4-1: Estimating means and standard deviations (ignoring the sign of the differences) for calculating the sample size

Generally speaking the sample size of 160 is enough for the purposes of this study for individual and group level comparisons. The recruitment of participants proved to be difficult as is often the case with clinical populations. Increasing the sample size was not possible with the resources available at the time.

Conclusion

The main aim of this research study was to identify the level of agreement between patients and nurses about cancer patients' QoL. As QoL is a multi-dimensional construct, a range of quantitative statistical tools were used to better understand the complex nature of QoL. In this chapter details in relation to the method of the first phase were discussed. Despite the difficulties in relation to the best choice of the QoL tool, the WHOQoL-BREF questionnaire was selected as an appropriate tool for this research study. A proxy version of the WHOQoL-BREF questionnaire was also developed by making small modifications to the original one. The WHOQoL-BREF questionnaire has acceptable psychometric properties and has been developed across different countries and cultures. This tool might be broad enough to assess QoL for different cancer patients with a variety of health status. However, the appropriateness of the WHOQoL-BREF questionnaire tool for patients and nurses was tested further in this research study. An explanation of how the WHOQoL-BREF questionnaire was used to collect data from patients and nurses across three major hospitals was also provided. Finally, the sample size at individual and group levels was justified. The following chapter the results of the first phase are described.

CHAPTER FIVE: RESULTS OF THE FIRST PHASE

Introduction

In the previous chapter it was argued that the WHOQoL-BREF questionnaire is an appropriate tool for the first phase of the research study. Then, methods related to data collection of the first phase were introduced and a justification for selection of the sample size was provided.

This chapter outlines the statistical tests used in order to analyse patients' and nurses' QoL scores and report on the outcomes. The statistical tests consist of assessments of the level of agreement between individuals; between each patient-nurse score (Proportion of exact agreement; Pearson correlation; Intraclass Correlation Coefficient; and Bland-Altman test), and an assessment of the level of agreement at the group level (paired-t test). However, the Bland-Altman test was also used to assess the association between the level of agreement (the difference between each patient-nurse QoL score) and the patient's level of QoL. The influence of patients' and nurses' demographic and clinical variables on the level of agreement (the difference between patients' and nurses' QoL scores) was also assessed using Bivariate (Pearson correlation) and Multivariate analysis (Standard multiple regression).

As well as these statistical tests in relation to the major aim of the study (measuring the level of agreement), other statistical tests were used including: descriptive statistical tests concerning patients' and nurses' demographic and clinical variables; descriptive statistical tests related to the amount of missing data; test of the reliability of the WHOQoL-BREF questionnaire; and exploratory factor analysis to explore the complex structure of domains of the WHOQoL-BREF questionnaire. It was beyond the scope of this research study to explore the effect of patients' and nurses' demographic and clinical variables on the QoL mean domain scores of patients and nurses separately.

Firstly, this chapter describes the Statistical Package for the Social Sciences (SPSS)

data file that was used in this research. Results will then be presented in different sections using illustrative tables or figures. The statistical tests used in this research study (introduced above) are explained in more detail at the beginning of each section before reporting the actual results.

Preparing the SPSS file

The quantitative data were entered into the SPSS version 12 software using syntax as recommended by the WHO for this specific questionnaire (The Australian Centre for Posttraumatic Mental Health 2003).

QoL survey items in the questionnaire(s) have specific labels and are classified into four domains: physical, psychological, social relationship, and environmental. Because three questions (three, four, and 26) out of the 26 questions were constructed in a negative form, their scores were reversed. Then, the QoL mean domain scores (MD1=Physical, MD2=Psychological, MD3=Social relationship, and MD4=Environmental) calculated. While the following steps are not obligatory but recommended in the syntax, QoL mean domains were then multiplied by four. The reason for this is that the WHOQoL-100 consists of 24 facets, each of which consists of four items. Only one item of each facet was used to construct the WHOQoL-BREF questionnaire. The WHOQoL-BREF also consists of 26 items but two items (general health and overall QoL) are not included in the scoring (Skevington, S.M., Lotfy & O'Connell 2004). Therefore, multiplication by four allows the WHOQoL-BREF questionnaire to be directly comparable with scores derived from the WHOQoL-100, giving domain scores ranging from four to 20. The same process was used for nurses, leading to four QoL mean domain scores (nMD1=Physical, nMD2=Psychological, nMD3=Social relationship, and nMD4=Environmental). Refer to Appendix I for further information.

The WHOQoL domain scores were then transformed to a 0 to 100 scale by the recommended syntax. This latest information was only reserved on the research data file and has the potential usage in the future for purposes of comparison of data with other validated instrument tools and with published community norms. As this

83

research study did not aim to compare the results with other tools or norms, QoL main domain scores were used on the bases of 4-20 rather than 0-100.

Each item in the WHOQoL-BREF questionnaire used a Likert-type scale (e.g. ranging from very poor, to poor, to neither poor nor good, to good, and to very good) and are categorical in nature. However, as real numbers are allocated to these categorical responses (e.g. 1 for very poor, 2 for poor, 3 for neither poor nor good, 4 for good, and 5 for very good) they can be deemed as continuous variables (Ferrell et al. 1992). Tabachnick and Fidell (2001) argue that making a distinction between categorical and continuous variables is not always a straightforward task. However, as recommended by the WHO and applied in previous QoL research studies, the items in the questionnaire were considered as continuous (not categorical) variables and analysed accordingly.

Data analysis and results

Parametric tests (such as Pearson correlation) were selected. Generally speaking, parametric tests tend to be considered superior to non-parametric tests as they use continuous data which in fact are more informative than categorical data (Story 2004). For example, for categorical variables (like sex or marital status) it does not make sense to calculate summary statistics like mean, median or standard deviation whereas for continuous variables (like age) it does (Greenwood 2004). Pallant (2004) points out that parametric tests are more powerful than non-parametric tests when it comes to identifying the differences between groups provided that conditions such as normal distribution and large sample size have been satisfied.

The assumption of normality was assessed at the beginning of data checking which generally was satisfactory. For example, different QoL mean domain scores for both patients and nurses groups appeared to be reasonably distributed as shown by a bell-shape pattern in the histogram. It is also suggested that when the sample size is large (greater than 100), the assumption of normal distribution is met (Katz, MH 2006). The sample size of 166 was also considered large enough for purposes of the study. No specific criterion exists to identify a large sample size. However, as a general

guide, precision tends to increase steadily up to sample sizes of 150 to 200; after that there are only small gains of accuracy deriving from increasing sample sizes (Fowler 1988).

However, in order to be on the safe side results identified from parametric tests were also checked using non-parametric tests. Since the results were nearly similar, only the parametric results are reported. Story (2004) comments that it is a good strategy to use both parametric and non-parametric tests when researchers do not want to check the assumptions of the parametric tests or parametric assumptions are doubtful. If the results of non-parametric tests are consistent with parametric ones, then only the latter are reported.

Outcomes of descriptive tests relating to missing data

It is common to have a proportion of missing data particularly when human beings are involved in the research study (Pallant 2004). Such missing data can also occur in QoL research studies (Ratcliffe et al. 2005; Tang & McCorkle 2002b). Two kinds of missing data in QoL research studies have been stated by Movsas (2003). One that is generally less important refers to situations when a QoL questionnaire is not filled in completely and can be managed by mean substitution. In this research study, for example, missing data related to questionnaire items were managed using a procedure that was recommended in the WHO data management. Where there were intermittent missing data in patient and nurse questionnaire items, the missing values were imputed using horizontal mean substitution (Department of Psychiatry of the University of Melbourne 2007). In other words, QoL missing items were substituted by the mean of other QoL items on that domain within-subject. No action was taken for missing data related to patients' and nurses' clinical and demographic variables.

The second important kind of missing data is called 'non-ignorable' missing data. In this scenario, a whole questionnaire is missed due to the patient's death or poor health condition. Such problems can lead to biased findings. Steel et al. (2005) suggested that missing data in QoL studies range between 2% and 46%. For example, as reported by Ballatori (2001) about 12% of cancer patients did not fill in their questionnaire. Ganz et al. (1988) found that in patients with metastatic lung

cancer, 13% did not fill out the QoL tool in the baseline assessment. While no universally accepted solution can be suggested for the second cause of missing data, the use of a proxy might be helpful (Bernhard et al. 1998; Bush et al. 1995; Tamim, McCusker & Dendukuri 2002; Tang & McCorkle 2002b). Outcomes of this research study can show if nurses can reliably substitute missing data arising from non-respondent patients or not. Tables 5-1, 5-2 and 5-3 show the number of missing data related to 26 items of the questionnaire and for patients' and nurses' clinical and demographic variables.

Table 5-1 description: The number of missing data based on QoL items of the questionnaire for patients and nurses. The items 'Negative feelings' (with 6 cases) and 'Sex life' (with 9 cases) have the highest number of missing data for patients and nurses, respectively.

Domains	Patients (N)		Nurses (N)	
	Valid	Missing	Valid	Missing
Physical				
"Pain prevents doing"	165	1	166	0
"Need medical treatment"	161	5	163	3
"Energy for daily life"	165	1	166	0
"Able to get around"	164	2	166	0
"Sleep"	164	2	166	0
"Daily living activities"	165	1	166	0
"Capacity for work"	164	2	166	0
Psychological				
"Enjoy life"	166	0	166	0
"Life is meaningful"	166	0	166	0
"Concentrate"	166	0	166	0
"Accept bodily appearance"	166	0	166	0
"Satisfied with yourself"	166	0	166	0
"Negative feelings"	160	6	158	8
Social relationship				
"Sex life"	162	4	157	9
"Friends"	163	3	162	4
"Personal relationship"	161	5	159	7
Environmental				
"Safe in your daily life"	166	0	165	1
"Healthy physical environment"	166	0	165	1
"Money to meet needs"	166	0	165	1
"Information available"	166	0	166	0
"Opportunity for leisure activity"	166	0	166	0
"Conditions of living place"	166	0	165	1
"Access to health services"	166	0	165	1
"Transportation"	166	0	165	1
Overall items	161	5	165	1
"Rate QoL"	101	5	105	1
"Health satisfaction"	161	5	163	3

Table 5-1: The number of missing data based on QoL items of the WHOQoL-BREF questionnaire for patients and nurses

Table 5-2 description: The number of missing data based on patients' demographic and clinical variables. The highest number of missing data (15 cases) refers to patients' highest level of education.

Characteristics	N	
	Valid	Missing
Patient's gender	165	1
Patient's first language	160	6
Patient's marital status	161	5
Patient's highest level of education	151	15
Patient's current cancer diagnosis	154	12
Patient's current treatment	156	10
Patient's treatment setting	164	2
Patient's performance status	161	5
Patient's age	156	10

Table 5-2: The number of missing data based on patients' demographic and clinical variables

Table 5-3 description: The number of missing data based on nurses' demographic and clinical variables. The highest number of missing data (23 cases) relates to the time the nurse believes he/she spends with the patients/shift.

Table 5-3: The number of missing data based on nurses' demographic and clinical variables

Characteristics	N	
	Valid	Missing
Nurse' quality of life	160	6
Nurse's age	160	6
Nurse's educational level	162	4
Nurse's gender	165	1
Nurse's language	162	4
Nurse's marital status	159	7
Nurse clinical experience	161	5
Nurse clinical experience with cancer patients	158	8
Time nurse spends with patients/shift	143	23

Outcomes of descriptive tests related to patients' demographic and clinical variables

Figures 5-1 to 5-8 and Table 5-4 summarise the descriptive statistics about patients' variables.

Figure 5-1 description: The number of patients involved based on their gender. These data show that 88 patients (53.3%) were males and 77 patients (46.7%) were females.



Patients' gender

Figure 5-1: The number of patients involved based on their gender

Figure 5-2 description: The number of patients involved based on their first language. These data show that 145 patients (90.6%) used English as their first language and 15 patients (9.40%) spoke a non-English language.



Patients' first language

Figure 5-2: The number of patients involved based on their first language use

Figure 5-3 description: The number of patients involved based on their marital status. These data show that the majority of patients 101 (62.7%) were married.



Figure 5-3: The number of patients involved based on their marital status

Figure 5-4 description: The number of patients involved based on their highest level of education. It is evident that the highest number of patients 101 (66.9%) had secondary school as their highest level of education.



Patients' highest level of education

Figure 5-4: The number of patients involved based on their highest level of education

Figure 5-5 description: The number of patients involved based on their current treatment. These data reveal that the highest number of patients 86 (55.1%) received chemotherapy as their current treatment.



Figure 5-5: The number of patients involved based on their current treatment

Figure 5-6 description: The number of patients involved based on their treatment setting. It is indicated that 72 (43.9%) and 92 (56.1%) patients were treated in outpatient and inpatient wards, respectively.



Patients' treatment setting

Figure 5-6: The number of patients involved based on their treatment setting

Figure 5-7 description: The number of patients involved based on patients' performance status*. The highest number of patients 59 (36.6%) were "Restricted but ambulatory" in their performance status.



Patients' performance status

Figure 5-7: The number of patients involved based on patients' performance status

*The patient's performance status was rated using the Eastern Cooperative Oncology Group (ECOG) performance status scale, ranging from 0 (fully active) to 4 (completely disabled). This scale is used by doctors and researchers to assess how a patient's disease is progressing, assess how the disease affects the daily living abilities of the patient, and determine appropriate treatment and prognosis (Eastern Cooperative Oncology Group (ECOG) Performance Status).

Figure 5-8 description: The number of patients involved based on their cancer diagnosis*. The most common kind of cancer was revealed to be breast cancer (25 patients, 17%).



Figure 5-8: The number of patients involved based on their cancer diagnosis

*Cancer patients had a variety of diagnoses (26 types). However for purposes of more efficient reporting they were recoded into eight categories.

Table 5-4 description: Distribution of patients based on their number and their age. These data show that 166 cancer patients took part in the study with the average age of $63.7 (\pm 15.9 \text{ SD})$ ranging from 18-92.

Characteristics	Patients	
Number	166	
Age (Years)		
Mean	63.7	
Standard Deviation (SD)	15.9	
Range	18-92	

Table 5-4: Distribution of patients based on their number and their age

Outcomes of descriptive tests related to nurses' responses based on their clinical and demographic variables

Figures 5-9 to 5-13 and Table 5-5 summarise the descriptive statistics concerning nurses' responses based on their demographic and clinical variables. Descriptive statistics related to nurses' clinical and demographic variables were calculated based on all responses to the nurses' questionnaire, and therefore some nurses are included in the sample more than once. On average, one nurse completed two survey questionnaires.

Figure 5-9 description: The number of nurses' responses in this study based on their gender. The overwhelming majority of responses 159 (96.4%) was made by female nurses and only 6 (3.6%) responses were made by male nurses.



Figure 5-9: The number of nurses' responses based on their gender

Figure 5-10 description: The number of nurses' responses based on their first language. These data show that 146 (90.1%) responses were made by nurses who had English as their first language and 16 (9.9%) responses were by those nurses who spoke a non-English language.



Nurses' first language

Figure 5-10: The number of nurses' responses based on their first language

Figure 5-11 description: The number of nurses' responses based on their marital status. These data show that the highest number of nurses' responses, 97 (61.0%), were made by those who were married.



Figure 5-11: The number of nurses' responses based on their marital status

Figure 5-12 description: The number of nurses' responses based on their highest level of education. It appears that the highest number of nurses' responses, 55 (34.0%), emanated from nurses with the RN certificate as their highest level of qualification.



Nurses' highest level of education

Figure 5-12: The number of nurses' responses based on nurses' highest level of education

Figure 5-13 description: The number of nurses' responses based on their rating of their own quality of life. These data show that the highest number of nurses' responses 89 (55.6%) was made by those nurses who rated their own quality of life as 'very good'.



Nurses' quality of life

Figure 5-13: The number of nurses' responses based on their rating of their own quality of life

Table 5-5 description: Distribution of nurses' responses based on their number, their clinical experience, and time spent with cancer patients/shift. These data also show that 95 nurses participated in this study with an average age of 37.5 (\pm 8.30 SD), ranging from 21-55. The mean time the nurse spent for providing care for the given patient (hour/shift) was 2.63 (\pm 1.86 SD) with a range of 0.08-8 hour. The mean clinical experience of nurses was 14.1 years (\pm 9.60 SD) with a range of 0.16-37 years. The nurses' mean clinical experience with cancer patients was 8.15 years (\pm 6.78SD) with a range of 0-22 years.

Characteristics	Nurses
Number	95
Age (Years)	
Mean	37.5
Standard Deviation (SD)	8.30
Range	21-55
Nurse's clinical experience	
Mean	14.1
Standard Deviation (SD)	9.60
Range	0.16-37.0
Nurse's clinical experience with cancer patients	
Mean	8.15
Standard Deviation (SD)	6.78
Range	0-22
Time nurse spends with cancer patients (Hour/Shift)	
Mean	2.63
Standard Deviation (SD)	1.86
Range	0.08-8

Table 5-5: Distribution of nurses' responses based on their number, their clinical experience and time they spent with cancer patients/shift

Outcomes of assessing the reliability of the WHOQoL-BREF questionnaire (patients and nurses)

As noted earlier, one important psychometric test characteristic of QoL tools is reliability testing. To assess the reliability of the tool, internal consistency was measured using the Cronbach alpha coefficient for patients' and nurses' QoL mean domain scores. The ideal acceptable level for the Cronbach alpha coefficient is 0.70 (Pallant 2004) and was used for this study.

Table 5-6 shows the reliability of the WHOQoL-BREF questionnaire based on patients' and nurses' QoL mean domain scores. More details of the reliability test can be seen in **Appendix J**.

Table 5-6 description: The reliability test of the WHOQoL-BREF questionnaire completed by patients and nurses based on QoL mean domain scores. These data show that the Cronbach's alpha value for patients' responses were 0.83, 0.80, 0.37, and 0.69, for the physical, psychological, social relationship and environmental domain, respectively. The corresponding data for nurses were 0.85, 0.78, 0.60, and 0.80, respectively.

QoL domain	Cronbach alpha	
	Patients	Nurses
Physical (7 items)	0.83	0.85
Psychological (6 items)	0.80	0.78
Social Relationship (3 items)	0.37	0.60
Environmental (8 items)	0.69	0.80
Overall (26 items together)	0.89	0.92

Table 5-6: Test of reliability of the WHOQoL-BREF questionnaire completed by patients and nurses based on QoL mean domain scores

Outcomes of measuring the level of agreement between patients and nurses about cancer patients' QoL

Patient and proxy agreement is usually assessed at an individual level to measure the similarity of scoring of each item between the two rater groups and at the group level to identify any systematic bias between two raters (i.e. proxies underestimate or overestimate patients' QoL) (von Essen 2004). The following pages indicate how different measures have been used to measure agreement at the different levels (i.e. individual and group levels).

Correlations (r and ICC)

Correlation tests are used to describe the strength and direction of a linear relationship between two variables. One of the correlation methods is the Pearsonmoment coefficient (r) that has been designed to test for an association between two continuous variables. The result of the test (r) is indicated by a value of between -1 and +1. The + or - at the front indicates whether there is a positive correlation (as one variable increases, so too does the other) or a negative correlation (as one variable increases, the other decreases). The size of the absolute value provides an indication of the strength of the relationship. A perfect correlation of -1 or +1 indicates that the value of one variable can be determined exactly by knowing the value of the other variable (Pallant 2004).

Pearson correlation was conducted in the first step of this analysis despite the belief that it might lead to inaccurate conclusions about the level of agreement between two groups (Tang & McCorkle 2002b). For example, sometimes proxies provide a rating that is completely higher or lower than the patients' rating, but still a high correlation between two raters might exist (Moinpour et al. 2000).

An Intraclass Correlation Coefficient (ICC) was therefore calculated for determining the level of agreement between patients and nurses of continuous data (Nelson et al. 1990) that best accounts for chance agreement between the groups (Lee, Koh & Ong 1989). While the correlation (r) indicates whether paired scores go up and down together (or level of covary) (Lobchuk & Degner 2002), the ICC is the appropriate way for determining the level of agreement (equality of ratings or concordance) on continuous data between the two groups (Sneeuw et al. 1998).

There are criteria for making a judgment about the results of correlation coefficient values, For example, Cohen (1988) states that the results of correlation (r), irrespective of the sign in front, should be interpreted as below:

r=0.1 to 0.29 indicates a small correlation

r=0.3 to 0.49 indicates a medium correlation

r=0.5 to 1.00 indicates a large correlation

Landis and Koch (1977) also proposed criteria to judge the results of kappa statistics across studies as follows: less than or equal to 0.20, poor; 0.21-0.40, fair; 0.41-0.60, moderate; 0.61-0.80, substantial; 0.81-1, almost perfect. These criteria can also be used for correlation coefficient and ICC as well (Tang & McCorkle 2002b; von Essen 2004). The latter was used to judge the results of both r and ICC in this research study. Tables 5-7 and 5-8 show the results of correlation (r) and ICC. The ICC values between patients' and nurses' scores for all 26 items on the WHOQoL-

BREF questionnaire are included in Appendix K.

Table 5-7 description: Pearson correlations between different mean domain scores of patients and proxies (nurses). These data show that there was moderate significant correlation in the physical domain (r=0.50, p<0.01), a poor significant correlation in the psychological domain (r= 0.19, p<0.05), and a fair significant correlation in the environmental domain (r=0.34, p<0.01) between patients and nurses. There was also a non-significant and poor correlation in the social relationship domain (r=0.08, p=0.34) between patients and nurses.

Table 5-7: Pearson correlations between different mean domain scores of patients and nurses

Domains	Patient-Nurse correlation (r)	Significance*
Physical	0.50	0.00
Psychological	0.19	0.02
Social relationship	0.08	0.34
Environmental	0.34	0.00

*Only significant differences at (p<0.05) are important

Table 5-8 description: Intraclass correlations between the QoL mean domains of patients and nurses. There was moderate significant correlation in the physical domain (ICC=0.48, p<0.05), a poor significant correlation in the psychological domain (ICC= 0.19, p<0.05), and a fair significant correlation in the environmental domain (ICC=0.30, p<0.05). There also was a non-significant and poor correlation in the social relationship domain (ICC=0.06, p=0.22) between patients and nurses.

Table 5-8: Intraclass correlations between QoL mean domain scores of patients and nurses

Domains	Patient-Nurse ICC*	Significance **
Physical	0.48	0.00
Psychological	0.19	0.01
Social relationship	0.06	0.22
Environmental	0.30	0.00

*Results of single measure are reported

*Only significant differences at (p<0.05) are important

The proportion of exact agreement between patients and nurses

Even with a moderate association between patients and nurses, both ratings might be quite similar (von Essen 2004). Therefore, the proportion of exact agreement is suggested as a complementary test for measuring the level of agreement (Lobchuk & Degner 2002; Sneeuw et al. 1998; Tang & McCorkle 2002b). This test was also conducted to measure the level of agreement between cancer patients and nurses at the individual level. Exact agreement is defined as those cases where the response category chosen by the patient and the nurse for a given item is identical (Sneeuw et al. 1997). Percentage agreement is calculated by adding up the number of cases that received the same rating by both judges and dividing that number by the total number of cases rated by the two judges (Stemler 2004). While this test allows an estimation of the magnitude of agreement based on a calculation of the proportion of exact responses (Lobchuk & Degner 2002), it also is criticised as having no attention to agreement by chance (Magaziner et al. 1988; Sneeuw et al. 1997).

Unfortunately, there is not a cut-off point above which we can say that there is a good exact agreement between patients and nurses in their responses to each QoL questionnaire item. However, previous research studies (Fisch et al. 2003; Sneeuw et al. 1997) suggest that at least 60% of agreement between patients and proxies in a QoL tool items is satisfactory. This criterion was also used in this research study.

The proportion of exact agreement between patients' and nurses' scores was calculated for all 26 items on the WHOQoL-BREF questionnaire and reported in **Appendix L**. The exact agreement for each domain was then calculated as an average of exact agreement of corresponding items related to that specific domain. The average of exact agreement for all 26 items together was also calculated. The results of the exact agreement are shown in Table 5-9.

Table 5-9 description: The proportion of exact agreement between patients and nurses for different QoL domains of the WHOQoL-BREF questionnaire. These data show that the average proportion of exact agreement between patients and nurses is 35.5%. The proportion of the exact agreement between the two groups was 34.9%, 34.5%, 33.8%, and 36.9% for the physical, psychological, social relationship, and environmental QoL domains.

QoL domain	Proportion of exact agreement
Physical	34.9%
Psychological	34.5%
Social relationship	33.8%
environmental	36.9%
Average	35.5%

Table 5-9: The proportion of exact agreement for different QoL domains between patients and nurses

Bland-Altman test

As noted earlier, the Pearson correlation (r) is not the most robust indicator for assessing the level of agreement between patients and their nurses. Bland and Altman proposed an alternative way that is named the Bland-Altman test (Bland & Altman 1986). In this test instead of using a simple correlation between two methods/raters, the difference between two raters is plotted against the average of two QoL scores (Bland & Altman 1986). In this research study, for example, the difference between the patient and nurse QoL scores was plotted against the average of two scores for each patient-nurse pair.

In this plot, two important issues need to be considered. The first point is the acceptability of the overall width of the scatter (upper and lower levels of agreement between the two groups). This is an interval measure and 95% of differences (in this case patient-nurse difference) would fall within this interval (Moinpour et al. 2000). Unfortunately, there is no cut-off point regarding the percentage difference between upper and lower levels of agreement that can be considered a significant difference. The acceptable level of difference between upper and lower levels of agreement was set at 10 %. The researcher referred to another research study (Sloan et al. 1998) in which a difference of 10 points out of 100 for the Spitzer scale was not considered an

acceptable difference between patients and proxies.

Another point of interest is whether there is a pattern in the scatter of points. In fact, this method finds out if agreement varies across a range of QoL scores by visual examination of a scatter plot (Lobchuk & Degner 2002). More details about how Bland-Altman plots were drawn for different QoL mean domain scores can be located in **Appendix M.** See Figures 5-14 to 5-17 for actual results of the Bland-Altman test.

Figure 5-14 description: Bland-Altman plot* indicating average physical QoL mean domain scores of patients and nurses against differences in their mean domain scores. These data show that the location of the upper and lower limits of agreement of two raters (patients and nurses) differs by around 6 points. This figure also shows that there is not an obvious pattern between the difference of means and the average of means in the physical QoL domain.



*(MD1+nMD2)/2: This shows the average of mean domain scores of two groups in the horizontal (X) axis. (MD1-nMD2): The difference between mean domain scores of two groups in the vertical (Y) axis is shown. There are three lines in the plot. The first one in the middle (near to the point 0.00) shows the mean of differences. The line above this (near to the point 5.00) and the line below (near to the point - 5.00) are upper and lower limits of agreement, respectively.

Figure 5-14: Bland-Altman plot indicating average physical QoL mean domain scores of patients and nurses against differences in their mean domain scores

Figure 5-15 description: Bland-Altman plot* indicating average psychological QoL mean domain scores of patients and nurses against differences in their mean domain scores. These data show that the location of the upper and lower limits of agreement of two raters (patients and nurses) differs by around 6 points. This figure also shows that there is no obvious pattern between the difference of means and the average of means in the psychological QoL domain.



*(MD1+nMD2)/2: This shows the average of mean domain scores of two groups in the horizontal (X) axis. (MD1-nMD2): The difference between mean domain scores of two groups in the vertical (Y) axis is shown. There are three lines in the plot. The first one in the middle (near to the point 0.00) shows the mean of differences. The line above this (near to the point 5.00) and the line below (near to the point - 5.00) are upper and lower limits of agreement, respectively.

Figure 5-15: Bland-Altman plot indicating average psychological QoL mean domain scores of patients and nurses against differences in their mean domain scores

Figure 5-16 description: Bland-Altman plot* indicating average social relationship QoL mean domain scores of patients and nurses against differences in their mean domain scores. These data show that the location of the upper and lower limits of agreement of two raters (patients and nurses) differs by around 7 points. There is no obvious pattern between the difference of means and the average of means in the social relationship QoL domain.



*(MD1+nMD2)/2: This shows the average of mean domain scores of two groups in the horizontal (X) axis. (MD1-nMD2): The difference between mean domain scores of two groups in the vertical (Y) axis is shown. There are three lines in the plot. The first one in the middle (near to the point 0.00) shows the mean of differences. The line above this (near to the point 5.00) and the line below (near to the point - 5.00) are upper and lower limits of agreement, respectively.

Figure 5-16: Bland-Altman plot indicating average social relationship QoL mean domain scores of patients and nurses against differences in their mean domain scores

Figure 5-17 description: Bland-Altman plot* indicating average environmental QoL mean domain scores of patients and nurses against differences in their mean domain scores. These data show that the location of the upper and lower limits of agreement of two raters (patients and nurses) differs by around 5 points. This figure also shows that there is not an obvious pattern between the difference and average of means in the environmental QoL domain.



*(MD1+nMD2)/2: This shows the average of mean domain scores of two groups in the horizontal (X) axis. (MD1-nMD2): The difference between mean domain scores of two groups in the vertical (Y) axis is shown. There are three lines in the plot. The first one in the middle (near to the point 0.00) shows the mean of differences. The line above this (near to the point 5.00) and the line below (near to the point - 5.00) are upper and lower limits of agreement, respectively.

Figure 5-17: Bland-Altman plot indicating average environmental QoL mean domain scores of patients and nurses against differences in their mean domain scores
Paired t-test for comparing means of two group responses

The above mentioned statistical tests (r, ICC, the proportion of the exact agreement, and the Bland-Altman test) identify the amount of association or agreement between two groups of patients and nurses but cannot show how nurses rate their patients as having a higher or lower level of QoL than did the patients themselves. In this situation, comparing the means of two groups of patients and nurses can help in identifying such differences more clearly (Lobchuk & Degner 2002; Sneeuw et al. 1998; Tang & McCorkle 2002b).

In this research study, a paired t-test was used to compare the means of two groups of patients and nurses in order to find out the statistically significant differences between different QoL mean domain scores. The significance level was set at 0.05 (p<0.05). The effect size is also reported below in the description of Table 5-10. Effect sizes provide an indication of the magnitude of the differences between groups. There are different ways to obtain effect size but one of them which is calculated in this research study is the Eta squared:

Eta squared= $t^2/t^2 + N - 1$

Cohen (1988) classifies 0.01 as a small effect, 0.06 as a medium effect and 0.14 as a large effect. These criteria were used in this research study. Table 5-10 shows the results of the t-test.

Table 5-10 description: Paired samples t-test values between different QoL mean domain scores of patients and nurses. These data show that there were significant differences between social relationship QoL mean domain scores of patients and nurses [t (161) =2.27, p< 0.05] and between environmental mean domain scores of patients and nurses [t (165) =4.39, p< 0.01]. For the social relationship and the environmental QoL domains, the Eta squared (as an indicator of effect size) was 0.03 and 0.10, indicating small and medium effect size, respectively. There were no significant differences between patients and nurses in the physical [t (165) =-1.14] and the psychological [t (165) =1.25] QoL mean domains.

Table 5-10: Paired samples t-test values between QoL mean domain scores of patients and nurses

QoL domain	Patients		Nurses		Patient-Nurse difference		Patient-Nurse t-test	
	Mean	SD	Mean	SD	Mean	SD	t	d.f
Physical	12.4	3.25	12.6	2.70	-0.27	3.04	-1.14	165
Psychological	14.3	2.66	14.0	2.28	0.31	3.16	1.25	165
Social relationship	15.4	2.86	14.8	2.41	0.64	3.60	2.27*	161
Environmental	15.5	1.99	14.7	2.03	0.79	2.30	4.39**	165

*Statistically significant difference between patient and nurse score (p<0.05)

**Statistically significant difference between patient and nurse score (p<0.01)

Outcomes of assessing the relationship/effect of demographic and clinical variables of patients and nurses on the level of agreement using Bivariate and Multivariate analysis

One major issue in using Bivariate tests such as t-test or correlation as previously reported is that they overlook the influence that might exist between variables themselves (Sneeuw et al. 1998). Using Multivariate tests such as Multiple regression was introduced to identify if this is a problem. In fact, such tests are useful for determining the relative contributions of different causes to a single event (Katz 2006).

One of the aims of this study was to conclude if any patients' and nurses' clinical and demographic variables (independent or predictor variables) can significantly

influence the level of agreement between patients and nurses (dependent variable), and whether this variable is still able to affect the level of agreement when the effects of another variable(s) are controlled for. Multiple regression is a very sophisticated statistical test that can be used to answer such questions (Tabachnick & Fidell 2001). The Standard multiple regression was used for Multivariate analysis. Unlike other studies that chose the Hierarchical multiple regression (Rothman et al. 1991), this study did not have any theory to support entering the data into the equation in a given order.

At first, a Bivariate correlation (r) was conducted between patient-nurse absolute QoL mean domain differences as the dependent variable with patients' and nurses' clinical and demographic variables as the independent variable. Bivariate correlation identified only one or two significant correlations at 0.05 between patients' and nurses' clinical and demographic variables with absolute QoL mean domain differences. Therefore, in order to conduct Standardised multiple regression analysis, those correlations with the significance level less than 0.10 simultaneously entered into the equation. It is worth stating that both Bivariate and Multivariate analysis require one continuous dependent variable and two or more continuous independent variables (or categorical variables with only two levels) (Pallant 2004). Therefore, at both Bivariate and Multivariate levels those categorical variables had more than two levels translated to dummy variables before entering into the equation. Altogether 15 dummy variables were created (see Table 5-11).

At Multivariate level, Beta coefficients with p values of 0.05 or less were considered significant. Moreover, Adjusted R-square was used to determine the proportion of variance in patient–nurse differences that could be explained by the patient and nurse variables combined. There is no cut-off point to judge if outcomes of Adjusted R-square are trivial or not. The significance of these outcomes will be discussed along with the results of the second phase.

Table 5-11 below shows Bivariate and Multivariate correlation between patients' and nurses' clinical and demographic variables and absolute difference between patients and nurses QoL mean domain scores.

Table 5-11 description: Bivariate and Multivariate correlations between patients' and nurses' clinical and demographic variables and absolute difference between patients and nurses QoL mean domain scores. At the Bivariate level only a few variables were identified to be significantly associated with the absolute difference between patients' and nurses' QoL mean domain scores (being an indicator of agreement). Larger differences between patient and nurse physical QoL domain scores were noted when the nurses had less clinical experience with cancer patients. In the psychological and social relationship QoL domains, greater differences between nurses and their patients were found to exist for inpatients. Finally, larger differences between patient and nurse environmental QoL mean domain scores were noted for inpatients and nurses with lower levels of QoL.

However, using Multivariate analysis, the Adjusted R-square indicated that these variables explained only 4%, 2%, 3%, and 6% of the variance in absolute patientnurse differences in the physical, psychological, social relationship and environmental QoL mean domain scores, respectively. Where the patients were receiving treatment (β =0.20) was the only statistically significant predictor of differences between patient and nurse scores obtained for the social relationship and psychological QoL domain scores. In the environmental domain, the only statistically significant predictor of differences between patient and nurse was the nurse's QoL (β =-0.20).

Notes for Table 5-11

* Correlation is significant at the 0.05 level (2-tailed).

** Correlation is significant at the 0.01 level (2-tailed).

*** Correlation is significant at a level between 0.05 and 0.1 (2-tailed). These significant correlations also imported into the equation in order to better conduct the regression. Also, in all four equations, patient's first language, patient's treatment setting, nurse clinical experience, nurse clinical experience with cancer patients, and nurse' quality of life imported into the equation.

 $^{\circ}$ As these categorical variables had more than two categories, they were entered into equation after creating dummy coding variables. Some variables recoded before creating dummy coding variables.

•In the parenthesis in front of all variables, the reference groups are identified. For example, male vs. female indicates that male is the reference group. This is the situation for other variables.

Table 5-11: Bivariate and Multivariate correlations between patients' and nurses' clinical and demographic variables and absolute difference between patients and nurses QoL mean domain scores

Patients and nurses characteristics•	Absolute difference							
	(Physical)		(Psychological)		(Social		(Environmental)	
	()		()8)		relationship)		()	
		D.		D.	relations	nip)		D
	r	Beta	r	Beta	r	Beta	r	Beta
Patient's gender (male vs. female)	0.03		0.12		-0.01		0.02	
Patient's age (year)	-0.01		-0.11		0.06		-0.13	
Patient's first language (English vs. other)	0.13***	0.12	0.03	0.01	0.12	0.10	-0.04	-0.07
Patient's treatment setting (outpatient vs. inpatient)	0.15***	0.08	0.20 *	0.20*	0.16*	0.20*	0.19*	0.17
Patient performance status (From fully active to completely disabled)	0.10		0.09		0.07	0.13	-0.05	
Nurse's gender (male vs. female)	-0.08		-0.04		0.09		-0.06	
Nurse's age (year)	-0.10		-0.04		0.05		-0.09	
Nurse's language (English vs. other)	-0.10		0.02		-0.04		0.01	
Time nurse spends with patients/shift	-0.02		0.11		-0.00		0.07	
Nurse' clinical experience (year)	-0.14 ***	-0.02	-0.10	0.05	0.00	0.13	-0.14***	-0.15
Nurse' clinical experience with cancer	-0.17*	-0.14	-0.13	-0.08	-0.04	-0.06	-0.05	0.15
Nurse' quality of life (from very poor to very good)	0.10	0.13	0.06	0.10	0.10	0.13	-0.22**	-0.20*
Patient marital status	0.03		0.04		0.08		0.10	
Patient marital status	-0.10		0.06		-0.00		0.11	
Patient educational level	0.07		0.02		-0.10		-0.04	
Patient educational level	0.01		0.05		-0.01		0.05	
Cancer diagnosis	0.08		0.01		0.05		0.00	
Cancer diagnosis	-0.09		-0.09		-0.07		-0.09	
(Breast cancer vs. Reproductive) = Cancer diagnosis	0.03		0.07		0.03		0.10	
(Breast cancer vs. NHL) ^a Cancer diagnosis	-0.00		-0.03		-0.01		-0.04	
(Breast cancer vs. GI) • Cancer diagnosis	-0.03		-0.02		0.06		0.04	
(Breast cancer vs. lung cancer) ^o Cancer diagnosis	0.02		-0.04		-0.02		0.04	
(Breast cancer vs. other kind) ^o	0.07		0.11		0.01		0.02	
(Married vs. Single) $^{\circ}$	-0.07		-0.11		-0.01		0.02	
Nurse's marital status	0.05		0.03		0.13		0.04	
Nurse's educational level	0.12		0.05		0.01		0.03	
(KN vs. BN) ² Nurse's educational level	-0.08		-0.06		-0.05		021	
(RN vs. Graduate diploma) ^o	0.02		0.00		0.02		0.02	
Nurse's educational level (R.N vs. other) ^o	-0.02		-0.08		-0.02		-0.02	
Patient 's current cancer treatment (chemotherapy vs. other)	0.08		-0.01		0.12		-0.04	
Adjusted R-square		4%		2%		3%		6%

Outcomes of the exploratory factor analysis showing the structure of domains of the WHOQoL-BREF questionnaire completed by cancer patients

Factor analysis is a very complex statistical test that takes a large set of variables and tries to summarise them to a smaller set of factors or components (Pallant 2004). This test is helpful for researchers who want to see how items in a tool or scale can be grouped together and create smaller subscales or tools. While factor analysis needs at least 300 cases, when the sample is strong and reliable and contains only a small number of factors, a smaller sample may be chosen (Harris & Heard 2004). Tabachnick and Fidell (2001) also recommended a sample size of 300 for more comfortable results. But they also suggested that having 5 cases for each item of the tool or scale is adequate in most cases. The WHOQoL-BREF questionnaire contains 26 items. However, two items (health satisfaction and rate QoL) are not included in the scoring. Given that 5 cases are needed for each item analysis, a sample of at least 120 (24 multiply 5) is enough for this analysis. In this research study the sample size of patients is 166, which would be enough to conduct an exploratory factor analysis.

Having considered the complexities and differences in conducting and reporting the outcomes of an exploratory factor analysis, the method as suggested by Pallant (2004) was conducted using SPSS version 12 software. Following these guidelines, at the first step, items of the WHOQoL-BREF questionnaire were assessed for suitability of the data for factor analysis. For example, inspection of the correlation matrix supported conducting this analysis because there were many coefficients of 0.3 or above. The Kaiser-Meyer-Olkin measure of sampling adequacy was 0.83, exceeding the recommended value of 0.60. Bartlett's test of Sphericity reached statistical significance, supporting the factorability of the correlation matrix for patients.

At the second step, factor extraction was conducted using Principal Components Analysis (PCA). In this step, for each component or factor (in this case each item of the questionnaire) the eigenvalue was calculated (named Total in the table) which shows the amount of the total variance explained with that factor (% of Variance in the table). However, as recommended as a rule of thumb by Harris & Heard (2004), the researcher generally considered seriously factors with eigenvalues greater than 1.

Table 5-12 shows the eigenvalues more than 1 and related variances for the WHOQOL-BREF questionnaire completed by patients.

Component	Initial Eigenvalues					
	Total	% of Variance	Cumulative %			
1	6.60	27.5	27.5			
2	2.79	11.6	39.1			
3	1.55	6.44	45.6			
4	1.38	5.74	51.3			
5	1.23	5.13	56.4			
6	1.08	4.49	60.9			
7	1.01	4.20	65.1			

 Table 5-12: The Eigenvalues over one and related variances for the WHOQoL-BREF questionnaire completed by patients

Extraction Method: Principal Component Analysis

Table description: Table 5-12 shows the eignvalues more than 1 and related variances for the WHOQoL-BREF questionnaire completed by patients. Principal components analysis revealed the presence of 7 components eigenvalues exceeding 1, explaining 27.5%, 11.6%, 6.44%, 5.74%, 5.13%, 4.49%, and 4.20% of the variance, respectively.

Conclusion

In the first phase of this study, QoL scores of patients and nurses were analysed using parametric tests because the data were considered continuous and generally met perquisites of a normal distribution. Descriptive outcomes related to demographic and clinical variables of patients and nurses were presented through a number of tables and figures. Several important findings were found as a result of different statistical tests:

- 1. Correlations of r and ICC ranged from 0.06 in the social relationship domain up to 0.50 in the physical domain indicating a poor to moderate association between patients' and nurses' QoL scores.
- 2. Bland-Altman test revealed a 6 point (out of 20) difference between each patient-nurse score (around 30 % differences).

- 3. Bland-Altman test also revealed that there was no obvious association between the level of agreement and the patients' level of QoL.
- 4. Paired sample t-test indicated that nurses' scores were significantly lower than patients' scores in the social relationship and environmental domains.
- 5. Bivariate correlation (Pearson correlation) identified that a number of patients' and nurses' variables were associated with the level of agreement. However, at the Multivariate level (using Standard multiple regression) only a limited number of these variables were found to be associated with the level of agreement including (a) where the patients were treated (outpatients versus inpatients) was associated by the level of agreement in psychological and social relationship domains; and (b) nurses' rating of their own QoL was associated with the level of agreement in the environmental domain.
- The test of the reliability of the WHOQoL-BREF questionnaire using the internal consistency test showed that except the social relationship domain, Cronbach alpha values approached or were above 0.70 for both patients and nurses scores.
- 7. The exploratory factor analysis for patients' scores using the WHOQoL-BREF questionnaire identified 7 factors (or domains) with the eignvalues more than 1. In other words, based on patients' responses items of the WHOQoL-BREF questionnaire can be grouped into 7 domains.

No further interpretations of empirical findings are made in this chapter. The above important empirical findings will be used in the discussion and the concluding chapters, which will also incorporate interpretive outcomes that seek to answer the research questions, make conclusions, and provide suggestions for future research studies.

CHAPTER SIX: METHODS OF THE SECOND PHASE

Introduction

In the previous chapter, results of the first phase using different statistical tests were presented. These statistical tests were those that measured the level of agreement at the individual level (between each patient-nurse score) and at the group level (patients as a group with nurses as a group).

This chapter explains and justifies the methods that were used in the second phase in order to deeply understand nurses' perceptions about cancer patients' QoL. Two important issues need to be addressed here. Firstly, while this chapter is in fact a method chapter it can also be seen as an audit trail chapter that shows what steps were undertaken in the second phase. It is important for researchers to elaborate on methods and processes they used in interpretive studies so that other researchers can also scrutinise them. In the next chapter the results of the second phase will be provided. This chapter is mainly concerned with the methods used in the second phase. However, these methods incorporated some features and outcomes of the first phase. In other words, some outcomes of the first phase were used to construct the second phase. These issues are further elaborated on throughout this chapter.

Aim of the second phase

In the second phase, the major aim of the study is to explore why nurses' perceptions about cancer patients' QoL is different from those of cancer patients' own perceptions.

Research methods

As explained in the methodology chapter, the second phase of the study was based on the principles of a classical version of grounded theory. While the processes of data collection and analysis in an interpretive study are interwoven, for the purpose of easy understanding they are discussed in two separate sections.

Data collection

In the first phase, the WHOQoL-BREF questionnaire was used in order to collect data, and a complementary interview method was used in the second phase. In the following section, a justification is given as to why the interview has been chosen as the data collection method in this phase of the study.

Why use interviews

While interviews or participant observation or a combination of them are the main way of data collection in a research study based on a grounded theory methodology (Stanley 2006), there is no limitation on the kind of data to gather because different kinds of data give the analyst different views (Leedy & Ormrod 2005). Such different views are called "slices of data" (Glaser & Strauss 1967, p.65). However, an interview was found to be more suitable for this research study because it allowed cancer patients and nurses to talk more openly about their QoL perceptions. In fact, in a QoL research study researchers might combine QoL tools (that was used in the first phase) with interviews for a variety of reasons. Interviewing is a good way for respondents to express individual information that may not be revealed in a questionnaire (Aaronson 1988; Grant & Rivera 1988). It is particularly applicable to a variety of patients such as elderly, poorly educated, and moderately or severely ill groups (Aaronson 1988, 1989; Mandelblatt & Eisenberg 1995) and therefore can mitigate the problem of missing data (Aaronson 1989).

Cox (2003) in a study that used interview and questionnaires to compare QoL of cancer patients, discussed reasons why these two ways of collecting data do not always match with each other, probably related to the fact that questionnaires usually ask patients to express their feelings over a specific time, for example one week or two. By then researchers may be able to compare participants' responses better because they were based on experiences that occurred in a similar time frame. Conversely, interviewing aims to go beyond such limitations and allows patients to talk widely and deeply about their experiences.

Anderson et al. (1986) in a study that compared self versus interviewer modes, states that self-report in QoL assessment is a good approach in terms of saving time and cost; however, the validity and reliability of results should also be considered. They concluded that an inquiring interview mode is more sensitive than a questionnaire to assess Qol accurately. Gill and Feinstein (1994) are of the opinion that along with standard QoL tools, some supplemental items should be provided so that clients have the opportunity to express their individual opinion and reactions further. Cella (1996) emphasised that it is reasonable to use a self-report QoL questionnaire with an exploring interview whenever possible, because the richness and completeness of interviewed data can never be compared with self-report information.

As the research study was about patients' and nurses' perceptions about cancer patients' QoL, the researcher realised that participant observation cannot add any more to his understanding than interviews. It also was not feasible to make observations throughout a variety of inpatient and outpatient oncology wards and across different hospitals. Given the fact that oncology patients are usually sick and undergoing palliative or curative cures or treatments, observation might be considered intrusive. Therefore, in the second phase of the study the interview was chosen in order to find out why nurses are different in their perceptions about cancer patients' QoL.

Ethical considerations

Like the first phase, for conducting this phase of the research study, ethics committee approval was gained from relevant ethics committees of two hospitals. These committees are linked to Flinders University and approval from them is considered as Flinders University Ethics Committee approval. Both hospitals are located in South Australia. The major difference between the two hospitals is that in one of them cancer patients receive services in a specialised oncology ward whereas in the other one such patients were distributed across different wards (non-specialist oncology wards).

Similar to the first phase, the letters of support from the Director of Oncology and Director of Nursing of two hospitals were accepted and sent to the ethics committees in order to get approval to conduct the research study. Given the fact that the aim of the study was to include the cancer patients having palliative care, another letter of support was accepted from the palliative system.

Recruiting the participants was conducted by the researcher (see pages 124 and 125 for more details). Before starting the interviews, patients and nurses received a participant information sheet. More verbal information was given to participants if they so requested. The aim was to be sure that participants were completely aware of all aspects of the study. The major issues discussed with them were about the aim and main research questions, significance of the study as a postgraduate study, voluntary participation in the study, ability to withdraw from the study at any stages, and confidentiality of personal information. If they were satisfied, then they were asked to sign another consent form. The participant information sheets that were used in one of hospitals are shown in **Appendices C** and **D** for patients and nurses separately. The same sheet with only small modifications was used in the other hospital.

In order to prevent possible emotional distress, supportive care was negotiated so that the Clinical Nurse Consultant or a more experienced nurse was available to talk with participants. While there was no obvious report of such distress from patients and nurses, minor emotional changes might have occurred within the participants due to the nature of face-to-face interviews.

Interview questions and strategies used in the research study

During the interview semi-structured questions were used. However, one of the most important and challenging parts of the interview process was to prepare the interviewee's questions. As explained earlier, the major aim of the second phase was to explore why nurses differ in their perceptions about cancer patients' QoL with those of the patients. In other words, what is actually going on between cancer patients and nurses that make their perceptions different?

General open-ended questions were prepared that guided interviewees through the process that fitted in with the grounded theory approach on the interaction between

cancer patients and nurses. It is more common to use the 'how' and 'what' questions in interpretive studies particularly those based on the principles of a grounded theory methodology. This is because the focus is mainly on the processes (Stanley 2006; Wilson & Hutchinson 1991). 'Why' questions are asked cautiously in interpretive studies because "the question of why things happen the way they do can lead to inferential leaps and empirical speculations that propel qualitative analysis far from its stock-in-trade" (Gubrium & Holstein 2000, p.502). Therefore, in this research study research questions were based on more 'what' and 'how' questions which were then named as *essential questions*. These questions for patients and nurses were relatively the same with only small modifications. Some of these questions are stated below:

Nurses

- 1-When you hear the concept quality of life what do you think of? What comes to mind?
- 2- How do you usually assess cancer patients' quality of life?
- 3-What factors do you think influence your assessment of cancer patients' quality of life?
- 4-How do you know whether your assessment of a patient's quality of life is accurate or not?
- 5-What are the outcomes if your assessment of cancer patients' QoL is similar or different from that of cancer patients' own perceptions?

Patients

- 1-When you hear the term quality of life, what does it mean to you?
- 2-How do nurses usually assess your quality of life?
- 3-How can nurses have a better understanding of your quality of life?
- 4- In what ways do you think nurses' understanding of your quality of life affected your actual quality of life?

As well as essential questions, it was also necessary to ask some *extra questions* due to the nature of the grounded theory (see the following sections about theoretical saturation and theoretical sensitivity). For example, in the first interviews it was identified that interviewees were talking about the individuality of the QoL concept. The researcher was sensitive to understanding if the next interviewees would also express their ideas on whether the QoL concept has different meanings for different

people. As Minichiello et al. (2004, p.422) point out, researchers must not "rely too heavily on the initial interview guide and ask the same questions and in the same order".

Because the second phase of the study is based on the principles of grounded theory methodology, it also helped the researcher to incorporate and elaborate on some significant results from the first phase. For example, the researcher was interested in whether patients and nurses agree that nurses with more clinical experience with cancer patients have a better understanding of some aspects of such patients' QoL (one of the significant results of the first phase) and if yes, why such differences exist. Therefore, the *elaborating questions* were prepared. Some examples from results of the first phase were applicable to the second, particularly when talking with participants about influencing factors of agreement. For example, when the researcher asked nurses about factors influencing their assessment of cancer' patients' QoL, he also queried their ideas about the effect of nurses' experiences. The researcher asked, for example: 'how does your experience as a nurse affect your assessment of cancer patients' QoL?'

Probing questions were also developed. These questions help an interviewer to stay on track while participants are encouraged to talk widely about the topic(s) in question. Probing questions are those that ask participants for: elaboration (providing more details); clarification (resolve confusions); completion (finish a talk); communication (keep talking); evidence (providing more facts); and attention (seeking participant attention) (Liamputtong & Ezzy 2005). Given the fact that the researcher, as an interviewer, had English as a second language, the clarification probes were the most used. For example, sometimes the researcher restated or rephrased interviewees' unclear statements to ensure that his understanding was correct. Such probing questions were quite necessary as the participants sometimes used slang words or phrases that did not make sense to the researcher.

At the beginning of interviews, some *demographic questions* were also asked so that participants could provide a background about themselves, such as their age and their clinical experience. However, if participants for any reasons did not want to talk

about some of their demographics in detail, the researcher did not repeat the question.

In order to ensure that participants were completely relaxed, open and did not have any concerns during the interview, some *assuring questions* or statements were used to ease the situation and to make the interview atmosphere friendlier, for example: "If there is anything or any concern at the beginning, please feel free to let me know otherwise we can start". Toward the end of the interview, participants were encouraged to feel relaxed and open and tell the interviewer about anything which was related to the issue under study. For example, the researcher stated, "This is now the end of the interview. If there is anything else you want to discuss, please let me know".

In some cases the interviewees emphasised that they were not sure if their experiences were important or relevant. The researcher advised them that they were. This was an important strategy to remind the interviewee that what is important or relevant are what the participant thinks is important. The following statement can also be very helpful:

The world is a serious place where people who are directly involved in it can know completely what it is like. You are an expert and I meekly beseech your help in gaining a more complete - never complete - understanding of it. (Liamputtong & Ezzy 2005, p.62)

An attempt was also made to be silent as much as possible and try not to interrupt the interviewees when they were talking. Altogether, the research questions consisted of essential, extra, elaborating, probing, demographic, and those questions or statements that ensured that interviews were going smoothly and participants felt relaxed.

Participants

After finishing the data collection for the first phase, three cancer patients and 10 nurses were selected to take part in the second phase of the research study. It should be noted that while the number of nurse interviewees was three times more than patient interviewees, the interview period for patients was significantly more than that with nurses. Given that the main focus of the interview questions was on the

QoL assessment as performed by nurses, generally it appears to the researcher that nurses can provide more detailed information about the process of QoL assessment that what patients might do.

The main aim of the second phase was to further understand why nurses differ in their perceptions about cancer patients' QoL with patients' perceptions and to elaborate on some results emanating from the first phase. Therefore, conducting the qualitative phase in a sequential order after the quantitative phase proved more suitable than doing both phases simultaneously. Although sequential sampling takes time it is a very useful approach, particularly for purposes of elaborating on results of a preliminary phase (Teddlie & Tashakkori 2003).

Patients had been diagnosed with prostate, lung, and breast cancer with metastasis to some organs. Two were male and one was female. One of the nurses was male and nine were female with clinical experience ranging from only one year to around 30 years. The nurses worked across different inpatient and outpatient services including the palliative system. One patient experienced both oncology and palliative areas. Seven nurses also worked in both oncology and palliative areas. While some PhD theses have described participants in more detail (Long 1998; Power 1998), this research chose not to do so. While choosing a number or a pseudonym can help to maintain confidentiality, Zollo (2002) has pointed out readers may be able to make accurate guesses about the participants' identity, for example, through their demographic characteristics.

The sampling strategies

In this research study two strategies were used by the researcher to access participants. The first sampling strategy used in this research study was convenience sampling. The researcher's first contacts were with participants of the first phase because he had communicated with them for more than one year and built a good rapport and relationship with them. Glaser & Strauss (1967) noted that rapport is an important aspect of sampling in a grounded theory research study that is time-consuming. Moreover, a number of participants of the first phase were also very thoughtful, articulate and wanted to talk about their experiences in the second phase.

More experienced participants proved to be good interviewees due to their willingness to talk, knowledge and experience (Morse 1991, cited in Cutcliffe 2000).

Along with the first strategy, snowball or chain sampling was used in which the researcher asked the initial more experienced participants to nominate other interviewees who might be suitable and happy to take part in the study. As well as establishing a good rapport between the researcher and more experienced participants, they claimed that they were also aware of the importance of the research study for cancer patients. Cutcliffe (2000, p.1478) pointed out that more experienced participants can work as a "gatekeeper" to introduce subsequent interviewees.

On reflection, the researcher personally also found that more experienced and knowledgeable participants can provide more information. They are in the environment for quite a long time and have more experience in talking about the focus compared with the less experienced. They also may work better at introducing the subsequent interviewees as a gatekeeper. Introducing a participant by the previous participants may be criticised as leading to less diversity in the sample. However, the researcher personally found this technique very helpful, particularly as an international student who was very new to the sampling environment.

Theoretical sampling

While the main strategy for data collection used in this research study was convenience sampling, the framework of sampling in the second phase consisted of theoretical sampling. As the categories emerged through the first interviews, a decision was made about who was the best person to interview during the next step and where the researcher could find them. For example, in the first interviews, nurse interviewees articulated that the best way for QoL assessment is to improve the relationship and rapport with patients rather than using QoL tools even for palliative patients. This encouraged the researcher to select one patient in the next interviews from the palliative area to elaborate on that issue further. Theoretical sampling is the main feature in a study based on the principles of grounded theory and "is the process of data collection for generating theory whereby the analyst jointly collects, codes, and analyses his data and decides what data to collect next and where to find

them, in order to develop his theory as it emerges" (Glaser 1978, p.36; Glaser & Strauss 1967, p.45). Therefore, theoretical sampling does not necessarily mean sampling persons; it also means sampling "situations, events or processes which are regarded as theoretically relevant to the emerging concepts" (Llewellyn, Sullivan & Minichiello 2004, p.215). Theoretical sampling does not result in success unless researchers are theoretically sensitive.

Theoretical sensitivity

During the whole process of theoretical sampling, researchers need to be "sufficiently theoretically sensitive so that he can conceptualise and formulate a theory as it emerges from the data" (Glaser & Strauss 1967, p.46). In other words, to achieve theoretical sensitivity researchers must enter the field by putting aside any preconceived ideas and generate a theory based on the data (Glaser 1978; Jeon 2004). To achieve this sensitivity as much as possible, the following strategies were used in the second phase of the research study.

The first strategy to achieve more sensitivity during the process of data collection and analysis was to put aside any preconceived ideas and let the data speak. The researcher conducted a line by line analysis more than any other method. Glaser believes that theoretical sensitivity is a process achievable only by immersion in the data, line by line, comparison by comparison, memo by memo, and code by code and let the data speak (Walker & Myrick 2006).

The second strategy that improved theoretical sensitivity in the second phase was information arising from conducting the literature review relevant to the topic. This information was very useful in order to be familiar with current concepts and to develop other conceptual concepts during the coding process. To achieve this, a comprehensive literature review was conducted toward the end of the second phase in the coding process. The research study had a preliminary empirical phase and in order to narrow the field, find existing gaps, construct research questions, and justify using a grounded theory approach, conducting a literature review at the beginning of the research study was completely necessary. Glaser and Strauss (1967) highlighted that although a preliminary literature review may be conducted before a research

study using grounded theory methodology, a full literature review is done at the coding stage in order to allow the theory to arise completely from the actual data and not to be influenced by pre-existing knowledge or perceptions of truth.

Altogether, in the second phase theoretical sensitivity was mainly achieved by relying on the actual data, by conducting a literature review of both phases, and by entering the field and the knowledge arising from the first phase. Theoretical sensitivity helps researchers to achieve to the point of theoretical saturation.

Theoretical saturation

Theoretical saturation is another feature of a grounded theory approach whereby:

no additional data are being found whereby the sociologist can develop properties of the category...He goes out of his way to look for groups that stretch diversity of data as far as possible, just to make certain that saturation is based on the widest possible range of data on the category. (Glaser & Strauss 1967, p.61)

In fact, when researchers are satisfied that the data are rich enough, then the sample is large enough (Llewellyn, Sullivan & Minichiello 2004). In this research study, after nine interviews, the researcher felt that for a few categories no new or additional information would arise. From this point onward only some interviews were conducted to ensure that the categories were saturated. There were time constraints that did not allow for all categories to be saturated. As explained in Chapter Three (see page 59), in this research study a core category was not sought (i.e. a theory was not being developed). Therefore, the researcher considered the saturation only for important categories that were in line with research study aims.

Strauss and Corbin (1998, p.136) argued that in reality if a person looks "long and hard" in the data there is always a possibility for the "new" to emerge but at the point of data saturation such "new" cannot "add much more to the explanation at this time" (Strauss & Corbin 1998, p.136). In other words, as researchers proceed through their theoretical sampling, they will develop different categories and not all of them have the same importance. "Core categories with the most explanatory power should be saturated as completely as possible" (Glaser & Strauss 1967, p.70). Saturating less relevant categories depends on the resources available. Therefore, researchers need

to balance the depth and breadth they are looking for in the saturation process with the resources of time and money available, and make a judgment about additional sampling (Liamputtong & Ezzy 2005). For example, researchers might limit their study in order to make a final conclusion (Llewellyn, Sullivan & Minichiello 2004) and leave the results to be developed in future research studies.

Inclusion and exclusion criteria

All adult cancer patients under different treatments and in different wards (inpatients or outpatients) who were able to articulate well in English about their experiences and had consented to be interviewed were included in the research study. Included nurses also were all registered (RNs and ENs) to work in South Australia, spoke in English and had consented to take part in the study. According to ethics guidelines, the participants were able to withdraw from the study at any time.

The interview process

Before starting the interviews with cancer patients and nurses, the researcher conducted interviews with three university lecturers at Flinders University with the aim of improving his interview abilities. This was the first experience of the researcher doing an interpretive study using the interview method, and provided him with the opportunity to receive feedback from more experienced people and to improve his interview technique. Generally speaking the feedback was positive and therefore the researcher felt that he could start his actual interviews. This is a good technique particularly for less experienced people using semi-structured interviews and is comparable with a pilot study in an empirical research study.

Data were collected from each participant through a semi-structured interview during a nine month period from June 2006 to February 2007. Patients and nurses were generally interviewed in oncology units. There were some exceptions in which cancer patients or nurses were interviewed in their offices as they requested. Generally speaking, the interviews were conducted in a quiet and private atmosphere at a time suitable for participants. Only one patient was interviewed while he was under chemotherapy with ward curtains around to make the environment more private. Also, when the researcher interviewed two nurses in the oncology wards, we

128

had to change the location due to interruptions. After that, the interview proceeded well and the interviewees apologised for such occurrences. The researcher did not have any issues regarding the conduct of interviews in the clinical area. Participants were generally talkative and were interested in expressing their thoughts and feelings.

The patients' interviews lasted approximately one hour. Before interviews were conducted, the general interview questions were given to some participants so that they had enough time to think about them deeply. While the researcher was taking some notes during the interviews, the interviews were audio recorded on an MP3 player which enabled the researcher to transfer the audio files into the computer and transcribe using Express Scribe software and link into the NVivo software. There was also on the MP3 player a pause button that the interviewe could press whenever they wished throughout the interview if they did not want certain comments recorded. While no participant used this button, this is a helpful strategy that allows interviewees to feel free and to talk about any things if they can stop recording (Grbich 1999).

The researcher also arranged a time before and after recording the interviews for interviewees to talk. The time before recording is in fact a settling-down period whereas post-recoding time is one that provides interviewees with the best possible opportunity to talk more about those issues that they did not want recorded. Some valuable comments that interviewees make may come before or after an interview begins (Grbich 1999). For example, a couple of nurse interviewees briefly pointed out to problems related to the health system that might cause a QoL assessment to be conducted inappropriately. This made the researcher theoretically more sensitive to trace the issue more fully in the next interviews.

Data analysis

Transcribing the interviews

The interviews were transcribed verbatim by the researcher. Transcription is a timeconsuming activity and might take around 6-10 hours of typing for each hour of interview tape (DePoy & Gitlin 1998). As English was the researcher's second language, transcriptions took around 16 hours per one hour interview. There was no record of non-verbal effects such as change in the voice of participants and possible emotional changes in transcriptions. The interviewee could make any modifications to the transcript and return it to the researcher. Six of them had positive responses and requested to see the transcribed interview. They made minor changes such as punctuations, adding or deleting some words or small sentences, and then returned it to the researcher.

Using NVivo software to facilitate the data analysis

The NVivo version 7 computer software was used to analyse the textual data. However, the researcher also found it useful to do some aspects of the data analysis such as coding and categorising concepts manually. This software is one of the products of the Qualitative Research Solutions International (QSR). The original name of the NVivo is in fact Non-numerical Unstructured Data (NUD). This programme was then improved and named as NVivo. NVivo utilises participants' own words to name categories during the coding process (Bringer, Johnston & Brackenridge 2004).

Using computer software like NVivo to analyse the interpretive data is a controversial issue (Richards 1999). Charmaz (2000) stated that using such computer software might impose limitations on the study particularly in a grounded theory study. This includes a poor understanding of grounded theory methods, using the software to "legitimate" the study rather than to conduct it, its usefulness more for objectivist modes of grounded theory rather than constructivist versions, and finally reducing rich interpretation of qualitative data to a set of procedures (Charmaz 2000, p.520). Occasionally the researcher also thought that such computer software might hinder his rich interpretations of data by constructing a distance between him and the actual data. In order to avert this, the transcribed interviews were read several times and the data coded by hand as well as with software.

On the other hand, soon after starting the data analysis, the researcher was faced with a bulk of textual data that appeared very difficult to analyse manually. It was difficult for the researcher to even imagine how he could do the coding of the textual data on the floor by cutting and collecting the same texts with each other. Charmaz (2000, p.520) stated that such programs are useful for "data management" due to huge amount of qualitative data. Moreover, having the capacity to integrate all information together, this therefore can also maximise the "transparency" of the audit trail particularly within a grounded theory study (Bringer, Johnston & Brackenridge 2004, p.247). It seems that this software is particularly consistent with the grounded theory methodology (Lonkila 1995, cited in Bringer, Johnson & Brackenridge 2004), which generally needs a line by line coding and analysis. The memo option on this software, which is an essential part of the grounded theory method, in particular helped the researcher to write his memos during the coding process.

Comparative data analysis

The data analysis process was based on the principles of the Glaserian (classical) version of grounded theory at the same time as data collection, using comparative data analysis. Comparative data analysis is another feature of a grounded theory approach in which a researcher collects the data, makes comparisons between data through the analysis, and aims for more data collection if it is necessary in order to reach the point of theoretical saturation (Browne 2004). In other words, in grounded theory it is impossible to do theoretical sampling and reach theoretical saturation without coding and analysing at the same time (Glaser & Strauss 1967). This is in contrast to other ways of data analysis in which it can be done with already collected data or after finishing the data collection. For example, in content analysis the categories are identified prior to coding or in post-structural and in semiotic analysis the focus is more how a text is produced rather than its content (Liamputtong & Ezzy 2005).

Therefore, the following steps were conducted:

1-Providing a reflective journal: The researcher prepared a reflective journal and wrote down any ideas that came to his mind such as personal thoughts, theoretical ideas, and any concerns about the research study before data collection and analysis. It is very important to write reflective thoughts as soon as possible otherwise they may be forgotten. Grbich (1999) commented that a reflecting process at least allows

one to think about their beliefs and values and how they might affect their interactions and the interpretations in the research setting.

2-Memo writing: The notes from the reflective journal were then transferred into the NVivo software as memos and the process of memo writing was continued all through the coding process using the NVivo software. Memo writing is one of the major features of grounded theory. Glaser (1967) points out that the coding process or making categories can produce conflict in researchers' minds. He suggests in these situations that researchers "stop coding and record a memo" on their ideas (p.107). This can help researchers prevent possible breaks in their logic (Strauss & Corbin 1998), develop more "abstract thoughts into more clear thought-out ones" (Bringer, Johnston & Brackenridge 2004, p.254) and reach the point of theoretical codes (Jeon 2004). When a researcher has codes and available memos, this will help them write their analysis (Charmaz 2000). Consider, for example, the following two memos the researcher wrote during stages of the analysis process:

October 2006

I realised that the majority of participants in some parts are talking about the individuality of the QoL concept. It has also been addressed in the literature that QoL is an individual experience. I asked myself what they mean by the individuality. It was necessary to ask in later interviews what do they mean by the individuality or do they believe that QoL is an individual concept.

February 2007

During the phase of coding, I understood that participants are talking about the QoL concept in any way. At first three subcategories under the category of QoL concept emerged which were named QoL characteristics, QoL aspects, and QoL definitions. In the subcategory of QoL characteristics, I had properties of a complex issue, a broad statement, an individual concept, and an ever changing type event. I was confused if subcategories of QoL aspects and QoL definitions can be merged into another category of QoL characteristics or not. Then, I realised that in fact in the subcategory of QoL aspects, participants are talking about different things that increase the breadth of the QoL concept. Therefore, I merged the subcategory of QoL aspects into the property of a broad statement. The subcategory of QoL definitions at this moment can stand for itself.

3- Naïve reading of the transcribed interviews: The transcribed interview files were naïvely read several times. The first reading coincided with the listening to the relevant audio-recorded file. While the major aim was to become immersed in the data, this also helped the researcher to ensure that interviews were transcribed

verbatim. As pointed out by DePoy and Gitlin (1998), it is very important for transcriptions to be read several times so that during the analytical process the researcher is accurately immersed in the data.

4- Importing of the transcribed interviews into NVivo: The transcribed files were then imported into the NVivo software for further analysis.

5- The coding process: Coding is another feature of grounded theory (Kendall 1999; Walker & Myrick 2006). While in Strauss' version, the coding consists of three stages of open coding, axial coding, and selective coding (Strauss & Corbin 1998), in the Glaser version, none of the above coding systems can be identified (Glaser & Strauss 1967). Instead, they describe four stages for the constant comparative method: (1) comparing incidents applicable to each category; (2) integrating categories and their properties; (3) delimiting the theory; and (4) writing the theory. While in the first and second stages a researcher tries to code data and develop categories and their properties, in the third stage, a researcher attempts to modify and reduce categories by constant comparison method reaching to a core category. When a researcher is confident about categories, in the final stage the theory is written based on categories and related memos (Glaser & Strauss 1967).

Glaser (1978) in his later publication used two major coding systems to describe these stages: "substantive and theoretical coding". Despite the ambiguity of this coding, this is more understandable compared with the previous one. The substantive coding consists of open and selective coding which allows the analyst to "conceptualize the empirical substance of the area of research. Theoretical codes conceptualize how the substantive codes may relate to each other as hypothesis to be integrated into the theory" (Glaser 1978, p.55). What is clear is that there is no axial coding in Glaser's mode and in fact it seems that axial coding is integrated into the substantive 2006).

When the researcher reflected on the coding process, it proved to be the hardest part of the second phase. This has been reported in other PhD studies. Redden (2005, p.108), for example, in her thesis reported that "deciding how best to use grounded theory to analyse the data for this work was not easy". The researcher found coding, conceptualising, and categorising very difficult, time consuming and sometimes completely frustrating. The words of participants sometimes seemed to have different meanings. It was only near to the end of the coding and categorising that the researcher got a comprehensive and meaningful understanding of the data. Anxiety about nothing emerging at the beginning of coding is a source of concern for many students that will change later (Glaser 1978). Auerbach and Silverstein (2003, p.32) point out that sometimes beginners feel they are "immobilised" because they assume that there is no right way to interpret data. They might feel that nothing really is directly related to one's research concerns. This stage is named finding yourself "a drift in a sea of data".

The process of coding in this study consists of the first step (substantive) of the Glasererian (classical) version. In other coding, the following steps were conducted:

5-A) Open coding: This process consists of three stages: reading the first transcription and identifying the relevant text, conceptualisation, and using comparative data analysis and memo writing for subsequent transcriptions.

During the first stage, after reading the first transcription, the researcher tried to delete unrelated textual materials and identify the relevant text. The relevant text also named the relevant incidents/ideas/events that in fact were related to research questions. This process in NVivo was named creating 'free nodes'.

In the second stage, the researcher tried to conceptualise the incidents in the first interview and this meant giving them a name. In order to name concepts, the researcher usually used 'in vivo codes', which in essence means taking the concepts' names from the participants' words (Strauss & Corbin 1998). However, there were some cases where the researcher chose concepts' names from his own words or from the literature. For example, one interviewee said to me that 'quality of life would be to enjoy life'. The researcher coded it as 'having enjoyment' which is in fact driven from the interviewee's own words.

Finally, open coding of the subsequent interview transcriptions was then conducted based on comparative data analysis and related memos. Similar names were given to

the emerging codes if it was applicable when the researcher compared them with the previously coded data, otherwise the researcher used new names. It was also necessary sometimes to rename previous concepts. Generally speaking, the outcome of open coding was having numerous concepts and reducing the textual data. The process of open coding can be seen in Figure 6-1.



Figure 6-1: Open coding process

5-B) Selective coding: This step was in fact similar to the previous one but instead of finding the relevant text (incident) and comparing it with other incidents, an attempt was made to compare the concepts with concepts and try to reduce their number by collapsing them based on similar characteristics. This finally led to a list of categories and subcategories. This was called selective coding because it is when the researcher focused on the list of major or core categories and tried to reach saturation point. This in turn, becomes a guide to further data collection and theoretical sampling (Glaser 1978). This step of coding is more in line with creating 'tree nodes' in the NVivo. In fact, the researcher tried to link smaller concepts in order to create a broader category.

For example, 'nurses are fantastic in assessing quality of life' and 'nurses are exemplary in assessing quality of life' are collapsed into one category named 'nurses are the best people to assess QoL'. Or in a later stage of theoretical coding, categories like 'having freedom', 'having enjoyment', and 'having choice' were classified under a broader category 'quality of life meanings'. The process of open coding can be seen in Figure 6-2.



Figure 6-2: Selective coding process

6-Improving the credibility (rigour) of findings:

The issue of rigour in interpretive studies is a controversial one in that rigour is more about covering all aspects of the phenomenon being investigated with less disturbance to the setting and a clear interpretation and presentation of the results (Grbich 1999). Auerbach and Silverstein (2003, p.78) proposed using the concept of "justifiability" to show the rigour in an interpretive study instead of reliability and validity, which are used in empirical studies. By justifiability, they mean that data need to be transparent, communicable, and coherent.

For "transparency", a researcher needs to show how the results are created so that other researchers can also follow how they did it. It does not mean that others agree with your interpretation but presents how you arrived at your interpretations. Data have "communicability" if categories you create make sense not only to other researchers but also to participants. Finally, if your theoretical constructs fit together so that you can tell a story about them, then your data are "coherent" (Auerbach & Silverstein 2003, pp. 84-85). In order to address the issue of transparency in this research study, the detailed process of the second phase, coding and conceptualisation have been addressed in this chapter. Moreover, the categories explored in this study (presented in the next chapter) were discussed and made sense to other PhD researchers as a linked and sound review.

In summary, the issue of rigour in the second phase of the research study has been addressed in relation to general criteria (transparency, communicability, and coherency).

Conclusion

In this chapter the methods used in the second phase of the research study in order to collect and analyse data were discussed. It was argued that semi-structured interviews provided the researcher with this opportunity to get an in-depth understanding of why nurses' perceptions about patients' QoL are different from patients' own perceptions. Moreover, conducting interviews allowed the researcher to elaborate on some significant results of the first phase. An attempt was made to clarify the steps undertaken for data collection and analysis. These steps incorporated the process of comparative data analysis undertaken by the researcher. These steps show how the researcher arrived at the results of the second phase, i.e. the conceptual categories, are addressed.

CHAPTER SEVEN: RESULTS OF THE SECOND PHASE

Introduction

The purpose of this chapter is to describe the categories derived from the qualitative data using an interpretive approach based on the principles of a grounded theory study. In the next chapter, these categories will be discussed in conjunction with the quantitative findings, and with reference to the literature, where this is possible

In this study, exemplars are provided to elaborate on the categories. This allows the features of the experience to be portrayed (Benner 1985). It helps to provide an understanding of how conceptualisations have been made from the qualitative data. A number was allocated as an identifier for each participant, rather than a pseudonym, and these numbers are stated when exemplars of interviews are presented. 'Patient' was used to refer to patient interviewees and similarly 'Nurse' for nurse interviewees. The exemplars numbers do not relate to the chronological order in which interviews were performed. For example, 'Patient 1' does not mean it was conducted before 'Patient 2'.

Selected exemplars naturally cannot reflect all of the nuances of a category or subcategory. Moreover, these conceptualisations have not been made solely on the basis of interview data. Other inferences emerging, for example, from the quantitative data of the first phase, from memoing and reflecting on the texts as a whole and from the literature also helped the researcher in the process of coding and conceptualisation. The exemplars presented in this chapter have been chosen because they exemplify or reflect a particular feature of that category.

Results

After data collection and analysis, the following main interpretive categories were identified:

1-QoL aspects;

- 2-QoL meanings;
- 3-Cues-based QoL assessment;
- 4-Purpose-based QoL assessment;
- 5-Facilitators of QoL assessment; and
- 6-Barriers to QoL assessment.

The following sections present each category and sub-category. Each sub-category section ends with a summary followed by a conclusion at the end of the particular category.

QoL aspects

In response to questions put to the interviewees, participants expressed that individuals' QoL influenced by a whole range of aspects in life including physical, psychological, social relationships, family interactions, financial, environmental and spirituality issues. Consider, for example, the following extracts:

They [nurses] didn't ask me about my quality of life, they did not ask me where I lived or anything about my quality of life. In fact there was a lot happening in my life at the time which it would help me if they had asked me about. Who was my primary contact person, where I am living or how is it going for me? What is happening in my life? Who do I want information shared with? What's been happening in my life?...There is a whole range of things around quality of life. (Patient 3)¹

Immediately I think it [quality of life] is a very big and too generalised concept. When you look at quality of life a little more deeply, I have to say that it would have to be based on how or what you getting out of life from the experience that you are having at a particular point of time, whether it would be something that is influenced by your physical health, whether it is something influenced by emotional things going on in your life, whether it is a combination of things influencing the quality of life for you. (Nurse 3)

Both nurses' and patients' interviews articulated that individuals' QoL can be influenced by a range of factors. Some nurses in the first phase commented that they did not agree with several items in the WHOQoL-BREF questionnaire or they did want to see some other themes noted in the tool. This encouraged the researcher to write a memo about items in the questionnaire and to further search for QoL aspects in the second phase. Consider, for instance, this memo that the researcher wrote:

¹ Words in the [] added by the researcher for clarification of the textual segment.

It is apparent from the literature that QoL is a multi-dimensional concept. It is also clear that participants share this view about the complexity of the QoL concept. For example, participants described the term QoL with characteristics like very big and very broad. QoL has different aspects but do participants of the study agree with them or suggest other aspects? Do participants value a specific aspect more than others? It is therefore important to focus on QoL aspects and ask participants further questions to elaborate on the broadness of the QoL concept. (29/02/07)

The category 'QoL aspects' was created to reflect on how different aspects may impinge on an individuals' QoL. These aspects are explained in the following sections.

Physical aspects

An important aspect of the QoL concept highlighted by participants was physical abilities:

Eating, normal eating, going to the toilet, mobility, survival basically...I can function. (Patient 1)

I can still get out and about and live almost at the same level as I had in the past. There are some restrictions as far as what my capabilities in some situations although I still do quite a lot of work. (Patient 2)

Physical functioning at the level acceptable to individual patient participants is an important aspect of QoL. This can range from basic activities such as eating, mobilising and elimination, to more complex activities identified by patients.

Nurses also saw physical functioning as an important part of QoL:

[Quality of life] for the patients would be ultimate wellness, feeling well, and being physically able to do, being financially stable, and having support from your family, being able to eat and drink comfortably, being able to have usual bowel and bladder comfortably, not restriction in your life, just being able to function normally like every one else. (Nurse 3)

Get out of the bed, pop on the car, go to work, do the work, return home... This is the physical part of quality of life. (Nurse 7)

Therefore, an important aspect of QoL for both patients and nurses is physical functioning. In every day life individuals are doing a number of basic activities like

eating, mobilising, and elimination that contribute to their QoL. It is very important for participants to work and have function as normally as possible.

Psychological aspects

Even though the physical aspect is a very basic and important aspect of QoL for participants, it was evident from the interviews that other aspects of QoL were also important. Some participants pointed out the psychological aspects as an essential part of their QoL. In the subsequent exemplars, nurses expressed that:

I think that quality of life has facts of well being, mentally or psychologically feeling well, physically feeling well, financially feeling well, and support from your family (Nurse 1).

Persons' quality of life, their perception is going to change daily; sometimes several times maybe in a day according to how they are feeling about their physical and psychological health (Nurse 2).

Similarly, patient interviewees also think that psychological aspect is an important part of their lives but they used a slightly different language to express it. One patient, for example, in response to the research's question about the psychological aspect, said:

To be honest I have no idea about this [psychological aspect]. What I can say is that I am feeling good, I am extremely fortunate with my situation. Yes I know other people that they had cancer and they had all sort of complications and concerns (Patient 2).

Another patient remarked:

My feelings keep changing, good, bad, good, bad but I am feeling good now. It is all stayed stable and now I know what is going on exactly. I am amazed how well I can cope on my own and now I let people in (patient 1).

Patients might use plain language to articulate their psychological status by referring to their concerns, feelings and emotions during the cancer journey. This might be different with the professional language that nurses might use or expect to hear from patients, such as mentally or psychologically feeling well. It is interesting that patients expressed that they are generally happy with their current situation even though their feelings to some extent were variable during the cancer journey.

Spiritual aspects

Nurses expressed the view that spiritual aspects of life are also important contributors

to individuals' QoL. For example, participants stated:

There is also the spiritual part of quality of life and the psychosocial part of the quality of life with a lot of layers. (Nurse 3)

Quality of life gets right down to the basic of comfort, and I guess at the next level we need to be spiritually nourished and respected and valued. All of the aspects that make us and help us to feel there is a sense of quality, have to be met on a day to day basis. (Nurse 7)

The spiritual needs to come into it [quality of life]. So they [patients] will talk a lot about existential stuff, things like death and dying and how patients actually want that to happen, whether they actually believe in certain things that might happen after they die. (Nurse 9)

The bigger picture of spiritual life or meaning does not come into it until I got a relationship with the patient. (Nurse 10)

Similar to psychological aspects, patients did not use an explicit language to talk about their spirituality; however, nurses explained that their patients talked a lot about spirituality. They realised that spirituality is both multifaceted with a lot of layers and pervasive aspects of patients' lives. From participants' statements it appears that spirituality is more about the meaning of life and what will happen after people die. In other words, how they have lived and what will happen for them as a result of this kind of living after they die. Spiritual issues can be defined as a belief system which creates meaning of life and being for a person and can include religious as well as non-religious beliefs (MacDonald 2001, p.383).

Nurse interviewees also mentioned that there are meanings in lives that often people take-for-granted and do not think about them in a great depth:

I think there are many aspects to quality of life in everyday life that you take them for granted. (Nurse 7)

Most of us live our lives without deep consideration of what deeply satisfies us. When people come into crisis in hospital sometimes it is a new consideration for them. (Nurse 10)

It is very interesting to see that participants pointed out taken-for-granted aspects of life. As addressed in some studies (Kwok & Sullivan 2007; Nelson 2007), participants highlighted that taken-for-granted aspects of life are those that people

rarely attend to, like their health, their immediate environment and their social relationships. Often people pay attention to taken-for-granted aspects of their lives when their lives are challenged, for example, being in an emergency situation. The term 'shattered assumptions' has been used in some studies (Berzoff 2006; Davies 1997; Matthews & Marwit 2006) to show how taken-for-granted aspects or accepted meanings of life shatter when people are in unusual and challenging circumstances. Many people, for example, assume that the world is a safe place to live in and they can look after themselves. Many people suppose that they are invulnerable and cancer or other threatening diseases will not happen to them. However, suddenly a problem or trauma emerges such as being mugged in the street or diagnosed with a threatening disease like cancer. This causes individuals to reconsider the taken-for-granted assumptions that hitherto they had and rebuild their value systems.

Environmental aspects

Environmental issues are also an important aspect of QoL. This can be seen, for instance, in the following extracts:

Just being in a different environment, what is usual, when a person comes to the hospital their quality of life is changed and you have to be mindful of that. (Nurse 2)

If you are left in the hospital room all the time for reasons beyond your care that it can have a negative impact on quality of life because it is a little room, regardless if four to five people are in it or not. You lose your privacy. You lose your ability to have your own things. That has to impact on your quality of life. (Nurse 3)

Nurse interviewees valued the relationship between environmental issues and patients' QoL. Here patients did not talk explicitly about environmental issues but nurses became a voice for them. Nurse interviewees stated that when a person lives in an environment that is different from their usual place of residence, for instance, it can influence individuals' QoL. This is particularly important when a person is hospitalised and especially for a long period of time. Putting aside problems related to the disease and the treatment, just being in the hospital might have a negative impact on individuals' QoL. The privacy and usual habits of individuals are hindered in the hospital and health care professionals, particularly nurses, need to be aware of that all the time. For example, leisure opportunities and choices of patients

(MacDonald 2006) and their dietary regimen (McMillan et al. 2005) might be altered when they are in an another environment other than their home place.

Social aspects

Relationship with family and significant others are important contributors to individuals' QoL as it can be identified, for example, in following extracts:

My small research [as a nurse] showed me that dying is a very lonely experience. There are lots of sub-themes that come with that [research]. One is when a person has gone through a long trajectory of illness and the family has already buried him. They've decided that he's outlived his prognosis. Sometimes the family has already separated and, as a health care professional, it is very traumatic. I see them [patients] realising they are a burden on their own remaining family members because they didn't die in the time frame that the doctors had told them. (Nurse 8)

What is happening at the home, have the family dealing with them, are they living on their own, are they not getting out at the proper time when they leave us. (Nurse 4)

If that person has a very supportive family come in [to the hospital] and is with that person and I can see there is an interaction going there then my estimation [of patients' quality of life] goes up slightly. If on the other hand the same person in the same situation has no family, no visitor, I assess his quality of life lower than the person who does. (Nurse 2)

Here again nurse interviewees highlighted that family issues and other interactions are quite important contributors to patients' QoL. People do not live in isolation and interactions with other people, particularly those who love them are a critical issue. Interactions might also indicate how much support patients receive in the hospital or in the home particularly when they are dying. A chronic disease like cancer might bring patients and their family together through a long journey. Patients' families might exhaust their own psychological resources in order to support patients. The patient-family interactions and supports that patients receive from their family will affect how nurses think about patients and their QoL.

Conclusion for the category: QoL aspects

As highlighted by other researchers (Cella et al. 2002; Kaasa & Loge 2003), it was discussed that many aspects contribute to an individual's QoL. These include: physical, psychological; spiritual; environmental; and family aspects and other social interactions. Both patients and nurses stated that physical and psychological aspects
are important parts of individuals' QoL. For other aspects nurses generally became a voice for patients. More importantly, participants stated that the spirituality in itself is multifaceted and there might be aspects in the life that can be taken-for-granted. This can imply that there might be more than above identified aspects that contribute to individuals' QoL although they did not become evident in this research.

These findings also show that nurses generally are aware of importance of a number of aspects in the construction of individual's QoL when they consider patients' QoL. However, this does not mean that they have enough understanding of the current state of these aspects when they consider individual patient participants.

QoL meanings

Participants also explained that QoL is an individual concept with many different meanings. Consider, for instance, the following extracts:

Quality of life would be different for everyone. Every one would determine quality of life differently. (Nurse 1)

It [quality of life] is an individual thing at a particular time. It is probably the thing that you may be thought when you are 20, you may thought some quality of life things were as a most quality factor in your life, but when you are 34, and it is a different situation, you completely reconsider what are you doing, and even when you are 34 and 6 months down the track, it might change again, because something else is influencing it (Nurse 3).

My view could be different. It [quality of life] is personal belief or thought...Everyone's would be different. (Patient 3)

Both patients and nurses expressed their views that the QoL concept has different meanings. At this stage, the researcher wrote a memo about QoL meanings. This can be seen as follows:

I realised that the majority of participants in some parts are talking about the individuality of the QoL concept. It has also been addressed in the literature that QoL is an individual experience. But is there any shared or common definition of this concept amongst participants? It was necessary to ask in later interviews what do they mean by the QoL concept. (23/10/06)

The category 'QoL meanings' was created to explain different meanings participants

allocated to the QoL concept. These QoL meanings are explained in the following sections.

Happiness and enjoyment

It became clearer during the research process that one important meaning of QoL is related to being happy in life:

Quality of life for me is just to be happy and calm like we are here [in the shade in the courtyard in the ward]. When I come here and sit down here is like heaven. I get the same relaxation being here in courtyard as I get from my normal recreation. (Patient 1)

Quality of life for me means how happy some one is to live, how enjoyable I found life. (Nurse 6)

I ask different questions to assess patients' quality of life. Are you happy? Are you happy within yourself? Are you happy to come in to us? Are you happy to go home again? (Nurse 4)

Both patient and nurse interviewees expressed their opinions that the QoL concept means having happiness and enjoyment in life. Activities that create enjoyment might lead to happiness but there might still be other sources for happiness (Joly et al. 2002; Ventegodt, Merrick & Andersen 2003). In the first phase of the research study it was also identified that cancer patients have a better QoL than nurses might think about their QoL (nurses' underestimation of patients' QoL). There were different reasons or sources that make patients happy, for example according to some patients:

Do not really care [about the enjoyment], does not matter, just happy and being relaxed, like we are here, beautiful...I always have a frame on my mind, that is, do not worry, just work, just do it, do not worry about cancer. (Patient 1)

The quality of life of the patient was not what I thought. They seemed quite happy, no issues. Even though they lived alone, on the pension, did not have any friends, be in pain, cannot work, but they did not seem bothered by that. They thought that their lives are meaningful. (Nurse 5)

'Patient 1' has cancer at an advanced level with different metastasis to the lung and kidney. However, he said that he is happy because he is relaxed at this moment (in the shade in the courtyard in the ward when he was smoking) even though it might appear that he does not have any specific enjoyment. It can also imply that sometimes even simple things can make patients feel happy. Another reason for happiness of

this patient seems to be the acceptance of the current situation because the patient, at least to some extent, tried to put aside his worries particularly about cancer and attempted to become relaxed. In the interview with 'Nurse 5', this matter was again emphasised, that cancer patients she experienced were happy although she did not think in this way. More importantly, 'Nurse 5' pointed out another reason for happiness when she expressed that patients still think that they have a meaningful life even though it appears that they do not have positive emotions or specific enjoyments. This implies that for individuals to feel happy it is important to have positive emotions and enjoyment in their lives (e.g. through their relationships with others). In addition, meanings of such emotions and how individuals reflect on them are also critical issues. Altogether, a comparison between patients' and the nurses' perceptions implies that maybe the reason for such perceptual differences is that nurses are not completely aware of different sources that make patients happy.

Satisfaction

Satisfaction in life is another meaning that nurse participants referred to as a QoL concept. The following excerpts, for instance, are from two nurse interviewees:

First of all I think almost in the academic sense what does that mean as a definition and this is all those things which provide satisfaction...an individual feels has deep value and satisfies them. (Nurse 10)

So quality of life to me I guess means ability to fulfil all aspects which is important for me is in term of all aspects of quality of life. (Nurse 7)

In the above extracts, QoL was defined by nurse participants as having satisfaction and fulfilment in aspects that are important for them. Patient participants did not talk that much about the issue of satisfaction. Later, Nurse 10 explains that such a thing that can satisfy the person might have a deep value. Some people might think, for example, that money is something that has deep value and satisfies them whereas many people might look at the issue differently. This is, for instance, what can be seen in the following extract:

> I guess in Western society there is often some doubt that money can buy a lot of things but it cannot buy good health and if you have a diagnosis that has a poor prognostic outcome and you are dying. (Nurse 7)

This nurse participant emphasised that people might doubt that money is what life is

all about and can solve all manner of problems. This situation might be identified by individuals when a crisis happens. When individuals are faced, for example, with a dying person or they experience a threatening disease, they might deeply recognise that money is not as important as they thought. How can a dying person buy a new life? Where is that life-saving drug that advanced cancer patients can buy to get rid of their long months or years of suffering? Who is that doctor who can alleviate their pains?

Having choice

Nurse participants also defined QoL as having choice. Consider, for instance, the following extract:

It [quality of life] would to be to enjoy life, to be able to do the things you choose to do, to have the choice. (Nurse 1)

Having choice is important when the person is having an unpleasant experience:

Quality of life is what gives us a framework for what we are going to do and if it's going to be very unpleasant, we might make another decision, another choice. (Nurse 8)

It is therefore quite important that the health care professionals can explore and support patients' choices:

We had a patient who came in with a Fungating Vulval Carcinoma who had terrible pain, terrible odour from the cancer. Her chief complaint when she came in, was that her cat needed to be relocated and not that her odour needed to be contained or her pain managed but her cat that she adored needed to find a new home because she was going to die. So the first thing we did was to get the social worker in to see her, not the doctor and when you put a structure in place for the patient you are actually supporting their quality of life choices. (Nurse 8)

Sometimes a minor issue becomes a major concern for the patient. Health care professionals need to consider the broad range of patients' choices and try to meet them as much as they can. Keeping pets like cats and dogs, for example, is a relatively common custom in Australian culture. This can become a major concern for some people when they cannot do it. In some other cultures, however, this is not an issue.

Conclusion for the category: QoL meanings

In the previous category (QoL aspects), it was explained that QoL is a broad concept that incorporates a variety of categories. In the first stance it appears that QoL is a complicated concept and is difficult to understand. However, participants in this research study teased this concept out and gave some common definitions to this individual concept that facilitates an understanding of this broad concept. Patient interviewees used a simple language and defined the QoL concept mainly as being happy in life whereas nurse interviewees considered QoL to be more complicated. They allocated a broad range of meanings to it including happiness, enjoyment, satisfaction and having choices.

Participants stated that cancer patients might be quite happy, cheerful and satisfied with their lives even though no specific enjoyment can be identified by caring people. In fact, depending on the situation even little things like being relaxed might make the patient feel satisfied and happy. It appears from participants' statements that cancer patients' acceptance of the real situation, particularly at more advanced levels, facilitates the process of satisfaction and happiness. As well as these meanings, participants expressed that the QoL concept means having choice especially in unpleasant circumstances. Patients' QoL choices need to be supported by health care professionals, particularly nurses. Nurse participants also stated that they need to be mindful that when considering patients' QoL, patients might talk about QoL in a simple language, such as being happy in their lives.

Cues-based QoL assessment

In response to the research questions, it was found that nurses pick up on simple cues during their relationships with patients as a way for QoL assessment. This is what one nurse participant highlighted below:

> I think a lot of time when nurses do assess quality of life they do not do it in a structured official capacity. This is more something that they are picking up during their encounter with patients. They pick up on cues of cancer patients rather than usually doing a formal assessment of the patient's quality of life. (Nurse 9)

In the first phase of the research study it was also identified that nurses differ in their

perceptions about cancer patients' QoL with patients' own perceptions using the WHOQoL-BREF questionnaire in oncology wards. The category 'cues-based QoL assessment' was created to show how participants think about the kind of QoL assessment that nurses actually perform in oncology wards and the reasons and consequences of this kind of assessment.

The category 'cues-based QoL assessment' consists of the following sub-categories: 1) Just simple cues; and 2) Assessment with uncertainty. These sub-categories are discussed in the following sections.

1) Just simple cues

This sub-category was created to demonstrate how nurses actually are doing a cuesbased QoL assessment and how it varies in oncology wards. Nurses expressed that in oncology wards they pick up on cues of cancer patients when they communicate with them:

> So you pick up on cues of people, like I mentioned before, up or down or stressful a pain or whatever but they might tell you, they might actually tell I've had enough, yes, constantly looking at people and try to assess where they are. (Nurse 2)

> You can over the years, pick up on little ways and little things that patients do especially when you look after them for a while and you think, oh, that does not seem quite right. There are some things a little off colour. They are coming and they are quite introverted when they normally would be out spoken, happy to see you full of hope for their treatment, no I feel much better off the last time, and they come suddenly right down at the bottom...just simple little things. (Nurse 4)

So just a clue is the first thing [for quality of life assessment] and then later in the day, I will often ask how things are going. (Nurse 10)

Similarly, patients in response to research questions as how nurses assess your QoL stated that:

[Nurses] generally ask me how are you feeling, generally they are concerned...The only time that I can think of it [a formal quality of life assessment], was not here but my local doctor, whose one of his nurses came to my home and did a survey of my situation. (Patient 2)

There might be some odd questions here and there [that nurses ask for quality of life assessment] but not in a structured way. (Patient 3)

It is clear from both patient and nurse interviewees that QoL assessment in oncology wards is something that each nurse will do individually. One nurse, for example, might only consider pain as an important cue to QoL whereas another nurse may focus on patients' general feeling of happiness. Moreover, such assessments are being conducted informally because nurses do it during their communications with cancer patients and not based on any agreed process or structure.

Even though nurses might assess cancer patients in different ways, they focus mainly on physical issues. Nurse participants stated:

I think when you [nurses] assess the patient you do not broaden or do more global assessment. You only assess what you require to assess. If you ask about nutritional status or if you ask about bowel management or something like that this is the only thing you assess. (Nurse 3)

From the very beginning when you train to become a nurse, it is a physical thing to address all the way through, physical, physical, all the time. We deal with a bit, with psychological. (Nurse 4)

And similarly a patient participant stated that:

They [nurses] use a medical model. The medical model is a medical approach where they see you [patient] as somebody who has had a left lumpectomy and 12 nodes removed. You have a drain in situ, therefore you need these kinds of observations; to watch out for infection you'll need these antibiotics for 3 days and then we'll take the drain out and then we'll send you home. That's medical model. The doctor will come in and check your site, your wound; the drain will be removed. If you don't spike a temperature you'll be sent home. That's a medical model. It's not a social health. (Patient 3)

Both patient and nurse participants stated that a QoL assessment could focus on aspects that nurses think they are required to do, i.e. physical aspects of the job. It is interesting that a more educated patient, Patient 3, introduced the 'medical model' term and believes that physical aspects are often a focus of nurse assessments.

Another important character of a cues-based QoL assessment highlighted by participants is that more experienced nurses often better pick up on QoL cues. Nurses' clinical experience was also a significant finding in the first phase affecting the level of agreement between cancer patients and nurses about cancer patients' QoL, which was further elaborated in the second phase under another category (see nurses' experiences as a facilitator to QoL assessment in page 172).

Summary

One way for QoL assessment in specialist and non-specialist oncology wards is that nurses pick on cues. There are some important characteristics to this assessment. In this assessment nurses informally pick up little things that they individually think are important indicators for patients' QoL. The focus is more on physical issues even though more experienced nurses might assess a broader spectrum of QoL cues such as social and emotional status.

2) Assessment with uncertainty

This sub-category was created to explain the uncertainty that nurses feel about a cues-based QoL assessment, and what could be the possible reasons and consequences of this uncertainty. Participants stated:

You probably do not assess it too much. You probably subconsciously assess quality of life for patients but [you are] not sure if it is quality of life assessment or not. (Nurse 5)

It [quality of life assessment] is like a gut feeling. It is what you get with most nurses. (Nurse 4)

I'm not sure they [nurses] know how to assess and unpack it [quality of life]. What quality of life means and how it is influenced by different things? It is being taught but even the people teaching it don't understand it, therefore, the outcome is less than optimal. (Patient 3)

Here both patient and nurse participants demonstrated they are not certain if nurses are doing a QoL assessment or not. Throughout the interviews nurses frequently expressed that they are not sure if what they have done is a QoL assessment or not.

The reason for this uncertainty might be that health care professionals including nurses often do not have enough information about patients' situations. For instance, participants said:

Whether really as a health professional we have actually any idea about what quality of life means to our patients at all is questionable. (Nurse 9)

You [as a nurse] just make some assumptions which sometimes may not be right...You can never be totally sure that you are going to be spot on and really know how this person is feeling. (Nurse 2)

Here, 'Nurse 9' explicitly questioned interpretations that health care professionals are making about patients' QoL. This implies that she is not sure if she, as a health care professional, has the correct idea of patients' QoL or not. 'Nurse 2' highlighted more specifically that nurses' interpretations of patients' QoL might not be correct.

Another reason for this uncertainty could be that nurses generally do not use the professional language of QoL. This is identified below:

Regardless of thinking what it [quality of life] is, I do it but I do not think about it too much, just as a part of my work. You do not think of it in terms of quality of life. That is a term you don't really use. You tossed it around but it is not something that you focus on. This term is a bit confusing. Nursing staff do not really know what quality of life [concept] means. (Nurse 3)

It [quality of life assessment] is really hard because it is just a sort of do it but I don't really think about it too much as how I assess every one that comes in. I don't know really [if my assessment of patients' quality of life is correct or not]. I guess from the feedback I get from patients. (Nurse 5)

It appears that QoL is not a concept that nurses normally use in their professional conversations. They do not know what QoL assessment means exactly so that they can only evaluate it if they have done a QoL assessment previously.

QoL assessment with uncertainty has problematic consequences for patients. For example, nurses often anticipate more problems for patients:

We probably think their [patients] illness is dominating their life but it may not be. We anticipate problems. We anticipate that the human is full of problems. Therefore we probably underestimate their quality of life. (Nurse 10)

The frame of reference for nurses is to predict more problems for patients. This will

affect nurses' perceptions of cancer patients' QoL. They may not imagine that a patient with cancer still can live happily as other people do. This was also identified in the first phase of the research study in which it was identified that patients' QoL scores were higher than nurses' scores (as patients' proxies) indicating that nurses underestimated cancer patients' QoL. The sub-category 'QoL has different meanings' shows that cancer patients are generally happy with their lives.

Summary

Participants in the second phase stated that a cues-based QoL assessment, as seen in specialist and non-specialist oncology wards, is an experience shrouded in uncertainty. This scenario has reasons and consequences. The possible reasons for this uncertainty would be that health care professionals including nurses often do not have enough information about patients' situations and therefore they are not sure how accurately they understand patients' QoL. Moreover, nurses generally do not use the professional language of the QoL concept and QoL assessment. Therefore, they are not sure what they need to assess as a means for QoL assessment and specific criteria they need to evaluate if their assessment of patients' QoL is correct or not. Outcomes of such an assessment might be only assumptions about patients' QoL, which is more like anticipating problems for patients and underestimating patients' QoL.

Conclusion for the category: Cues-based QoL assessment

Participants emphasised that QoL assessment in specialist and non-specialist oncology wards is conducted informally, and based on nurses' relationships with patients, during which nurses pick up on patients' cues. QoL assessment in oncology wards does not appear to be conducted using any agreed process or structure, but by individual nurses picking up on cues of cancer patients. The main focus of such an assessment is the physical aspect, but the experience seems to play a role in the choice of indicators utilised. Another important feature of a cues-based QoL assessment is that nurses feel uncertain if what they are doing is really a QoL assessment or not. The reason for this uncertainty appears to be that nurses know that they do not have comprehensive knowledge about the patients' situation and must make some assumptions about patients' QoL. Moreover, it is not clear to nurses what the QoL concept means for the patient and what they need to consider in undertaking their QoL assessment. Outcomes of a cues-based QoL assessment may be to underestimate the QoL of those patients coping with cancer.

Purpose-based QoL assessment

There were participants who had experiences specifically with palliative care or had a shared experience in both oncology and palliative care areas. They expressed the view that generally QoL assessment in the oncology wards differs from that in the palliative situation:

I believe that down the track in palliative care the situation changes. From the little bit I've seen and heard of palliative care because I have looked into it... and I actually did have an interview with somebody in the palliative care area and I had to say the interview was totally different, all about quality of life. (Patient 3)

In palliative care quality of life is important because the aim is to have the patient live more comfortably and that frames how we think and how we structure things, whereas in acute care services the aim is to cure even if the patient endures a bit of discomfort. (Nurse 8)

From both patient and nurse interviewees it appears that QoL assessment in the palliative care area differs from that conducted in oncology wards. The researcher wrote the following memo:

QoL assessment in the palliative area seems to be performed differently from that in the oncology wards. In the palliative area the final aim is for patients to live more comfortably whereas in the oncology area the aim of QoL assessment is to respond to patients' expectations for cure and recovery. There might be some overlaps in a way that QoL assessment is conducted between oncology wards and the palliative area. However, it appears that some characteristics of QoL assessment in the palliative is unique and needs to be searched further. (30/02/07)

The category 'purpose-based QoL assessment' was created to reflect on the usefulness of this approach to QoL assessment in the palliative care area.

One important character of QoL assessment in the palliative care area, as explained by nurses, is that it is being conducted using a specific guideline. One nurse, for example, in the palliative care area brought for the interviewer the assessment and the care plan sheet specifically used in the palliative area. She explained how nurses assess patients' QoL in the palliative care area:

We actually write down how they feel, we assess all their problems, discuss the background of their family, how much help they have in the home, that sort of thing and we also ask them to make a comment about their quality of life. They actually scale it from one to ten. We also do background assessment on other areas that might impinge on that. (Nurse 6)

The Likert score gives you a line in the sand. There are a whole lot of things we do. We do a thorough symptom list and find out the impact that symptom has on the patient's quality of life...you've got a fairly comprehensive symptom assessment which addresses a lot of that emotional, spiritual and psychological suffering as well as physical suffering. (Nurse 8)

From the above extracts, it can be seen that in the palliative care area, nurses fill out a QoL tool and a symptom checklist directly with the patients and these are universal across the palliative care area. Moreover, information is gathered from different aspects of patients' lives such as physical, psychological, and spiritual issues. These portray a fuller picture of patients' QoL.

Further to this, in the palliative care area QoL information is gathered from a range of heath care professionals:

A team meeting will put everybody's assessment together. So we'll start putting the jigsaw puzzle together and that the social worker will have elements of this patient's story to contribute and the doctors and the pastoral care person and when you put the picture together, you might say this is not a good existence for this person. It might make you refocus your attention to go back and find out what is it that they actually want? Are we doing the right thing? (Nurse 8)

In the palliative care area there is an opportunity for QoL information to be discussed by the professionals involved in the patients' care. This is very helpful in order to do a more comprehensive analysis and to improve the impact of strategies designed to support patients' QoL. Such discussions between health care professionals might also happen in the specialist and non-specialist oncology wards but it appears that they are not regular occurrences. The following example can elaborate further about how the situation is managed in the palliative area:

When there is a family issue which is beyond our power to manage, we have palliative care social workers who will become involved especially

about things like legal arrangement, emotional support, financial issues, and so on. We have also the pastoral care team who come and support people in a spiritual way. They are actually non-denominational group of people, all the patients and denomination of pastoral and priest can come. So, all these ways will address the quality of life. (Nurse 6)

As well as using a QoL tool and a symptom checklist to assess patients' QoL, nurses also pick up on cues for QoL assessment particularly for patients who cannot communicate, for example those who are dying. Participants stated:

If they [patients] are unconscious then you can assess quality of life by picking up on cues like facial expression and general relaxation. Even the day before they die, a lot of patients are able to make some kinds of facial expression, or just squeezing the hand. (Nurse 6)

[For those patients who cannot communicate] I think nurses assess quality of life in terms of their [patients] comfort. For example, if they [patients] are furrowing their brow or fidgeting or grimacing, there needs to be an intervention put in place. We don't have the verbal cue but we do have the strong assessment skills. (Nurse 8)

Here it can be seen that as well as using a guideline for QoL assessment in the palliative care area, nurses also assess patients' QoL by picking up on cues particularly in relationship to those patients who cannot communicate appropriately.

Conclusion for the category: Purpose-based QoL assessment

Both patient and nurse participants stated generally that QoL assessment in the palliative care area is conducted differently from the oncology wards. Nurse participants explained that in the palliative care area, a guideline exists to assess QoL because the goal of treatment is for patients to live more comfortably and to improve their QoL. In the palliative care area nurses assess patients' QoL across several areas including the physical, the emotional, and the spiritual dimensions. Moreover, information is gathered from different health care professionals and patients' families. In the palliative care area nurses also pick up on cues of patients as one important way for QoL assessment, particularly for those patients who have communication deficits. Thus, a purpose-based QoL assessment is much more holistic than a cues-based QoL assessment because the former assesses QoL using specific guidelines as well as by picking up on QoL cues through interactions and communications with patients. Palliative care nurses are well aware of the process they are undertaking rather than using an informal and intuitive assessment as can be

seen from the data gathered from the nurses working in oncology wards.

Facilitators of QoL assessment

As well as explaining the current state of QoL assessment in oncology wards and the palliative care area, during the course of interviews participants (mainly nurses) suggested a number of issues that can facilitate nurses' understanding of cancer patients' QoL. The category 'facilitators of QoL assessment' was created to incorporate the following sub-categories: 1) Relationship and rapport; 2) Doing a QoL assessment; 3) Nurses' experiences; 4) Nurses' education and training; and 5) Using alternative criteria. These sub-categories are discussed in the following sections.

1) Relationship and rapport

During the process of comparative data analysis it was found that participants address a number of concepts like openness, enthusiasm, rapport and communication to facilitate nurses' understanding of patients' QoL. Participants stated:

There is a certain amount of openness and rapport or interest or enthusiasm that I am not sure how I can describe that I think could be the key factor [for nurses' understanding of patients' quality of life]. (Nurse 6)

The only way is if you build a rapport with a person and you talk to them and you find some insights about what they are telling you, give you a little bit more idea how you see that person's situation. (Nurse 2)

These concepts were then connected under the sub-category 'relationship and rapport'. Establishing a good relationship and rapport can lead to effective communication (Corner 2002). The most important characteristics of this sub-category are considered in the following sections.

<u>Enthusiasm</u>

Enthusiasm directly or indirectly influences their relationships with patients (Sadovich 2005). Participants explained that enthusiasm or interest in health care professionals facilitates the process of establishing a good rapport with patients to assess their QoL. Consider, for example, the following remark:

If you've got a genuine interest in the person you're looking after [that might allow you to have a better understanding of the patient's quality of life]. Willingness to engage and spend time voluntarily...It's the emotional and psychic energy to be involved with another person and possibly with a person who is going through something painful. (Nurse 10)

Interest in health care professionals developing a good relationship with patients is very important because they need to engage with patients who are suffering from pain or other physical and psychological issues. Such experiences are not pleasant, so nurses need to be highly motivated and psychologically strong for such an engagement (Finfgeld-Connett 2008). In the following quote, the nurse described an unpleasant experience:

If you have a young mother with two young children admitted who is going to die with us, that in itself can be quite traumatic. The staff can identify with this. They may have two young children, what if that was me? And that can actually get in the way of a good therapeutic relationship. (Nurse 8)

In these situations, more interested nurses make a therapeutic relationship with patients to help them.

More enthusiastic nurses might search for a common interest in order to communicate with patients:

Just chatting with patients about what you learn, what are you do, are you knitting or sewing or have a dog. Once you have a common interest, you can talk about. (Nurse 4)

It is also possible that a more interested nurse even spends time voluntarily with the patient in and outside of the hospital in order to communicate with patients better:

As most nurses do I have been guilty of after hours, I go around and walk the dogs or clean for them. (Nurse 4)

When they [doctors] told me that I had full bone cancer after all other cancer, there was a student nurse...she is standing next to me, after doctor told me what has gone on, and I grabbed her hand, I was in tears, she is starting on tears, that sort of things you just cannot replace, and she was just student nurse, and she rang later on that night to see how I was. That sort of thing is fantastic, communication. When I go home I miss nurse, miss the care. (Patient 1)

In contrast, less interested nurses might only respond to the surface issues to make their work and duties easier:

Sometimes there are days when I just want to do the work and [I am not interested] to be bothered by the complexities of quality of life...If this person [patient] looks bright and fairly happy, then I think that is OK, that is good. This is a little bonus. This person does not seem to have problems. I won't go looking for trouble here. I'll just accept this as one of the easy patients. (Nurse 10)

More interested nurses try to establish patients' problems and to solve them, whereas less interested ones only try do their job and earn money. This will mitigate against a holistic QoL assessment. One patient alluded to this issue as follows:

> Nurses can be judgmental without fully understanding [patients] because nurses see themselves as the expert. That's very hard for them to put themselves in the patient's shoes. If they try and find out what the patient's view is, the patient will say possibly something quite different from the nurse...Some nurses just need to get a job. They might say: Right! No money therefore they're no hopers...I could be judgmental there too because not all nurses would do that...The nurses are doing an allocation of duties because management are trying to save money. There's a lot of that going on too. (Patient 3)

To facilitate nurses' relationships with patients, nurses need to explore in what areas they would actually like to work:

The beauty of being a nurse is that you can find your special niche for your personality type. The theatre if you're more of a production line type. If you have a great ability and interest to relate to people then you are drawn to places like palliative care, medical units where you have the opportunity to develop relationships with people. It's about your natural sense of which you are. (Nurse 8)

In oncology and palliative care wards, patients are often hospitalised for a longer time and the situation is not critical. Here, nurses need to be more therapeutic in their relationships so that they can relate to patients, understand their situation and improve their level of comfort and QoL. Conversely, in acute areas such as surgical and emergency wards, there is not much time for a detailed relationship with patients but rather nurses need to have an ability to make quick decisions to manage sudden problems.

Openness

Openness is also very important during the relationship and rapport with patients. It is the quality of being able to think, accept or listen to different ideas or people. Participants remarked:

You need to be aware of what information [patients] got and discuss with them openly. (Nurse 4)

That's why I keep coming back to communication through this entire interview. It's the key. Not being restricted but being ready for whatever might come up. (Nurse 8)

I think having an inquisitive mind and seeking new knowledge and being open to new ideas and new thoughts [is a factor to better understand patients' quality of life]. (Nurse 9)

Here participants expressed the idea that nurses need to be prepared to enquire, listen and discuss issues with their patients if they are to have a better picture of patient QoL. An open-minded nurse is ready to hear and/or discuss about everything that might happen during the course of a relationship and communication with patients. Nurses cannot restrict themselves to hear specific things they want to; instead they need to be prepared to listen to a variety of thoughts and opinions. Moreover, nurses need to encourage patients to talk about all issues even if they think these ideas would be unpleasant and/or contrasting with them and their ideas.

Be fit and well

The first phase of the research study explored nurses' QoL, which influences their understanding of patients' QoL. This issue was further elaborated in the second phase. One nurse, for example, said:

You have to make sure you are fit and well and ready to listen. (Nurse 6)

It is important that nurses who are looking after patients are at a high level of wellness otherwise this will affect their ability to communicate with patients:

Things like stress and so on would decrease the amount of listening power they [nurses] have. (Nurse 6)

The happier they [nurses] are, the happier they are with the people. If they are happy with themselves, they more switched on. (Patient 1)

If I [nurse] have poor quality of life, that is going to colour my perception against the people that I am looking after...I suppose you can look at the issue in this way. If I come to work and I have got headache that is impacting on my quality of life. Is not it? on that point of time, and yes I would be more grouchy than usual, I might not feel like opening up to the people, and encouraging me to talk and build up a rapport, therefore I think in that way so every one knows it is a bad day. (Nurse 2)

Both patient and nurse interviewees expressed that nurses who are not happy or with lower levels of QoL might be too engaged with themselves and cannot focus on the patients and listen to them appropriately. Nurses' feelings may colour their perceptions of what other people are feeling.

Patients who took part in this study also expressed their ideas about how happy nurses are when they interact with them in their clinical practice:

Nurses here are fantastic. Very few not good, and they always have a reason, they can say they have stress or they have too many shifts, or they are grumpy, but generally I have happy people around all the time, happy people, I like that...They do not show anxiety and depression. (Patient 1)

It appears that patients are observing nurses generally as happy people. This can confirm results of the first phase indicating that nurses generally rated their QoL as good or very good. It is interesting to see that patients appear to have a reasonable understanding of nurses' QoL.

Consider patients' characteristics

Participants pointed out that some patients' characteristics are very important when nurses want to develop rapport with them. Gender is one such characteristic:

My opinion is that females have been more open about the issue and they also want to know about their condition whereas I sometimes found that males are just, do it I do not want to know about it, but it is not for every one because there are some people that want to know. But I do find in general females are more open in the discussion. (Nurse 5)

Sometimes female patients are more comfortable with female nurses and sometimes male patients with male nurses. (Nurse 2)

Female patients are seen to be more interested about detailed discussions with nurses compared with male patients. Patients also may feel prefer to talk with a nurse from

their same gender. If these issues can be taken into account by nurses, they may better communicate with patients.

Another issue is the age of patients when nurses communicate with patients:

Younger people seem more frightened and older patients may be more accepting. (Nurse 5)

May be a older person who has come to the realisation that accept the process that I have this disease, I am 80, 90, and I am tired, and I am ready to go, such acknowledgement makes it easy compared to a 22 year old who has leukaemia and is dying, that is more challenging, confronting and emotional for nurses compared to a older person who is ready to go to the heaven. (Nurse 7)

Here it can be seen that dealing with older cancer patients might be generally easier compared with younger ones because older people might have accepted and acknowledged their diagnosis. Nurses need to be aware that communication with a younger cancer patient might be more challenging.

Nurses also need to be aware of patients' language and culture:

Patients having English as a first language are more communicating...There is a lady there and she suffers from dementia or something, she is a foreigner and she speaks in a language and she gets angry and she wants to wander around, she is a bloody nuisance and that is hard for the nurses to function properly. (Patient 1)

I suppose it is not only the better language skills to communicate with but you might have more cultural background for example when the person comes from the similar culture you have a few more cues as to what person thinking but when the person comes from different culture or background you can find yourself thinking maybe I should not have said that. You have to be sort of mindful of the culture. (Nurse 2)

Nurses need to be mindful that Australian English is considered difficult to understand for those who have a different first language or are more conversant with British or American accents:

> Specifically, Australians have a lot of funny words. We have an awful lot of funny words for different things, for example, down in the dumps means feeling low, and a lot of people, we have now a Chinese girl, she just looked at me a couple of times. It has a big impact; this is the way that Australian people say things. (Nurse 4)

Accordingly, nurses must be very careful in their communication with second language patients and arrange an interpreter if or when necessary:

Non-English language patients, they are a bit harder, but most people when they are having treatment, they have still the same reactions to the treatment, and you can pick up, and if I cannot understand, I get some one who can, so involving a family member if I can because they can discuss openly. Sometimes they prefer to have an outsider to come in and do some translations and someone who gives them better understanding. (Nurse 4)

Also, according to type of patients' treatment, the communication might differ:

They [patients] react in different ways depend on what treatment they are given. Like chemotherapy they walk on egg shells. (Patient 1)

Chemotherapy is the major type of cancer treatment but has many side effects that can hinder patient-nurse communication, something that nurses need to be mindful of.

Nurses also need to be aware of the stage of cancer if they want to be more in tune in their communication style:

Understanding the cycle of cancer, understanding the type of cancer they have, understanding the impact these cancers have on their quality of life, knowing the full cycle, what really gets to you because you know the cycle of the cancer and you know, all right at the best you have three years, you can be much more in tune or aware of the stages they are at, because you are aware of the stages of the cancer and how it works and its impact on the patients. (Nurse 4)

When communicating with patients, nurses need to consider the demographic and clinical characteristics of patients such as age, gender, language, culture, treatment and stage of cancer. It is interesting to note that a couple of these characteristics were found (in phase one) to be associated with the level of agreement between patients and nurses about cancer patients' QoL using Bivariate analysis (such as Pearson correlation).

<u>Be non-judgmental</u>

When nurses communicate with patients, they need to be aware of differences that

exist between their perceptions about a patient' QoL with those of the patient. Nurses need to approach patients in a non-judgmental way. Participants stated:

I think it is very important that in a non-judgmental way the person ask me, are you happy for someone to assess your quality of life because it has been shown people who are being given some support going through this journey do better, have better outcomes. (Patient 3)

That is part of nursing try not to put your own. Remember that is your patients' view that you are considering and to understand that they do have different views to you...aware of differences between patients and yourself. (Nurse 1)

You might have your own assumptions about what that person may be feeling, just looking at from your own point of view, but at the same time you need to be mindful of trying to see how that person would see things from their perspective. (nurse 2)

Here it can be concluded that in order to better understand patients' QoL, nurses need to put aside their own assumptions and not be judgmental about patients' QoL. Nurses with a non-judgmental approach toward patients are often more supportive in their roles:

I have to be just there and provide supportive capacity to try to listen, at least he found some one to communicate his feeling with in non judgmental way, somebody not to turn around and push his feeling aside, just got to go with it. (Nurse 2)

Quite often more experienced nurses who observed different patients and situations might be less judgmental in their approaches:

I suppose you see enormous range of people of all walks of life, a lot of diversity and it probably teaches you to be very accepting and non judgmental and tolerant and not to be biased but we still measure quality of life to a degree from our own experiences and so going through the process of caring for people. (Nurse 7)

Altogether, for an effective communication and rapport nurses need to approach patients in a non-judgmental way. This is an approach allowing nurses to be more supportive in their roles.

Involve the family and other health care professionals

Participants in this research study also commented that relationship and rapport with

patients need to go beyond the patient and include the family:

If you establish a very close rapport with the person you are caring for and their family in a very accelerated manner and I think that you have an opportunity to be witness to very private and deep and important relationship issues that maybe give you a sense of what is important for that person. So, there are opportunities that they are made available to facilitate, opportunities to explore quality of life. (Nurse 7)

This is a good example of how the family is important and health care professionals need to recognise, through their communication, if patients have a supportive family or not. This is particularly important when patients are in advanced levels of a chronic disease or when they are dying because at these stages patients are more dependent on their families.

Participants in this research study also recognised the family as an important source of information when patients cannot communicate or are in the palliative care area:

I guess a lot of time they are in the palliative situation and are not able to comprehend the questions or if they are they are not able to verbalise their feeling. I guess quality of life can be determined by speaking to their family. (Nurse 1)

We had an interesting situation only a few days ago...We actually spoke to the family about using a drug that we hadn't used before...They said, "We know our mother would want to receive that."...That woman stopped twitching the minute that infusion started. And that family was delighted because she looked a whole lot more comfortable and peaceful. They became the voice for their mother. They kept saying, "I know this is what my mother would want." We'd had these conversations so we do use the family in that way. (Nurse 8)

The patient's family can be used as a good source of QoL information (proxy) particularly when patients cannot communicate well. This also supports findings of the first phase indicating that nurses generally are not good proxies for patients and families are more suitable for such purposes. Nurse 8 explained how the family can be useful to provide QoL information instead of patients.

In order to better understand patients' situation, communication with other health care professionals is also important:

When something comes up that you [nurse] don't have the skills to

manage and that's where having a multi disciplinary team is so vital. You might have to say to your patient, this is an issue that I can't help you with, but I have a colleague who can, a social worker. Would you like me to get them to speak with you? It all adds to a greater understanding and why they behave the way they do, why they react as they do, why they put the coping mechanisms in place that they do. (Nurse 8)

I think discussing the patient in a multidisciplinary group would be useful as well to bring out issues and I think it would be useful overall to bring out issues that need to be discussed about the patient care. (Nurse 9)

The relationship and rapport need to be established not only with patients but also by patients' family and other health care professionals. Developing a relationship with the family is especially very important because they have a supportive role for patients and they can work as a proxy in some situations.

A holistic approach

An effective rapport is also one that attempts to understand all aspects of individuals' QoL such as physical, psychological, and social relationships. This is referred to in some studies as the holistic approach (Bishop & Griffin 2006; Saylor 2004). Participants, however, identified what a holistic approach meant to them:

I think holistic is very over rated and I think it locks people in. We all strive for the ideal, that Utopian way of life. If we say that it's holistic care, therefore we are doing the right thing. Unless you get in and really understand your patient on a deep level of communication, holistic care is just a concept. (Nurse 8)

A holistic approach in fact is a relationship and rapport with the patient that is deep enough to understand all aspects that may impinge on patients' QoL:

Make sure that you care for all parts of patients...Make sure you listen to what is happening, you look after the spiritual, emotional needs as well...I think it needs more intensive evaluation than evaluating physical capacity. (Nurse 6)

Unless you've interviewed them [patients] at length and done a thorough psycho-social assessment, I don't think any nurse would advocate for them. (Nurse 8)

So often it is physical care and emotional support that it is the first response of what the patient needs. The bigger picture of spiritual life or meaning does not come into it until I got a relationship with the patient. (Nurse 10)

In a comprehensive approach all aspects of patients' needs to be taken into consideration. Although the physical issues are an important aspect of individuals' QoL, the psychological and spiritual aspects of QoL cannot be overlooked.

In a comprehensive approach, personal issues such as sexual aspects need to be taken into consideration:

It wouldn't be common that patients talk about those personal aspects. But yes, it could be beneficial. Even just expression of these issues with somebody else is useful... The only ways that those discussions come about with a patient are when a relationship has been well established involving trust. (Nurse 10)

We do not really do any sexual assessment, urology patients, we do not tend to. Even though they have a need for sexual assessment, and it is an area that nursing staff find particularly difficult to deal with because they do not know how to talk about it with patients. But because we are working in haematology/oncology there is a need to address that because the patients are under chemotherapy. (Nurse 3)

Without considering all aspects of QoL including private ones such as sex life or financial issues, it is very difficult to achieve a holistic and comprehensive picture of individuals' QoL. These areas are difficult for nurses to deal with unless a close relationship and rapport is established between patients and nurses.

Summary

The issue of relationship and rapport is important in facilitating QoL assessment. As pointed out earlier, one of the main characteristics of a cues-based QoL assessment used in both oncology wards and the palliative care area is that it is very dependent on relationships. There are very important figures in such a relationship. Participants stated that nurses needed to be genuinely interested in developing an open relationship with patients in a non-judgmental way. This rapport needs to go beyond the patient and include other health care professionals as well as the patient's family. This is particularly important when patients are at advanced levels of the disease or when they are dying. Finally, these relationships need to go beyond the physical issues and take into consideration other aspects such as sexual and financial issues.

2) Doing a QoL assessment

In response to some probing questions, participants pointed out ways that a QoL assessment can be conducted. The sub-category 'doing a QoL assessment' was one in which participants (mainly nurses) commented about using QoL tools and interviews.

<u>QoL</u> tools or guidelines

In the first phase of the research study, a QoL tool was used to assess patients' QoL. The researcher wrote a memo indicating that the use of QoL tools appears to be a challenge for nurses and needed to be investigated further in the second phase. Some nurses in response to the researcher's questions about using QoL tools in the clinical area, explained their ideas:

We need to have a tool. We need to have a guideline. (Nurse 3)

However, the interest for using tools or guidelines varied across experienced and junior nurses:

We do not use them [quality of life tools] in the clinics routinely...It [tool] just might provide some reminders for more experienced nurses that already are becoming more professional and aware of what kind of questions to discuss but there is not a thing wrong with a reminder and for the more junior nurse it could provide some guidance as the type of things that might be useful to ask the patient or useful to consider when you are looking after a cancer patient. (Nurse 9)

I think with younger nurses, they like to have a guideline to work with. (Nurse 4)

The above nurse interviewees had been in the profession for more than ten years. They generally recommended using QoL tools for more junior or novice nurses.

Participants also highlighted their ideas about questions that needed to be included in a QoL tool or guideline and how such questions need to be asked and interpreted:

I think it is hard to get agreement on what right questions to ask...Obviously screening type tools would not incorporate everything you want to ask; it would be good to have some good trigger questions there that would highlight areas that might be problematic and then you can look at them deeply if you needed to; I think once those are sorted out, there have to be some things that ask patients about how important that actual item was to them, and how much it actually impacted on their functioning, so not just about something that existed but whether it actually impacted on them or not and whether there are some things that they want to deal with or not. I think the other thing is interpreting the results; who actually is going to interpret the results and if the person who is interpreting is in the position that they could do something about those results or working on strategies how we can then deal with the results. (Nurse 9)

In the above extract, it has been identified that the first issue is to formulate the kinds of questions to ask in a QoL tool or guideline. Moreover, the importance of each question for patients also needs to be questioned and the interpretations should be conducted in a way that is useful.

These tools or guidelines needs to have several characteristics as suggested by participants:

With a lot of older people, there is no point using the fancy statements and to use a tool as a guideline because those people do not really understand what is that. (Nurse 4)

I think if we are looking to use these quality of life tools for patients on a routine basis this has to be something that is easy for patients to complete but it must also be relevant to the clinic needs at the time as well. (Nurse 9)

Questions in a QoL tool need to be easy for patients to understand and particularly for elderly people. These tools need to be comprehensive enough to reflect on some important aspects of patients but in the meantime they need to be brief. Lengthy guidelines or tools might be very difficult for patients to complete:

Our patients get incredibly tired with a lengthy document or a couple of tools. They might say: My quality of life would be greatly improved if you just walked away and stopped asking me questions. (Nurse 8)

The frequency of QoL assessment using QoL tools is also an important matter:

I think doing a once-off quality of life assessment is useless and it would be better to do a longitudinal tool so that we can see changes to quality of life over the time. (Nurse 9)

The above participants remarked that the QoL data of patients need to be assessed longitudinally, and measured repeatedly during the admission, if nurses want to better understand patients' QoL.

A longitudinal QoL assessment might be particularly very useful in obtaining some ideas when patients cannot communicate:

When they're unconscious, it's too late and you have fighting relatives over the will. It needs to happen much earlier and even maybe in the general practice area with a GP nurse so that a quality of life tool has been completed from a very early stage of life and then is developed. I think we leave things too late. (Patient 3)

Thus, frequent QoL assessment may provide some background information about patients' QoL. This can provide health care professionals with a comparative basis to understand patients' QoL better and if patients' QoL has changed recently, how this has occurred and what interventions or care strategies may have influenced this change. This is highly important when patients are in situations in which they cannot communicate and report on their QoL.

Interviewing patients

Even though a few participants explained that QoL tools might be useful in the clinical area, participants also commented about interviewing with patients as an important way for QoL assessment:

It would be hard to quantify quality of life without talking to the patients themselves. (Nurse 8)

From such extracts it can be seen that an interview is an essential way for QoL assessment.

Even though QoL interviews can be used formally by allocating a specific time to take a seat and talk with people, it is very common for nurses to use these questions informally when they are doing their duties:

I think you can do those assessments in more informal settings, when you're washing a patient you can be talking to them about things that matter to them, not keeping the conversation neutral or selective, like, "It's a lovely day outside," but, "How did you sleep last night? Do you have anything on your mind?" Give them the opportunity to offload. (Nurse 8)

Because nurses are doing this kind of assessment during their relationship and work

with patients, this implies that this kind of assessment is very dependent on establishing a good rapport with patients (see the category 'relationship and rapport').

Moreover, more experienced nurses are more interested in using interviews for a QoL assessment:

I am not definitely a specific tool person. I can probably have a look at some and utilise some of those questions. (Nurse 4)

The above excerpt is from an interview with a nurse with around 30 years' clinical experience. It appears that she wants to use some of the questions in QoL tools but in an informal way. This raises an interesting problem in introducing tools to the care setting because some less experienced practitioners may rely on the tool as it stands while more experienced practitioners may have a more sophisticated understanding and seek the required information in other ways.

Summary

Participants who took part in this research study noted that QoL assessments need to be conducted in the clinical area using QoL tools or interviews. Both novice and expert nurses generally prefer to use interviews. However, novice nurses think that a formal QoL assessment using a complete QoL tool might also help them to understand patients' QoL. Expert nurses generally prefer to use some QoL tool items to initiate an informal conversation with patients when they are doing their work. Participants stated that QoL tools for the clinical area need to be brief and include items that are easily understood by patients.

3) Nurses' experiences

In response to questions put to participants about what nurses' characteristics might affect nurses' understanding of patients' QoL, both patient and nurse participants talked about a number of nurses' characteristics. Nurses' experiences were one of those variables that facilitated their understanding of cancer patients' QoL. This can comprise life and clinical experience:

Life experience can be a great deal when we deal with ill people, so

probably with our patient population, older nurses have better insights into what is going on in patients' life and from more personal point of view because a lot of them are dealing with elderly parents. (Nurse 2)

More specifically, however, it could be clinical experience:

I think the more patients you see and the more patients that you interact with, the better understanding you have of general factors that might affect quality of life, but also you get a better understanding of how different quality of life is for each patient because you have more experience with more patients to notice the differences. (Nurse 9)

I think years of service, years of work, how many years they have been here... I think the more broad spectrum of life as a nurse, the better help they are to the patients...I think if anything nurses should get experience in more than nursing. (Patient 1)

Several reasons were suggested why nurses with more clinical experience have a better understanding of patients' QoL:

They would have done some further education and in- depth reading and writing. They would have done case studies on their clients therefore they would have a better understanding of how quality of life might impact on client outcomes, absolutely. (Patient 3)

As you become more experienced, your assessment skills will improve and you learn to read faces better and learn to assess all the problems plus listen better and so on. (Nurse 6)

You need to have high degrees of clinical experience and proficiency to have a high degree of comfort in the workplace. Then you can have a better understanding of the person's experience. If something goes wrong in the middle of chemo-therapy, you will have the clinical know how to troubleshoot. If you're a junior nurse and something goes wrong your heart stops. What have I done? Your stress comes up and you're not actually able to get to the patient's zone because you're too busy being in your own. (Nurse 8)

I still think that probably someone with more experience may pick up on more things but it is just my personal opinion. (Nurse 5)

Patient 3 expressed her view that nurses with more clinical experience have a better understanding of patients' QoL because they have superior education and training. Nurse 6 stated that more experienced nurses have better assessment skills. As pointed out by Nurse 8, nurses with more clinical experience are more comfortable doing their work and can better understand patients' QoL. Finally, clinical experience can help nurses to better pick up on cues of patients. This is quite important because

picking up on cues is a common technique for QoL assessment particularly in oncology wards. This was addressed previously under the category "Cues-based QoL assessment".

Clinical experience might be one reason why nurses in outpatient chemotherapy units were found in the first phase to have a better understanding of patients' QoL:

There is another factor that might explain the differences between inpatient and outpatient units, especially in chemotherapy day unit nurses have got usually a reasonable amount of experience behind them, they are working in a specialised unit and have special training and I think that nurses in a day unit would have more clinical oncology experience compared with inpatient nurses. (Nurse 9)

The above statement elaborates on one significant result of the first phase, about why nurses in outpatients' chemotherapy units have a better understanding of patients' QoL compared with nurses working in inpatients' oncology wards. It may be contended that nurses in outpatient chemotherapy units might have a better clinical experience compared with inpatient nurses. One reason may be that managers sometimes allocate more senior nurses to outpatient wards so that they can better manage sudden problems occurring in this area. Other reasons why nurses in outpatient chemotherapy units have a better understanding of patients' QoL can be seen in pages 178 and 180.

Summary

Participants stated that nurses' experience with cancer patients can work to facilitate better understanding of patients' QoL. They highlighted that these nurses have developed better communication and assessment skills and can better notice the differences that might exist between patients. They are more professional and can pick up on cues better. Participants also explained that nurses in outpatient chemotherapy units have had more clinical experience. This can explain the significant findings of the first phase concerning why nurses in outpatient chemotherapy units have a better understanding of patients' QoL compared with inpatient nurses.

4) Nurses' education and training

Participants also highlighted that nurses do need education and training in order to better understand cancer patients' QoL:

We need education on quality of life. It is probably something to work to probably nursing training when they are training, that it is an aspect that they need to start looking at. (Nurse 3)

We all need reminders of all kinds of care, physical care, emotional care...I think it is a very relevant. (Nurse 6)

I am sure. I think continuing education and training in any way can be beneficial to the patients. (Nurse 1)

Here it can be concluded that nurses need to be educated about the QoL concept.

This education needs to teach nurses how to assess patients' QoL:

I am not sure about the formal training but may be a good idea to tune to assess quality of life... Yes I think could be a good idea but I do not know how we can go about it, if we can run a course. (Nurse 5)

Thus, QoL assessment is another focus of continuing education. Participants also indicated that such education needs to include other health care professionals as well as nurses:

There needs to be some reorientation, education of the surgeons, the oncologists and the radiologists. It's not just the nurses and they are often being directed by these medical people. (Patient 3)

Summary

Education and training are useful strategies to improve nurses' understanding of patients' QoL. The most important areas for education are clarification about the QoL concept and how to conduct QoL assessment and analyse the findings. Even though such education and training focuses on nurses, it also needs to include other health care professionals.

5) Using alternative criteria

Participants also suggested using alternative criteria when it is appropriate for QoL assessment:

When I was at the Psycho Conference in Venice in October last year they decided to replace the term with Patient Reported Outcomes which means they are no longer meaning quality of life but Patient Reported Outcome because that is more tangible. They can tell you if they have a benefit they can put into their management plan if they can get up and walk to the toilet this morning whereas yesterday they couldn't or they couldn't walk more than three feet. But for them that's an achievement, a realistic goal for them and they can say, "I've had some improvement today," and that, in itself, informs us of their own assessment of their quality of life. (Nurse 8)

Obviously there are other things that can be measured, not just something that a patient has to say, so it could be looking at things like independence in activities of daily living because quite often it can indicate the overall person's health status, ..., some other things that we could be thinking as a part of life, face-grimacing in pain, bringing knees on to the chest because it can bring relief from the abdominal pain, and I think it could be fair to assume that that pain would decrease quality of life". (Nurse 9)

Nurse 8 stated the Patient Reported Outcomes, implying that she may be encouraged to use this criterion to measure QoL. Nurse 9 clearly suggested measuring other criteria like pain or the activity of daily living for QoL assessment when it is appropriate.

Summary

Participants commented on some other measures that can be used to obtain some ideas about patients' QoL particularly for non-communicating patients. This consisted of the Patient Reported Outcomes, pain assessment, and the activity of daily living.

Conclusion for the category: Facilitators of QoL assessment

Participants in this study commented about some facilitators that might be useful to improve nurses' understanding of cancer patients' QoL. Firstly, relationship and rapport with patients is highly important. This relationship and rapport starts with a real enthusiasm and proceeds in an open and non-judgmental way in which patients' characteristics are considered. Moreover, this relationship is comprehensive and focuses on all aspects. This communication goes beyond the patients and includes family and other health care professionals. Nurses need to actually assess patients' QoL using the relevant QoL tools or interviews. Educating nurses and other health

care professionals can facilitate the process of assessing patients' QoL. In some circumstances other measures may help nurses to understand patients QoL better such as measuring the pain or Patient Reported Outcomes.

Barriers to QoL assessment

As well as facilitators of QoL assessment, participants also expressed their opinions about those issues that might work as barriers to QoL assessment. The category 'barriers to QoL assessment' consists of the following sub-categories: 1) Focus on tasks; 2) Time limitation; and 3) Fragmentation.

1) Focus on tasks

One underlying barrier influencing nurses' assessment of patients' QoL and particularly in oncology wards is that nurses focus on other tasks:

Quality of life is ultimate for everybody. I think it is very important. If we do not do it it's because we concentrate on what we have to do. (Nurse 3)

I think the nurses were very task oriented. CNC going up the hierarchy would sometimes get more involved in quality of life issues but not in a great depth. (Patient 3)

There's a whole culture in nursing that still hasn't been addressed fully and it hangs over from the old days when everything was regimented and people had time frames in which to do things. So, often it's my needs as a nurse overrides your needs as a patient, so that I look good to my colleagues and the patient is thinking "I might be clean but I'm miserable". (Nurse 8)

Although nurses are too task-oriented the situation is different in the palliative area:

We [in the palliative area] have an orientation manual that says the patient is not to be woken for breakfast and mealtimes but things are done in the patient's time frame so that is quite different from the philosophy of an acute care ward. My nurses are able to say at the end of the day, 'I haven't done this patient's wash because they didn't want it. I've done everything else, teeth, incontinence pad etc.' and that is the level of patient care we strive for and I think, on the whole, we're pretty successful at it. (Nurse 8)

In the palliative setting, nurses' priorities should be on patients' needs and not just

completing their duties. Nurses are able to say that we did not do this task because the patient did not want it.

Summary

Here it can be seen that nurses have different tasks and they focus more on doing them rather than assessing patients' QoL thoroughly. Maybe the CNC has more time to explore QoL issues further. However, in the palliative area the situation is different where nurses are less task-oriented and priorities are devoted to patients' needs.

2) Time limitation

During the interviews many participants emphasised that nurses do not have enough time to thoroughly assess patients' QoL. This is evident in the following extracts:

Never, never enough time to build as good a rapport as we would like...we have time limited, we have very limited time to sit and talk with patients. (Nurse 2)

They are very busy people, this morning is very quiet but sometime they are extremely busy. They do not have time to sit or stand here and ask me how are you keeping...I think they do extremely good job under these circumstances. (Patient 2)

Nurse 2 explicitly expressed how time limitations influenced nurses' communication in the oncology wards. This matter can also be seen during the interview with Patient 2 in which he expressed the view that nurses are generally very busy.

The degree of time limitation varies across inpatient oncology wards and outpatient chemotherapy units:

In outpatients because you focused on that patient for that period of time, that is why you give them your all, because you are allocated for that patient that you look after them, this is what is happening whereas in inpatients you have 5-6 patients, to try to do everything done for them, there is not enough time. (Nurse 4)

In the above comment, Nurse 4 said that nurses in outpatient chemotherapy units have more time to spend with one patient compared with inpatient nurses. This can elaborate further on the significant results of the first phase as to why outpatient chemotherapy nurses have a better understanding of patients' QoL compared with those nurses working in inpatient wards. Other reasons why nurses in outpatients' chemotherapy units have a better understanding of patients' QoL can be seen on pages 174 and 180.

Even though time constraints are a problem, how nurses use their allocated time needs to be considered. This can be seen, for example, in the following extract from a nurse's comment in the palliative care area:

As a manager, yes, I have enough time the way the staff is allocated. In an ideal world, no. The psychosocial assessment is very important and it takes time...The danger we get into health care with and bureaucracy is there is a group who, even if they had extra time, would not spend it with a patient but at the desk. By spending more time with patients they can only benefit. (Nurse 8)

From such comments it can be stated that health care professionals may be using their allocated time ineffectively. One reason for this might be the bureaucracy and demand to follow procedures in the health care system.

Summary

Time limitation is another important barrier that influences nurses' assessment of patients' QoL. However, in outpatient chemotherapy units the situation is rather different because nurses there may spend more time with patients. Moreover, how nurses use their hours more effectively is also a matter that needs to be taken into consideration.

3) Fragmentation

Fragmentation or discontinuity works as a barrier against health care professionals wanting to assess patients' QoL:

The way that our health system is at this moment...inpatient care, hospice care, community care. They are fragmented. There is not a lot of crossing of the boundaries, and even say for palliative care, they used to have palliative care teams and they would cross these boundaries, they look at the patient in the community, when the patient comes to the hospital the same nurse will come and see them in the hospital, but it does not happen any more; either there is a community care nurse and hospital nurses. I think we can get a better idea about patients' quality of

life if there is some crossing of boundaries but it has to made in the professional capacity. (Nurse 9)

It [the health system] was very fragmented. I think care should be seamless. Information should be passed on. This is about seamless care, integrated care but I went from the surgeon, another hospital where I saw the oncologist and then the radiologist. It felt like they didn't talk to each other...The outcome is disenfranchisement. You feel alone and I know that my family were increasingly worried about me but because there was no one really to contact I was out on a limb. I wasn't supported. (Patient 3)

Here participants expressed that there is a fragmentation in the whole health care system. This mainly refers to the number of disciplines, each of which has its own boundary and ways of doing things. This might prevent health care professionals from providing a seamless or integrated care system.

This fragmentation can also be seen more specifically in the nursing system:

I was in a breast cancer ward. The assessment they did on clients when they came in was...very brief, basically medical, a little bit of quality of life stuff, then it was seeing the social worker, the physiotherapist or the psychologist's job to deal with inherent issues that may come up during the treatment. At discharge there was a discharge plan but it was again handed over to a discharge planning nurse. (Patient 3)

The patient cited above expressed her idea about fragmentation by pointing out that her assessment was conducted by different people. It implies that she expected her assessment by nurses to be better integrated rather than being shared around by many health care professionals. The fragmentation also varies across inpatient and outpatient wards:

[In outpatient chemotherapy units] you have more time to spend with them[patients] and you see them in a more regular basis, whereas in the wards you might see them once and then you move around different patients, even if they are for a month you can only look after them once. It is very small contact, continuity of care definitely. (Nurse 5)

In inpatients wards the care received might be more fragmented compared with outpatient chemotherapy units. This again elaborates the significant result of the first phase regarding why nurses in outpatient chemotherapy units may have a better understanding of patients' QoL compared with inpatient nurses. Other reasons for such a difference can be seen in pages 174 and 178.
Summary

Fragmentation in the health care system might prevent patients from receiving a seamless or an integrated care, such as in the nursing system. Discontinuity of care might be seen more in inpatient wards compared with outpatient chemotherapy units.

Conclusion for the category: Barriers to QoL assessment

As well as facilitators to QoL assessment, there are some issues that make QoL assessment difficult to impossible. Nurses have different tasks and may not have enough time to thoroughly assess patients' QoL. Fragmentation exists in the health care system generally and in the nursing system in particular, and these can work against nurses who want to assess their patients' QoL more thoroughly.

Conclusion

Interpretive aspects of this research study found six major categories that can better portray why cancer patients and nurses differ in their perceptions about QoL. These categories consist of: QoL aspects; QoL meanings; Cues-based QoL assessment; Purpose-based QoL assessment; Facilitators of QoL assessment, and Barriers to QoL assessment.

Each of these categories consists of sub-categories. Some examples from the data have been added to help readers to understand the sub-categories and their characteristics. This might also shed light for readers to understand how the conceptualisations have been made. These categories are briefly reviewed below.

QoL was found in this research study to be connected to a whole range of life aspects such as physical, psychological, spiritual, social interactions, and environmental factors. Participants interpreted this concept in their own way to broadly include happiness, enjoyment, satisfaction, and having choice in life. In order to understand the QoL experiences of cancer patients in oncology wards, nurses currently are doing a cues-based QoL assessment in which they pick up on cues that they individually think are salient to patients' QoL. These assessments are mainly focused on the physical aspects and more experienced nurses can assess these cues better. This kind of assessment is tainted by uncertainty and it cannot provide nurses with a fully holistic picture of the patient's condition.

However, the situation to some extent varies in the palliative area in which nurses are doing a purpose-based QoL assessment in which nurses use QoL tools and guidelines as well as picking up on cues. In outpatient chemotherapy units nurses also do a cuesbased QoL assessment but this will be done in a broader way compared with the inpatient oncology wards. QoL assessment is also hindered by nurses' focus on tasks, time constraints, and fragmentation in the health care system generally and in the nursing system in particular.

Participants also suggested some points that will be useful for nurses to develop their understanding of patients' QoL. Relationship and rapport are very important variables. Such a rapport starts with a real interest and enthusiasm and proceeds in an open and non-judgmental way. Moreover, this relationship needs to take into consideration patients' characteristics and be conducted in a comprehensive way and include patients' families and other health care professionals. Formal QoL assessment by novice nurses using tools or guidelines is recommended. In the meantime, interviewing patients in a less formal way can also be used with a focus on all aspects when nurses communicate with patients. Nurses might use some other measures such as pain measurement alone or a complementary approach involving other measures to obtain a better idea of an individual patient's QoL. Training nurses and other health care professionals can facilitate nurses' understanding of patients' QoL.

In the next chapter, which discusses the findings, an attempt is made to connect all the relevant data arising from the two phases of this study and answer the research questions.

CHAPTER EIGHT: DISCUSSION OF THE RESULTS

Introduction

In Chapter Five, results of the first phase were presented. The outcomes of the first phase were empirical in which patients' and nurses' perceptions about cancer patients' QoL were compared with each other using different statistical tests. In Chapter Seven, outcomes of the second phase were explored using an interpretive approach. Outcomes of this phase were presented in six major categories and supported by using quotes from the actual data.

In this chapter information arising from the two research phases will be compared and discussed in order to answer the research questions with regard to nurses' perceptions of cancer patients' QoL. However, the discussion of the results needs to be seen in the framework of the limitations of the research study which are provided at the end of the chapter.

Presenting the results of a mixed research study is challenging (Cox 2003). One way that was chosen here is to present the basic results of each phase of the study in a separate chapter and then combine the overall discussion into a singular chapter organised under the headings of the research questions (Thomas 2000). The use of one discussion chapter will help the reader to identify the answers from the results when both quantitative and qualitative results are examined together. Each question is introduced and discussions appear in the subsections regarding 'key findings' by referring to the research phase outcomes and by drawing support from the relevant literature.

'The level of agreement between patients and nurses about cancer patients' QoL' and 'nurses' understanding of cancer patients' QoL' are two terms which are used to present empirical and interpretive findings with regard to similarities and differences of QoL perception between cancer patients and nurses in the following sections.

Question One: What differences and/or similarities are there between cancer patients' and nurses' perceptions of cancer patients' QoL?

This question is answered by considering findings of the first phase. However, empirical findings of the first phase are also supported by categories found in the second phase. Three trends were found as a result of this inquiry: (a) at the individual level (comparing each patient-nurse score) the level of agreement ranges from poor in the social relationship domains up to moderate in the physical domain; (b) at the group level (comparing nurses' scores as a group with patients' scores as a group) nurses' mean domain scores are similar with those of patients in physical and psychological domains; and (c) at the group level (comparing nurses' scores as a group) patients' scores as a group with patients' scores as a group with patients' scores as a group with patients' scores as a group patients' scores as a group with patients' scores as a group patients' scores were significantly higher than those of nurses in the social relationship and environmental domains. These findings will be considered in turn in following sections.

(a) At the individual level, the level of agreement ranges from poor in the social relationship domain up to moderate in the physical domain

Despite a variety of statistical tests that can assess the level of agreement at the individual level (Altman & Bland 1983; Lobchuk & Degner 2002; Sneeuw et al. 1998), there is little consensus about what statistical methods are most suitable to analyse rater agreement (Uebersax 2003). In these situations using more than two statistical tests is usually recommended for research studies (Lampic & Sjoden 2000). Therefore, Pearson correlation (r) and Intraclass Correlation Coefficient correlation (ICC), the proportion of exact agreement, and the Bland-Altman test were used to assess the level of agreement between patients and their nurses.

Results of r and ICC identified moderate, poor, poor, and fair agreement between patients and their nurses in the physical, psychological, social relationship, and environmental QoL domains, respectively. In other words, the results showed no substantial agreement between each patient and nurse ratings about a cancer patient's QoL. These findings are similar to general trends found in many other research studies where the level of agreement between patients and proxies (nurses or others) were assessed using QoL tools other than the WHOQoL questionnaires (Larsson, von Essen & Sjoden 1998; Molzahn, Northcott & Dossetor 1997; Wilson et al. 2000). The outcomes are also very similar to two other studies (Becchi et al. 2004; Herrman, Hawthorne & Thomas 2002) where WHOQoL questionnaires (WHOQoL-BREF or WHOQoL-100) were used but with non-cancer patient populations.

Herrman, Hawthorne and Thomas (2002) in their Australian study compared psychosis patients and their case managers as patients' proxies using a set of questionnaires including the WHOQoL-BREF questionnaire. Results identified that case managers' and patients' scores correlated moderately; correlations ranged from 0.31 in the social domain up to 0.47 in the physical domain. They concluded that a significant difference exists between patients and proxies. These outcomes are generally supported by another study conducted by Becchi et al. (2004) in which QoL of patients with schizophrenia was compared by proxies using the WHOQoL-100 questionnaire. Of the proxies, 52.7% were relatives whereas 47.3% were non-relatives (e.g. friends, social workers, and nurses). The outcomes of ICC ranged between 0.26 for the psychological area to 0.42 for the physical area, indicating a poor agreement between patients and proxies.

A comparison between the results of the above studies and the research project indicates that nurses had a better understanding of cancer patients' QoL in the physical domain. This was greater than agreement in the psychological, social relationship, and environmental domains. In order to better understand why nurses' perceptions about cancer patients' QoL vary across the QoL domains, a discussion of each domain based on QoL items that constitute each domain will follow in the next sections.

Physical domain

Items in the physical QoL domain consist of: pain; medical treatment; energy for everyday life; ability to get around; sleep; capacity for work; and daily living activities. Compared with other QoL items, these circumstances generally appeared to be more concrete, objective and related directly to the patients' clinical care or recovery program. Therefore, nurses can better understand what is going on in relation to these situations when they encounter patients in the clinical area.

'Energy for everyday life' (ICC=0.42%), for example, is one of those items that nurses had a better understanding of compared with other QoL items. Having energy for conducting daily living activities is generally an objective situation. Patients' having energy to conduct daily living activities can be assessed by nurses when they are working with patients, for example, giving their medication or serving their food. Such trends can generally be applied to other physical QoL items mentioned above.

Psychological domain

The psychological domain consists of items including enjoyment in life, having a meaningful life, ability to concentrate, accepting bodily appearance, satisfaction with one's self, and having negative feelings. The common thread amongst these items is their subjectivity. Without a deep level of communication, this is not an easy task for nurses, for example, to understand if life is really meaningful for patients (ICC=0.05) or if they have negative feelings (ICC=0.05).

Social relationship domain

Social relationship domain encompasses three items: having a sex life, having friends, and having personal relationships. The general trend dominating these items is their private nature. It is not surprising that nurses had a very low or poor understanding of how patients have received support from their friends (ICC=0.05). Nurses may not know how to ask patients about their sex life and unless a major issue exists in this area, it is not acceptable for nurses to query patients about it.

In the empirical phase of the research study, social relationship items also had the most missing data of all 26 items. This was the case not only for patients but particularly for nurses. This again suggests difficulty for nurses to estimate the patients' actual situation in these private aspects. As the social relationship domain heading suggests, these activities mainly refer to patients' situations when they are interacting with society. Without a good relationship and a thorough baseline assessment, it is less likely that nurses accurately understand patients' social situations.

Environmental domain

The environmental domain includes being safe in one's daily life; having a healthy

physical environment; having money to meet needs; information at hand; having the opportunity for leisure activities; one's own living place; having access to health services; and transport. These items also tend to be relatively subjective and private but less so than psychological and social relationship items. This is why empirical evidence generally rated nurses' understanding of the environmental QoL items better than psychological and social relationship items. For example, safety in daily living activities (ICC=0.14) and a healthy physical environment (ICC=0.19), at least partly, can be addressed by nurses. Nevertheless, there are still items like patients' conditions of living (ICC=0.05) that are very problematic for nurses to understand.

Altogether, empirical findings emphasised that nurses generally have a better understanding of the physical QoL items compared with other aspects. This was further supported by findings of the interpretive phase which found that nurses in oncology wards focus more on physical aspects. Consider the following extract obtained from a nurse interviewee on an oncology ward:

From the very beginning when you train to become a nurse, it is a physical thing to address all the way through, physical, physical, all the time. We deal a bit, with psychological. (Nurse 4)

The next measurement of the level of agreement between patients and nurses at the individual level, evaluated the proportion of the exact agreement between ratings of two parties. This test identified that in 35.5% of cases a similar response category had been chosen by both patients and nurses in the questionnaire. It means that, for example, in answering item one in the questionnaire, 35.5% of both patients and nurses chose the category 'satisfied' or 'very satisfied' for that item. This was not considered to represent a substantial agreement because it was below the acceptable level (60%) which was set for this study. Moreover, the decision was also made on the basis of other complementary tests conducted (r, ICC, and Bland-Altman test) and these confirmed each other. Other researchers may have their own interpretations of the results of the proportion of the exact agreement.

In a research study conducted by Sneeuw et al. (1999), the QoL of cancer patients was assessed and compared to the perspectives of significant others, physicians, and nurses using a QoL tool. Part of the outcomes identified that 41% of all comparisons

were in exact agreement. While the level of agreement was far from 60%, the authors concluded that "judgments made by significant others and professional caregivers about general aspects of cancer patients' QoL are reasonably accurate" (Sneeuw et al. p.93). The authors explained that these interpretations were based on calculations of the proportion of approximate (global) agreement as well as the proportion of exact agreement. In calculating the proportion of approximate (global) agreement, when both patients' and nurses' responses differ from each other on the response scale by one category in either direction, differences can be interpreted as small. Only differences of more than one category are considered large. The proportion of approximate (global) agreement in Sneeuw's study (1999) was 43% and in regard to outcomes of the exact agreement (41%), it was concluded that a reasonable level of agreement exists between patients and nurses.

Finally, in order to confirm the findings of correlations and the proportion of exact agreement, the Bland-Altman scatter plots were checked. This method is one of the most accurate ways of measuring the level of agreement in which difference between scores of two raters can be seen visually (Bland & Altman 1986). The scatter plots also confirmed around 6 out of 20 points (or around 30%) of difference between a patient's QoL score and a nurse's score in different domains. This difference also indicates that no substantial agreement exists between each patient and nurse about each cancer patient's QoL.

Checking the Bland-Altman plots can also be helpful to identify whether the level of agreement is related to the patients' level of QoL or not. This is an important point because those patients for whom outcomes of the proxy research studies may be vital, usually cannot take part due to their poor health status (non-respondent participants). For example, those patients at an advanced level or those who are dying from cancer cannot participate. Research studies cannot confirm if an agreement at the moderate level between patients and nurses, for instance, can be generalised exactly to this non-respondent group of patients.

Having checked the plots in this study, no obvious pattern was identified that can imply that the level of agreement between patients and nurses about cancer patients' QoL was better for cancer patients with a specific level of QoL. It was therefore concluded that the results of this study that no substantial agreement exists between cancer patients and nurses about cancer patients' QoL may not be generalised to non-respondent patients. These results differ from a study by Sneeuw et al. (1998) who reported more scatter (poorer agreement) at moderate levels of patient' QoL, with less scatter at either extreme. In other words, in Sneeuw et al.'s study, proxies had a better understanding of QoL of those patients with either higher or lower rating of their QoL compared with those in the middle range.

The empirical outcomes of this study indicate that at the individual level each nurse differs in their perceptions of each patient's QoL. This is a new finding in Australia indicating that cancer patients and their nurses differ in their perceptions. Such findings are also found in the interpretive phase of this study. Generally, nurses had difficulty defining their patients' QoL and were uncertain about the QoL assessment they conducted. Consider, for example, the following statement:

You probably do not assess it [quality of life] so much, you probably subconsciously assess quality of life for patients ... your assessment of quality of life might be a general statement of your perception for that person that would be completely wrong. (Nurse 3)

When a nurse differs in his/her perceptions of each patient's QoL, it is most likely that their decisions do not meet a patient's needs. Conversely, when a nurse has a reasonable understanding of a cancer patient's QoL, their decision-making is relevant to a patient's needs. In turn, the nursing care they will provide may improve patients' QoL and care can be more individualised (King 2006).

Even though each nurse differs in their perceptions of each patient's QoL at the individual level, a comparison was also made between the QoL mean domain scores of patients and nurses using t-test. The aim was to understand how patients differ from nurses at the group level. This comparison revealed two important points. Firstly, nurses' mean domain scores are similar to those of patients in the physical and psychological aspects of patients' QoL. Secondly, nurses underestimated patients' QoL in the social relationship and environmental domains. These important issues are discussed below.

(b) At the group level, nurses' mean domain scores are similar to those of patients in physical and psychological domains

One important point arising from comparing the means of the two groups is that in the physical and psychological domains, the QoL mean domain scores of nurses were very close to those of their patients. In clinical research trials, the similarity of QoL mean domain scores of patients with those of proxies is more important than individual scores and may be useful in substitution of missing data arising from non-respondent patients (Sneeuw et al. 1997b; Sneeuw et al. 1999; Tripoliti et al. 2007).

The reason for this is that in such research studies groups of patients are compared with each other rather than as individuals. If comparison between patients' and nurses' scores only is made on the basis of mean scores, empirical findings of the research study indicates that QoL mean domain scores of patients are close to those of patients in physical and psychological domains. So researchers could substitute QoL mean scores of nurses with those of patients, of course with the necessary caution, in order to manage missing data in their research studies. For example, a researcher may want to assess the effect of a new chemotherapy regimen on survival rate and the QoL of a group of cancer patients. Because some patients are in an advanced stage of cancer, they cannot fill out the QoL questionnaire. Due to similarity between nurses' scores and patients' scores in physical and psychological domains, researchers may ask nurses who know the patients well to act as patients' proxies in this situation. However, the substitution of patients' scores with nurses' scores might be even more accurate in the physical domain because not only the means of the two groups are the same but also the correlations (r and ICC) are stronger in the physical domain compared with the psychological domain (0.48 versus 0.19).

Although QoL mean domain scores of patients were similar with those of nurses as patients' proxies in the physical and psychological domains, the comparison between QoL mean domain scores of patients with those of nurses in the social relationship and environmental domains had different outcomes. This finding is discussed next.

(c) At the group level, nurses underestimated patients' QoL in the social relationship and environmental domains

Another important point emerging from comparing means of two groups (patients and nurses) is that patients' QoL domain scores were significantly higher than that of nurses in the social relationship and environmental domains. In other words, nurses significantly underestimated patients' QoL in the social relationship and environmental domains. This has been reported in many other studies in which proxies reported lower levels of QoL than patients themselves (Pierre et al. 1998; Sprangers & Aaronson 1992; Wilson et al. 2000). Because nurses did not have a clear idea about the social relationship and environmental domains, they underestimated patients' QoL. There are several underlying reasons why nurses may have underestimated cancer patients' QoL.

The first reason refers to differences between patients' and nurses' expectations, standards, and goals. There is more evidence now that cancer patients go through a process named 'response shift' during which they change their internal standards and expectations (Breetvelt & Van Dam 1991; Schwartz & Sprangers 2002). Through this process they adapt themselves to a stressful situation (King et al. 1995). This does not mean that they have no issues at all but they can cope and manage issues to the level that satisfies them. This can lead to re-evaluation and what they deem as a reasonable level of QoL.

In contrast, staff may not understand clearly such internal changes (King et al. 1995; Molzahn, Northcott & Dossetor 1997; Tamim, McCusker & Dendukuri 2002). They may judge patients' QoL based on their own expectations, standards, and goals. Nurses may also compare a patient's circumstances with other patients and make a judgment as to which ones would have more problems. Even though comparing individuals' QoL with each other as a way to assess one's QoL might work occasionally, this approach is generally problematic because each person is different in terms of their expectations and goals.

The second reason for nurses' underestimations of cancer patients' QoL relates to differences between patients' and nurses' knowledge and experiences. Staff generally have more information about the cancer and its process than the patients. During their

relationship with cancer patients, nurses might develop a perception that cancer is a devastating problem that dominates all aspects of patients' lives. They might see more people dying from cancer rather than surviving (Lampic et al. 1996). This may lead to more negative and pessimistic thoughts in the staff, which are also labelled as a 'clinician's illusion' (Breetvelt & Van Dam 1991; Larsson, von Essen & Sjoden 1998). On the other hand, cancer patients may not be given as much detailed information about their prognosis by health care professionals. This might lead to the outcome that patients generally have more positive views about their lives and their QoL after being diagnosed with cancer compared with health care professionals.

The third reason for such underestimations is for nurses to "reassure themselves that their value systems are still meaningful: a phenomenon called the requirement of mourning" (Jennings & Muhlenkamp 1981, p.485). Consider, for example, the following extract that describes the requirement of mourning:

When a person has a need to safeguard his values, he will either (1) insist that the person he considers unfortunate is suffering (even when he seems not to be suffering) or (2) devaluate the unfortunate person because he ought to suffer and does not. This implies that the devaluer wants the unfortunate person to suffer. He wants him to suffer as a sign that the values denied the unfortunate person are still worthy and important and good. Especially if this security depends upon maintaining these values will he insist that the unfortunate person admit his suffering. (Wright 1960, cited in Jennings & Muhlenkamp 1981, p. 1981)

In this study, for example, nurses underestimated patients' QoL in the social relationship and environmental domains. Nurses implied that they are actually aware of how these domains are still important for patients even though they do not have enough information about these domains.

Finally, patients might rate their own QoL better than the actual situation in order to have a better sense of well-being. This is named 'self-deception' during which patients try to retain a belief in order to feel more comfortable and satisfied even though they may know that it is not true (de Sousa 1998, cited in Larsson, von Essen & Sjoden 1998). This might work as a coping mechanism for patients, particularly at difficult stages. For example, when the patients' QoL has significantly deteriorated following hearing the cancer diagnosis, this strategy might be used by patients.

Patients may also overrate their own QoL in order to please health care professionals (Schipper & Levitt 1985). Health care professionals may feel better when they assume that their clients generally are satisfied with the kind of treatment or care they are receiving. Then patients might receive more attention from the health care professionals (Westbrook & Nordholm 1986, cited in Lampic et al. 1996). Patients might also overrate their QoL because they do not want to be considered a burden (Rubenstein et al. 1984). They may want to be perceived as someone who is happy and coping.

Therefore, different reasons are involved in nurses' underestimation of patients' QoL. Whatever the reason, these underestimations by nurses have negative consequences for patients. Nurses are usually with the patients from the beginning of their diagnosis to the end of their lives. Patients may try to adapt themselves to the cancer, and underestimations in QoL by nurses might postpone this process. So this impacts on nurses' therapeutic relationships with patients and their role may not be as supportive as it may otherwise be (King 2006a).

It can be concluded from the outcomes of this part that often individual nurses will differ in their perceptions about each cancer patients' QoL. However, nurses as a group have a reasonable understanding of the physical and psychological aspects of cancer patients' QoL, but they tend to downgrade patients' QoL in the social relationship and environmental domains. This is also portrayed by nurses in the interpretive phase:

The quality of life of the patient was not what I thought. They seemed quite happy, no issues. Even though they lived alone, on the pension, did not have any friends, be in pain, cannot work, but they did not seem bothered by that. They thought that their lives are meaningful. (Nurse 5)

Although the above general trends were identified about the level of agreement, different conditions or variables can influence variations in patients' and nurses' perceptions of cancer patients' QoL. These situations are explained in the following section to answer the second question.

Question Two: Why do differences and/or similarities exist between cancer patients' and nurses' perceptions about cancer patients' QoL?

Why patients and nurses differ in their perceptions of cancer patients' QoL, is mainly answered by considering categories that were found in the second phase of this study. However, interpretive outcomes are also supported by some findings in the first phase. The reasons for the difference in perceptions are discussed in the following sections in relationship to: (a) Emergence of spirituality, (b) How nurses assess cancer patients' QoL, (c) Barriers to QoL assessment, and (d) Patients' and nurses' clinical and demographic characteristics.

(a) Emergence of spirituality

Even though most of the QoL domains found in this research study are the same as those introduced by the WHO and applied in the construction of the WHOQoL-BREF questionnaire (such as physical, psychological, social relationship, and environmental factors), participants in the interpretive phase emphasised that spirituality is an important issue. In the WHOQoL-100 questionnaire, spirituality is a separate domain. Spirituality was modified in the construction of the WHOQoL BREF questionnaire (Department of Psychiatry of the University of Melbourne 2000) and is measured mainly by the item 'having a meaningful life' under the psychological domain. It is interesting that participants highlighted spirituality as an important aspect of QoL. Other researchers (Hinds & King 2003; Taylor 2003) also proposed that spirituality is a significant element of QoL and that it is important to measure it in great depth.

Spirituality may focus on a belief in, or a relationship with, a higher power which gives purpose, meaning, and direction to life and may include religion (Baker 2003). Even though people might conflate spirituality with religiosity, spirituality is a broader concept that deals with both religious and non-religious thoughts and behaviours (Holley 2007). Due to the importance of spirituality in this thesis, it will be discussed in the following sections in more detail along with outcomes of the second phase of the research study.

Religious aspects of spirituality

Religiosity is an important aspect of spirituality. Taylor (2003) explains that religiosity is a narrower concept than spirituality and shapes the individuals' worldview to answer questions related to the meaning of life. Participants in this research study also highlighted that issues related to the meaning of life and what will happen after death are quite important. Consider, for instance, the following remark that was put forward by a nurse who took part in the second phase:

The spiritual needs to come into it [quality of life]. So they [patients] will talk a lot about existential stuff, things like death and dying and how patients actually want that to happen, whether they actually believe in certain things that might happen after they die. (Nurse 9)

This study did not aim to explore how religiosity might be beneficial for cancer patients. However, an ample amount of literature exists indicating how religious thoughts and beliefs might help people in their lives and make them happier. For example, from the literature it can be seen that a relationship and connection with God is an important source of strength for many people (Burkhardt 1994) particularly in more stressful situations (Maton 1989) such as when individuals experience cancer (Jenkins & Pargament 1988). Gall (2004) conducted an empirical research study involving thirty-four men with prostate cancer to explore the relationship with God and its influence on QoL issues. It was identified that relationship with God is a significant predictor of role, emotional, and social functioning for these participants. For example, those patients who considered that prostate cancer was in any way related to God perceived that they have a greater control over the illness and its management. They also viewed God with a variety of positive characteristics which would support them particularly in stressful situations. In another study involving a group of breast cancer patients, religion was identified as a very important factor in coping with breast cancer (Johnson & Spilka 1991).

Religious texts argue that the connection with God is very important and highly related to individuals' fulfilment. For example, in Islamic literature a number of Islamic Persian philosophers like Ibn Seena (commonly known in English by his Latinised name Avicenna) (cited in Morewedge 1973) argue that all means and states of happiness and satisfaction are helping human beings reach God and individuals

find a true sense of happiness in connection with God (Avicenna 2005). This idea can further be seen in the following extracts from the Holy Quran:

[O, People]Be sure that real tranquillity for the hearts rests in Allah's Remembrance [both in thought and action] (Quran, 13:28)

Those who avoided worshipping false deities and turned to Allah, seeking his forgiveness, for them will be glad-tidings [happiness], so [O, Messenger] give the good news to my obedient worshippers! (Quran 39:17)

[And the Devoted Friends of Allah are] those who believe in the oneness of Allah and constantly act piously. For them is glad-tidings [happiness] in the life of this world and in the Hereafter. No change can be there in the words and promises of Allah; this is indeed a great achievement. (Quran 10:63-64) (Saffarzadeh 2001)

These verses from the Holy Quran, show that God (Allah) guarantees happiness, now and forever, for God's believers. We learn from the Holy Quran that happiness is a quality of the soul. A person who attains all materialistic desires such as money, power, and reputation may end up being an unhappy person. Other religions and Western philosophers also discussed similar issues. For example, Western philosophers like Aristotle (cited in McKeon 1947) long ago explained that:

> ...the question is asked, whether happiness is to acquired by learning or by habituation or some sort of training, or comes in virtue of some divine providence or again by chance. Now if there is any gift of the gods to men, it is reasonable that happiness should be god-given and most surely god-given of all human things inasmuch as it is the best. But this question would perhaps be more appropriate to another inquiry; happiness seems, however, even if it is not god-sent but comes as a result of virtue and some process of learning or training, to be among the most god-like things; for that which is the prize and end of virtue seems to be the best thing in the world, and something godlike and blessed. (Aristotle cited in McKeon 1947, p. 322)

Similar issues also pointed out by participants who took part in this research study:

I guess in Western society there is often some doubt that money can buy a lot of things but it cannot buy good health and if you have a diagnosis that has a poor prognostic outcome and you are dying. (Nurse 7)

Participants in the second phase of the research study emphasise that people value health and relationships and the spiritual life and doubt that material riches are important to happiness or able to solve all problems. Accordingly, one important aspect of spirituality for many people is that religiosity needs to be considered in a QoL assessment. But non-religious aspects of spirituality are also important and worthy of discussion as explained below.

Non-religious aspects of spirituality

Spirituality is not limited only to religious beliefs and thoughts (Mahlungulu & Uys 2004). Mahoney and Graci (1999) conducted a research study to explore the possible differences existing between religiosity and spirituality. Two groups of experts in the fields of death studies and spiritual studies responded to 54 statements in a questionnaire. One main finding of the research study was that both groups identified themselves to be spiritual but not religious. The main categories related to spirituality explored in this research study by both groups were charity, community or connectedness, compassion, forgiveness, hope, meaning and mortality.

The above findings relating to the broader aspects of spirituality were also put forward by Taylor (2003), specifically for cancer patients. She proposed several positive explanations for individuals who experience cancer. Cancer patients might perceive changes in themselves (e.g. development of new skills and confidence), changes in relationships with other people (e.g. more receptivity, sensitivity, and compassion for others) and changes in their philosophy of life (e.g. more spiritual awareness). Such changes might allow cancer patients to conclude "cancer was the best thing that happened to me" or "cancer made me a better person" (p.107). Such comments are consistent with the outcomes of other studies arguing that cancer helps people to have a greater understanding of their lives, develop a greater appreciation of daily activities, or understand the importance of their supportive family and reliable friends (Hinds & King 2003; McGrath 2004).

Similar issues related to spirituality were also remarked on by some participants in the second phase but in a slightly different language:

I think there are many aspects to quality of life in everyday life that you take them for granted. (Nurse 7)

Participants used the language regarding the 'taken-for-granted aspects' of QoL to show that they had stopped to reflect on the meaning of life. In other words, participants highlighted that many of them live without a deep consideration as to what really satisfies them. On occasions when individuals' QoL is challenged, for example, when they have a life-threatening disease like cancer, they might then think deeply about issues that contribute to their QoL. Generally, individuals may think about their health when they are ill or their valuable lives when death is approaching.

One main reason for differences existing between patients' and nurses' perceptions about cancer patients' QoL may be linked to different aspects that influence individuals' QoL. For example, spirituality is an important issue for cancer patients that may not be assessed in great depth using a QoL tool or even an interview. How nurses actually assess cancer patients' QoL is discussed further in the following section.

(b) How do nurses assess cancer patients' QoL?

Another underlying reason for differences between cancer patients' and nurses' perceptions about cancer patients QoL can be seen in how nurses actually assess the QoL. The level of agreement between cancer patients' and nurses' perceptions about cancer patients' QoL in the first phase of the research study was assessed using the WHOQoL-BREF questionnaire. Such tools have been structured to measure QoL more rigorously across specific domains or dimensions. However, categories identified in the second phase indicated that nurses' assessments of patients' QoL in specialist and non-specialist oncology wards (the major sampling environment) were mainly conducted informally when nurses communicated with patients. Findings of the interpretive phase also indicated that such assessments focused on the physical aspects.

Findings of the study are similar to a research study conducted by King et al. (2002) that explored QoL of cancer patients from the perspective of oncology nurses. In this research study a focus group was used involving 24 oncology nurses in the United States of America. When the interviews were transcribed and after coding, five main categories and a conceptual model were identified. The major outcome of the study was that nurses' assessments of patients' QoL depended on the strength of their relationships and nurses with a better relationship with patients could better assess cancer patients' QoL.

This does not indicate that a QoL assessment achieved by nurses picking up on cues is not an appropriate method of assessment. In fact, even though nurses do not use formal QoL assessment language they do focus on and assess QoL (King 2006b). In research conducted by Fitch (1998), 25 oncology nurses were interviewed in order to explore their perceptions, values and behaviours in relation to understanding QoL. The major categories explored in this study were: defining QoL; assessment of QoL; the role of nurses in quality of life; conflicts surrounding QoL; and factors that help or hinder that understanding. The study concluded that while nurses may not use the language of QoL, they are dealing with QoL issues. Another point highlighted in this research was that the ability of nurses to achieve QoL assessment goals and thereby build a close relationship with patients.

The research concludes that nurses generally appear to assess the QoL of patients more informally during interactions with them rather than through the application of QoL tools. As an outcome of the second phase, it was noted that the state of QoL might change over time and that is relatively dependent on individual priorities and feelings. QoL tools may not be able to identify these changes unless they are performed longitudinally and frequently, and often this is not practical. Therefore, the best way for nurses to assess patient QoL is to improve their rapport and ability to develop a relationship. However, the research identified a notable exception in that less experienced junior nurses might have some interest in using QoL tools for more formal assessment of patient QoL as guidelines while they develop better skills and gain additional experience of assessing patients in their care. Nurses can use some QoL tools as trigger questions to initiate conversation about patients' QoL. Several publications (Caton & Klemm 2006; Sherman & Dyess 2007) addressed the importance of guidelines in dealing with issues that novice nurses face in clinical practice.

In Benner's seminal work, *From Novice to Expert* (1984, pp.13-34), nursing proficiency was defined using five levels. These levels consist of novice, advanced beginner, competent, proficient, and expert. Novice nurses generally rely on rules, guidelines, and standards rather than experience based knowledge. Even though discussions have occurred regarding Benner's idea over the years, it is still generally accepted that experience is an important issue in the clinical area in order to provide

better care. An interpretive research study by Taylor (2002) observed how novice and expert nurses access different sources of information before meeting a patient. One main finding of this research study was that expert nurses were more likely to access multiple sources of information than novices. Novice nurses in this research study also commented that they base their care on previous contacts with a patient. Taylor remarked that novice nurses who rely heavily on previous contacts must to do the next step, and this is consistent with the classification of novice-expert nurses (Benner 1984).

This thesis found that differences existing between cancer patients' and nurses' perceptions about cancer patients' QoL are strongly related to poor relationship and rapport between patients and nurses. However, there are some barriers that may hinder nurses establishing a good relationship with patients and thoroughly assessing patients' QoL.

(c) Barriers to QoL assessment

Interpretive findings of this study found that there are some important barriers that prevent nurses from doing a thorough QoL assessment. One of these barriers is time limitation or constraint. The interpretive phase of the research study found that nurses generally have limited time to do a thorough QoL assessment. Interpretive findings were further supported by empirical findings of the first phase, indicating that nurses generally spent around 2.5 hours per shift with patients. A thorough QoL assessment is time-consuming and such time limitations can prevent nurses from becoming familiar with all aspects of patients' QoL (Frost, Brueggen & Mangan 1997).

Mohan et al. (2005) conducted a research study in two hospitals in the western suburbs of Sydney. The aim was to describe the experience of nurses caring for cancer patients in non-specialist wards using an interpretive approach. Six major categories were explored and one main category was lack of time. The researchers explained how time limitation compromised the quality of care delivered to cancer patients and their families. Nurses in this study generally wished to spend more time with patients and due to time limitations they felt frustrated and helpless. Even though time limitation is a problem for QoL assessment, this study also highlighted another issue about the appropriate use of time. For example, like many other research studies (Nasrabadi, Emami & Yekta 2003; van der Walt & Swartz 2002) this thesis suggests that nurses working in a time limited environment may concentrate on the specific tasks they need to do. If nurses have extra time left after completing essential patient care tasks, they may spend time on activities such as talking with patients and developing rapport.

As found in the second phase, fragmentation or discontinuity exists in the health system in general and specifically in the nursing system, and this can also work against nurses who want to establish a better relationship with patients and thoroughly assess their QoL. Relational discontinuity might emerge from infrequent contacts between patients and health care professionals or when quality of communication is poor (Dumont, Dumont & Turgeon 2005) and may lead to feelings of isolation and dissatisfaction in patients and their families (Hearn & Higginson 1998). This emerged during the second phase. Consider, for example, the following remark by one patient interviewee:

It [the health system] was very fragmented...The outcome is disenfranchisement. You feel alone and I know that my family were increasingly worried about me but because there was no one really to contact, I was out on a limb. I wasn't supported. (Patient 3)

Therefore, establishing a good relationship and rapport with patients and conducting better QoL assessment is also linked to managing problems nurses face such as time limitation and relational discontinuity. There are also some clinical and demographic variables concerning patients and nurses that can influence nurses' understanding of cancer patients' QoL.

(d) Patients' and nurses' clinical and demographic characteristics

Another aim of the study was to identify if any clinical and demographic characteristics (such as patients' and nurses' age, gender, marital status, level of education, patient's current treatment, treatment setting, nurse's QoL, educational level) may influence the level of agreement between patients and nurses about cancer patients' QoL. In the first phase clinical and demographic variables were collected using patients' and nurses' characteristics forms and analysed using Bivariate and

Multivariate analysis.

A number of variables were identified associated with the level of agreement between patients and nurses at the Bivariate level. Bivariate analysis does not take into account the interrelations existing amongst the variables themselves. So it might lead to misleading judgments. For example, at the Bivariate level it was identified that in the environmental domain, the level of agreement between patients and nurses decreased when nurses have lower levels of QoL or when patients are treated in inpatient departments. Using Multivariate analysis, only nurses' QoL was identified as a significant predictor of the level of agreement in the environmental QoL domain. During a similar process, Multivariate analysis showed that in the psychological and social relationship domains, the level of agreement is better when patients are treated in outpatient departments. In the physical domain, none of the variables was found to be a significant predictor of the level of agreement using Multivariate analysis. However, at the Bivariate level both the variables nurse clinical experience and nurse clinical experience with cancer patients were significant.

The Multivariate analysis also showed that in each domain patients' and nurses' clinical and demographic variables together explained around 5% of variance in differences between patient-nurse scores. In other words, clinical and demographic variables may not be considered significant predictors of the level of agreement at the Multivariate level. In a study by Sneeuw (1998) several demographic and clinical variables of the patients and their significant others were associated with the level of agreement using Multivariate analysis. However, they explained less than 15% of the variance in patient-proxy differences which was considered as a trivial proportion. Given the fact that only 5% of variance in differences between patient-nurse scores was explained by clinical and demographic variables at the Multvariate level, it was necessary to investigate other clinical and demographic variables influencing the level of agreement in the interpretive phase. Moreover, the interpretive phase provided an opportunity to elaborate on some significant findings of the first phase.

In the interpretive phase, patients' and nurses' clinical and demographic variables of the first phase again were discussed with participants. In the meantime participants were also asked to discuss any other variables which might influence nurses' understanding of cancer patients' QoL. In general, in the interpretive phase three variables (nurses' clinical experience with cancer patients, treatment setting, and nurses' QoL) were found to be salient to nurses' understanding of patients' QoL (the level of agreement).

Factors influencing nurses' understanding of cancer patients' QoL (the level of agreement) in both phases of the research study can be seen in **Table 8-1**.

Table 8-1: Three factors influencing the level of agreement between patients and nurses about cancer patients' QoL

Factors	Questionnaire survey findings		Interpretive findings
	Bivariate	Multivariate	
Nurses' clinical		-	
experience with cancer			
patients			
Treatment setting		\checkmark	
(Inpatients vs.			
Outpatients)			
Nurses' QoL			

 $\sqrt{}$: Significant factors

The above prominent factors salient to nurses' understanding of patients' QoL are discussed further in following sections.

Nurses' clinical experience with cancer patients

One important new finding in the first phase of the research study at the Bivariate level was that nurses with more clinical experience, particularly with cancer patients, had a better understanding of patients' physical and environmental QoL domain. Participants in the second phase also highlighted that nurses with some sort of experience may have a better understanding of cancer patients' QoL. Participants in the interpretive phase elaborated on possible reasons why more experienced nurses might better understand patients' QoL.

Participants reasoned that more experienced nurses possibly have developed better assessment and communication skills because they have communicated with a wider range of patients and can understand differences existing among patients in terms of their QoL. They might also have developed greater proficiency in their workplace and can manage a wide variety of tasks. Taylor (2002) stated that a highly developed level

of skill is recognised as essential in all expert practice. Even though education is an important factor to improve nurses' knowledge and skills, such nurses' qualities are often related to years of experience (Caton & Klemm 2006).

Participants who took part in the second phase of the study also commented that more experienced nurses can pick up more on 'little things' or cues as a way to assess patients' QoL. More experienced nurses themselves highlighted that they might assess patients' QoL by their 'gut feeling'. Literature is supportive of this notion that experienced nurses often use intuition or gut feelings to guide them in the clinical area even though sometimes they may not realise how their knowledge and practice have been influenced by such intuition (Hams 2000; Poh 1997). For example, Benner (1984, pp.13-34) explained that expert nurses do not rely only on rules and guidelines to connect their understanding of the situation to an appropriate action. They have an intuitive grasp of the situation and can concentrate on the problem without having to consider a large range of unproductive alternative diagnoses and solutions. When such nurses are asked, for example, why they acted in such a way, they may reply, "because it looked good". Ruth-Sahd (1993) commented that emphasising intuition does not indicate that expert nurses' knowledge and action are not supported by rationales. Rather, they use rationales but do not forget the importance of intuition.

MacLeod (1994b, p.363; 1994a) in two Scottish teaching hospitals analysed nursing care delivered to surgical patients by ten experienced nurses via participant observations and interviews. Nurses take note of simple signs of patients such as a change of focus of attention, interest in an outside event, interest in appearance or even being ready to wear false teeth. They also mentioned that sometimes there is no specific sign and they feel overall that patients "just look better". They also stated that sometimes their work consists of just doing "little things" such as making a handle for a chest drainage bottle or placing a bath mat differently. However, MacLeod argued that such simple assessments and practices are based on nurses' complex knowledge and skills they have gained through their experiences.

These results are very similar to another grounded theory research study conducted by Atkins (1998) in Australia. The main aim of the study was to discover and conceptualise nurses' views about QoL of patients with severe multiple impairments.

One of the main findings of this study was that nurses discovered that QoL of this group of patients for them meant "just little things" (p.134). However, this research study argued that even though nurses use the language of "just little things", the complex nature of nurses' knowledge and skills explains this. To address the complexity of these little things, Atkins explained that nurses become more intimate with people who have multiple impairments. Only after that do nurses realise that these patients are human beings and that their everyday lives need to be supported (Atkins 1998).

It can be concluded that more experienced nurses might better understand cancer patients' QoL because they not only have more skills but also may pick up more effectively on cues using their intuition.

Patients' treatment setting

Another important issue identified in the first phase of the research study was that nurses in outpatients' oncology units have a better understanding of patients' QoL, particularly in the psychological and social relationship domains. Participants of the second phase commented that the main reason why nurses in outpatient oncology wards have a better understanding of patients' QoL is that they are generally more experienced. Moreover, outpatient wards included in this research study were mainly outpatient chemotherapy units. The style of communication in these wards is that nurses sit and talk with patients while delivering chemotherapy drugs. This encourages nurses to develop a closer relationship with patients and better unpack or understand private and personal issues. Finally, participants mentioned that patients come to the outpatient wards on a regular basis, for example, five days per week for 3-4 weeks continuously. Each day patients might stay for a couple of hours. So, this might provide nurses with more relational continuity and better understanding of patients' QoL.

There are some issues arising from reviewing the literature that also shed light on why working in inpatient wards might hinder nurses' understanding of cancer patients' QoL. Although prolonged contact with the patient in inpatient departments might potentially increase the level of agreement (Magaziner et al. 1988), it can also lead to disagreement when a nurse is under pressure to do different tasks (von Essen 2004). It

might be the case that patients in inpatient oncology wards usually are sicker than patients in oncology chemotherapy units. Therefore, nurses might need to spend more time and do more tasks for such patients and it may be more difficult to communicate and to assess their QoL. Outpatients may have a better QoL than inpatients (Isikhan et al. 2001) and the level of agreement between patients and nurses about cancer patients' QoL could be better either when patients are very sick or when they are very healthy (Sneeuw et al. 1998; Sneeuw et al. 1999).

The findings of this research study demonstrate some important characteristics that exist in chemotherapy oncology units and may facilitate nurses' understanding of cancer patients' QoL. If these characteristics can be implemented in inpatient wards, it may improve nurses' understanding of cancer patients' QoL.

Nurses' QoL

Empirical findings identified that the majority of nurses rated their QoL as being good. It was also found that nurses with better QoL could better assess patients' QoL in the environmental domain. It might be argued that when nurses have lower levels of QoL and personally experienced tougher lives, they might better understand patients' QoL because they might have more empathy with patients who are assumed to be in a parlous situation. Even though such an empathic understanding can work in some instances, this may not be enough for a deeper level of understanding. Nurse interviewees who took part in the second phase explained that when they are happier with their lives they often make more open relationships with patients and will be more therapeutic as a result. They stated that they might be more 'switched on' and listen to patients better. This is interesting to note that the literature review could not identify any material that specifically focused on nurses' QoL and how it can influence their understanding of cancer patients' QoL. However, the literature shows that while nurses try to improve patients' QoL, their own QoL may be impacted. Consider, for example the following comment by Baycroft (2002, p.3):

As nurses, we are deeply concerned with maintaining and improving the quality of life of those we care for. It is ironic that in pursuing this career, my own quality of life has taken a severe nosedive...a shift can leave me so drained that my own health and well being is neglected and the emphasis on getting the tasks done, leaves little or no time for the rewarding and satisfying aspects of the job.

When nurses' QoL falls, this in turn can influence their understanding of patients' QoL. A literature review by Neumann, Araki and Gutterman (2000) revealed that proxies who reported more caregiver responsibilities and subjective stress from caregiver duties provided more negative assessments of subjects' health and wellbeing. In another study, family proxies' QoL responses were compared with those of patients. One part of the results indicated that agreement was better when caregivers reported lower levels of care-giving burden (Williams et al. 2006). These findings are similar to another proxy research study suggesting that when caregivers have more pressures, it is most likely that they will overestimate the older person's disability (Long, Sudha & Mutran 1998).

Thus it appears that how nurses feel about their own QoL can influence their understanding of cancer patients' QoL.

Limitations

The findings of this research study must be interpreted within its limitations.

Limitations of the first phase

The first phase was conducted in three major hospitals across several wards including two specialist oncology wards, five non-specialist oncology wards, three outpatient chemotherapy units, one radiotherapy centre and one palliative care area. The sample size drawn from this population of patients and nurses was calculated based on the minimum number required to identify the level of agreement between patients and nurses. This can be considered as a representative sample size and results having external validity to the wider Australian community. However, as is the case with many other research studies, those patients and nurses who were selected from oncology wards will share a common work environment and might be influenced by each other (intra-group effect). Naturally, a sample of patients and nurses from a greater number of oncology wards may decrease intra-group effects and provide more rigorous findings.

Secondly, before conducting the first phase, nurses were well informed about the

importance of the research and the need to complete the questionnaires as accurately as possible. However, the study was conducted in busy wards and on some occasions this may have affected nurses' concentration while completing the questionnaires. This is a problem that has also been reported in other studies (Sneeuw et al. 1997b; Sneeuw et al. 1998).

Thirdly, in this research study an attempt was made to include as many nurses as possible so that each nurse assessed only one patient. However, the population of cancer patients varied from time to time whereas the nurses' population did not alter much. Taking into account that the response rate of patients was more than that of nurses, the research study was faced with a higher sample of patients who had not had a nurse pairing. Therefore, proceeding with only one nurse assessing one patient, significantly decreased both cohorts of patients and nurses. Despite the best efforts undertaken during this longitudinal research study, the number of nurses was only slightly more than half of the patients (57.3%). In other words, each nurse took part in the study nearly twice. So this might lead to the bias of assessing many patients with a few proxies (Broberger, Tishelman & von Essen 2005; Lampic & Sjoden 2000). This problem has been reported elsewhere with proxies not always equal to number of patients (Blazeby et al. 1995; Molzahn, Northcott & Dossetor 1997; Slevin et al. 1988; Sneeuw et al. 1998). In most of these studies there is no statement as to how many times proxies assessed patients as proxies. However, a comparison between patients and proxies who took part in these studies shows that such assessments were done more than twice by proxies. Fortunately, this is an advantage in this thesis compared to other studies because each nurse on average assessed two patients.

Fourthly, in the first phase of the research study patients' scores were considered as the best possible information available and nurses' scores were compared with them. In this study it was found that nurses generally tend to underestimate patients' QoL. However, it can be argued that patients' scores themselves are prone to some bias as stated previously. For example, patients may rate their QoL better to please health care professionals. If this is the case, patients in this study have rated their QoL higher than what they really felt. Therefore, nurses might be closer in their rating of patients' QoL with that of patients' own rating in a real situation. However, QoL perceptual differences refer to a number of factors as discussed in this chapter. Finally, nurses completed the QoL questionnaire for patients usually on the same day as patients did, but did not do so immediately after them due to work issues. However, participants in the second phase commented that their state of QoL might change over time. The pattern of changes and how frequently such changes might occur for patients has not been addressed yet in the literature. It is possible, however, that differences between patients' and nurses' scores were linked to changes occurring in the patients' level of QoL later when nurses assessed patients' QoL. Nurses might be closer in their rating of patients' QoL with that of patients' own rating if it was possible for them to assess patients' QoL immediately after patients did.

Limitations of the second phase

Firstly, the qualitative data was mined as extensively as possible and saturation reached to the researcher's satisfaction within the time available. There are other characteristics of categories that were not explored in this research study. This is a common feature in qualitative research studies including one based on the principles of grounded theory. No matter how often the data is investigated, new characteristics may constantly emerge. However, future research studies may be needed to more fully explore other characteristics.

Secondly, even though conceptual sampling was used in the interpretive phase of the research study, this sampling was implemented into a convenience sample of patients and nurses from two major public hospitals who were willing to take part. There was quite a good variation within the sample chosen. However, for transferability to be achieved more effectively, a random sample from patients and nurses from a wider population and more hospitals is necessary.

Finally, the researcher previously explained issues related to his background, culture and language as an international student. He also described what was conducted to manage these issues. However, the researcher's ability to deeply explore patients' and nurses' perceptions of cancer patients' QoL might still have been influenced by these factors. Psychological and spiritual aspects of QoL are particularly prone to be coloured more by the researcher's culture, background, and language issues.

Conclusion

In this chapter the findings of the research study were discussed. Question number one was answered using empirical findings to indicate that nurses generally differ in their perceptions to patients about cancer patients' QoL. Moreover, nurses tend to underestimate patients' QoL in the social relationship and environmental domains. The reasons for such differences were further investigated in order to answer the second question by referring to the outcomes of both phases, particularly those in the second phase. In general, participants commented that differences existing between patients' and nurses' perceptions about cancer patients' QoL were due to a poor relationship and rapport between cancer patients and their nurses. However, when nurses are more experienced and have better levels of QoL or when patients are treated in outpatient oncology wards, it is most likely that nurses can better understand cancer patients' QoL.

The next chapter concludes this thesis and summarises the key findings of the research. It also outlines implications for future research, education and clinical practice.

CHAPTER NINE: CONCLUSION

Introduction

In the previous chapter, the findings of the phases of the research study with regard to nurses' perceptions about cancer patients' QoL were discussed in the light of literature in order to answer the research questions.

This chapter presents the main implications of the research and concludes the thesis. These implications are provided under the themes of key issues discussed below.

Nurses' assessment of cancer patients' QoL in the clinical area

Empirical findings of the research study generally identified a fair (less than moderate) agreement between cancer patients' and nurses' scores on the WHOQoL-BREF questionnaire. This was further supported by interpretive outcomes in which participants stated that nurses mainly pick up on patients' cues as a way for QoL assessment when they communicate with patients in the clinical area rather than using QoL tools. Furthermore, nurses are, at times, uncertain if they are doing a QoL assessment and if their assessment of patients' QoL is correct or not. Participants also believed that the main reason for differences that exist between patients' and nurses' perceptions about cancer patients' QoL is a poor relationship and rapport between them and their nurses.

Therefore, this supports a need for nurses to develop a more holistic relationship and stronger rapport with patients to underpin the assessment of cancer patients' QoL through various cues. In the interpretive phase of this thesis the participants commented on how nurses can develop such a relationship and rapport with patients. Participants stated that what makes nurses have a stronger relationship and rapport with patients is a genuine interest in and their desire to nurture people. They believed that this genuine interest might encourage nurses to spend more time with patients and communicate with patients in an open, receiving and non-judgmental way.

Empirical findings supported by interpretive outcomes also highlighted that nurses' understanding of cancer patients' QoL, particularly is lower in the psychological, social relationship and environmental aspects. So in order to document these aspects, nurses need to develop a deeper level of communication and rapport with patients. Participants mentioned that as a result of such a well-developed relationship and rapport, nurses might empathise with a number of issues, some of which are quite personal.

Moreover, this research study found that spirituality is a prominent aspect in cancer patients' QoL. QoL tools like the WHOQoL-BREF questionnaire might marginalise spirituality because it does not measure it as a separate domain. Given the importance that participants placed on spirituality, higher levels of understanding of spirituality can be achievable only through a more in-depth relationship and rapport with patients.

Despite the importance of a holistic relationship and rapport to assess cancer patients' QoL, participants in the second phase commented that more junior nurses are interested in using QoL tools as guidelines for QoL assessment. QoL tools like the WHOQoL-BREF might be used as guidelines for such nurses to assess cancer patients' QoL rather than relying heavily on their perceptions and intuitions. However, this study found that there are some problems in using QoL tools in the clinical area. For example, one reason for not using QoL tools in the clinical area is that they are lengthy and may cause undue pressure not only for nurses but also for patients. Secondly, nurses generally face severe time limitations or constraints. However, this study has found a basis for a modified version of such tools being used in nursing practice. This is what can be seen in some palliative care areas in which nurses are using both formal (one item QoL tool plus a symptom check list) and an informal QoL assessment.

Therefore, for the purpose of QoL assessment in oncology wards as a first step, a modified version of QoL assessment as seen in the palliative area is useful. In other words, nurses might use some items of the WHOQoL-BREF questionnaire to initiate a conversation about QoL. These items would be those found in this research study in which nurses had less understanding of cancer patients' QoL (sex life, personal

relationships, living a meaningful life, accepting bodily appearance, and having negative feelings). In the meantime, however, nurses need to improve the breadth and depth of their relationships with patients to ensure that the holistic nature of QoL assessment remains intact.

Nurses as proxies in QoL research studies

Empirical findings of the research study arising from a comparison of patients' and nurses' scores on the WHOQoL-BREF questionnaire have important implications for researchers who are interested in knowing whether nurses may play a proxy role for non-respondent patients in their QoL research studies or not. This research study in general suggests that individual nurses cannot perform this role successfully. However, nurses as a group might be used as proxies for patients but mainly in the physical domain. This is due to the similarity of QoL mean domain scores of patients and nurses and stronger correlations in this area.

The next question is whether there are nurses with specific demographic and clinical characteristics who can better fill the proxy job. This may help in selecting a suitable proxy from those available (Tang & McCorkle 2002b). This research study suggests that nurses with more clinical experience with cancer patients and those who have a better level of QoL may provide a closer rating of cancer patients' QoL. Moreover, the findings of this study indicated that often nurses think that patients' families are a better source of QoL information for non-respondent patients and therefore might rely on them. This implies that proxy information provided by the patient's family in some instances can be used as complementary data, particularly for more private and personal aspects of QoL such as psychological, social relationship and environmental domains. Cancer has an effect on the whole family and they are generally engaged with the patient more than health care professionals and are familiar with the patients' situation much better than nurses (Sigurdardottir, Brandberg & Sullivan 1996).

Another important finding identified in this study is that assessment of QoL in others and even for ourselves is complex and at times difficult. It also appears from the literature that an adequate replacement for measuring QoL has not been found. However, some participants in this research study suggested that nurses can use other measures to help them to obtain a more accurate or useful idea about cancer patients' QoL and to provide more therapeutic nursing care. Such measures are particularly useful when patients cannot communicate adequately and report on their QoL. For example, the Patient Reported Outcome (PRO) was recommended as a complementary measure by several participants in this study. PRO is a measure of 'progress' or 'recovery' rather than a QoL assessment but can provide general information about the patients' situation and abilities that may be useful for health care professionals to evaluate if they have achieved their nursing goals or not, for example, when nurses help a patient up who has been lying in bed for a prolonged period and send them to the hydrotherapy pool. At the end of two weeks of doing this the patient may actually walk by themselves to their therapy session. This patientreported outcome demonstrates an improvement in the patient. Altogether, it appears that the idea of 'replacement' for QoL measurement needs to be considered seriously by researchers. Such measures, while reflecting the complexities of the QoL concept, need to be as simple as possible to be understood and performed.

Nurses' education support

One main finding in this research study is that nurses are often unskilled and uncertain about the QoL assessment they are conducting. They expressed that they need education specifically about how to do a QoL assessment. The main way for QoL assessment in oncology wards is picking up on cues when nurses communicate with patients. Education may therefore focus on issues like how nurses may build better relationships and improve communication with patients (Sivesind et al. 2003). The focus of this 'relationship program' may be applicable to issues discussed in this thesis, such as how nurses can have an open relationship with patients, what cues they need to look for and how to interpret such cues.

This education also needs to focus on the QoL concept. "Clinical oncology nurses must understand the term quality of life" (King 2006, p.5) as it applies to their patients. Participants also expressed the idea that they need education about the QoL

concept. QoL was a complex concept for participants and nurses generally had difficulty defining QoL for their patients. So introducing the QoL concept more specifically might help nurses to think more and better understand complexities of the QoL concept such as its broadness and individuality. They might be able to conceptualise QoL for specific groups of cancer patients or those who are under specific treatments and define it in a practical way when assessing QoL. Such operationalisations are quite necessary to bring perceptions of two parties (patients and nurses) much closer to each other.

Moreover, education for improving nurses' assessment of patients' QoL needs to include not only nurses but also other health care professionals. In the interpretive phase of the research study it was found that nurses are in a close relationship with other health care professionals when caring for and treating patients. They need to have a common understanding of QoL and are able to communicate effectively with patients. It is quite important that more than theoretical issues in relation to the QoL concept should be incorporated into education programmes. The more experienced nurses have a better understanding of cancer patients' QoL and their insights should be included in an education program, as reflected in the two phases of this study. Qualitative data suggested that more experienced nurses might have better communication and assessment skills. They are also better able to pick up on cues and can explain what QoL cues they look for and how they interpret them.

One direction for education is to have focus groups that include both experienced and junior nurses so that they can interact with each other. This will encourage more experienced nurses to talk about their experiences as how they have actually assessed patients' QoL. Such sessions might initiate junior nurses to develop communication skills. Another way might be that more experienced nurses conduct their communication with patients in front of more junior nurses in the clinical area. Junior nurses might then play those roles with actual cases with the more experienced nurses supervising them and make comments as to how they may improve their communication skills.

Future research studies

Some important implications emerged from conducting this research study and are useful for future research studies. Firstly, the WHOQoL-BREF questionnaire is a reliable and valid tool as recommended by the WHO. The reliability of the tool was tested again and supported by this research study. Cronbach alpha for both patients and nurses QoL domain scores (except of the social domain) approached or was above the acceptable level of 0.70. In general, a high internal consistency may not be necessary for a multi-domain tool (Cella & Tulsky 1990) like the WHOQoL-BREF questionnaire.

The low internal consistency of the social relationship domain might be related to the low number of items (only three) and their content (sex life, personal relationships, and friends) that had been reported in another research study (Hanestad et al. 2004). Generally speaking with only three items in the social QoL domain, the outcomes of Cronbach alpha may not be reliable (The WHO Group 1998a). Moreover, interpretive outcomes of the research study revealed that participants gave more weight to some domains such as spirituality. For those researchers specifically interested in the social relationship or spiritual domains of QoL, the WHOQoL-100 may be more suitable. O'Carroll et al. (2000) also argued that for more sensitive areas of QoL like social relationship domain, the WHOQoL-100 is more sensitive than the WHOQoL-BREF.

An exploratory factor analysis was also conducted to explore the structure of domains of the WHOQoL-BREF questionnaire completed by patients. Outcomes indicated that items of the WHOQoL-BREF questionnaire can be grouped into seven domains. This would be in line with the interpretive findings where participants agreed that spirituality is an important domain and there might be other taken-for-granted aspects. It appears that the WHOQoL-BREF questionnaire does not enquire about spirituality issues in as great a depth as required. Such findings imply that maybe the WHOQoL-BREF can not comprehensively assess cancer patients' QoL. However, exploratory factor analysis in this research study was conducted by a sample that might be considered small by some researchers (Tabachnick & Fidell 2001). Therefore, it is suggested that the construct validity of the WHOQoL-BREF
be explored further for cancer patients in the clinical area, using factor analysis with a larger sample size (eg. 300 or more). This would indicate if this tool in its complete form can really assess those QoL issues which are important for cancer patients.

Secondly, empirical findings of the research study identified that QoL of nurses has an influence on their assessment of patients' QoL. However, QoL of nurses was assessed against one global question because practically it was very difficult for nurses to fill out a QoL tool for themselves as well another form and a QoL tool for patients. As recommended by Haberman and Bush (2003) a single-item that measures global perceptions of QoL can be useful in some contexts. For example, in this research study, assessing nurses' QoL with a global item can be considered an initial step in searching for the possible relationships between nurses' rating of their own QoL and their understanding of patients' QoL. However, it is recommended that nurses' QoL be measured in a separate research study using a comprehensive QoL tool. The influence of nurses' QoL on the nurses' understanding of patients' QoL can then be explored further using an interpretive approach.

Thirdly, compared with other proxy research studies, this research study used a reasonably large sample size of patients and proxies (nurses) in the first phase. However, as stated previously, the number of nurses included in the first phase of the study was slightly more than half the number of patients (57.3%). This limitation in part was compensated for by conducting an interpretive study that generally supported findings of the questionnaire survey phase. However, it is still recommended that the results of nurses' understanding about cancer patients' QoL be examined in larger samples particularly with a more representative sample of nurses in Australian populations.

Fourthly, empirical findings of the research study identified that more experienced nurses have a better understanding of patients' QoL. Some nurse interviewees also stated that they assess patients' QoL using gut feeling or intuition. How the more experienced nurses use their intuition in the clinical area to assess patients' QoL deserves more investigation in another interpretive research study.

Fifthly, in the interpretive aspects of the research study it was found that nurses in

the palliative area are doing both formal and informal QoL assessment because the purpose of the ward is for patients to live more comfortably and therefore the focus is on QoL issues. It may therefore be logical that palliative nurses form more accurate ideas about patients' QoL compared with nurses in other wards. Unfortunately, only five pairs of patient-nurse from the palliative setting took part in the first phase. This number is very limited to make a comparison between nurses in the palliative area and oncology wards regarding which setting has a better understanding of patients' QoL. Consequently, a comparison of cancer patients' and nurses' understanding of cancer patients' QoL between the palliative area and other wards can reveal if the purpose of QoL assessment, i.e. to support better palliation or to maintain expectations of recovery - can really make a difference in nurses' perceptions' of cancer patients' QoL.

Sixthly, this research study identified that nurses in outpatient chemotherapy oncology units have a better understanding of cancer patients QoL compared with nurses working in inpatient wards (specialist and non-specialist). This research study also found some underlying reasons for these differences. However, a research study that is conducted which compares only specialist inpatient wards with the outpatient chemotherapy units would address this question. Specialist oncology wards may focus more on QoL issues than non-specialist wards. This research needs to be followed up by an interpretive study exploring what kinds of facilitators and drawbacks exist in the context of specialist inpatient oncology wards and outpatient chemotherapy units.

Finally, this research study found that patients and nurses differ in their perceptions of cancer patients' QoL. However, this research study did not have enough time and resources available to completely understand, from patients' and nurses' perspectives, how such differences might impact on patients, particularly their QoL. For example, in this research study it was identified that nurses underestimated patients' QoL. Therefore, it would be very useful in an interpretive study to explore how nurses' underestimations of patients' QoL influenced patients' QoL and other related issues, like patients' coping styles, in more depth.

The QoL concept

With regard to the QoL concept, this research study confirmed that it is a broad and intangible concept. After many years of study and investigation there is still no universally agreed definition of QoL and no general consensus on those aspects that constitute it. The majority of individuals spend their lives without a deep consideration of what issues really contribute to their QoL unless their QoL is challenged. What sorts of things satisfy human beings and make them really happy? Is there anything in the physical world or the social world that can really feed the true sense of happiness or we do need to search for more than materialistic or secular answers?

These questions are important and need to be answered. A deep consideration about the 'taken-for-granted' aspects of life is quite necessary particularly for those whose lives are challenged. Answers to these questions at the individual level might then help researchers to make better definitions of QoL and thus enable health care professionals to be more therapeutic. It is suggested that discussions about the QoL concept focus more on issues like 'self-identification' (who we are?) and the 'meaning of life' (why we are here? and where do we go?). In other words, research results from this thesis suggest that discussion upon the QoL concept focus more on spirituality. It is most likely that such a focus on spirituality issues, which has been started recently, can resolve many ambiguities already existing in the field of QoL. It is predicted that future research studies will focus more on the notion of spirituality.

The researcher will conclude this thesis with the following statements written by a famous Persian poet, Molana Jalal-e-Din Mohammad Molavi Rumi, and translated into English by Shahriari (1998):

Everyday I meditate upon this, and every night I groan Why is my own existence to myself the least known? Whence have I come, why this coming here? Where to must I go, when will my home to me be shown? I am in desperate awe, why was I ever created? For this, my creation, whatsoever was the reason?

Conclusion

This research study identified differences that exist between cancer patients and nurses about cancer patients' QoL. It found that such differences are mainly due to the individuality of the QoL concept – different for each person – and its breadth as well as QoL assessment being dependent on the strength of the relationship and rapport between patients and nurses. This research study used a mixed approach and an overlap in the findings of both phases of the research study was identified. The transferability of outcomes were found to have potential applicability. Some outcomes of the research study, particularly in the interpretive phase (see Chapter Seven) may have transferability to other populations, particularly the researcher's country of origin, Iran.

This research study accomplished its aims to understand nurses' perceptions about cancer patients' QoL. It addressed several important issues that may contribute to the current knowledge of QoL and push the boundaries of practice further. It also raised questions and opened doors for future researchers who are interested in this field of endeavour. The researcher hopes that the outcomes of the thesis make it more likely that those experiencing a life-threatening illness such as cancer will receive the support they need to maintain or improve their quality of life.

We must always keep in mind that cancer is not a disease that 'others' suffer from. We might be one of those future sufferers of this disease.

APPENDICES

Appendix A: The relevant article accepted for publication in *Contemporary Nurse,* vol. 29, no. 1, pp. 67- 69.

Patients' quality of life: A comparison of patient and nurse perceptions

Received 23 November 2006

Accepted 16 January 2008

Masoud Bahrami² Ph.D candidate School of Nursing & Midwifery Flinders University Adelaide SA, Australia

Dr Steve Parker Senior Lecturer School of Nursing & Midwifery Flinders University Adelaide SA, Australia

Dr Ian Blackman Lecturer School of Nursing & Midwifery Flinders University Adelaide SA, Australia

Background: Quality of Life (QoL) is a subjective perception a person has of their position in life. Tailoring care to a patient's unique needs requires nurses and patients to have a similar understanding of a patients' QoL.

Aim: This study aimed to identify: (a) the level of agreement between patients and nurses about cancer patients' QoL; and (b) variables that may affect the level of agreement between them.

² Masoud has a scholarship from Isfahan University of Medical Sciences, Isfahan, Iran (Islamic Republic), to complete his PhD in nursing.

Methods: Cancer patients (n=117) and nurses (n=49) from a public hospital were invited separately to complete the World Health Organisation Quality of Life Brief (WHOQoL-BREF) questionnaire. This assesses QoL in physical, psychological, social relationship and environmental domains, or dimensions.

Findings: Intraclass Correlation Coefficient (ICC) revealed a moderate agreement between nurses' and patients' scores in the physical QoL domain but lower agreement on other domains. A paired t-test identified patients' QoL domain scores were significantly higher than that of nurses in social relationship and environmental domains. Multivariate analysis using Standard Multiple Regression analysis demonstrated that agreement between patients and nurses was higher: (a) in the physical QoL domain with nurses who have greater clinical experience with cancer patients; (b) in the social relationship QoL domain when patients are treated in outpatient departments.

Conclusion: These results imply that differences exist between patients' and nurses' perceptions about cancer patients' QoL and nurses tend to underestimate patients' QoL in social relationship and environmental domains. Higher clinical experience with cancer patients may contribute toward a better understanding by nurses of cancer patients' QoL.

Keywords: nursing; quality of life; oncology patients; oncology nurses; World Health Organization Quality of Life Brief questionnaire; nursing care

INTRODUCTION

Quality of Life (QoL) issues have been of particular interest to many researchers, particularly in the area of oncology (Giesler 2000; He & Liu 2005). QoL has been researched with different purposes in mind such as curative, palliative, improving

symptom relief, care or rehabilitation, facilitating communication with patients, awareness of patient preferences, understanding of late problems of psychological adaptation, and for medical decision-making (Fayers & Machin 2000).

QoL is defined in this research project as "individuals' perception of their position in life in the context of the culture and value systems in which they live and in relation to their goals, expectations, standards, and concerns" (WHO Group 1997: 1). Such a definition, implies that asking patients is an ideal way to assess this individual concept (Ballatori 2001; von Essen 2004). However, there are important reasons that perceptions of heath care professionals, including nurses, about patients' QoL are also important.

The main reason is that in the clinical area heath care professionals usually form perceptions about QoL of patients based on different sources of information. Such perceptions might even be formed for those patients who can communicate, and whose QoL can be directly assessed by consulting with the patient. Such perceptions are used as a basis by heath care professionals in order to make clinical decisions, modify or even change treatments or programs for patients (King et al. 2002).

Another reason is that when a research study is conducted in the clinical area, some patients may not personally want to take part in the research study. Another group may wish to be part of the study but be unable to adequately communicate so that an assessment of their QoL may be carried out (McPherson & Addington-Hall 2003). Therefore, heath care professionals might be asked to assess a patient's QoL. In these situations, where a person other than the patient assesses a patient's QoL, the person is identified as a proxy. There are numerous QoL research studies (Slevin et al. 1988; Fisch et al. 2003) that have described situations in which heath care professionals were requested to act as proxies for such patients.

Even though family members or significant others can also fulfil a proxy role, for purposes of this research study only nurses were chosen. Nurses play a key role in the process of making decisions about patients' care. Their significance is derived from the fact that nursing care is holistic in its approach – a necessary characteristic when it comes to assessing a patient's QoL (Rebollo et al. 2004; Tallis 2005). A nurse's care is also supportive in nature (Gray et al. 2002), particularly in oncology settings where physicians are very busy and may rely on nurses for information about the patients' QoL.

Whatever the circumstances or reasons, nurses need to make such assessments of patients' QoL to provide better supportive care for patients. In other words, if they have a better understanding of patients' QoL, it may contribute to better support for patients (Pickard & Knight 2005) otherwise the assessments might lead to negative consequences for the patients. It follows that a reasonable degree of agreement between patients' and nurses' perceptions of cancer patients' QoL may improve health outcomes. Given this, assessing the level of agreement between the patient and the nurse over the patient's QoL is considered important and worthy of investigation.

Previous research studies, which mainly used 'proxy rating', identified some general trends (McPherson & Addington-Hall 2003; Pickard et al. 2004a). For example, proxies appear to have a better understanding of objective aspects of QoL, like the physical QoL, rather than subjective aspects such as the psychological QoL. However, there are still some inconsistencies in the findings, particularly when the effect of patients' and proxies' characteristics on the level of agreement were assessed (Sneeuw, Sprangers & Aaronson 2002; Tang, ST & McCorkle 2002b). For example, in one research study, the degree of QoL agreement between patients and

their proxies was influenced by the patients' performance status (Horton 2002). Another study yielded an inconsistent pattern of such a relationship (SneeuwAaronson et al. 1997a).

A literature search yielded no Australian research that compared nurses' perceptions of patients' QoL with cancer patients' own perceptions of their QoL. Nor is it clear whether the findings of studies carried out in other countries hold true for Australia due to the culturally influenced nature of QoL (The WHO Group 1995; WHO Group 2000). Moreover, only a few research studies assessing the level of agreement between patients and proxies used more accurate statistical tests like Multivariate analysis. Therefore, the current investigators were interested to know how accurately nurses perceive and report patients' QoL in the clinical area of cancer patients in an Australian population.

AIMS OF THE STUDY

The primary aim of this study was to (a) measure the level of agreement between patients' and nurses' perceptions of the patients' QoL as the patients' proxies³. This study also sought to address the following secondary aims: (b) the reliability of patients' and nurses' QoL scores using the World Health Organisation Quality of Life Brief (WHOQoL-BREF) questionnaire; and (c) the influence of patients' and nurses' variables on the level of agreement between patients and nurses.

METHODS USED IN THE STUDY

Research Participants

A convenience sample of cancer patients and their nurses were recruited from one inpatient oncology ward, one outpatient chemotherapy unit and one radiotherapy centre at a large Australian teaching hospital. Any adult cancer patients undergoing

³From hereon, the term 'nurse' or its variations will be used to refer to the nurse as a proxy, except where specifically stated.

treatment were able to take part in the study provided that they could read the WHOQoL-BREF questionnaire and write their responses. Patients differed in their health status, disease severity and treatment modalities. This increased the variability in QoL ratings and allowed outcomes of agreement to be generalised to a wider group of patients. Nurses selected for the research were those who provided the primary care for a patient and were, therefore, familiar with them.

Instrument

The WHOQoL-BREF questionnaire was used for this research. This test uses 26 items which assess the QoL for four domains or dimensions, including physical (7 items), psychological (6 items), social relationship (3 items), and environmental (8 items) domains, and 2 items measuring overall quality of life and general health. All 26 items are rated on a 5-point Likert-type scale (The WHO Group 1998a).

The WHOQoL-BREF questionnaire was selected for this study as it was considered suitable by WHO to assess QoL in different situations such as routine clinical work, epidemiological studies and clinical research trials (WHO Group 1996). In this study, the Australian version of the questionnaire was completed in the framework of a clinical research study.

Two other separate forms, Patients' and Nurses' characteristics forms, were also created by the investigators. They were used as main instruments to collect demographic and clinical variables of patients and nurses.

Data Collection

The research study was completed during a six month period from July 2005 to December 2005 in a major public hospital in Adelaide, South Australia.

In the majority of cases, the principal researcher introduced the research study to each patient-nurse pair. If they agreed to take part in the study, demographic

and clinical variables of each patient-nurse pair such as age, gender, patient's cancer diagnosis, and nurse's clinical experience were recorded. The principal investigator gave the patient the WHOQoL-BREF questionnaire to complete and the nurse then filled out a WHOQoL-BREF separately, based on his/her understanding of the patient's QoL. These were collected by the researcher and placed in an envelope. There were a few cases when, due to practical issues, nurses gave questionnaires to the patients. However, the nurses were well informed beforehand about the research process and that completion of the questionnaires should be done on their own without asking the patient any questions.

Ethical Considerations

This research was approved by the appropriate Clinical Research Ethics Committees. The same number was recorded on a Patient's and Nurse's characteristic form as well as the WHOQoL-BREF questionnaire, to be completed by patients and nurses, so that the information from the patient's and nurse's forms were able to be properly matched and the data compiled, while participant anonymity was assured. Verbal and written information about the research project was provided for both patients and nurses. Agreement to complete the questionnaire was considered as consent for both patients and nurses and they were informed this was the case, as well as of their right to withdraw from the study at any time if they so desired. In order to deal with patients' possible emotional distress, supportive care in the form of counselling was negotiated with the Clinical Nurse Consultant. Nurses were not expected to experience any emotional stress by filling out a QoL questionnaire for patients.

DATA ANALYSIS AND FINDINGS

To calculate the sample size, the standard deviation of differences between patients and proxies from two previous research studies that used the WHOQoL-BREF

questionnaires with a sample of people with psychosis (Herrman, Hawthorne & Thomas 2002) and schizophrenia (Becchi et al. 2004) was calculated. This provided a standard deviation range from 1.00 to 1.50 units on the total WHOQoL instrument. Setting a significance level of 5% and power 80%, a sample of 117 was considered sensitive enough to detect a difference in means of 0.30-0.40 units.

The data were entered into SPSS 12 software and analysed using a syntax recommended by the World Health Organization (WHO) for use with this specific questionnaire (The Australian Centre for Posttraumatic Mental Health 2003). The patients' and nurses' QoL mean domain scores were calculated by multiplying the mean of all items included within the domain by four to be directly comparable with scores derived from WHOQOL-100, giving domain scores ranging from 4 to 20.

Patients' and nurses' clinical and demographic variables

Demographic variables of patients are shown in Table 1.

	Patients			
	No.	%		
Patient's gender				
Male	60	51.3		
Female	57	48.7		
Patient's age				
Mean	59.8			
Standard Deviation	15.3			
Patient 's first language				
English	102	91.1		
Other	10	8.90		
Patient's marital status				
Single	12	10.5		
Married	76	66.7		
Divorced/widowed	22	19.3		
Other	4	3.5		

 Table 1: Patients demographic variables

One-hundred and seventeen patients took part in this study with the mean age of 59.8 $(\pm 15.3 \text{ SD})$ years. The highest percentage of patients were married (66.7 %), male

(51.3%), used English as their first language (91.1%), and secondary school was their highest level of education (71.7%). Patients had a range of cancer diagnoses, with breast cancer (23.1%), colorectal cancer (12.0%), and non-Hodgkin lymphoma (10.2%) being the most prevalent. The highest percentage of patients completing the questionnaire were from outpatient departments (54.8%), with chemotherapy (68.2%) as their major treatment. The highest percentage of patients (42.5%) were classified as 'restricted but ambulatory' in their performance status.

Results also indicated that from the 49 nurses who took part in the study, the highest percentage of them were female (96.6%), married (61.3%), and had English as their first language (90.3%). Nurses had a range of qualifications with Registered Nurse certificate (30.8%), Graduate Diploma (26.5%) and Bachelor of Nursing (20.5%), being the most prevalent. The mean age of nurses was 37.4 (\pm 7.10 SD) with a range of 21-55 years. The mean time nurses spent providing care for a given patient (hour/shift) was 2.45 (\pm 1.50 SD) hours with a range of 0.08-6.00 hours. The mean of nurses' clinical experience was 15.5 years (\pm 8.90 SD) with a range of 0.16-30 years. The mean of nurses' clinical experience with cancer patients was 9.40 years (\pm 7.00 SD) with a range of 0-22 years.

In the nurses' Characteristics Form, nurses were asked to rate their own quality of life against one question (How would you rate your quality of life?). The aim was to assess the effect of nurses' QoL on the level of agreement. The highest percentage of nurses (53.5%) rated their own quality of life as 'very good'.

Reliability of patients' and nurses' QoL scores using the WHOQoL-BREF

To assess the reliability of the tool, internal consistency was measured using the Cronbach alpha coefficient for patients' and nurses' QoL mean domain scores. The ideal acceptable level was 0.70 (Pallant 2004). The results of Cronbach alpha are shown in Table 2.

QoL domain	Cronbach alpha			
	Patients	Nurses		
Physical (7 items)	0.84	0.87		
Psychological (6 items)	0.76	0.80		
Social Relationship (3 items)	0.35	0.60		
Environmental (8 items)	0.67	0.86		
Overall (26 items together)	0.90	0.93		

Table 2: Test of reliability of the WHOQoL-BREF

Agreement between patients and nurses on patients' QoL mean domain scores

Different tests were used to measure the level of agreement. Firstly, Pearson Correlation (r) and Intraclass Correlation Coefficient (ICC) were calculated. The criteria to judge the results of correlation coefficient and ICC were as follows: less than or equal to 0.20, poor; 0.21-0.40, fair; 0.41-0.60, moderate; 0.61-0.80, substantial; 0.81-1, almost perfect (Tang, ST & McCorkle 2002b; von Essen 2004).

The results of agreement between patients' and nurses' QoL domain scores using correlation statistical tests are reported in Table 3. The correlation (r) between QoL domain scores of patients and nurses ranged from 0.16 to 0.50. With the exception of the physical QoL domain, Intraclass Correlations were lower than the correlations ranging from 0.14 to 0.50.

QoL domain	Patient-Nurse Correlation			
	r	ICC		
Physical	0.50	0.50		
Psychological	0.26	0.24		
Social relationship	0.16	0.14		
Environmental	0.37	0.29		

Table 3: Patient-Nurse agreement on the WHOQoL-BREF using correlation tests

Secondly, a graphical approach described by Bland and Altman (1986) was conducted. In this test the difference between the patient and nurse QoL scores was plotted against the average of two scores for each patient-nurse pair. The main point is the acceptability of the overall width of the scatter (upper and lower levels of agreement). This is an interval in which 95% of patient-nurse differences would fall. Another point to be considered is whether there is a pattern in the scatter of points.

Figure 1 shows that for the physical QoL domain the location of the upper and lower limits of agreement of patients and nurses differs by around 6 points. For other QoL domains, the scatter plots (not included) were relatively similar to the physical QoL domain. Moreover, there is not an obvious pattern between the difference and the mean of physical QoL domain scores.



Figure 1: Bland-Altman plot between average of physical QoL mean domain scores of patients and nurses and difference between patients and nurses physical QoL mean domain scores

Thirdly, the QoL mean domain scores of nurses were compared to those of the patients using a paired t-test. The significance level was set at 0.05. The results of paired t-tests between physical, psychological, social relationship, and environmental domains of patients and nurses are reported in Table 4. Findings revealed that there were significant differences between social relationship domain scores of patients and nurses (t=2.33, p< 0.005) (d.f=114) and between environmental domain scores of patients' (t=4.78, p< 0.005) (d.f=116). In these two domains, patients'

QoL domain scores were significantly higher than that of nurses, suggesting that nurses rated their patients as having an inferior QoL to what the patients rated themselves.

QoL domain	Patients		Nurses		Patient differ	-Nurse rence
	Mean	SD	Mean	SD	Mean	SD
Physical	12.93	3.21	12.93	2.73	0.00	3.00
Psychological	14.58	2.63	14.03	2.35	0.55	3.05
Social relationship	15.64	2.73	14.92	2.26	0.72*	3.26
Environmental	15.85	1.86	14.84	2.17	1.01*	2.28

Table 4: Patient-Nurse agreement on the WHOQoL-BREF using t-test

*Statistically significant difference between patient and nurse score (P<0.05) using paired t-test

Influence of patients' and nurses' characteristics on the level of agreement between patients and nurses

At first, a Pearson correlation was conducted between patient-nurse absolute QoL mean domain differences (as the dependent variable) with patients and nurses' variables (as the independent variable). In line with a similar research study (Sneeuw et al. 1998) those with significant correlations (p value 0.05 or less) were then entered simultaneously into the equation and analysed using Standardized Multiple Regression analysis. Beta coefficients with p values of 0.05 or less were considered significant.

At the bivariate level using the Pearson correlation, several variables were identified to be significantly associated with the absolute difference between patients' and nurses' QoL domain scores. This is represented in Table 5. Larger differences between patient and nurse physical domain scores were noted when the nurses had less clinical experience with cancer patients, and for patients receiving non-chemotherapeutic treatments. In the psychological domain, greater differences between nurses and their patients were found to exist for inpatients, and patients who

were either in the level of 'ambulatory but unable to do work activities' or in the levels other than 'restricted but ambulatory' in their performance status. In the social relationship domain, greater differences were found for inpatients.

Finally, larger differences between patient and nurse environmental domain scores were noted for inpatients, non-married patients, younger nurses and nurses with less clinical experience. However, with Multivariate analysis, the nurses' length of clinical experience with cancer patients (β =-0.20), where the patients were receiving treatment (β =0.20), and the patients' marital status (β =-0.20) were the only statistically significant predictors of differences between patients and nurses scores obtained for the physical, social relationship and environmental QoL domain scores, respectively.

DISCUSSION

The major aims of this study were to identify the level of agreement between patients and nurses on patients' QoL and to identify those variables that could affect the degree of agreement between patients' and nurses' QoL scores.

First, the WHOQoL-BREF questionnaire was identified as a reliable tool for assessing cancer patients' QoL. Cronbach alpha of both patients' and nurses' QoL domain scores (except the social relationship domain) approached or was above the acceptable level of 0.70. In general, a high internal consistency may not be necessary for a multi-domain tool (Cella & Tulsky 1990) like WHOQoL-BREF questionnaire. The low internal consistency of the social relationship domain might be related to the low number of items (only three) and the content of the items (sex life, personal relationships, and friends) which was reported in another research study (Hanestad et al. 2004).

In general, results r and ICC support that there is no substantial agreement between each patient and nurse about a patient's QoL. This can be confirmed by checking the Bland-Altman scatter plots which shows around 6 points of difference (out of 20) between a patient's QoL score and a nurse's score.

Patients' and nurses' characteristics	Absolute difference							
	Phy	sical	Psychological		So relati	cial onship	Enviro	nmental
	r	ß	r	ß	r	ß	r	ß
Patient's gender Male (R) Female	0.02		0.10		0.02		-0.01	
Patient's age	0.02		-0.01		0.03			-0.01
Patient 's first language English(R) Other	0.13		-0.03		0.10			0.02
Patient's marital status• Single Married Divorced/Widowed Other (R)							-0.23*	-0.20*
Patient's current treatment• Chemotherapy Radiotherapy Surgery Other therapy More than one treatment (R)	-0.19*	-0.20						
Patient's treatment setting Outpatients (R) Inpatients	0.15		0.20*	0.14	0.20*	0.20*	0.27*	0.10
Patient performance status• Fully active Restricted but ambulatory Ambulatory but unable to do work activities Limited capability or disabled (R)			-0.20* 0.24*	-0.10 0.20				
Nurse's gender Male (R) Female	0.08		-0.05		0.03		-0.08	
Nurse's age	-0.15		-0.09		-0.09		-0.25*	-0.10
Nurse's language English(R) Other	-0.12		0.04		0.07		-0.09	
Time nurse spends with patients/shift	0.01		0.11		-0.02		0.05	
Nurse clinical experience	-0.18		-0.11		-0.06		-0.27*	-0.14
Nurse clinical experience with cancer patients	-0.23*	-0.20*	-0.18		-0.09		-0.13	

Table 5: Bivariate(r) and Multivariate (B) correlation coefficients between patients and nurses characteristics and absolute difference between patients' and nurses' QoL mean domain

•As these categorical variables had more than two categories, they were entered into the equation after creating dummy coding variables. To simplify the table, patients' and nurses' characteristics where none of their dummy coding variables showed any significant relationship with difference between patients and nurses QoL mean domain scores at the bivariate level were deleted (patients' highest level of education, patients' current cancer diagnosis, nurses' marital status, nurses' educational level, nurses' QoL)

*significant; R: Reference group; r: Bivariate correlation; ß: Beta coefficient

As other research studies have also reported (McPherson & Addington-Hall 2003; von Essen 2004) nurses had better agreement with their patients in QoL domains that consisted of more objective items such as the physical domain. In contrast, in the social relationship and psychological domains, which consisted of more private and subjective items such as questions about the patient's 'sex life' or 'friends', nurses were less successful in correctly rating patients' QoL. This may also show that the QoL is very subjective and individual in nature and might change from time to time according to patients' expectations, hopes and desires. It may not be achievable for nurses to accurately assess the patients' level of QoL because it might change frequently. However, other reasons such as how the QoL of patients may be important to them and their level of skill and expertise in relation to the assessment of patients' QoL may also be considered.

Moreover, nurses have different professional responsibilities as well as time limitations in providing nursing care. These time limits might prevent nurses from becoming adequately familiar with some aspects of patients' QoL like psychological and social relationships and environmental ones (Frost, Brueggen & Mangan 1997; Mohan et al. 2005).

A comparison of QoL domain scores of patients with nurses also revealed two important points. Firstly, except for the physical and psychological domains, nurses significantly underestimated patients' QoL. A reason might be that nurses do not have a clear understanding of these aspects of patients' QoL like social relationship and environmental domains and assumed the worst situation for the patients in these areas. This has been reported in many other research articles in which proxies reported lower levels of QoL than patients themselves did (Pierre et al. 1998; Sprangers, M.A.G. & Aaronson 1992). Nurses are usually around the patients

during their cancer journey. Patients may try to adapt themselves to the cancer, and underestimations in QoL by nurses might postpone this process, thus their role may not be as supportive as it may be otherwise.

The second point is that, except for the environmental QoL domain in which a large effect size between patients and nurses was identified, the QoL mean domains of nurses were the same or very close to those of patients (particularly the physical and psychological domains). Therefore, in line with the suggestion made by (Sneeuw et al. 1999), for those researchers who are interested in preventing missing data in QoL clinical trials comparing groups rather than individuals, means of QoL rating from nurses may provide a useful source of information.

No obvious pattern was identified from the scatter plot that may have implied that agreement between patients and nurses was better for a patient group with a specific level of QoL. This result differed from a study by Sneeuw et al. (1998) who reported more scatter (poorer agreement) at moderate levels of patient' QoL, with less scatter at either extreme. In this latter study outcomes indicated that proxies had a better understanding of QoL in those patients with either higher or lower rating of their QoL compared with those at the middle.

Several variables that may have influenced the level of patient-nurse agreement were also explored using Multivariate analysis. There was greater agreement between nurses with more clinical experience with cancer patients and patients in the physical QoL domain. Why a nurse who has more experience with cancer patients has a better understanding of a patients' QoL needs to be explored more. However, these nurses may have developed better assessment and communication skills in order to provide closer ratings on patients' QoL. In the social relationship domain, there was more agreement between patients and nurses for patients treated in outpatient departments compared with inpatients. Despite the fact that a close and prolonged relationship with the patient (for example in inpatient departments) might increase the agreement (Magaziner et al. 1988) it could also lead to disagreement when a nurse was under the pressure of care giving (von Essen 2004). Outpatients might be coming to the clinic on a regular basis and this might allow nurses to get to know them quite well due to the continuity of care. In general, outpatients also had a better health status than inpatients and, as noted earlier, the level of agreement between patients and nurses might be better either when patients were very sick or very healthy (Sneeuw et al. 1998; Sneeuw et al. 1999). There may, of course, be other variables that are unique to the outpatient department that influenced the agreement.

Implication of findings for nursing practice and research

The main implication of the findings of this study is that nurses may need to enhance their understanding of patients' QoL. Nurses have an important supportive role toward cancer patients which includes physio-psycho-social concerns and needs. For achieving holistic health support, nurses need to understand a patient's QoL to assist them to make appropriate decisions in relation to patient care. For example, in decision making about what supportive services may suit a patient's needs, the following strategies may help nurses.

Firstly, if nurses are aware that their perceptions of QoL may differ to patients, then providing tools such as the WHOQoL-BREF questionnaire in the clinical area, may allow nurses to fully assess all aspects of patients' QoL instead of relying on their perceptions about QoL. This may lead to a better understanding of patients' QoL and a better standard of care.

Secondly, nurses may need to consider other sources of information about patients' QoL such as family, particularly their spouse, friends, and other health care professionals like doctors and social workers. Such sources may provide complementary information or may be useful when clients are not able to communicate appropriately with nurses. Of course, nurses need to consider the congruency between all sources of information.

Finally, the results that more experienced nurses have higher rating agreement with patients' QoL rating may be due to better communication and assessment skills. Therefore, to facilitate quality nursing care, training that allows nurses to develop a better understanding of the QoL concept, its importance to patients' QoL and how these tools may be used to facilitate systematic assessment of patients' QoL, may be useful. Alternatively, nurses may be able to help family care givers use the QoL tools to maximise standard of care in this area as well.

Study Limitations

The findings of this research study must be interpreted within its limitations. Firstly, an attempt was made to include as many nurses as possible in the research study (Broberger, Tishelman & von Essen 2005)however, like other research studies (Molzahn, Northcott & Dossetor 1997; Sneeuw et al. 1998) the number of nurses included was less than half of patients (42%). Nurses were well informed about the importance of the research and the need to complete the questionnaires as accurately as possible. However, the study was conducted in a very busy oncology environment and on some occasions this may have affected the nurses' concentration while completing the questionnaires.

Secondly, while some important variables of patients and nurses that may influence agreement were identified as significant, there may be other variables influencing this that were not included in this study.

Future research

This study needs to be conducted further with larger samples, particularly with a more representative sample of nurses. How nurses may have a better understanding of cancer patients' QoL, and how such perceptions may affect patients' QoL in either positive or negative ways, need to be explored further. These questions are presently being investigated in an interpretive study with oncology patients and nurses. Why nurses sometimes underestimate patients' QoL, why those nurses who work in outpatients cancer clinics and those who have more clinical experience with cancer patients may have a better understanding of patients' QoL may also be further investigated.

CONCLUSION

This study shows that differences exist between nurses' and patients' perceptions of cancer patients' QoL which may impact on the quality of patient care and support. To address this, strategies need to be devised that increase nurses' understanding of patients' QoL, particularly in the most neglected areas of psychological, social relationship, and environmental QoL domains. It is also important that nurses learn how to assess cancer patients' QoL thoroughly, possibly by using standardised QoL tools instead of relying heavily on their own intuitive perceptions which may lead to misunderstandings and misjudgements of a patients' QoL. Ultimately, increasing the level of agreement between nurses and their patients about patients' QoL makes it more likely that those experiencing a life-threatening illness will receive the care they need to maintain or improve their quality of life.

Acknowledgment

Thanks go to Professor John Keeves and Ms Kylie Lange from Flinders University

for their comments and advice about the statistical issues used in this research study.

The authors are also grateful to the nursing staff and cancer patients for their

cooperation in this study.

References

- Ballatori E (2001) Unsolved problems in evaluating the quality of life of cancer patients, *Annals of Oncology* 12: 11-13.
- Becchi A, Rucci P, Placentino A, Neri G and Girolamo G (2004) Quality of life in patients with schizophrenia: comparison of self-report and proxy assessments, *Social Psychiatry and Psychiatric Epidemiology* 39(5): 397-401.
- Bland J and Altman D (1986) Statistical methods for assessing agreement between two methods of clinical measurement, *Lancet* 1(8476): 307-310.
- Cella, DF and Tulsky DS (1990) Measuring quality of life today: methodological aspects, *Oncology (Huntington)* 4(5): 29-38; discussion 69.
- Fayers P and Machin D (2000) *Quality of Life: Assessment, Analysis and Interpretation*, John Wily & Sons Ltd, Chichester.
- Fisch MJ, Titzer ML, Kristeller JL, Shen J, Loehrer PJ, Jung S, Passik SD and Einhorn LH (2003) Assessment of quality of life in outpatients with advanced cancer: the accuracy of clinician estimations and the relevance of spiritual well-being- a Hoosier Oncology Group study, *Journal of Clinical Oncology* 21(14): 2754-2759.
- Frost MH, Brueggen C and Mangan M (1997) Intervening with the psychosocial needs of patients and families: perceived importance and skill level, *Cancer Nursing* 20(5): 350-358.
- Giesler RB (2000) Assessing the quality of life of patients with cancer, *Current Problems in Cancer* 24(2): 55-92.
- Gray R, Goel V, Fitch M, Franssen E and Labrecque M (2002) Supportive care provided by physicians and nurses to women with breast cancer, *Supportive Care in Cancer* 10(8): 647-652.
- Hanestad BR, Rustoen T, Knudsen O, Lerdal Jr A and Wahl AK (2004)
 Psychometric properties of the WHOQOL-BREF questionnaire for the
 Norwegian general population, *Journal of Nursing Measurement* 12(2): 147-159.

- He GMD and Liu SMSNRN (2005) Quality of life and coping styles in Chinese nasopharyngeal cancer patients after hospitalization, *Cancer Nursing* 28(3): 179-186.
- Herrman H, Hawthorne G and Thomas R (2002) Quality of life assessment in people living with psychosis, *Social Psychiatry and Psychiatric Epidemiology* 37(11): 510-518.
- Horton R (2002) Differences in assessment of symptoms and quality of life between patients with advanced cancer and their specialist palliative care nurses in a home care setting, *Palliative Medicine* 16(6): 488-494.
- King CR, Hinds P, Dow KH, Schum L and Lee C (2002) The nurse's relationshipbased perceptions of patient quality of life, *Oncology Nursing Forum* 29(10): E118-126.
- Landis JR and Koch GG (1977) The measurement of observer agreement for categorical data, *Biometrics* 33(1): 159-174.
- Magaziner J, Simonsick EM, Kashner TM and Hebel JR (1988) Patient-proxy response comparability on measures of patient health and functional status, *Journal of Clinical Epidemiology* 41(11): 1065-1074.
- McPherson CJ and Addington-Hall JM (2003) Judging the quality of care at the end of life: can proxies provide reliable information?, *Social Science & Medicine* 56(1): 95-109.
- Mohan S, Wilkes LM, Ogunsiji O and Walker A (2005) Caring for patients with cancer in non-specialist wards: the nurse experience, *European Journal of Cancer Care* 14(3): 256-263.
- Molzahn AE, Northcott HC and Dossetor JB (1997) Quality of life of individuals with end stage renal disease: perceptions of patients, nurses, and physicians, *American Nephrology Nurses' Association (ANNA) Journal* 24(3): 325-333; discussion 34-5.
- Pallant JF (2004) SPSS survival manual: a step by step guide to data analysis using SPSS, 2nd edn, Allen & Unwin, Crows Nest, NSW.
- Pickard AS and Knight SJ (2005) Proxy evaluation of health-related quality of life a conceptual framework for understanding multiple proxy perspectives, *Medical Care* 43(5): 493-499.
- Pierre U, Wood-Dauphinee S, Korner-Bitensky N, Gayton D and Hanley J (1998) Proxy use of the Canadian SF-36 in rating health status of the disabled elderly, *Journal of Clinical Epidemiology* 51(11): 983-990.
- Slevin ML, Plant H, Lynch D, Drinkwater J and Gregory WM (1988) Who should measure quality of life, the doctor or the patient? *British Journal of Cancer* 57(1): 109-112.

- Sneeuw KC, Aaronson NK, Sprangers MA, Detmar SB, Wever LD and Schornagel JH (1998) Comparison of patient and proxy EORTC QLQ-C30 ratings in assessing the quality of life of cancer patients, *Journal of Clinical Epidemiology* 51(7): 617-631.
- Sneeuw KC, Aaronson NK, Sprangers MAG, Detmar SB, Wever LDV and Schornagel JH (1999) Evaluating the quality of life of cancer patients: assessments by patients, significant others, physicians and nurses, *British Journal of Cancer* 81(1): 87-94.
- Sneeuw KC, Sprangers MA and Aaronson NK (2002) The role of health care providers and significant others in evaluating the quality of life of patients with chronic disease, *Journal of Clinical Epidemiology* 55(11): 1130-1143.
- Sneeuw KC, Aaronson NK, Sprangers MA, Detmar SB, Wever LD and Schornagel JH (1997) Value of caregiver ratings in evaluating the quality of life of patients with cancer, *Journal of Clinical Oncology* 15(3): 1206-1217.
- Sprangers MA and Aaronson NK (1992) The role of health care providers and significant others in evaluating the quality of life of patients with chronic disease: a review, *Journal of Clinical Epidemiology* 45(7): 743-760.
- Tang ST and McCorkle R (2002) Use of family proxies in quality of life research for cancer patients at the end of life: A literature review, *Cancer Investigation* 20(7-8): 1086-1104.
- The Australian Centre for Posttraumatic Mental Health (2003) World Health Organization Quality of Life (WHOQoL) Project, accessed at <u>http://www.acpmh.unimelb.edu.au/whoqol/default.html</u> on 21 February 2006
- von Essen L (2004) Proxy ratings of patient quality of life Factors related to patient-proxy agreement, *Acta Oncologica* 43(3): 229-234.
- World Health Organization Group (1995) The World Health Organization quality of life assessment (WHOQOL): Position paper from the World Health Organization, *Social Science & Medicine* 41(10): 1403-1409.
- World Health Organization Group (1996) WHOQOL-BREF: Introduction, administration, scoring and generic version of the assessment, accessed at <u>http://www.who.int/mental_health/media/en/76.pdf</u> on 25 February 2005.
- World Health Organization Group (1997) WHOQoL: Measuring Quality of Life, accessed at <u>http://www.who.int/mental_health/media/en/68.pdf</u> on 13 June 2007.

World Health Organization Group (1998) Development of the World Health Organization WHOQOL-BREF Quality of Life Assessment, *Psychological Medicine* 28(3): 551-558

Appendix B: The detailed process of one strategy for data collection in the first phase

1. Introduction

Dear Nursing staff member,

Thank you for your cooperation and also for your participation in this research study which is called: **'A comparison of patients' and nurses' perceptions of cancer patients' Quality of Life (QoL)'**

Any adult cancer patient (above 18 years old) with diagnosis of any kind of cancer (haematology or oncology) inpatient or outpatient and under any cancer treatment is welcome to participate in the research study provided that she (he) can read and circle the items in the questionnaire.

Any nurse is eligible to take part in the study if he (she) is familiar with the patient and is happy to complete the forms and the questionnaire as mentioned below. Your voluntary completion of the forms and questionnaire will be considered as **consent**.

2. Method: what you need to do

Please do the following steps:

A) If your patient has met the above inclusion criteria, would you please give him (her) **'The patient information sheet'** which introduces the research study and ask if she (he) is prepared to fill out a questionnaire.

B) If the patient **verbally** confirms her (his) consent to participate with this survey, would you please complete '**The patient clinical and demographic characteristic form**' and pass this form '**The WHOQoL-BREF questionnaire**' to the patient from the set that has the **same number**.

C) Please ensure that the **questionnaire is completed by the patient on their own without any help** and returned to you together with **'The participant information sheet'** on that day.

D) Please ask the patient to fold and staple the questionnaire before giving it back to you.

E) Please complete '**The nurse characteristic form**' and the nurse survey '**The WHOQoL-BREF questionnaire**' yourself from the set with the **same number** for the patient based on your perception about patient's quality of life. Please do this **without** seeking the patient's opinion.

F) You can complete the questionnaire immediately after the patient has done so or at any time on that day which is convenient for you.

G) Please return the completed set (The patient information sheet read by patient + The patient clinical and demographic characteristic form completed by you + The WHOQoL-BREF questionnaire completed by patient + The nurse characteristic form completed by you + The WOQoL-BREF questionnaire completed by you for a patient) preferably as a stapled set to the principal researcher. Alternatively, you can deliver it to the Clinical Nurse Consultant in your Unit.

If you have queries about this survey, you can always contact me on (08) 8201 2071.

Yours sincerely, Masoud Bahrami, Principal researcher

Appendix C: Participant Information Sheet (Patient specific for the second phase)



'A comparison of patients' and nurses' perceptions of cancer patients' Quality of Life (QoL)'

As a student at Flinders University, enrolled in a PhD degree, I am seeking your assistance to take part in a study to compare and contrast the evaluation of Quality of Life (QoL) of cancer patients and nurses and the implications of the findings for nursing clinical practice.

If you choose to participate, you will be asked to participate in an interview about Quality of Life and this interview will be no longer than 60 minutes (which can also be done in two parts). This interview will be at a time and place of your convenience. These are the kinds of questions I will be asking: What does the term QoL mean to you? How do you usually think about your own QoL? What things do you think affect the way you think about your own quality of life?

Your participation in the study is entirely voluntary and you have the right to withdraw at any time. If you decide not to participate in this study or if you withdraw, you may do this freely without prejudice to any treatment at

All records containing personal information will remain confidential and no information that could lead to your identification will be released.

If you, as a participant of this research, suffer injury, compensation may, at the discretion of Flinders University, be paid without litigation. However, compensation is not automatic and you may have to take legal action in order to receive payment.

Study results may be published in scientific journals at a later date. It is possible that the results may not be published because of commercial, scientific or other reasons. Also, the research results can be delivered to responsible persons at theto be used as a source for better care and support of cancer patients.

Should you require further details about the project, before, during or after the study, you may contact me on (08) 201 2071. Any further queries concerning the project can also be directed to my supervisor, Dr Steve Parker on (08) 201 3402.

This study has been reviewed by theClinical Research Ethics Committee. Should you wish to discuss the project with someone not directly involved, in particular in relation to matters concerning policies, your rights as a participant, or should you wish to make a confidential complaint, you may contact the Administrative Officer - Research,

Yours sincerely, Masoud Bahrami

Appendix D: Participant Information Sheet (Nurse specific for the second phase)



'A comparison of patients' and nurses' perceptions of cancer patients' Quality of Life (QoL)'

As a student at Flinders University, enrolled in a PhD degree, I am seeking your assistance to take part in a study to compare and contrast the evaluation of Quality of Life (QoL) of cancer patients and nurses and the implications of the findings for nursing clinical practice. Your participation in the study is entirely voluntary and you have the right to withdraw at any time.

If you choose to participate, you will be asked to participate in an interview about Quality of Life and this interview will be no longer than 60 minutes (which can also be done in two parts). These are the kinds of questions I will be asking: What does the term QoL mean for you? How do you usually evaluate patients' QoL? What factors do you believe may affect your perception in evaluating a patient's QoL? How do you incorporate evaluations of a patient's quality of life in your practice?

This interview will be at a time and place of your convenience. You may also be asked to participate in a focus group (no longer than one hour duration) to discuss finding ways to improve the Quality of Life for patients with cancer and the results of this study.

All records containing personal information will remain confidential and no information that could lead to your identification will be released.

If you, as a participant of this research, suffer injury, compensation may, at the discretion of Flinders University, be paid without litigation. However, compensation is not automatic and you may have to take legal action in order to receive payment.

Study results may be published in scientific journals at a later date. It is possible that the results may not be published because of commercial, scientific or other reasons. Also, the research results can be delivered to responsible persons at theto be used as a source for better care and support of cancer patients.

Should you require further details about the project, before, during or after the study, you may contact me on (08) 201 2071. Any further queries concerning the project can also be directed to my supervisor Dr Steve Parker on (08) 201 3402.

This study has been reviewed by theClinical Research Ethics Committee. Should you wish to discuss the project with someone not directly involved, in particular in relation to matters concerning policies, your rights as a participant, or should you wish to make a confidential complaint, you may contact the Administrative Officer - Research,....

Yours sincerely, Masoud Bahrami

Appendix E: The patient clinical and demographic characteristics form

Number of Form

1. Patient's gender							
2. Patient's ageyears							
3. What is patient's first language? (Please describe)							
4. Patient's marital status Single Married Divorced/Widowed Other							
5. Patient's highest level of education \Box Primary school \Box Secondary school							
Tertiary school							
6 Patient's current cancer diagnosis (Please describe in full words)							
7. Patient's most common current treatment (tick more than one if applicable)							
Chemotherapy Radiotherapy Surgery Immunotherapy Other							
8. Treatment Setting Outpatient Inpatient							
9. Patient performance status (Circle the appropriate Grade for the patient)							
Grade Status							
0 Fully active, able to carry on all pre-disease performance without restriction							
1 Restricted in physically strenuous activity but ambulatory and able to carry out work of a light or sedentary nature, e.g., light house work, office work							
2 Ambulatory and capable of all self-care but unable to carry out any work activities. Up and about more than 50% of waking hours							
3 Capable of only limited self-care, confined to bed or chair more than 50% of waking hours							
4 Completely disabled. Cannot carry on any self-care. Totally confined to bed or chair							

Appendix F: The WHOQoL-BREF questionnaire for patients

Number of Questionnaire

Instructions

This assessment asks how you feel about your quality of life, health, & other areas of your life. **Please answer all the questions**. If you are unsure about which response to give to a question, **please choose the one** that appears most appropriate. This can often be your first response.

Please keep in mind your standards, hopes, pleasures and concerns. We ask that you think about your life in the **last two weeks**.

	Not at all	Slightly	Moderately	Very	Complete		
Do you get the kind of support from others that you need?	1	2	3	4	5		
You would circle the number 4 if in the last two weeks you got a great deal of support from others							
	Not at all	Slightly	Moderately	very	Complete		
		\bigcirc	2	1	5		

Thank you for your help.

Now turn to the back of this page

Please read each question and assess your feelings, for the last two weeks, and circle the number on the scale for each question that gives the best answer for you.

	Very poor	Poor	Neither Poor nor Good	Good	Very Good
1. How would you rate your quality of life?	1	2	3	4	5

	Very Dissatisfied	Fairly Dissatisfied	Neither Satisfied nor Dissatisfied	Satisfied	Very Satisfied
2. How satisfied are you with your health?	1	2	3	4	5

The following questions ask about how much you have experienced certain things in the **last two weeks.**

	Not at all	A Small amount	A Moderate amount	A great deal	An Extreme amount
3. To what extent do you feel that physical pain prevents you from doing what you need to do?	1	2	3	4	5
4. How much do you need any medical treatment to function in your daily life?	1	2	3	4	5
5. How much do you enjoy life?	1	2	3	4	5
6. To what extent do you feel your life to be meaningful?	1	2	3	4	5

	Not at all	Slightly	Moderately	Very	Extremely
7. How well are you able to concentrate?	1	2	3	4	5
8. How safe do you feel in your daily life?	1	2	3	4	5
9. How healthy is your physical environment?	1	2	3	4	5

	Not at all	Slightly	Somewhat	To a great extent	Completely
10. Do you have enough energy for every day life?	1	2	3	4	5
11. Are you able to accept your bodily appearance?	1	2	3	4	5
12. Have you enough money to meet your needs?	1	2	3	4	5
13. How available to you is the information you need in your daily life?	1	2	3	4	5
14. To what extent do you have the opportunity for leisure activities?	1	2	3	4	5

	Not at all	Slightly	Moderately	Very	Extremely
15. How well are you able to get around	1	2	3	4	5
physically?					

The following questions ask you to say how good or satisfied you have felt about various aspects of your life over **the last two weeks.**

	Very Dissatisfied	Fairly Dissatisfied	Neither Satisfied nor Dissatisfied	Satisfied	Very Satisfied
16. How satisfied are you with your sleep?	1	2	3	4	5
17. How satisfied are you with your ability to perform your daily living activities?	1	2	3	4	5
18. How satisfied are you with your capacity for work?	1	2	3	4	5
19. How satisfied are you with yourself?	1	2	3	4	5
20. How satisfied are you with your personal relationships?	1	2	3	4	5
21. How satisfied are you with your sex life?	1	2	3	4	5
22. How satisfied are you with the support you get from your friends?	1	2	3	4	5
23. How satisfied are you with the conditions of your living place?	1	2	3	4	5
24. How satisfied are you with your access to health services?	1	2	3	4	5
25. How satisfied are you with your transport?	1	2	3	4	5

	Never	Infrequently	Sometimes	Frequently	Always
26. How often do you have negative feelings such as blue mood, despair, anxiety, depression?	1	2	3	4	5

THE	END
Appendix G: The nurse characteristics form

1. Nurse's gender 🗌 Male 🗌 Female
2. Nurse's ageyears
3. What is your first language? (Please describe)
4. Nurse's marital status Single Married Divorced/Widowed Other
5. Nurse's highest level of qualification \Box RN Certificate \Box Dip.N \Box B.N
Master PhD Other (Please describe)
6. Approximate how much time you have spent providing care for this patient per
shift?hour/s
7. How much clinical experience have you had?years
8. How much experience have you had with nursing patients with cancer?years
9. Please answer the following question in which you as a nurse ask to assess your
own quality of life (Please circle the appropriate number about your life for the
last two weeks)

Number of Form

	Neither Poor						
	Very poor	Poor	nor Good	Good	Very Good		
How would you rate your quality of life?	1	2	3	4	5		

Appendix H: The WHOQoL-BREF questionnaire to be completed by nurses for a patient

Number of Questionnaire				
-------------------------	--	--	--	--

Instructions

This assessment asks how you feel about your patient's quality of life, health, & other areas of their life. We are asking you to complete this survey specifically about your patient and the quality of life (QoL) you believe this patient has. **Please answer all the questions**. If you are unsure about which response to give to a question, please choose the one that appears most appropriate. This can often be your first response.

Please keep in mind the patient's standards, hopes, pleasures and concerns. We ask that you think about the patient's life in the **last two weeks.**

	Not at all	Slightly	Moderately	Very	Completel
Does the patient get the kind of support from others that he or she needs?	1	2	3	4	5
You would circle the num patient received a great de	ber 4 if yo al of suppo	ou think i ort from o	n the last tw	vo weeł	ts the
You would circle the num patient received a great dea	ber 4 if yo al of suppo	ou think i ort from o	n the last tw thers	vo weel	cs the
You would circle the num patient received a great dea	ber 4 if yo al of suppo Not at all	ou think i ort from o Slightly	n the last tw thers Moderately	vo weeł Very	Completely

Thank you for your help.

Now turn to the back of this page Please read each question and assess your feelings, for the last two weeks, and circle the number on the scale for each question that provides the best answer.

	Very poor	Poor	Neither Poor nor Good	Good	Very Good
1. How would you rate the patient's quality of life?	1	2	3	4	5

	Very Dissatisfied	Fairly Dissatisfied	Neither Satisfied nor Dissatisfied	Satisfied	Very Satisfied
2. How satisfied is the patient with his(her) health?	1	2	3	4	5

The following questions ask about how much the patient has experienced certain things in the **last two weeks**.

	Not at all	A Small amount	A Moderate amount	A great deal	An Extreme amount
3. To what extent do you feel that physical pain prevents the patient from doing what she(he) needs to do?	1	2	3	4	5
4. How much does the patient need any medical treatment to function in his or her daily life?	1	2	3	4	5
5. How much does the patient enjoy his or her life?	1	2	3	4	5
6. To what extent do you feel the patient's life to be meaningful?	1	2	3	4	5

	Not at all	Slightly	Moderately	Very	Extremely
7. How well is the patient able to concentrate?	1	2	3	4	5
8. How safe does the patient feel in his or her daily life?	1	2	3	4	5
9. How healthy is the patient's physical environment?	1	2	3	4	5

	Not at all	Slightly	Somewhat	To a great exte nt	Completely
10. Does the patient have enough energy for every day life?	1	2	3	4	5
11. Is the patient able to accept his or her bodily appearance?	1	2	3	4	5
12. Has the patient enough money to meet his or her needs?	1	2	3	4	5

13. How available is information that is needed for patient's daily life?	the	1	2	3	4	5
14. To what extent does the patient have the opportu for leisure activities?	inity	1	2	3	4	5
	Not at all	Slightly	Modera	tely	Very	Extremely
15. How well is the patient able to get around physically?	1	2	3		4	5

The following questions ask you to say how good or satisfied the patient has felt about various aspects of his or her life over the **last two weeks**.

	Very Dissatisfied	Fairly Dissatisfied	Neither Satisfied nor Dissatisfied	Satisfied	Very Satisfied
16. How satisfied is the patient with his or her sleep?	1	2	3	4	5
17. How satisfied is the patient with his or her ability to perform his or her daily living activities?	1	2	3	4	5
18. How satisfied is the patient with his or her capacity for work?	1	2	3	4	5
19. How satisfied is the patient with himself or herself?	1	2	3	4	5
20. How satisfied is the patient with his or her personal relationships?	1	2	3	4	5
21. How satisfied is the patient with his or her sex life?	1	2	3	4	5
22. How satisfied is the patient with the support he or she gets from his or her friends?	1	2	3	4	5
23. How satisfied is the patient with the conditions of his or her living place?	1	2	3	4	5
24. How satisfied is the patient with his or her access to health services?	1	2	3	4	5
25. How satisfied is the patient with his or her transport?	1	2	3	4	5
	Never	Infrequently	Sometimes	Frequently	Always
26. How often does the patient have negative feelings	1	2	3	4	5

such as blue mood, despair, anxiety, depression?

sion? 1 2 5 THE END

Appendix I: Labelling the WHOQoL-BREF items and calculating the mean domains for patients and nurses as proxies

- W1 "Rate QoL"
- W2 "Health satisfaction"
- W3 "Pain prevents doing"
- W4 "Need medical treatment"
- W5 "Enjoy life"
- W6 "Life is meaningful"
- W7 "Concentrate"
- W8 "Safe in your daily life"
- W9 "Healthy physical environment"
- W10 "Energy for everyday life"
- W11 "Accept bodily appearance"
- W12 "Money to meet needs"
- W13 "Information available"
- W14 "Opportunity for leisure activities"
- W15 "Able to get around"
- W16 "Sleep"
- W17 "Daily living activities"
- W18 "Capacity for work"
- W19 "Satisfied with yourself"
- W20 "Personal relationships"
- W21 "Sex life"
- W22 "Friends"
- W23 "Conditions of your living place"
- W24 "Access to health services"
- W25 "Transportation"
- W26 "Negative feelings".

Physical domain consists of items: (W3, W4, W10, W15, W16, W17, and W18). Psychological domain consists of items: (W5, W6, W7, W11, W19, W26). Social domain consists of items: (W20, W21, W22). Environment domain consists of items: (W8, W9, W12, W13, W14, W23, W24, and W25).

MD1 = 4*(Mean (W3r, W4r, W10, W15, W16, W17, W18)). MD2 = 4*(Mean (W5, W6, W7, W11, W19, W26r)). MD3 = 4*(Mean (W20, W21, W22)). MD4 = 4*(Mean (W8, W9, W12, W13, W14, W23, W24, W25)). MD=Mean Domain r=Reversed nMD1 = 4*(Mean (nW3r, nW4r, nW10, nW15, nW16, nW17, nW18)). nMD2 = 4*(Mean (nW5, nW6, nW7, nW11, nW19, nW26r)). nMD3 = 4*(Mean (nW20, nW21, nW22)). nMD4 = 4*(Mean (nW8, nW9, nW12, nW13, nW14, nW23, nW24, nW25)).

N=Nurse r=Reversed

Appendix J: Outcomes of SPSS for reliability testing (patients and nurses)

Reliability (Patients)

Reliability Statistics

Cronbach's Alpha	N of Items
.885	26

Item Statistics

	Mean	Std. Deviation	N
"Rate QoL"	3.6645	.96212	155
"Health satisfaction"	2.8774	1.19161	155
"Enjoy life"	3.6710	.97441	155
"Life is meaningful"	3.7032	.98807	155
"Concentrate"	3.5355	.83181	155
"Safe in your daily life"	3.6839	.86606	155
"Healthy physical environment"	3.7097	.82153	155
"Energy for daily life"	3.0000	1.09900	155
"Accept bodily appearance"	3.7484	1.09065	155
"Money to meet needs"	3.6194	1.09468	155
"Information available"	3.8387	.82560	155
"Opportunity for leisure activity"	2.9226	1.14832	155
"Able to get around"	3.3871	1.02809	155
"Sleep"	3.1419	1.22442	155
"Daily living activities"	3.0323	1.18661	155
"Capacity for work"	2.5806	1.22671	155
"Satisfied with yourself"	3.4452	1.11741	155
"Personal relationship"	4.1806	.92210	155
"Sex life"	3.1097	1.34632	155
"Friends"	4.3355	.82397	155
"Conditions of living place"	4.4516	.78288	155
"Access to health services"	4.4645	.74970	155
"Transportation"	4.3419	.79314	155
"Pain prevents doing"	3.4000	1.23583	155
"Need medical treatment"	3.0903	1.08931	155
"Negative feelings"	3.4194	.96612	155

		Scale	Corrected	Cronbach's
	Scale Mean if	Variance if	Item-Total	Alpha if Item
"Data Oal "	Item Deleted	Item Deleted	Correlation	Deleted
"Rate QoL"	88.6903	1/0.436	.541	.879
"Health satisfaction"	89.4774	165.186	.599	.877
"Enjoy life"	88.6839	170.620	.525	.880
"Life is meaningful"	88.6516	171.865	.467	.881
"Concentrate"	88.8194	171.954	.564	.879
"Safe in your daily life"	88.6710	170.352	.612	.878
"Healthy physical environment"	88.6452	174.776	.437	.882
"Energy for daily life"	89.3548	163.932	.704	.875
"Accept bodily appearance"	88.6065	171.149	.442	.882
"Money to meet needs"	88.7355	177.131	.227	.887
"Information available"	88.5161	176.031	.376	.883
"Opportunity for leisure activity"	89.4323	166.273	.586	.878
"Able to get around"	88.9677	165.850	.681	.876
"Sleep"	89.2129	172.805	.331	.885
"Daily living activities"	89.3226	162.480	.696	.874
"Capacity for work"	89.7742	163.916	.622	.877
"Satisfied with yourself"	88.9097	164.992	.652	.876
"Personal relationship"	88.1742	176.833	.296	.885
"Sex life"	89.2452	176.277	.192	.890
"Friends"	88.0194	179.993	.193	.886
"Conditions of living place"	87.9032	178.322	.287	.885
"Access to health services"	87.8903	181.241	.156	.887
"Transportation"	88.0129	176.649	.364	.883
"Pain prevents doing"	88.9548	170.628	.396	.883
"Need medical treatment"	89.2645	174.196	.333	.884
"Negative feelings"	88.9355	172.282	.462	.881

Item-Total Statistics

Scale Statistics

Mean	Variance	Std. Deviation	N of Items
92.3548	184.945	13.59944	26

Reliability (Nurses)

Reliability Statistics

Cronbach's	
Alpha	N of Items
.918	26

ltem	Statistics
------	------------

	Mean	Std. Deviation	Ν
Rate QoL (Nurses)	3.5130	.87251	154
Health satisfaction (Nurses)	2.6039	.89602	154
Enjoy life (Nurses)	3.4481	.80879	154
Life is meaningful (Nurses)	3.8831	.79169	154
Concentrate (Nurses)	3.7208	.82046	154
Safe in your daily life (Nurses)	3.6883	.75424	154
Healthy physical environment (Nurses)	3.7662	.73912	154
Energy for everyday life (Nurses)	2.9675	.85878	154
Accept bodily appearance (Nurses)	3.4221	.82274	154
Money to meet needs (Nurses)	3.5195	.77730	154
Information available (Nurses)	3.8506	.81475	154
Opportunity for leisure activities (Nurses)	2.8377	.92511	154
Able to get around (Nurses)	3.4805	.87239	154
Sleep (Nurses)	3.1169	.85519	154
Daily living activities (Nurses)	3.2273	.90404	154
Capacity for work (Nurses)	2.7792	.93757	154
Satisfied with yourself (Nurses)	3.2857	.86102	154
Personal relationships (Nurses)	3.9416	.82634	154
Sex life (Nurses)	3.0000	.89296	154
Friends (Nurses)	4.0909	.67990	154
Conditions of your living place (Nurses)	3.9805	.70914	154
Assess to health services (Nurses)	4.0195	.69987	154
Transportation (Nurses)	3.8377	.79599	154
Pain prevents doing (Nurses)	3.4805	1.02403	154
Need medical treatment (Nurses)	3.0130	.99664	154
Negative feelings (Nurses)	3.1299	.82209	154

	Scale Mean if	Scale Variance if	Corrected Item-Total	Cronbach's Alpha if Item
Rate QoL (Nurses)	86.0909	141.312	.701	.912
Health satisfaction	87.0000	143.725	.562	.914
Enjoy life (Nurses)	86.1558	141.583	.747	.911
Life is meaningful (Nurses)	85.7208	145.248	.562	.914
Concentrate (Nurses)	85.8831	147.437	.426	.917
Safe in your daily life (Nurses)	85.9156	145.372	.586	.914
Healthy physical environment (Nurses)	85.8377	145.784	.576	.914
Energy for everyday life (Nurses)	86.6364	142.416	.656	.913
Accept bodily appearance (Nurses)	86.1818	146.385	.479	.916
Money to meet needs (Nurses)	86.0844	147.934	.427	.916
Information available (Nurses)	85.7532	148.658	.367	.917
Opportunity for leisure activities (Nurses)	86.7662	142.742	.588	.914
Able to get around (Nurses)	86.1234	143.194	.606	.913
Sleep (Nurses)	86.4870	147.899	.383	.917
Daily living activities (Nurses)	86.3766	140.707	.703	.912
Capacity for work (Nurses)	86.8247	141.178	.653	.912
Satisfied with yourself (Nurses)	86.3182	141.277	.713	.912
Personal relationships (Nurses)	85.6623	148.042	.392	.917
Sex life (Nurses)	86.6039	150.202	.256	.920
Friends (Nurses)	85.5130	149.585	.394	.917
Conditions of your living place (Nurses)	85.6234	148.498	.440	.916
Assess to health services (Nurses)	85.5844	149.003	.416	.917
Transportation (Nurses)	85.7662	146.102	.513	.915
Pain prevents doing (Nurses)	86.1234	143.285	.501	.916
Need medical treatment (Nurses)	86.5909	141.276	.605	.913
Negative feelings (Nurses)	86.4740	148.107	.391	.917

Item-Total Statistics

Scale Statistics

Mean	Variance	Std. Deviation	N of Items
89.6039	156.607	12.51426	26

Appendix K: The ICC between patients' and nurses' scores for all 26 items on the WHOQoL- BREF questionnaire

Number of item	ICC
Rate QoL	0.29
Health satisfaction	0.29
Pain prevents doing	0.32
Need medical treatment	0.24
Enjoy life	0.08
Life is meaningful	0.05
Concentrate	0.10
Safe in your daily life	0.14
Healthy physical environment	0.19
Energy for everyday life	0.44
Accept bodily appearance	0.04
Money to meet needs	0.18
Information available	0.19
Opportunity for leisure activities	0.15
Able to get around	0.30
Sleep	0.34
Daily living activities	0.28
Capacity for work	0.38
Satisfied with yourself	0.19
Personal relationships	0.12
Sex life	0.15
Friends	0.05
Conditions of your living place	0.05
Access to health services	0.07
Transportation	0.12
Negative feelings	0.05

Appendix L: The proportion of exact agreement between patients' and nurses' scores for all 26 items on the WHOQoL-BREF questionnaire

Number of item	Proportion of exact agreement between
	patients and nurses
Rate QoL	44.0%
Health satisfaction	32.0%
Pain prevents doing	35.0%
Need medical treatment	35.0%
Enjoy life	41.0%
Life is meaningful	33.0%
Concentrate	37.0%
Safe in your daily life	42.0%
Healthy physical environment	41.0%
Energy for everyday life	42.0%
Accept bodily appearance	31.0%
Money to meet needs	36.0%
Information available	36.0%
Opportunity for leisure activities	34.0%
Able to get around	36.0%
Sleep	32.0%
Daily living activities	30.0%
Capacity for work	34.0%
Satisfied with yourself	31%
Personal relationships	40.0%
Sex life	22.3%
Friends	39.0%
Conditions of your living place	33.0%
Access to health services	38.0%
Transportation	35.0%
Negative feelings	34.0%
Average	35.5%

Appendix M: Bland-Altman test for measuring agreement between patients and nurses

First, I create two new variables in the SPSS data file: 1-difference between mean domains as assessed by patients and proxies (say, diff) and 2- the average of the two mean domains as assessed by patients and nurses (say, mmean). I calculated this via Transform > compute.

Secondly, I calculate the standard deviation of the new diff variable via Analyse > Descriptive Statistics > Descriptive. The standard deviation (S) was 3.04, 3.16, 3.60, and 2.30 for physical, psychological, social, and environmental domains, respectively.

Thirdly, I create the scatter plot of the difference against the mean. I do this from Graphs > Scatter > Simple > Define. I put the diff in the Y-axis box and the mmean in the X-axis box. Then OK.

After creating the chart, it edits to include the limits of agreement. After a double click on the chart, the Chart Editor will be created. Then Chart >Add Chart Elements > Y Axis Reference Line > Reference Line. I need to add in Y Axis Position three numbers: mean - 2S, mean, and mean + 2 S. We would expect 95% of the differences to lie between mean plus two standard deviation and mean minus two standard deviation provided that differences are normally distributed. Such differences are likely to follow a normal distribution because we have removed a lot of the variation between subjects and are left with the measurement error. The averages of the difference for physical, psychological, social, and environmental domains were - 0.27, 0.31, 0.64, and 0.79, respectively .Therefore, the other two "limit of agreement" calculates as below:

Physical domain:

Mean $-2S = -0.27 - (2 \times 3.04) = -6.35$, Mean $+2S = -0.27 + (2 \times 3.04) = 5.81$

Psychological domain:

Mean $-2S = 0.31 - (2 \times 3.16) = -6.01$, Mean $+2S = 0.31 + (2 \times 3.16) = 6.63$ Social relationship domain: Mean $-2S = 0.64 - (2 \times 3.60) = -6.56$, Mean $+2S = 0.70 + (2 \times 3.26) = 7.22$ Environmental domain: Mean $-2S = 0.79 - (2 \times 2.31) = -3.83$, Mean $+2S = 0.79 + (2 \times 2.31) = 5.41$

For each number I use the Apply > Close. After entering the three numbers, close the Chart Editor.

There is also an option which is named "precision of estimated limits of agreement (Bland & Altman 1986). It states that sometimes we want to use standard errors and confidence intervals to see how precise our estimates are. By then calculating 95% confidence intervals for the upper and lower limits of agreement we can derive the most optimistic (by using the lower confidence limit of the upper limit of agreement) and the most conservative (by using the upper limit of the upper limit of agreement) and the most confidence limit of the lower limit of the upper limit of agreement, and the lower confidence limit of the lower limit of agreement, and the lower confidence limit of the lower limit of agreement) and the most confidence limit of the lower limit of agreement) estimates of the range of bias. The standard error of the mean difference (i.e., the bias) is $\sqrt{S^2}/n$, where n is the sample size, and the standard error mean – 2S and mean + 2S is about $\sqrt{3S^2}/n$. 95% confidence intervals can be calculated by finding the appropriate point of the t distribution with n-1 degrees of freedom, on most tables the columns 5% or 0.05. Then the confidence interval will be from the observed value minus t standard errors to the observed value plus t standard errors.

REFERENCES

- Aaronson, NK 1988, 'Quality of life: what is it? How should it be measured?' *Oncology*, vol. 2, no. 5, pp. 69-74.
- ---- 1989, 'Quality of life assessment in clinical trials: Methodologic issues', *Controlled Clinical Trials*, vol. 10, no. 4 suppl, pp. 195S-208S.
- ---- 1991, 'Methodologic issues in assessing the quality of life of cancer patients', *Cancer*, vol. 67, no. 3 suppl, pp. 844-50.
- Addington-Hall, J & Kalra, L 2001, 'Measuring quality of life: Who should measure quality of life?' *British Medical Journal*, vol. 322, no. 7299, pp. 1417-20.
- Allison, PG, Guichard, C & Gilain, L 2000, 'A prospective investigation of dispositional optimism as a predictor of health-related quality of life in head and neck cancer patients', *Quality of Life Research*, vol. 9, no. 8, pp. 951-60.
- Altman, D & Bland, J 1983, 'Measurement in medicine: The analysis of method comparison studies', *Statistician*, vol. 32, pp. 307-17.
- Anderson, JP, Bush, JW & Berry, CC 1986, 'Classifying function for health outcome and quality-of-life evaluation. Self- versus interviewer modes', *Medical Care*, vol. 24, no. 5, pp. 454-69.
- Anderson, KL & Burckhardt, CS 1999, 'Conceptualization and measurement of quality of life as an outcome variable for health care intervention and research', *Journal of Advanced Nursing*, vol. 29, no. 2, pp. 298-306.
- Andrews, FM & Withey, SB 1976, Social indicators of well-being: Americans' perceptions of life quality, Plenum Press, New York.
- Annells, M 2003, 'Grounded theory', in Z Schneider, D Elliott, G LoBiondo-Wood & J Haber (eds), *Nursing research: methods, critical appraisal and utilisation*, 2nd edn, Mosby Publishers, Sydney.
- Annells, M 1996, 'Grounded Theory Method: Philosophical Perspectives, Paradigm of Inquiry, and Postmodernism', *Qualitative Health Research*, vol. 6, no. 3, pp. 379-93.
- ---- 1997, 'Grounded theory method, Part I: Within the five moments of qualitative research', *Nursing Inquiry*, vol. 4, no. 2, pp. 120-9.
- ---- 2006, 'Triangulation of qualitative approaches: hermeneutical phenomenology and grounded theory', *Journal of Advanced Nursing*, vol. 56, no. 1, pp. 55-61.
- ---- 2007, 'What's common with qualitative nursing research these days?' *Journal of Clinical Nursing*, vol. 16, no. 2, pp. 223-4.

- Arraras, JI, Arias, F, Tejedor, M, Pruja, E, Marcos, M, Martinez, E & Valerdi, J 2002, 'The EORTC QLQ-C30 (Version 3.0) Quality of Life questionnaire: Validation study for Spain with head and neck cancer patients', *Psycho-Oncology*, vol. 11, no. 3, pp. 249-56.
- Atkins, C 1998, 'Just little things: Nurses' perceptions of quality of life for people with severe multiple impairments', PhD thesis, The University of Sydney.
- Auerbach, CF & Silverstein, LB 2003, *Qualitative data: an introduction to coding and analysis*, Qualitative studies in psychology, New York University Press, New York.
- Australian Bureau of Statistics 2006, *Cancer in Australia: A Snapshot 2004-05*, viewed 19 June 2007, http://www.abs.gov.au/Ausstats/abs@.nsf/e8ae5488b598839cca25682000131612/8ddd5aed085834daca256f010077be4a!OpenDocument.
- Australian Bureau of Statistics 2007, *Causes of Death, Australia, 2005*, viewed 11 January 2008, http://www.abs.gov.au/ausstats/abs@.nsf/productsbytopic/2093DA6935DB 138FCA2568A9001393C9?OpenDocument>.
- Australian Cancer Research Foundation (ACRF) n.d., *About Cancer*, viewed 19 June 2007, http://www.acrf.com.au/page/about_cancer.html>.
- Australian Institute of Health and Welfare (AIHW) n.d., *Cancer*, viewed 19 June 2007, http://www.aihw.gov.au/cancer/index.cfm.
- Avicenna 1973, The Metaphysica of Avicenna (ibn Sina): a critical translationcommentary and analysis of the fundamental arguments in Avicenna's Metaphysica in the Danish nama-i 'ala'i (the book of scientific knowledge), Trans. P. Morewedge, Routledge & Kegan Paul, London.
- ---- 2005, *The metaphysics of The healing : A parallel English-Arabic text*, Trans. M.E. Marumura, Islamic translation series, Brigham Young University Press, Provo, Utah.
- Aziz, NM 2002, 'Cancer Survivorship Research: Challenge and Opportunity', *Journal of Nutrition*, vol. 132, no. 11, pp. 3494S-503.
- Baker, DC 2003, 'Studies of the inner life: The impact of spirituality on quality of life', *Quality of Life Research*, vol. 12, no. 1, pp. 51-7.
- Ballatori, E 2001, 'Unsolved problems in evaluating the quality of life of cancer patients', *Annals of Oncology*, vol. 12, suppl. 3, pp. S11-S13.
- Bardwell, WA, Major, JM, Rock, CL, Newman, VA, Thomson, CA, Chilton, JA, Dimsdale, JE & Pierce, JP 2004, 'Health-related Quality of Life in women previously treated for early-stage breast cancer', *Psycho-Oncology*, vol. 13, no. 9, pp. 595-604.

- Barofsky, I 1997, '4th Annual Conference of the International Society for Quality of Life Research', *Quality of Life Research*, vol. 6, nos. 7/8, pp. 613-747.
- Baycroft, P 2002, 'Time nurses improved their own quality of life', *Australian Nursing Journal*, vol. 10, no. 3, p. 3.
- Becchi, A, Rucci, P, Placentino, A, Neri, G & Girolamo, G 2004, 'Quality of life in patients with schizophrenia-comparison of self-report and proxy assessments', *Social Psychiatry and Psychiatric Epidemiology*, vol. 39, no. 5, pp. 397-401.
- Belcher, AE 1990, 'Nursing aspects of quality of life enhancement in cancer patients', *Oncology* (*Huntingt*), vol. 4, no. 5, pp. 197-9; discussion 207-8.
- Benner, P 1984, *From novice to expert: Excellence and power in clinical nursing practice*, Addison-Wesley, Menlo Park.
- ---- 1985, 'Quality of life: a phenomenological perspective on explanation, prediction, and understanding in nursing science', *ANS Advances in Nursing Science*, vol. 8, no. 1, pp. 1-14.
- Bernhard, J, Cella, DF, Coates, AS, Fallowfield, L, Ganz, PA, Moinpour, CM, Mosconi, P, Osoba, D, Simes, J & Hürny, C 1998, 'Missing quality of life data in cancer clinical trials: serious problems and challenges', *Statistics in Medicine*, vol. 17, no. 5-7, pp. 517-32.
- Berzoff, J 2006, 'Narratives of grief and their potential for transformation', *Palliative & Supportive Care*, vol. 4, no. 2, pp. 121-7.
- Bezjak, A, Ng, P, Skeel, R, DePetrillo, AD, Comis, R & Taylor, KM 2001, 'Oncologists' use of quality of life information: Results of a survey of Eastern Cooperative Oncology Group physicians', *Quality of Life Research*, vol. 10, no. 1, pp. 1-14.
- Bezjak, A, Taylor, KM, Ng, P, Macdonald, K & DePetrillo, AD 1998, 'Quality-oflife information and clinical practice: the oncologist's perspective', *Cancer Prevention and Control*, vol. 2, no. 5, pp. 230-5.
- Bishop, LP & Griffin, C 2006, 'Holistic healing methods positively advance patient care', *Nursing Management*, vol. 37, no. 7, pp. 30-6.
- Bland, J & Altman, D 1986, 'Comparing methods of measurement: why plotting difference against standard method is misleading', *Lancet*, vol. 346, pp. 1085-7.
- ---- 1986, 'Statistical methods for assessing agreement between two methods of clinical measurement', *Lancet*, pp. 307-10.
- Blazeby, JM, Brookes, ST & Alderson, D 2001, 'The prognostic value of quality of life scores during treatment for oesophageal cancer', *Gut*, vol. 49, no. 2, pp. 227-30.

- Blazeby, JM, Williams, MH, Alderson, D & Farndon, JR 1995, 'Observer variation in assessment of Quality-of-Life in patients with esophageal cancer', *British Journal of Surgery*, vol. 82, no. 9, pp. 1200-3.
- Bloom, JR, Stewart, SL, Chang, S & Banks, PJ 2004, 'Then and now: Quality of life of young breast cancer survivors', *Psycho-Oncology*, vol. 13, no. 3, pp. 147-60.
- Blumer, H 1969, *Symbolic interactionism: perspective and method*, Prentice-Hall, Englewood Cliffs, N.J.
- Boling, W, Fouladi, RT & Basen-engquist, K 2003, 'Health-related quality of life in gynecological oncology: Instruments and psychometric properties', *International Journal of Gynecological Cancer*, vol. 13, no. 1, pp. 5-14.
- Bonomi, AE, Patrick, DL, Bushnell, DM & Martin, M 2000, 'Validation of the United States' version of the World Health Organization Quality of Life (WHOQOL) instrument', *Journal of Clinical Epidemiology*, vol. 53, no. 1, pp. 1-12.
- Bottomley, A 2002, 'The cancer patient and quality of life', *The Oncologist*, vol. 7, no. 2, pp. 120-5.
- Bottomley, A, Efficace, F, Thomas, R, Vanvoorden, V & Ahmedzai, SH 2003, 'Health-related quality of life in non-small-cell lung cancer: Methodologic issues in randomized controlled trials', *Journal of Clinical Oncology*, vol. 21, no. 15, pp. 2982-92.
- Bottomley, A, Therasse, P, Piccart, M & Efficace, F 2005, 'Health-related quality of life in survivors of locally advanced breast cancer: an international randomised controlled phase III trial', *Lancet Oncology*, vol. 6, pp. 287-94.
- Breetvelt, IS & Van Dam, FS 1991, 'Underreporting by cancer patients: the case of response-shift', *Social Science & Medicine*, vol. 32, no. 9, pp. 981-7.
- Bringer, JD, Johnston, LH & Brackenridge, CH 2004, 'Maximizing transparency in a doctoral thesis: The complexities of writing about the use of QSR NVivo within a grounded theory study', *Qualitative Research*, vol. 4, no. 2, pp. 247-65.
- Broberger, E, Tishelman, C & von Essen, L 2005, 'Discrepancies and Similarities in How Patients with Lung Cancer and Their Professional and Family Caregivers Assess Symptom Occurrence and Symptom Distress', *Journal of Pain and Symptom Management*, vol. 29, no. 6, pp. 572-83.
- Browne, J 2004, 'Grounded theory analysis: coming to data with questioning minds', in V Minichiello, G Sullivan, K Greenwood & R Axford (eds), *Handbook of research methods for nursing and health science*, 2nd edn, Prentice Hall Health, Frenchs Forest, N.S.W.

- Brunelli, C, Costantini, M, Di Giulio, P, Gallucci, M, Fusco, F, Miccinesi, G, Paci, E, Peruselli, C, Morino, P, Piazza, M, Tamburini, M & Toscani, F 1998, 'Quality-of-life evaluation: When do terminal cancer patients and health-care providers agree?' *Journal of Pain and Symptom Management*, vol. 15, no. 3, pp. 151-8.
- Burkhardt, MA 1994, 'Becoming and connecting: elements of spirituality for women', *Holistic Nursing Practice*, vol. 8, no. 4, pp. 12-21.
- Bush, NE, Haberman, M, Donaldson, G & Sullivan, KM 1995, 'Quality of life of 125 adults surviving 6-18 years after bone marrow transplantation', *Social Science & Medicine*, vol. 40, no. 4, pp. 479-90.
- Butow, PN, Coates, AS & Dunn, SM 1999, 'Psychosocial Predictors of Survival in Metastatic Melanoma', *Journal of Clinical Oncology*, vol. 17, no. 7, p. 2256.
- Byar, K, Eilers, JE & Nuss, S 2005, 'Quality of life 5 or more years post-autologous hematopoietic stem cell transplant', *Cancer Nursing*, vol. 28, no. 2, pp. 148-57.
- Campbell, A, Converse, PE & Rodgers, WL 1976, *The Quality of American Life: Perceptions, Evaluations, and Satisfactions*, Russell Sage Foundation, New York.
- Carr, AJ, Gibson, B & Robinson, PG 2001, 'Measuring quality of life: Is quality of life determined by expectations or experience?' *British Medical Journal*, vol. 322, no. 7296, pp. 1240-3.
- Carr, AJ & Higginson, IJ 2001, 'Measuring quality of life: Are quality of life measures patient centred?' *British Medical Journal*, vol. 322, no. 7298, pp. 1357-60.
- Cassileth, BR, Lusk, EJ, Miller, DS, Brown, LL & Miller, C 1985, 'Psychosocial correlates of survival in advanced malignant disease?' *The New England Journal of Medicine*, vol. 312, no. 24, pp. 1551-5.
- Cassileth, BR, Walsh, WP & Lusk, EJ 1988, 'Psychosocial correlates of cancer survival: a subsequent report 3 to 8 years after cancer diagnosis', *Journal of Clinical Oncology*, vol. 6, no. 11, pp. 1753-9.
- Caton, AP & Klemm, P 2006, 'Introduction of novice oncology nurses to end-of-life care', *Clinical Journal of Oncology Nursing*, vol. 10, no. 5, pp. 604-8.
- Cella, D, Chang, CH, Lai, JS & Webster, K 2002, 'Advances in quality of life measurements in oncology patients', *Seminars in Oncology*, vol. 29, no. 3 suppl 8, pp. 60-8.
- Cella, D & Tulsky, DS 1990, 'Measuring quality of life today: methodological aspects', *Oncology (Huntingt)*, vol. 4, no. 5, pp. 29-38; discussion 69.

- Cella, D & Nowinski, CJ 2002, 'Measuring quality of life in chronic illness: the functional assessment of chronic illness therapy measurement system', *Archives of Physical Medicine and Rehabilitation*, vol. 83, no. 12 suppl 2, pp. S10-7.
- Cella, DF 1995, 'Measuring quality of life in palliative care', *Seminars in Oncology*, vol. 22, no. 2 suppl 3, pp. 73-81.
- ---- 1996, 'Quality of life outcomes: measurement and validation', *Oncology* (*Williston Park*), vol. 10, no. 11 suppl, pp. 233-46.
- Cense, HA, Visser, MRM, Van Sandick, JW, De Boer, A, Lamme, B, Obertop, H & Van Lanschot, JJB 2004, 'Quality of life after colon interposition by necessity for esophageal cancer replacement', *Journal of Surgical Oncology*, vol. 88, no. 1, pp. 32-8.
- Chaboyer, W & Forrester, K 2000, 'The use of proxy-generated health status assessments in ICU survivors: The issue of quality of life', *Journal of Law and Medicine*, vol. 8, pp. 157-63.
- Chancellor, J, Coyle, D & Drummond, MF 1997, 'Constructing health state preference values from descriptive quality of life outcomes: Mission impossible?' *Quality of Life Research*, vol. 6, no. 2, p. 0.
- Chang, VT, Thaler, HT, Polyak, TA, Kornblith, AB, Lepore, JM & Portenoy, RK 1998, 'Quality of life and survival: the role of multidimensional symptom assessment', *Cancer*, vol. 83, no. 1, pp. 173-9.
- Charmaz, K 2000, 'Grounded theory: Objectivist and Constructivist Methods', in NK Denzin & YS Lincoln (eds), *Handbook of qualitative research*, 2nd edn, Sage Publications, Thousand Oaks, Calif.
- Chaturvedi, SK, Tazaki, M, Fukuhara, S & Ikeda, S 1997, 'Quality of life research and activities in Asia: Current perspectives', *Quality of Life Research*, vol. 6, nos. 7/8, p. 65.
- Cheek, J 1996, 'Taking a View: Qualitative Research as Representation', *Qualitative Health Research*, vol. 6, no. 4, pp. 492-505.
- Clipp, EC & George, LK 1992, 'Patients with cancer and their spouse caregivers. Perceptions of the illness experience', *Cancer*, vol. 69, no. 4, pp. 1074-9.
- Coates, A, Gebski, V, Bishop, J, Jeal, P, Woods, R, Snyder, R, Tattersall, M, Byrne, M, Harvey, V & Gill, G 1987, 'Improving the quality of life during chemotherapy for advanced breast cancer. A comparison of intermittent and continuous treatment strategies', *The New England Journal of Medicine*, vol. 317, no. 24, pp. 1490-5.

- Coates, A, Porzsolt, F & Osoba, D 1997, 'Quality of life in oncology practice: prognostic value of EORTC QLQ-C30 scores in patients with advanced malignancy', *European Journal of Cancer*, vol. 33, no. 7, pp. 1025-30.
- Coates, A, Thomson, D, McLeod, GR, Hersey, P, Gill, PG, Olver, IN, Kefford, R, Lowenthal, RM, Beadle, G & Walpole, E 1993, 'Prognostic value of quality of life scores in a trial of chemotherapy with or without interferon in patients with metastatic malignant melanoma', *European Journal of Cancer*, vol. 29A, no. 12, pp. 1731-4.
- Coates, AS, Hurny, C, Peterson, HF, Bernhard, J, Castiglione-Gertsch, M, Gelber, RD & Goldhirsch, A 2000, 'Quality-of-Life Scores Predict Outcome in Metastatic but Not Early Breast Cancer', *Journal of Clinical Oncology*, vol. 18, no. 22, pp. 3768-74.
- Cohen, J 1988, *Statistical power analysis for the behavioral sciences*, Hillsdale, NJ: Erlbaum.
- Corner, J 2002, 'Nurses' experiences of cancer', *European Journal of Cancer Care*, vol. 11, no. 3, pp. 193-9.
- Costantini, M, Mencaglia, E, Giulio, PD, Cortesi, E, Roila, F, Ballatori, E, Tamburini, M, Casali, P, Licitra, L, Candis, DD, Massidda, B, Luzzani, M, Campora, E, Placido, SD, Palmeri, S, Angela, PM, Baracco, G, Gareri, R, Martignetti, A, Ragosa, S, Zoda, L, Ionta, MT, Bulletti, S & Pastore, L 2000, 'Cancer patients as `experts' in defining quality of life domains. A multicentre survey by the Italian Group for the Evaluation of Outcomes in Oncology (IGEO)', *Quality of Life Research*, vol. 9, no. 2, pp. 151-9.
- Cox, K 2003, 'Assessing the quality of life of patients in phase I and II anti-cancer drug trials: interviews versus questionnaires', *Social Science & Medicine*, vol. 56, no. 5, pp. 921-34.
- Creswell, J, PLano Clark, V, Gutmann, M & Hanson, W 2003, 'Advanced mixed methods research designs', in A Tashakkori & C Teddlie (eds), *Handbook of Mixed Methods in Social & Behavioral research*, Sage Publications, Thousand Oaks.
- Creswell, JW 2003, *Research design: qualitative, quantitative, and mixed methods approaches*, 2nd edn, Sage Publications, Thousand Oaks, Calif.
- Crossan, F 2003, 'Research philosophy: towards an understanding', *Nurse Researcher*, vol. 11, no. 1, pp. 46-55.
- Cutcliffe, JR 2000, 'Methodological issues in grounded theory', *Journal of Advanced Nursing*, vol. 31, no. 6, pp. 1476-84.
- Dancey, J, Zee, B, Osoba, D, Whitehead, M, Lu, F, Kaizer, L, Latreille, J & Pater, JL 1997, 'Quality of life scores: An independent prognostic variable in a general

population of cancer patients receiving chemotherapy', *Quality of Life Research*, vol. 6, no. 2, pp. 151-158.

- Davies, ML 1997, 'Shattered assumptions: time and the experience of long-term HIV positivity', *Social Science & Medicine*, vol. 44, no. 5, pp. 561-71.
- de Haes, JCJM & van Knippenberg, FCE 1985, 'The quality of life of cancer patients: A review of the literature', *Social Science & Medicine*, vol. 20, no. 8, pp. 809-17.
- de Liano, AD, Martinez, FO, Ciga, MA, Aizcorbe, M, Cobo, F & Trujillo, R 2003, 'Impact of surgical procedure for gastric cancer on quality of life', *British Journal of Surgery*, vol. 90, no. 1, pp. 91-4.
- Denzin, NK & Lincoln, YS (eds) 2000, *The handbook of qualitative research*, 2nd edn, Sage Publications, Thousand Oaks, Calif.
- Department of Psychiatry of the University of Melbourne 2000, Australian WHOQOL-100, WHOQOL-BREF and CA-WHOQOL instructions: user's manual and interpretation guide, viewed 13 June 2007, <http://www.psychiatry.unimelb.edu.au/qol/whoqol/instruments/manual.pdf>
- ---- 2007, Centre for Participant Reported Outcomes: incorporating the Australian WHOQoL Field Centre, viewed 24 July 2007, <http://www.psychiatry.unimelb.edu.au/qol/>.
- DePoy, E & Gitlin, LN 1998, *Introduction to research: understanding and applying multiple strategies*, 2nd edn, Mosby, St. Louis.
- Deschler, D, Walsh, KA, Friedman, S & Hayden, R 1999, 'Quality of Life Assessment in Patients Undergoing Head and Neck Surgery As Evaluated by Lay Caregivers', *Laryngoscope*, vol. 109, no. 1, pp. 42-6.
- Dorman, PJ, Waddell, F, Slattery, J, Dennis, M & Sandercock, P 1997, 'Are proxy assessments of health status after stroke with the EuroQol questionnaire feasible, accurate, and unbiased?' *Stroke*, vol. 28, no. 10, pp. 1883-7.
- Draper, P 1992, 'Quality of life as quality of being: an alternative to the subjectobject dichotomy', *Journal of Advanced Nursing*, vol. 17, no. 8, pp. 965-70.
- Dumont, I, Dumont, S & Turgeon, J 2005, 'Continuity of care for advanced cancer patients', *Journal of Palliative Care*, vol. 21, no. 1, pp. 49-56.
- Earl, HM, Rudd, RM, Spiro, SG, Ash, CM, James, LE, Law, CS, Tobias, JS, Harper, PG, Geddes, DM, Eraut, D & et al. 1991, 'A randomised trial of planned versus as required chemotherapy in small cell lung cancer: a Cancer Research Campaign trial', *British Journal of Cancer*, vol. 64, no. 3, pp. 566-72.

- Earlam, S, Glover, C, Fordy, C, Burke, D & Allen-Mersh, T 1996, 'Relation between tumor size, quality of life, and survival in patients with colorectal liver metastases', *Journal of Clinical Oncology*, vol. 14, no. 1, pp. 171-5.
- Eastern Cooperative Oncology Group (ECOG) Performance Status viewed 22 Feb, 18:15 pm 2006, http://www.ecog.org/general/perf_stat.html.
- Ecosse, E, Leplège, A & group., TWR 2005, 'Rasch model isolates Quality-of-Life construct in six WHOQOL-100 Data Sets ', in B Nikolaus (ed.), *Rasch measurement in health sciences*, JAM Press, Maple Grove, Minn.
- Edwards, B & Ung, L 2002, 'Quality of life instruments for caregivers of patients with Cancer: A review of their psychometric properties', *Cancer Nursing*, vol. 25, no. 5, pp. 342-9.
- Efficace, F, Bottomley, A & van Andel, G 2003, 'Health related quality of life in prostate carcinoma patients A systematic review of randomized controlled trials', *Cancer*, vol. 97, no. 2, pp. 377-88.
- Efficace, F, Bottomley, A, Vanvoorden, V & Blazeby, JM 2004, 'Methodological issues in assessing health-related quality of life of colorectal cancer patients in randomised controlled trials', *European Journal of Cancer*, vol. 40, no. 2, pp. 187-97.
- Epstein, AM, Hall, JA, Tognetti, J, Son, LH & Conant, L, Jr. 1989, 'Using proxies to evaluate quality of life. Can they provide valid information about patients' health status and satisfaction with medical care?' *Medical Care*, vol. 27, no. 3 suppl, pp. S91-8.
- Ergun, FS, Oran, NT & Bender, CM 2005, 'Quality of life of oncology nurses', *Cancer Nursing*, vol. 28, no. 3, pp. 193-9.
- Erickson, HC, Tomlin, EM & Swain, MAP 1983, *Modeling and role modeling: a theory and paradigm for nursing*, Prentice-Hall, Englewood Cliffs, N.J.
- Fallowfield, L 2002, 'Quality of life: a new perspective for cancer patients', *Nature Reviews Cancer*, vol. 2, no. 11, pp. 873-9.
- Fayers, P & Machin, D 2000, *Quality of life: Assessment, Analysis and Interpretation*, John Wily &Sons, LTD, Chichester.
- Fayers, PM, Curran, D & Machin, D 1998, 'Incomplete quality of life data in randomized trials: Missing items', *Statistics in Medicine*, vol. 17, no. 5-7, pp. 679-96.
- Fayers, PM, Hand, DJ, Bjordal, K & Groenvold, M 1997, 'Causal indicators in quality of life research', *Quality of Life Research*, vol. 6, no. 5, pp. 393-406.
- Fayers, PM & Jones, DR 1983, 'Measuring and analysing quality of life in cancer clinical trials: a review', *Statistics in Medicine*, vol. 2, no. 4, pp. 429-46.

- Ferlito, A, Rogers, SN, Shaha, AR, Bradley, PJ & Rinaldo, A 2003, 'Quality of life in head and neck cancer', *Acta Oto-Laryngologica*, vol. 123, no. 1, pp. 5-7.
- Ferrans, CE & Powers, MJ 1985, 'Quality of life index: development and psychometric properties', Advances in Nursing Science, vol. 8, no. 1, pp. 15-24.
- ---- 1992, 'Psychometric assessment of the Quality of Life Index', *Research in Nursing and Health*, vol. 15, no. 1, pp. 29-38.
- Ferrell, B, Grant, M, Schmidt, GM, Rhiner, M, Whitehead, C, Fonbuena, P & Forman, SJ 1992, 'The meaning of quality of life for bone marrow transplant survivors. Part 2. Improving quality of life for bone marrow transplant survivors', *Cancer Nursing*, vol. 15, no. 4, pp. 247-53.
- Ferrell, BR, Dow, KH & Grant, M 1995, 'Measurement of the quality of life in cancer survivors', *Qual Life Research*, vol. 4, no. 6, pp. 523-31.
- Finfgeld-Connett, D 2008, 'Meta-synthesis of caring in nursing', *Journal of Clinical Nursing*, vol. 17, no. 2, pp. 196-204.
- Finlay, L 2006, 'Mapping methodology', in L Finlay & C Ballinger (eds), *Qualitative research for allied health professionals: challenging choices*, John Wiley, Chichester.
- Fisch, MJ, Titzer, ML, Kristeller, JL, Shen, J, Loehrer, PJ, Jung, S, Passik, SD & Einhorn, LH 2003, 'Assessment of quality of life in outpatients with advanced cancer: the accuracy of clinician estimations and the relevance of spiritual well-being -- a Hoosier Oncology Group study', *Journal of Clinical Oncology*, vol. 21, no. 14, pp. 2754-9.
- Fitch, MI 1998, 'The 1997 Schering Lecture. Quality of life in oncology: nurses' perceptions, values and behaviours', *Canadian Oncology Nursing Journal*, vol. 8, no. 1, pp. 24-39.
- Fossa, SD, de Wit, R, Roberts, JT, Wilkinson, PM, de Mulder, PHM, Mead, GM, Cook, P, de Prijck, L, Stenning, S, Aaronson, NK, Bottomley, A & Collette, L 2003, 'Quality of life in good prognosis patients with metastatic germ cell cancer: A prospective study of the European Organization for Research and Treatment of Cancer Genitourinary Group/Medical Research Council Testicular Cancer Study Group (30941/TE20)', *Journal of Clinical Oncology*, vol. 21, no. 6, pp. 1107-18.
- Fowler, FJ 1988, *Survey research methods*, Rev. edn, Applied social research methods series, Sage Publications, Newbury Park, CA.
- Frost, MH, Brueggen, C & Mangan, M 1997, 'Intervening with the psychosocial needs of patients and families: perceived importance and skill level', *Cancer Nursing*, vol. 20, no. 5, pp. 350-8.

- Funk, GF, Karnell, LH, Smith, RB & Christensen, AJ 2004, 'Clinical significance of health status assessment measures in head and neck cancer - What do qualityof-life scores mean?' Archives of Otolaryngology-Head & Neck Surgery, vol. 130, no. 7, pp. 825-9.
- Galalae, RM, Loch, T, Riemer, B, Rzehak, P, Kuchler, T, Kimmig, B & Kovacs, G 2004, 'Health-related quality of life measurement in long-term survivors and outcome following radical radiotherapy for localized prostate cancer', *Strahlentherapie Und Onkologie*, vol. 180, no. 9, pp. 582-9.
- Gall, TL 2004, 'Relationship with God and the quality of life of prostate cancer survivors', *Quality of Life Research*, vol. 13, no. 8, pp. 1357-68.
- Ganiats, TG, Sieber, WJ, Barber, E & Barrett, E 1997, 'Initial comparison of four generic health-related quality of life instruments', *Quality of Life Research*, vol. 6, no. 7-8, p. 119.
- Ganz, PA, Haskell, CM, Figlin, RA, La Soto, N & La Soto, N 1988, 'Estimating the quality of life in a clinical trial of patients with metastatic lung cancer using the Karnofsky performance status and the Functional Living Index--Cancer', *Cancer*, vol. 61, no. 4, pp. 849-56.
- Garratt, A, Schmidt, L, Mackintosh, A & Fitzpatrick, R 2002, 'Quality of life measurement: bibliographic study of patient assessed health outcome measures', *British Medical Journal*, vol. 324, p. 1417.
- Geddes, DM, Dones, L, Hill, E, Law, K, Harper, PG, Spiro, SG, Tobias, JS & Souhami, RL 1990, 'Quality of life during chemotherapy for small cell lung cancer: assessment and use of a daily diary card in a randomized trial', *European Journal of Cancer*, vol. 26, no. 4, pp. 484-92.
- Gerber, R & Moyle, W 2004, 'The role of theory in health research', in V Minichiello, G Sullivan, K Greenwood & R Axford (eds), *Handbook of research methods for nursing and health science*, 2nd edn, Prentice Hall Health, Frenchs Forest, N.S.W., pp. xxii, 680.
- Giesler, RB 2000, 'Assessing the quality of life of patients with cancer', *Current Problems in Cancer*, vol. 24, no. 2, pp. 55-92.
- Gill, TM & Feinstein, AR 1994, 'A critical appraisal of the quality of quality-of-life measurements', *Journal of The American Medical Association*, vol. 272, no. 8, pp. 619-26.
- Glaser, BG 1978, *Theoretical Sensitivity: Advances in the Methodology of Grounded Theory*, Sociology Press, Mill Valley, California.
- Glaser, BG & Strauss, AL 1967, *The discovery of grounded theory; strategies for qualitative research*, Observations, Aldine Pub. Co., Chicago.

- Glaus, A 1993, 'Quality of life: A measure of the quality of nursing care', *Support Care Cancer*, vol. 1, pp. 119-23.
- Grant, M, Padilla, GV, Ferrell, BR & Rhiner, M 1990, 'Assessment of quality of life with a single instrument', *Seminars in Oncology Nursing*, vol. 6, no. 4, pp. 260-70.
- Grant, M & Rivera, M 1988, Evolution of Quality of Life in Oncology and Oncology Nursing .Quality of Life: From Nursing and Patient Perspectives.
- Grbich, CF 1999, *Qualitative research in health: an introduction*, Allen & Unwin, St Leonards, N.S.W.
- Greenwood, KM 2004, 'Measurement: concepts, tools and issues', in V Minichiello, G Sullivan, K Greenwood & R Axford (eds), *Handbook of research methods for nursing and health science*, 2nd edn, Prentice Hall Health, Frenchs Forest, N.S.W.
- Gridelli, C, Cortesi, E & Roila, F 2001, 'Survival and quality of life: Comparing end points in oncology', *Annals of Oncology*, vol. 12, suppl. 3, p. S1.
- Guba, EG 1990, 'The alternative paradigm dialog', in EG Guba (ed.), *The Paradigm dialog*, Sage Publications, Newbury Park, Calif., p. 424.
- Guba, EG & Lincoln, YS 1994, 'Comparing paradigms in qualitative research', in NK Denzin & YS Lincoln (eds), *Handbook of qualitative research*, Sage Publications, Thousand Oaks, pp. xii, 643.
- Gubrium, JF & Holstein, JA 2000, 'Analyzing interpretive practice', in NK Denzin & YS Lincoln (eds), *Handbook of qualitative research*, 2nd edn, Sage Publications, Thousand Oaks, Calif.
- Guren, MG, Dueland, S, Skovlund, E, Fossa, SD, Poulsen, JP & Tveit, KM 2003, 'Quality of life during radiotherapy for rectal cancer', *European Journal of Cancer*, vol. 39, no. 5, pp. 587-94.
- Guyatt, GH, Feeny, DH & Patrick, DL 1993, 'Measuring Health-related Quality of Life', *Annals of Internal Medicine*, vol. 118, no. 8, pp. 622-9.
- Haberman, MR & Bush, N 2003, 'Quality of life: Methodological and measurement issues', in CR King & PS Hinds (eds), *Quality of Life: From Nursing and Patient Perspectives: Theory, Research, Practice*, Sudbury, Massachsette, Jones and Bartlett Publishers.
- Hams, SP 2000, 'A gut feeling? Intuition and critical care nursing', *Intensive & Critical Care Nursing*, vol. 16, no. 5, pp. 310-8.
- Hanestad, BR, Rustoen, T, Knudsen, O, Jr., Lerdal, A & Wahl, AK 2004, 'Psychometric properties of the WHOQOL-BREF questionnaire for the

Norwegian general population', *Journal of Nursing Measurement*, vol. 12, no. 2, pp. 147-59.

- Harris, LM & Heard, R 2004, 'Using multivariate analysis', in V Minichiello, G Sullivan, K Greenwood & R Axford (eds), *Handbook of research methods for nursing and health science*, 2nd edn, Prentice Hall Health, Frenchs Forest, N.S.W.
- Hawighorst-Knapstein, S, Fusshoeller, C, Franz, C, Trautmann, K, Schmidt, M,
 Pilch, H, Schoenefuss, G, Kelleher, DK, Vaupel, P, Knapstein, PG & Koelbl,
 H 2004, 'The impact of treatment for genital cancer on quality of life and
 body image-results of a prospective longitudinal 10-year study', *Gynecologic Oncology*, vol. 94, no. 2, pp. 398-403.
- Hawthorne, G, Richardson, J & Day, NA 2001, 'A comparison of the Assessment of Quality of Life (AQoL) with four other generic utility instruments', *Annals of Medicine*, vol. 33, no. 5, pp. 358-70.
- Hawthorne, G, Richardson, J & Osborne, R 1999, 'The Assessment of Quality of Life (AQoL) instrument: a psychometric measure of health-related quality of life', *Quality of Life Research*, vol. 8, no. 3, pp. 209-24.
- Hays, RD, Vickrey, BG, Hermann, BP, Perrine, K, Cramer, J, Meador, K, Spritzer, K & Devinsky, O 1995, 'Agreement between self reports and proxy reports of quality of life in epilepsy patients', *Quality of Life Research*, vol. 4, no. 2, pp. 159-68.
- He, G & Liu, S 2005, 'Quality of life and coping styles in Chinese Nasopharyngeal cancer patients after hospitalization', *Cancer Nursing*, vol. 28, no. 3, pp. 179-86.
- Hearn, J & Higginson, IJ 1998, 'Do specialist palliative care teams improve outcomes for cancer patients? A systematic literature review', *Palliative Medicine*, vol. 12, no. 5, pp. 317-32.
- Hempel, CG 1965, Aspects of scientific explanation, and other essays in the philosophy of science, Free Press, New York.
- Hendry, F & McVittie, C 2004, 'Is Quality of Life a Healthy Concept? Measuring and Understanding Life Experiences of Older People', *Qualitative Health Research*, vol. 14, no. 7, pp. 961-75.
- Herrman, H, Hawthorne, G & Thomas, R 2002, 'Quality of life assessment in people living with psychosis', *Social Psychiatry and Psychiatric Epidemiology*, vol. 37, no. 11, pp. 510-8.
- Higginson, IJ & Carr, AJ 2001, 'Measuring quality of life: Using quality of life measures in the clinical setting', *British Medical Journal*, vol. 322, no. 7297, pp. 1297-300.

- Higgs, J & Llewellyn, G 1998, 'Framing the research question', in J Higgs (ed.), *Writing qualitative research*, Hampden Press in conjunction with the Centre for Professional Education Advancement, Sydney.
- Hinds, PS & King, CR 2003, 'Nursing and Patient Perspectives on Quality of Life', in CR King & PS Hinds (eds), *Quality of Life: From Nursing and Patient Perspectives: Theory, Research, Practice*, Jones and Bartlett Publishers, Sudbury, Massachusetts.
- Holley, UA 2007, 'Social isolation: a practical guide for nurses assisting clients with chronic illness', *Rehabilitation Nursing*, vol. 32, no. 2, pp. 51-6.
- Holmes, CA 1989, 'Health care and the quality of life: a review', *Journal of Advanced Nursing*, vol. 14, no. 10, pp. 833-9.
- Holmes, S & Dickerson, J 2003, 'The quality of life: design and evaluation of a selfassessment instrument for use with cancer patients', *International Journal of Nursing Studies*, vol. 40, no. 5, pp. 515-20.
- Horton, R 2002, 'Differences in assessment of symptoms and quality of life between patients with advanced cancer and their specialist palliative care nurses in a home care setting', *Palliative Medicine*, vol. 16, no. 6, pp. 488-94.
- Hunt, SM 1997, 'The problem of quality of life', *Quality of Life Research*, vol. 6, no. 3, p. 0.
- Isikhan, VP, Guner, PP, Komurcu, SMD, Ozet, AMD, Arpaci, FMD & Ozturk, BMD 2001, 'The relationship between disease features and quality of life in patients with cancer', *Cancer Nursing*, vol. 24, no. 6, pp. 490-5.
- Jacobsen, PB, Davis, K & Cella, D 2002, 'Assessing quality of life in research and clinical practice', *Oncology (Huntingt)*, vol. 16, no. 9 suppl 10, pp. 133-9.
- Jenkins, RA & Pargament, KI 1988, 'Cognitive appraisals in cancer patients', *Social Science & Medicine*, vol. 26, no. 6, pp. 625-33.
- Jennings, BM & Muhlenkamp, AF 1981, 'Systematic misperception: oncology patients' self-reported affective states and their care-givers' perceptions', *Cancer Nursing*, vol. 4, no. 6, pp. 485-9.
- Jeon, Y-H 2004, 'The application of grounded theory and symbolic interactionism', *Scandinavian Journal of Caring Sciences*, vol. 18, no. 3, pp. 249-56.
- Johnson, SC & Spilka, B 1991, 'Coping with breast cancer: The roles of clergy and faith', *Journal of Religion and Health*, vol. 30, no. 1, pp. 21-33.
- Joly, F, Heron, JF, Kalusinski, L, Bottet, P, Brune, D, Allouache, N, Mace-Lesec'h, J, Couette, JE, Peny, J & Henry-Amar, M 2002, 'Quality of life in long-term survivors of testicular cancer: A population-based case-control study', *Journal of Clinical Oncology*, vol. 20, no. 1, pp. 73-80.

- Julious, SA, George, S, Machin, D & Stephens, RJ 1997, 'Sample sizes for randomized trials measuring quality of life in cancer patients', *Quality of Life Research*, vol. 6, no. 2, p. 109-117.
- Kaasa, S & Loge, JH 2003, 'Quality of life in palliative care: principles and practice', *Palliative Medicine*, vol. 17, no. 1, pp. 11-20.
- Kamangar, F, Dores, GM & Anderson, WF 2006, 'Patterns of cancer incidence, mortality, and prevalence across five continents: defining priorities to reduce cancer disparities in different geographic regions of the world', *Journal of Clinical Oncology*, vol. 24, no. 14, pp. 2137-50.
- Karnofsky, DA & Burchenal, JH 1949, *The clinical evaluation of chemotherapeutic agents in cancer*, in MacLeod CM (ed), *Evaluation of Chemotherapeutic Agents*, Columbia University Press, New York.
- Katz, MH 2006, *Multivariate analysis: a practical guide for clinicians*, 2nd edn, Cambridge University Press, Cambridge.
- Katz, S, Ford, AB, Moskowitz, RW, Jackson, BA & Jaffe, MW 1963, 'Studies of illness in the aged: a standardized measure of biological and psychological function', *Journal of The American Medical Association*, vol. 185, pp. 914-9.
- Kemmler, G, Holzner, B, Kopp, M, Dunser, M, Margreiter, R, Greil, R & Sperner-Unterweger, B 1999, 'Comparison of two quality-of-life instruments for cancer patients: the functional assessment of cancer therapy-general and the European Organization for Research and Treatment of Cancer Quality of Life Questionnaire-C30', *Journal of Clinical Oncology*, vol. 17, no. 9, pp. 2932-40.
- Kendall, J 1999, 'Axial Coding and the Grounded Theory Controversy', Western Journal of Nursing Research, vol. 21, no. 6, pp. 743-57.
- King, C 2006, 'Advances in How Clinical Nurses Can Evaluate and Improve Quality of Life for Individuals With Cancer', *Oncology Nursing Forum*, vol. 33, suppl. 1, pp. 5-12.
- King, CR 2003, 'Overview of quality of life and controversial issues', in CR King & PS Hinds (eds), *Quality of Life: From Nursing and Patient Perspectives: Theory, Research, Practice*, Jones and Bartlett Publishers, Sudbury, Massachsette, vol. 1, p. 533.
- ---- 2006a, 'Advances in How Clinical Nurses Can Evaluate and Improve Quality of Life for Individuals With Cancer', *Oncology Nursing Forum*, vol. 33, no. 1, p. 5.
- ---- 2006b, 'Introduction: Improving Oncology Nursing Through Advances in Quality-of-Life Issues', *Oncology Nursing Forum*, vol. 33, no. 1, p. 3.

- King, CR, Ferrell, BR, Grant, M & Sakurai, C 1995, 'Nurses' perceptions of the meaning of quality of life for bone marrow transplant survivors', *Cancer Nursing*, vol. 18, no. 2, pp. 118-29.
- King, CR, Haberman, M, Berry, DL, Bush, N, Butler, L, Dow, KH, Ferrell, B, Grant, M, Gue, D, Hinds, P, Kreuer, J, Padilla, G & Underwood, S 1997, 'Quality of life and the cancer experience: the state-of-the-knowledge', *Oncology Nursing Forum*, vol. 24, no. 1, pp. 27-41.
- King, CR, Hinds, P, Dow, KH, Schum, L & Lee, C 2002, 'The nurse's relationshipbased perceptions of patient quality of life', *Oncology Nursing Forum*, vol. 29, no. 10, pp. E118-26.
- King, MT, Dobson, AJ & Harnett, PR 1996, 'A comparison of two quality-of-life questionnaires for cancer clinical trials: the functional living index--cancer (FLIC) and the quality of life questionnaire core module (QLQ-C30)', *Journal of Clinical Epidemiology*, vol. 49, pp. 21-9.
- Koller, M, Lorenz, W, Wagner, K, Keil, A, Trott, D, Engenhart-Cabillic, R & Nies, C 2000, 'Expectations and quality of life of cancer patients undergoing radiotherapy', *Journal of the Royal Society of Medicine*, vol. 93, no. 12, pp. 621-8.
- Kong, S & Gandhi, S 1997, 'Methodologic assessments of quality of life measures in clinical trials', *The Annals of Pharmacotherapy*, vol. 31, no. 7, pp. 830-6.
- Kopp, M, Schweigkofler, H, Holzner, B, Nachbaur, D, Niederwieser, D, Fleischhacker, WW, Kemmler, G & Sperner-Unterwerger, B 2000, 'EORTC QLQ-C30 and FACT-BMT for the measurement of quality of life in bone marrow transplant recipients: a comparison', *European Journal of Haematology*, vol. 65, no. 2, pp. 97-103.
- Kristjanson, LJ, Nikoletti, S, Porock, D, Smith, M, Lobchuk, M & Pedler, P 1998, 'Congruence between patients' and family caregivers' perceptions of symptom distress in patients with terminal cancer', *Journal of Palliative Care*, vol. 14, no. 3, pp. 24-32.
- Kwok, C & Sullivan, G 2007, 'The concepts of health and preventive health practices of Chinese Australian women in relation to cancer screening', *Journal of Transcultural Nursing*, vol. 18, no. 2, pp. 118-26.
- Lampic, C & Sjoden, PO 2000, 'Patient and staff perceptions of cancer patients' psychological concerns and needs', *Acta Oncologica*, vol. 39, no. 1, pp. 9-22.
- Lampic, C, von Essen, L, Peterson, VW, Larsson, G & Sjoden, PO 1996, 'Anxiety and depression in hospitalized patients with cancer: agreement in patient-staff dyads', *Cancer Nursing*, vol. 19, no. 6, pp. 419-28.
- Landis, JR & Koch, GG 1977, 'The measurement of observer agreement for categorical data', *Biometrics*, vol. 33, no. 1, pp. 159-74.

- Langendijk, JA, Ten Velde, GPM, Aaronson, NK, De Jong, JMA, Muller, MJ & Wouters, EFM 2000, 'Quality of life after palliative radiotherapy in non-small cell lung cancer: A prospective study', *International Journal of Radiation Oncology Biology Physics*, vol. 47, no. 1, pp. 149-55.
- Larsson, G, von Essen, L & Sjoden, PO 1998, 'Quality of life in patients with endocrine tumors of the gastrointestinal tract: Patient and staff perceptions', *Cancer Nursing*, vol. 21, no. 6, pp. 411-20.
- Lawler, J 1997, 'Letters to the Editor', *Qualitative Health Research*, vol. 7, no. 3, pp. 436-8.
- Lee, J, Koh, D & Ong, CN 1989, 'Statistical evaluation of agreement between two methods for measuring a quantitative variable', *Computers in Biology and Medicine*, vol. 19, no. 1, pp. 61-70.
- Leedy, PD & Ormrod, JE 2005, *Practical research: planning and design*, 8th edn, Merrill Prentice Hall, Upper Saddle River, N.J.
- Lewes, GH 1853, *Comte's philosophy of the sciences: being an exposition of the principles of the Cours de philosophie positive of Auguste Comte*, Bohn's Scientific Library, H. G. Bohn, Lond.
- Lheureux, M, Raherison, C, Vernejoux, JM, Nguyen, L, Nocent, C, De Lara, MT & Taytard, A 2004, 'Quality of life in lung cancer: does disclosure of the diagnosis have an impact?' *Lung Cancer*, vol. 43, no. 2, pp. 175-82.
- Liamputtong, P & Ezzy, D 2005, *Qualitative research methods*, 2nd edn, Oxford University Press, South Melbourne.
- Llewellyn, G, Sullivan, G & Minichiello, V 2004, 'Sampling in qualitative research', in V Minichiello, G Sullivan, K Greenwood & R Axford (eds), *Handbook of research methods for nursing and health science*, Prentice Hall Health, Frenchs Forest, N.S.W.
- Lincoln, YS & Guba, EG 2000, 'Paradigmatic controversies, contradictions, and emerging confluences', in NK Denzin & YS Lincoln (eds), *Handbook of qualitative research*, 2nd edn, Sage Publications, Thousand Oaks, Calif.
- Lindley, C 1992, 'Quality of life measurements in oncology', *Pharmacotherapy*, vol. 12, no. 4, pp. 346-52.
- Litwin, MS, Fitzpatrick, JM, Fossa, SD & Newling, DWW 1999, 'Defining an International Research Agenda for Quality of Life in Men With Prostate Cancer', *The Prostate*, vol. 41, no. 1, pp. 58-97.
- Llewellyn, G, Sullivan, G & Minichiello, V 2004, 'Sampling in qualitative research', in V Minichiello, G Sullivan, K Greenwood & R Axford (eds), *Handbook of*

research methods for nursing and health science, 2nd edn, Prentice Hall Health, Frenchs Forest, N.S.W.

- Lobchuk, MM & Degner, LF 2002, 'Patients with cancer and next-of-kin response comparability on physical and psychological symptom well-being Trends and measurement issues', *Cancer Nursing*, vol. 25, no. 5, pp. 358-74.
- Lobchuk, MM, Kristijanson, L, Degner, L, Blood, P & Sloan, JA 1997, 'Perceptions of symptom distress in lung cancer patients: I. Congruence between patients and primary family caregivers', *Journal of Pain and Symptom Management*, vol. 14, no. 3, pp. 136-46.
- Long, K, Sudha, S & Mutran, EJ 1998, 'Elder-proxy agreement concerning the functional status and medical history of the older person: the impact of caregiver burden and depressive symptomatology', *Journal of the American Geriatrics Society*, vol. 46, no. 9, pp. 1103-11.
- Long, LE 1998, 'Getting through radiation therapy: A hermeneutic inquiry of the experience of undergoing radiation therapy', PhD thesis, Flinders University.
- Lutgendorf, SK, Anderson, B, Ullrich, P, Johnsen, EL, Buller, RE, Sood, AK, Sorosky, JI & Ritchie, J 2002, 'Quality of life and mood in women with gynecologic cancer - A one year prospective study', *Cancer*, vol. 94, no. 1, pp. 131-40.
- MacDonald, BH 2001, 'Quality of life in cancer care: patients' experiences and nurses' contribution', *European Journal of Oncology Nursing*, vol. 5, no. 1, pp. 32-41.
- MacDonald, KC 2006, 'Family and staff perceptions of the impact of the long-term care environment on leisure', *Topics in Geriatric Rehabilitation*, vol. 22, no. 4, pp. 294-308.
- MacLeod, M 1994a, "It's the little things that count': the hidden complexity of everyday clinical nursing practice', *Journal of Clinical Nursing*, vol. 3, no. 6, pp. 361-8.
- ---- 1994b, "It's the little things that count': the hidden complexity of everyday clinical nursing practice', *Journal of Clinical Nursing*, vol. 3, no. 6, pp. 361-8.
- Magaziner, J, Simonsick, EM, Kashner, TM & Hebel, JR 1988, 'Patient-proxy response comparability on measures of patient health and functional status', *Journal of Clinical Epidemiology*, vol. 41, no. 11, pp. 1065-74.
- Magaziner, J, Zimmerman, SI, Gruber-Baldini, AL, Hebel, JR & Fox, KM 1997, 'Proxy reporting in five areas of functional status. Comparison with selfreports and observations of performance', *American Journal of Epidemiology*, vol. 146, no. 5, pp. 418-28.

- Maguire, P & Selby, P 1989, 'Assessing quality of life in cancer patients', *British Journal of Cancer*, vol. 60, no. 3, pp. 437-40.
- Mahlungulu, SN & Uys, LR 2004, 'Spirituality in nursing: an analysis of the concept', *Curationis: South African Journal of Nursing*, vol. 27, no. 2, pp. 15-26.
- Mahoney, MJ & Graci, GM 1999, 'The meanings and correlates of spirituality: Suggestions from an exploratory survey of experts', *Death Studies*, vol. 23, no. 6, pp. 521-8.
- Mandelblatt, JS & Eisenberg, JM 1995, 'Historical and methodological perspectives on cancer outcomes research', *Oncology*, vol. 9, no. 11 suppl, pp. 23-32.
- Mast, ME 1995, 'Definition and measurement of quality of life in oncology nursing research: review and theoretical implications', *Oncology Nursing Forum*, vol. 22, no. 6, pp. 957-64.
- Maton, KI 1989 'The Stress-Buffering Role of Spiritual Support: Cross-Sectional and Prospective Investigations', *Journal for the Scientific Study of Religion*, vol. 28 no. 3, pp. 310-23.
- Matthews, LT & Marwit, SJ 2006, 'Meaning reconstruction in the context of religious coping: rebuilding the shattered assumptive world', *Omega: Journal of Death and Dying*, vol. 53, no. 1-2, pp. 87-104.
- McCusker, J & Stoddard, AM 1984, 'Use of a surrogate for the Sickness Impact Profile', *Medical Care*, vol. 22, no. 9, pp. 789-95.
- McGrath, P 2004, 'Positive outcomes for survivors of haematological malignancies from a spiritual perspective', *International Journal of Nursing Practice*, vol. 10, no. 6, pp. 280-91.
- McKenna, SP & Doward, LC 2004, 'The Needs-Based Approach to Quality of Life Assessment', *Value Health*, vol. 7, suppl. 1, pp. S1-S3.
- McKeon, R (ed.) 1947, Introduction to Aristotle, Modern Library, New York.
- McMillan, AS, Leung, KCM, Pow, EHN, Wong, MCM, Li, LSW & Allen, PF 2005, 'Oral health-related quality of life of stroke survivors on discharge from hospital after rehabilitation', *Journal of Oral Rehabilitation*, vol. 32, no. 7, pp. 495-503.
- McMillan, SC 1996, 'Quality-of-Life Assessment in Palliative Care', *Cancer Control Journal*, vol. 3, no. 3, pp. 223-229.
- McPherson, CJ & Addington-Hall, JM 2003, 'Judging the quality of care at the end of life: can proxies provide reliable information?' *Social Science & Medicine*, vol. 56, no. 1, pp. 95-109.

- McVilly, KR, Burton, RM & Davidson, JA 2000, 'Concurrence between subject and proxy ratings of quality of life for people with and without intellectual disabilities', *Journal of Intellectual &Developmental Disability*, vol. 25, no. 1, pp. 19-39.
- Meyerowitz, BE 1993, 'Quality of life in breast cancer patients: the contribution of data to the care of patients', *European Journal of Cancer*, vol. 29A suppl 1, pp. S59-62.
- Michael, M & Tannock, IF 1998, 'Measuring health-related quality of life in clinical trials that evaluate the role of chemotherapy in cancer treatment', *Canadian Medical Association Journal*, vol. 158, no. 13, pp. 1727-34.
- Minichiello, V, Madison, J, Hays, T & Parmenter, G 2004, 'Doing qualitative indepth interviews', in V Minichiello, G Sullivan, K Greenwood & R Axford (eds), *Handbook of research methods for nursing and health science*, 2nd edn, Prentice Hall Health, Frenchs Forest, N.S.W.
- Miyake, H, Muramaki, M, Eto, H, Kamidono, S & Hara, I 2004, 'Health-related quality of life in patients with testicular cancer: A comparative analysis according to therapeutic modalities', *Oncology Reports*, vol. 12, no. 4, pp. 867-70.
- Mohan, S, Wilkes, LM, Ogunsiji, O & Walker, A 2005, 'Caring for patients with cancer in non-specialist wards: the nurse experience', *European Journal of Cancer Care*, vol. 14, no. 3, pp. 256-63.
- Moinpour, CM, Hayden, KA, Thompson, IM, Feigl, P & Metch, B 1990, 'Quality of life assessment in Southwest Oncology Group trials', *Oncology (Huntingt)*, vol. 4, no. 5, pp. 79-84, 9; discussion 104.
- Moinpour, CM, Lyons, B, Schmidt, SP, Chansky, K & Patchell, RA 2000, 'Substituting proxy ratings for patient ratings in cancer clinical trials: An analysis based on a Southwest Oncology Group trial in patients with brain metastases', *Quality of Life Research*, vol. 9, no. 2, pp. 219-31.
- Molzahn, AE, Northcott, HC & Dossetor, JB 1997, 'Quality of life of individuals with end stage renal disease: perceptions of patients, nurses, and physicians', *American Nephrology Nurses Association Journal*, vol. 24, no. 3, pp. 325-33; discussion 34-5.
- Montazeri, A, Gillis, C & McEwen, Jnr 1998, 'Quality of life in lung cancer patients. A review of literature from 1970-1995', *Chest*, vol. 113, pp. 467-81.
- Montazeri, A, Hole, D, Milroy, R, McEwen, J & Gillis, C 2004, 'Does knowledge of cancer diagnosis affect quality of life? A methodological challenge', *BMC Cancer*, vol. 4, no. 1, p. 21.

- Montazeri, A, Milroy, R, Hole, D, McEwen, J & Gillis, CR 2001, 'Quality of life in lung cancer patients As an important prognostic factor', *Lung Cancer*, vol. 31, no. 2-3, pp. 233-40.
- ---- 2003, 'How quality of life data contribute to our understanding of cancer patients' experiences? A study of patients with lung cancer', *Quality of Life Research*, vol. 12, no. 2, pp. 157-66.
- Morris, CW (ed.) 1962, *Mind, self, and society: from the standpoint of a social behaviorist*, The works of George Herbert Mead; v. 1, University of Chicago Press, Chicago.
- Morris, M 1977, *An Excursion into Creative Sociology*, Basil, Blackwell, Columbia University Press Oxford, New York.
- Mosconi, P, Colozza, M, Laurentiis, MD, Placido, SD & Maltoni, M 2001, 'Survival, quality of life and breast cancer', *Annals of Oncology*, vol. 12, suppl. 3, pp. 15-9.
- Movsas, B 2003, 'Quality of life in oncology trials: A clinical guide', *Seminars in Radiation Oncology*, vol. 13, no. 3, pp. 235-47.
- Muhlenkamp, AF & Joyner, J 1986, 'Arthritis Patients' Self-Reported Affective States and Their Caregivers' Perceptions', *Nursing Research*, vol. 35, no. 1, pp. 24-7.
- Mystakidou, K, Tsilika, E, Kouloulias, V, Parpa, E, Katsouda, E, Kouvaris, J & Vlahos, L 2004, 'The "Palliative Care Quality of Life Instrument (PQLI)" in terminal cancer patients', *Health and Quality of Life Outcomes*, vol. 2, no. 1, p. 8.
- Nasrabadi, AN, Emami, A & Yekta, ZP 2003, 'Nursing experience in Iran', International Journal of Nursing Practice, vol. 9, no. 2, pp. 78-85.
- Nekolaichuk, CL, Bruera, E, Spachynski, K, MacEachern, T, Hanson, J & Maguire, TO 1999, 'A comparison of patient and proxy symptom assessments in advanced cancer patients', *Palliative Medicine*, vol. 13, no. 4, pp. 311-23.
- Nelson, A 2007, 'Seeing white: a critical exploration of occupational therapy with Indigenous Australian people', *Occupational Therapy International*, vol. 14, no. 4, pp. 237-55.
- Nelson, LM, Longstreth, WT, Jr., Koepsell, TD & van Belle, G 1990, 'Proxy respondents in epidemiologic research', *Epidemiologic Reviews*, vol. 12, pp. 71-86.
- Neumann, PJ, Araki, SS & Gutterman, EM 2000, 'The use of proxy respondents in studies of older adults: lessons, challenges, and opportunities', *Journal of the American Geriatrics Society*, vol. 48, no. 12, pp. 1646-54.

- Novella, JL, Jochum, C, Jolly, D, Morrone, I, Ankri, J, Bureau, F & Blanchard, F 2001, 'Agreement between patients' and proxies' reports of quality of life in Alzheimer's disease', *Quality of Life Research*, vol. 10, no. 5, pp. 443-52.
- O'Carroll, RE, Smith, K, Couston, M, Cossar, JA & Hayes, PC 2000, 'A comparison of the WHOQOL-100 and the WHOQOL-BREF in detecting change in quality of life following liver transplantation', *Quality of Life Research*, vol. 9, no. 1, pp. 121-4.
- Olweny, CL 1992, 'Quality of life in developing countries', *Journal of Palliative Care*, vol. 8, no. 3, pp. 25-30.
- Organisation., WH 2006, *The WHO Global Info Base: The Impact of Cancer*, World Health Organisation, viewed 5 August 2007, http://www.who.int/ncd_surveillance/infobase/web/InfoBasePolicyMaker/re ports/ReporterFullView.aspx?id=5>.
- Osoba, D 1999, 'What has been learned from measuring health-related quality of life in clinical oncology', *European Journal of Cancer*, vol. 35, no. 11, pp. 1565-70.
- ---- 2000, 'Health-related quality-of-life assessment in clinical trials of supportive care in oncology', *Support Care Cancer*, vol. 8, no. 2, pp. 84-8.
- Osoba, D, Tannock, IF, Ernst, DS & Neville, AJ 1999, 'Health-Related Quality of Life in Men With Metastatic Prostate Cancer Treated With Prednisone Alone or Mitoxantrone and Prednisone', *Journal of Clinical Oncology*, vol. 17, no. 6, p. 1654.
- Padilla, GV, Ferrell, B, Grant, MM & Rhiner, M 1990, 'Defining the content domain of quality of life for cancer patients with pain', *Cancer Nursing*, vol. 13, no. 2, pp. 108-15.
- Pallant, JF 2004, SPSS survival manual: a step by step guide to data analysis using SPSS, 2nd edn, Allen & Unwin, Crows Nest, N.S.W.
- Palmer, BV, Walsh, GA, McKinna, JA & Greening, WP 1980, 'Adjuvant chemotherapy for breast cancer: side effects and quality of life', *British Medical Journal*, vol. 281, no. 6255, pp. 1594-7.
- Parker, KP 1997, 'A critical review of the research study: Quality of life of individuals with end stage renal disease: perceptions of patients, nurses, and physicians', *American Nephrology Nurses Association Journal*, vol. 24, no. 3, pp. 334-5.
- Payne, SA 1992, 'A study of quality of life in cancer patients receiving palliative chemotherapy', *Social Science and Medicine*, vol. 35, no. 12, pp. 1505-9.

- Perez, DJ, McGee, R, Campbell, AV, Christensen, EA & Williams, S 1997, 'A comparison of time trade-off and quality of life measures in patients with advanced cancer', *Quality of Life Research*, vol. 6, no. 2, p. 133-138.
- Phillips, WT, Alexander, JL, Pepin, V & Riley, C 2003, 'Cardiac Rehabilitation Patient Versus Proxy Quality-of-Life Perceptions', *Clinical Nursing Research*, vol. 12, no. 3, pp. 282-93.
- Phoenix 5 2002, *Definition of quality of life*, viewed 18 July 2007, http://www.phoenix5.org/glossary/Quality_of_Life.html>.
- Phungrassami, T, Katikarn, R, Watanaarepornchai, S & Sangtawan, D 2004, 'Quality of life assessment in radiotherapy patients by WHOQOL-BREF-THAI: a feasibility study.' *Journal of the Medical Association of Thailand*, vol. 87, no. 12, pp. 1459-65.
- Pickard, AS, Johnson, JA, Feeny, DH, Shuaib, A, Carriere, KC & Nasser, AM 2004, 'Agreement between patient and proxy assessments of Health-Related Quality of Life after Stroke using the EQ-5D and Health Utilities Index', *Stroke*, vol. 35, no. 2, pp. 607-12.
- Pickard, AS & Knight, SJ 2005, 'Proxy evaluation of health-related quality of life A Conceptual framework for understanding multiple proxy perspectives', *Medical Care*, vol. 43, no. 5, pp. 493-9.
- Pierre, U, Wood-Dauphinee, S, Korner-Bitensky, N, Gayton, D & Hanley, J 1998, 'Proxy use of the Canadian SF-36 in rating health status of the disabled elderly', *Journal of Clinical Epidemiology*, vol. 51, no. 11, pp. 983-90.
- Poh, L 1997, 'Clinical focus. Intuition and its relevance in expert nurses' practice', *Professional Nurse*, vol. 24, no. 2, pp. 8-10.
- Power, C 1998, 'Reconstituting self: a feminist poststructural analysis of women's narratives of domestic violence', PhD thesis, Flinders University.
- Ratcliffe, J, Young, T, Longworth, L & Buxton, M 2005, 'An Assessment of the Impact of Informative Dropout and Nonresponse in Measuring Health-Related Quality of Life Using the EuroQol (EQ-5D) Descriptive System', *Value Health*, vol. 8, no. 1, pp. 53-8.
- Rebollo, P, Alvarez-Ude, F, Valdes, C & Estebanez, C 2004, 'Different evaluations of the health related quality of life in dialysis patients', *Journal of Nephrology*, vol. 17, no. 6, pp. 833-40.
- Redden, M 2005, 'Interstitial Cytitis ', PhD thesis, Flinders University
- Richards, L 1999, 'Data Alive! The Thinking Behind NVivo', *Qualitative Health Research*, vol. 9, no. 3, pp. 412-28.
- Richardson Lear, G 2004, *Happy lives and the highest good: an essay on Aristotle's Nicomachean ethics*, Princeton University Press, Princeton, N.J.
- Roila, F & Cortesi, E 2001, 'Quality of life as a primary end point in oncology', *Annals of Oncology*, vol. 12, suppl. 3, pp. S3-6.
- Rothman, ML, Hedrick, SC, Bulcroft, KA, Hickam, DH & Rubenstein, LZ 1991,
 'The validity of proxy-generated scores as measures of patient health status', *Medical Care*, vol. 29, no. 2, pp. 115-24.
- Rubenstein, LZ, Schairer, C, Wieland, GD & Kane, R 1984, 'Systematic biases in functional status assessment of elderly adults: effects of different data sources', *Journal of Gerontology*, vol. 39, no. 6, pp. 686-91.
- Ruckdeschel, JC & Piantadosi, S 1994, 'Quality of life in lung cancer surgical adjuvant trials', *Chest*, vol. 106, no. 6 suppl, pp. 324S-8S.
- Rustoen, Moum, Wiklund & Hanestad 1999, 'Quality of life in newly diagnosed cancer patients', *Journal of Advanced Nursing*, vol. 29, no. 2, pp. 490-8.
- Ruth-Sahd, LA 1993, 'A modification of Benner's hierarchy of clinical practice: the development of clinical intuition in the novice trauma nurse', *Holistic Nursing Practice*, vol. 7, no. 3, pp. 8-14.
- Sadovich, JM 2005, 'Work excitement in nursing: an examination of the relationship between work excitement and burnout', *Nursing Economics*, vol. 23, no. 2, pp. 91-6.
- Saffarzadeh, T 2001, *The Holy Quran: English and Persian Translation with Commentary*, Honare Bidari, Tehran.
- Salek, MS (ed.) 1998, *Compendium of quality of life instruments*, John Wiley & Sons, New York.
- Sandelowski, M 2000a, 'Combining qualitative and quantitative sampling, data collection, and analysis techniques in mixed-method studies', *Research in Nursing and Health*, vol. 23, no. 3, pp. 246-55.
- ---- 2000b, 'Whatever happened to qualitative description?' *Research in Nursing & Health*, vol. 23, no. 4, pp. 334-40.
- Sands, LP, Ferreira, P, Stewart, AL, Brod, M & Yaffe, K 2004, 'What explains differences between dementia patients' and their caregivers' ratings of patients' quality of life?' *American Journal of Geriatric Psychiatry*, vol. 12, no. 3, pp. 272-80.
- Saxena, S, Carlson, D, Billington, R & Orley, J 2001, 'The WHO quality of life assessment instrument (WHOQOL-Bref): The importance of its items for cross-cultural research', *Quality of Life Research*, vol. 10, no. 8, pp. 711-21.

- Saylor, C 2004, 'The circle of health: a health definition model', *Journal of Holistic Nursing*, vol. 22, no. 2, pp. 97-115.
- Schalock, RL 2004, 'The concept of quality of life: what we know and do not know', *Journal of Intellectual Disability Research*, vol. 48, no. 3, pp. 203-16.
- Schipper, H & Levitt, M 1985, 'Measuring quality of life: risks and benefits', *Cancer Treatment Report*, vol. 69, no. 10, pp. 1115-25.
- Schumacher, M, Olschewski, M & Schulgen, G 1991, 'Assessment of quality of life in clinical trials', *Statistics in Medicine*, vol. 10, no. 12, pp. 1915-30.
- Schwartz, CE & Sprangers, MAG 2002, 'An introduction to quality of life assessment in oncology: the value of measuring patient-reported outcomes', *American Journal of Managed Care*, vol. 8, no. 18 suppl, pp. S550-9.
- Schwenk, W, Neudecker, J, Haase, O, Raue, W, Strohm, T & Muller, JM 2004, 'Comparison of EORTC quality of life core questionnaire (EORTC-QLQ-C30) and gastrointestinal quality of life index (GIQLI) in patients undergoing elective colorectal cancer resection', *International Journal of Colorectal Disease*, vol. 19, no. 6, pp. 554-60.
- Shadbolt, B, Barresi, J & Craft, P 2002, 'Self-Rated Health as a Predictor of Survival Among Patients With Advanced Cancer', *Journal of Clinical Oncology*, vol. 20, no. 10, pp. 2514-9.
- Shahriari, SVC, 1999 2005, 1998, *Rnmi 114*, viewed 26 February 2008, http://www.rumionfire.com/shams/rumi114.htm.
- Sherman, RO & Dyess, S 2007, 'Be a coach for novice nurses', *American Nurse Today*, vol. 2, no. 5, pp. 54-5.
- Sigurdardottir, V, Brandberg, Y & Sullivan, M 1996, 'Criterion-based validation of the EORTC QLQ-C36 in advanced melanoma: the CIPS questionnaire and proxy raters', *Quality of Life Research*, vol. 5, no. 3, pp. 375-86.
- Sivesind, D, Parker, PA, Cohen, L, DeMoor, C, Bumbaugh, M, Throckmorton, T, Volker, DL & Baile, WF 2003, 'Communicating with Patients in Cancer Care; What Areas Do Nurses Find Most Challenging?' *Journal of Cancer Education*, vol. 18, no. 4, pp. 202-9.
- Skevington, SM 2002, 'Advancing cross-cultural research on quality of life: Observations drawn from the WHOQOL development', *Quality of Life Research*, vol. 11, no. 2, pp. 135-44.
- Skevington, SM, Lotfy, M & O'Connell, KA 2004, 'The World Health Organization's WHOQOL-BREF quality of life assessment: Psychometric properties and results of the international field trial. A Report from the WHOQOL Group', *Quality of Life Research*, vol. 13, no. 2, pp. 299-310.

- Slevin, ML, Plant, H, Lynch, D, Drinkwater, J & Gregory, WM 1988, 'Who should measure quality of life, the doctor or the patient?' *British Journal of Cancer*, vol. 57, no. 1, pp. 109-12.
- Slevin, ML, Stubbs, L, Plant, HJ, Wilson, P, Gregory, WM, Armes, PJ & Downer, SM 1990, 'Attitudes to chemotherapy: comparing views of patients with cancer with those of doctors, nurses, and general public', *British Medical Journal*, vol. 300, no. 6737, pp. 1458-60.
- Sloan, J, Loprinzi, C, Kuross, S, Miser, A, O'Fallon, J, Mahoney, M, Heid, I, Bretscher, M & Vaught, N 1998, 'Randomized comparison of four tools measuring overall quality of life in patients with advanced cancer', *Journal of Clinical Oncology*, vol. 16, no. 11, pp. 3662-73.
- Sloan, JA, Aaronson, N, Cappelleri, JC, Fairclough, DL & Varricchio, C 2002, 'Symposium on quality of life in cancer patients. Assessing the clinical significance of single items relative to summated scores', *Mayo Clinic Proceedings.*, vol. 77, no. 5, pp. 479-87.
- Smith, K & Biley, F 1997, 'Understanding grounded theory: principles and evaluation', *Nurse Researcher*, vol. 4, no. 3, pp. 17-30.
- Sneeuw, KC, Aaronson, NK, Osoba, D, Muller, MJ, Hsu, MA, Yung, WK, Brada, M & Newlands, ES 1997, 'The use of significant others as proxy raters of the quality of life of patients with brain cancer', *Medical Care*, vol. 35, no. 5, pp. 490-506.
- Sneeuw, KC, Aaronson, NK, Sprangers, MA, Detmar, SB, Wever, LD & Schornagel, JH 1997, 'Value of caregiver ratings in evaluating the quality of life of patients with cancer', *Journal of Clinical Oncology*, vol. 15, no. 3, pp. 1206-17.
- ---- 1998, 'Comparison of patient and proxy EORTC QLQ-C30 ratings in assessing the quality of life of cancer patients', *Journal of Clinical Epidemiology*, vol. 51, no. 7, pp. 617-31.
- Sneeuw, KC, Aaronson, NK, Sprangers, MAG, Detmar, SB, Wever, LDV & Schornagel, JH 1999, 'Evaluating the quality of life of cancer patients: assessments by patients, significant others, physicians and nurses', *British Journal of Cancer*, vol. 81, no. 1, pp. 87-94.
- Sneeuw, KC, Sprangers, MA & Aaronson, NK 2002, 'The role of health care providers and significant others in evaluating the quality of life of patients with chronic disease', *Journal of Clinical Epidemiology*, vol. 55, no. 11, pp. 1130-43.
- Soni, MK & Cella, D 2002, 'Quality of life and symptom measures in oncology: an overview', *The American Journal of Managed Care*, vol. 8, no. 18 suppl, pp. S560-73.

- South Australian Cancer Registry 2004, *Cancer in South Australia 2004 with incidence projections to 2007*, South Australian Department of Health, viewed 3 August 2007, http://www.dh.sa.gov.au/pehs/cancerreport2004.htm.
- Spilker, B, Molinek, FR, Jr., Johnston, KA, Simpson, RL, Jr. & Tilson, HH 1990, 'Quality of life bibliography and indexes', *Medical Care*, vol. 28, no. 12 suppl, pp. DS1-77.
- Spitzer, WO 1987, 'State of science 1986: quality of life and functional status as target variables for research', *Journal of Chronic Disease*, vol. 40, no. 6, pp. 465-71.
- Sprangers, MAG 2002, 'Quality-of-life assessment in oncology. Achievements and challenges', *Acta Oncologica*, vol. 41, no. 3, pp. 229-37.
- Sprangers, MAG & Aaronson, NK 1992, 'The role of health care providers and significant others in evaluating the quality of life of patients with chronic disease: a review', *Journal of Clinical Epidemiology*, vol. 45, no. 7, pp. 743-60.
- Stanley, M 2006, 'A grounded theory of the wellbeing of older people', in L Finlay & C Ballinger (eds), *Qualitative Research for Allied Health Professionals. Challenging Choices*, John Wiley & Sons Ltd., London.
- Stavem, K, Lossius, MI, Kvien, TK & Guldvog, Br 2000, 'The health-related quality of life of patients with epilepsy compared with angina pectoris, rheumatoid arthritis, asthma and chronic obstructive pulmonary disease', *Quality of Life Research*, vol. 9, no. 7, pp. 865-71.
- Steel, JL, Geller, DA & Carr, BI 2005, 'Proxy ratings of health related quality of life in patients with hepatocellular carcinoma', *Quality of Life Research*, vol. 14, no. 4, pp. 1025-33.
- Stemler, SE 2004, 'A comparison of consensus, consistency, and measurement approaches to estimating interrater reliability', *Practical Assessment, Research & Evaluation*, vol. 9, no. 4, viewed February 1, 2006, http://PAREonline.net/getvn.asp?v=9&n=4 .>.
- Stephens, RJ, Hopwood, P, Girling, DJ & Machin, D 1997, 'Randomized trials with quality of life endpoints: are doctors' ratings of patients' physical symptoms interchangeable with patients' self-ratings?' *Quality of Life Research*, vol. 6, no. 3, pp. 225-36.
- Stineman, MG, Ross, RN, Maislin, G & Iezzoni, L 2004, 'Estimating health-related quality of life in populations through cross-sectional surveys', *Medical Care*, vol. 42, no. 6, pp. 569-78.
- Story, I 2004, 'A brief survey of univariate group comparison procedures', in V Minichiello, G Sullivan, K Greenwood & R Axford (eds), *Handbook of*

research methods for nursing and health science, 2nd edn, Prentice Hall Health, Frenchs Forest, N.S.W.

- Strain, JJ 1990, 'The evolution of quality of life evaluations in cancer therapy', *Oncology (Williston Park)*, vol. 4, no. 5, pp. 22-6.
- Strauss, AL & Corbin, JM 1998, *Basics of qualitative research: techniques and procedures for developing grounded theory*, 2nd edn, Sage Publications, Thousand Oaks.
- Stromgren, AS, Groenvold, M, Sorensen, A & Andersen, L 2001, 'Symptom recognition in advanced cancer. A comparison of nursing records against patient self-rating', *Acta Anaesthesiologica Scandinavica*, vol. 45, no. 9, pp. 1080-5.
- Sugarbaker, PH, Barofsky, I, Rosenberg, SA & Gianola, FJ 1982, 'Quality of life assessment of patients in extremity sarcoma clinical trials', *Surgery*, vol. 91, no. 1, pp. 17-23.
- Suppe, F 2001, 'Response to "Positivism and Qualitative Nursing Research"', *Scholarly Inquiry for Nursing Practice*, vol. 15, no. 4, pp. 389-97.
- Sweed, MR, Schiech, L, Barsevick, A, Babb, JS & Goldberg, M 2002, 'Quality of life after esophagectomy for cancer', *Oncology Nursing Forum*, vol. 29, no. 7, pp. 1127-31.
- Szalai, A 1980, 'The meaning of comparative research on the quality of life', in A Szalai & FM Andrews (eds), *The quality of life: Comparative studies*, Sage Publications, London; Beverly Hills, Calif.
- Tabachnick, BG & Fidell, LS 2001, *Using multivariate statistics*, 4th edn, Allyn and Bacon, Boston, Mass.
- Tallis, K 2005, 'How to improve the quality of life in patients living with end stage renal failure', *Renal Society of Australasia Journal*, vol. 1, no. 1, pp. 18-22.
- Tamburini, M 2001, 'Health-related quality of life measures in cancer', *Annals of Oncology*, vol. 12, pp. S7-S10.
- Tamim, H, McCusker, J & Dendukuri, N 2002, 'Proxy reporting of quality of life using the EQ-5D', *Medical Care*, vol. 40, no. 12, pp. 1186-95.
- Tang, S 2006, 'Predictors of the extent of agreement for quality of life assessments between terminally ill cancer patients and their primary family caregivers in Taiwan', *Quality of Life Research*, vol. 15, no. 3, pp. 391-404.
- Tang, ST & McCorkle, R 2002a, 'Appropriate time frames for data collection in quality of life research among cancer patients at the end of life', *Quality of Life Research*, vol. 11, no. 2, pp. 145-55.

- ---- 2002b, 'Use of family proxies in quality of life research for cancer patients at the end of life: a literature review', *Cancer Investigation*, vol. 20, no. 7-8, pp. 1086-104.
- Tannock, IF, Boyd, NF, DeBoer, G, Erlichman, C, Fine, S, Larocque, G, Mayers, C, Perrault, D & Sutherland, H 1988, 'A randomized trial of two dose levels of cyclophosphamide, methotrexate, and fluorouracil chemotherapy for patients with metastatic breast cancer', *Journal of Clinical Oncology*, vol. 6, no. 9, pp. 1377-87.
- Taylor, C 2002, 'Assessing patients' needs: does the same information guide expert and novice nurses?' *International Nursing Review*, vol. 49, no. 1, pp. 11-9.
- Taylor, EJ 2003, 'Spiritual Quality of Life', in CR King & PS Hinds (eds), *Quality of Life: From Nursing and Patient Perspectives: Theory, Research, Practice,* Jones and Bartlett Publishers, Sudbury.
- Teddlie, C & Tashakkori, A 2003, 'Major issues and controversies in the use of mixed methods in the social and behavioural sciences', in A Tashakkori & C Teddlie (eds), *Handbook of Mixed methods in social and behavioural research*, Sage, Thousand Oaks, Calif.
- Testa, MA & Simonson, DC 1996, 'Assessment of Quality-of-Life Outcomes', *The New England Journal of Medicine*, vol. 334, no. 13, pp. 835-40.
- Thapa, K & Rowland, LA 1989, 'Quality of life perspectives in long-term care: staff and patient perceptions', *Acta Psychiatrica Scandinavica*, vol. 80, no. 3, pp. 267-71.
- The Australian Centre for Posttraumatic Mental Health 2003, *World Health Organization Quality of Life (WHOQoL) Project*, viewed 21 Feb, 1 pm 2006, http://www.acpmh.unimelb.edu.au/whoqol/default.html.
- The WHO Group 1995, 'The World Health Organization quality of life assessment (WHOQOL): Position paper from the World Health Organization', *Social Science & Medicine*, vol. 41, no. 10, pp. 1403-9.
- ---- 1998a, 'Development of the World Health Organization WHOQOL-BREF Quality of Life Assessment. '*Psychological Medicine*, vol. 28, no. 3, pp. 551-8.
- ---- 1998b, 'The World Health Organization quality of life assessment (WHOQOL): Development and general psychometric properties1', *Social Science & Medicine*, vol. 46, no. 12, pp. 1569-85.
- The WHOQOL Group 1994, 'Development of the WHOQOL: rationale and current status', *International Journal of Mental Health*, vol. 23, pp. 24–56.
- Thomas, SA 2000, *How to write health science papers, dissertations, and theses,* Churchill Livingstone, New York.

- Thorne, S, Kirkham, SR & MacDonald-Emes, J 1997, 'Interpretive description: a noncategorical qualitative alternative for developing nursing knowledge', *Research in Nursing and Health*, vol. 20, no. 2, pp. 169-77.
- Tripoliti, E, Campbell, C, Pring, T & Taylor-Goh, S 2007, 'Quality of life in multiple sclerosis: should clinicians trust proxy ratings?' *Multiple Sclerosis*, vol. 13, no. 9, pp. 1190-4.
- Trompenaars, F, Masthoff, ED, Heck, GLV, Hodiamont, PP & Vries, JD 2005, 'Content validity, construct validity, and reliability of the WHOQOL-Bref in a population of Dutch adult psychiatric outpatients', *Quality of Life Research*, vol. 14, no. 1, pp. 151-60.
- Uebersax, J 2003, *Statistical methods for rater agreement*, viewed 17th of January 2006, http://ourworld.compuserve.com/homepages/jsuebersax/agree.htm#basics>.
- van der Walt, HM & Swartz, L 2002, 'Task orientated nursing in a tuberculosis control programme in South Africa: where does it come from and what keeps it going?' *Social Science & Medicine*, vol. 54, no. 7, pp. 1001-9.
- Varricchio, CG 1990, 'Relevance of quality of life to clinical nursing practice', *Seminars in Oncology Nursing*, vol. 6, no. 4, pp. 255-9.
- Velikova, G, Booth, L, Smith, AB, Brown, PM, Lynch, P, Brown, JM & Selby, PJ 2004, 'Measuring quality of life in routine oncology practice improves communication and patient well-being: a randomized controlled trial', *Journal of Clinical Oncology*, vol. 22, no. 4, pp. 714-24.
- Ventegodt, S, Merrick, J & Andersen, NJ 2003, 'Quality of life theory I. The IQOL theory: an integrative theory of the global quality of life concept', *ScientificWorldJournal*, vol. 3, pp. 1030-40.
- Verma, BL 1997, 'Quality of life measures in health care: What makes a good measuring instrument', *Quality of Life Research*, vol. 6, no. 7-8, p. 405.
- Vigano, A, Donaldson, N, Higginson, IJ, Bruera, E, Mahmud, S & Suarez-Almazor, M 2004, 'Quality of life and survival prediction in terminal cancer patients: A multicenter study', *Cancer*, vol. 101, no. 5, pp. 1090-8.
- von Essen, L 2004, 'Proxy ratings of patient quality of life Factors related to patient-proxy agreement', *Acta Oncologica*, vol. 43, no. 3, pp. 229-34.
- Wainwright, SP 1997, 'A new paradigm for nursing: the potential of realism', *Journal* of Advanced Nursing, vol. 26, no. 6, pp. 1262-71.
- Waldron, D, O'Boyle, CA, Kearney, M, Moriarty, M & Carney, D 1999, 'Quality-of-Life Measurement in Advanced Cancer: Assessing the Individual', *Journal of Clinical Oncology*, vol. 17, no. 11, pp. 3603-11.

- Walker, D & Myrick, F 2006, 'Grounded Theory: An Exploration of Process and Procedure', *Qualitative Health Research*, vol. 16, no. 4, pp. 547-59.
- Weaver, K & Olson, JK 2006, 'Understanding paradigms used for nursing research', *Journal of Advanced Nursing*, vol. 53, no. 4, pp. 459-69.
- Williams-Cox, D 2004, 'A mixed-method study into quality of life for bladder cancer patients', *Professional Nurse*, vol. 19, no. 6, pp. 343-7.
- Williams, LS, Bakas, T, Brizendine, E, Plue, L, Tu, W, Hendrie, H & Kroenke, K 2006, 'How valid are family proxy assessments of stroke patients' healthrelated quality of life?' *Stroke*, vol. 37, no. 8, pp. 2081-5.
- Wilson, HS & Hutchinson, SA 1991, 'Triangulation of Qualitative Methods: Heideggerian Hermeneutics and Grounded Theory', *Qualitative Health Research*, vol. 1, no. 2, pp. 263-76.
- Wilson, HS, Hutchinson, SA & Holzemer, WL 2002, 'Reconciling Incompatibilities: A Grounded Theory of HIV Medication Adherence and Symptom Management', *Qualitative Health Research*, vol. 12, no. 10, pp. 1309-22.
- Wilson, KA, Dowling, AJ, Abdolell, M & Tannock, IF 2000, 'Perception of quality of life by patients, partners and treating physicians', *Quality of Life Research*, vol. 9, no. 9, pp. 1041-52.
- World Health Organisation 1946, 'The constitution of the world Health organization', *American Journal of Public Health and the Nations' Health*, vol. 36, no. 11, pp. 1315-23.
- ---- 1996, WHOQOL-BREF: Introduction, administration, scoring and generic version of the assessment, WHO, viewed 25 February 2005, http://www.who.int/mental_health/media/en/76.pdf>.
- ---- 1998, Annotated Bibliography of the WHO Quality of Life Assessment Instrument - WHOQOL, WHO, viewed 25 February 2005, http://www.who.int/evidence/assessment-instruments/qol/documents/WHOQOL-BIBLIOGRAPHY.pdf>.
- ---- 2007, Cancer viewed 20 July 2007, <http://www.who.int/cancer/en/>.
- World Health Organization 1993, WHO—Quality of life study protocol: The development of the World Health Organization quality of life assessment instrument, World Health Organization, Geneva, Switzerland.
- Wright, EP, Selby, PJ, Crawford, M, Gillibrand, A, Johnston, C, Perren, TJ, Rush, R, Smith, A, Velikova, G, Watson, K, Gould, A & Cull, A 2003, 'Feasibility and Compliance of Automated Measurement of Quality of Life in Oncology Practice', *Journal of Clinical Oncology*, vol. 21, no. 2, pp. 374-82.

- Yancik, R 2005, 'Population aging and cancer: a cross-national concern', *Cancer Journal*, vol. 11, no. 6, pp. 437-41.
- Zhan, L 1992, 'Quality of life: conceptual and measurement issues', *Journal of Advanced Nursing*, vol. 17, no. 7, pp. 795-800.
- Zhao, H, Kanda, K, Liu, S-J & Mao, X-Y 2003, 'Evaluation of quality of life in Chinese patients with gynaecological cancer: Assessments by patients and nurses', *International Journal of Nursing Practice*, vol. 9, no. 1, pp. 40-8.
- Zollo, JA 2002, 'Mainstreaming palliative care: hetoric or reality: a qualitative investigation of the perceptions of key palliative care stakeholders in relation to the 'mainstreaming' of hospice/palliative care', PhD thesis, Flinders University.